

Power Frequency Electromagnetic Fields, Melatonin and the Risk of Breast Cancer

Report of an independent Advisory Group on Non-ionising Radiation



Cover illustration: photomicrograph of the breast tissue of a premenopausal woman showing groups of normal resting lobular alveolar units within fibrous tissue and fat.

Reproduced by courtesy of Professor W M Miller, Western General Hospital, Edinburgh

Power Frequency Electromagnetic Fields, Melatonin and the Risk of Breast Cancer

Report of an independent Advisory Group on Non-ionising Radiation

Documents of the Health Protection Agency Series B: Radiation, Chemical and Environmental Hazards February 2006

Contents

Foreword				
Power Frequency Electromagnetic Fields, Melatonin and the Risk of Breast Cancer				
Advisory Group on Non-ionising Radiation				
Su	mmary	y .	Electromagnetic Fields, Melatonin east Cancer 1 on-ionising Radiation 3 on the report 3 or the report 7 he report 8 or of power frequency EMFs 11 on of power frequency EMFs 11 ssment 17 ronments 23 on Melatonin 23 gy 31 trol of melatonin synthesis in animals and man 40 onin in photoperiodic seasonal functions in animals 45 head in mammalian circadian rhythms 48 ed administration of melatonin in animals 48 numan physiology and pathology 50 genous melatonin in humans 61 f melatonin action 65 her hormones and breast cancer 68	
1	Intro	7		
	1.1	Background to the report	7	
	1.2	Structure of the report	8	
	1.3	References	10	
2	Sour	11		
	2.1	Characterisation of power frequency EMFs	11	
	2.2	Sources of power frequency EMFs	11	
	2.3	Exposure assessment	17	
	2.4	Exposure environments	23	
	2.5	Summary	26	
	2.6	References	27	
3	The	31		
	3.1	Basic physiology	31	
	3.2	Light–dark control of melatonin synthesis in animals and man	40	
	3.3	Role of melatonin in photoperiodic seasonal functions in animals	45	
	3.4	Role of the pineal in mammalian circadian rhythms	48	
	3.5	Effects of timed administration of melatonin in animals	48	
	3.6	The pineal in human physiology and pathology	50	
	3.7	Effects of exogenous melatonin in humans	61	
	3.8	Mechanisms of melatonin action	65	
	3.9	Melatonin, other hormones and breast cancer	68	
	3.10	Summary	72	
	3.11	References	73	

4	Effe 4.1 4.2 4.3 4.4 4.5	cts of EMFs on Melatonin <i>In vitro</i> studies <i>In vivo</i> studies Human experimental studies Epidemiology References	91 91 97 103 110 114		
5	Mel	atonin and Breast Cancer	121		
	5.1	In vitro studies	121		
	5.2	In vivo studies	122		
	5.3	Epidemiology: breast cancer risk in relation to melatonin levels	131		
	5.4	Epidemiology: breast cancer risk in relation to light exposure	134		
	5.5	References	142		
6	Effe	149			
	6.1	<i>In vitro</i> studies	149		
	6.2	<i>In vivo</i> studies	150		
	6.3	Epidemiology	153		
	6.4	References	154		
7	Con	157			
	7.1	Sources and exposure assessment	157		
	7.2	The pineal gland and melatonin	158		
	7.3	Effects of EMFs on melatonin	159		
	7.4	Melatonin and breast cancer	159		
	7.5	Effects of EMFs on breast cancer	160		
	7.6	Overall summary and conclusions	161		
8	Research Recommendations				
	8.1	Effects of EMFs on melatonin	163		
	8.2	Melatonin and breast cancer	164		
	8.3	EMFs and breast cancer	165		
Glossary					
Арр	Appendix Publications of the independent Advisory Group on				
	Non-Ionising Radiation				

Foreword

Until 1 April 2005, the National Radiological Protection Board (NRPB) had a statutory responsibility for advising UK government departments on health effects and standards of protection for exposures to ionising and non-ionising radiations. This responsibility now lies with the Radiation Protection Division (RPD) of the Health Protection Agency (HPA).

In 1990, to provide support for the development of advice on non-ionising radiations, the Director of the NRPB set up an Advisory Group on Non-ionising Radiation with terms of reference:

'to review work on the biological effects of non-ionising radiation relevant to human health and to advise on research priorities'

The Advisory Group was reconstituted in 1999 as an independent body and now reports directly to the Board of the HPA. Its current membership is given on page 3 of this report. For details of its current work programme, see the website www.hpa.org.uk.

The Advisory Group has, to date, issued a number of reports concerned with exposures to electromagnetic fields. It has considered their possible association with an increased risk of cancer, including childhood leukaemia. It has also reported on health effects related to the use of visual display units, on neurodegenerative disease and on corona ions and increased particle deposition near power lines. The Advisory Group has also considered the potential health effects of radiofrequency fields. Details of publications by the Advisory Group are given in an appendix.

In this report the Advisory Group considers the available scientific evidence from studies with humans, animals and cells relating to power frequency electromagnetic fields, melatonin and the risk of breast cancer.

Power Frequency Electromagnetic Fields, Melatonin and the Risk of Breast Cancer

Report of an independent Advisory Group on Non-ionising Radiation *Chairman* Professor A J Swerdlow

This report from the independent Advisory Group on Non-ionising Radiation reflects understanding and evaluation of the current scientific evidence as presented and referenced in this document.

Advisory Group on Non-ionising Radiation

CHAIRMAN

Professor A J Swerdlow Section of Epidemiology, Institute of Cancer Research, London

MEMBERS

Professor L J Challis University of Nottingham

Professor D N M Coggon University of Southampton

Dr L A Coulton University of Sheffield

Professor S C Darby Clinical Trials Service Unit, University of Oxford

Professor P A Gowland *(member from 1 December 2004)* University of Nottingham

Professor P Haggard Institute of Cognitive Neuroscience, University College, London

Professor D J Lomas *(member from 1 December 2004)* Addenbrooke's Hospital, Cambridge

Professor D Noble University of Oxford

SECRETARIAT

Dr S M Mann Health Protection Agency, Chilton

OBSERVER

Dr H Walker Department of Health, London

HPA REPRESENTATIVES

Dr A F McKinlay Health Protection Agency, Chilton

Dr C R Muirhead Health Protection Agency, Chilton

Dr J W Stather Health Protection Agency, Chilton

CONSULTANTS

Mr S G Allen Health Protection Agency, Chilton

Dr Z J Sienkiewicz Health Protection Agency, Chilton

AGNIR Melatonin Subgroup

CHAIRMAN

Professor A J Swerdlow Section of Epidemiology, Institute of Cancer Research, London

MEMBERS

Professor J Arendt University of Surrey

Dr L A Coulton University of Sheffield

Professor W R Miller University of Edinburgh

SECRETARIAT

Dr Z J Sienkiewicz Health Protection Agency, Chilton

OBSERVER

Mr G Hooker Department of Health, London

HPA REPRESENTATIVE

Dr J W Stather Health Protection Agency, Chilton

CONSULTANT

Dr M P Maslanyj Health Protection Agency, Chilton

Acknowledgements

We would like to thank Professor C Blakemore (University of Oxford) and Professor M Dowsett (Institute of Cancer Research, London) for contributing towards the early work of the Subgroup.

Summary

Exposure to power frequency electromagnetic fields (EMFs) is ubiquitous in modern life. The hypothesis that chronic exposure to EMFs may increase the risk of breast cancer, via a reduction in secretion of the hormone melatonin from the pineal gland, was first made almost 20 years ago, and has led to a great deal of research. To review this hypothesis, this report addresses evidence on three issues, namely, whether:

- (a) EMFs affect the production or action of melatonin,
- (b) melatonin affects the risk of breast cancer,
- (c) EMFs affect the risk of breast cancer.

Investigations using cells, animals and humans have not given consistent or convincing evidence that EMF exposure affects melatonin production or action. However, there are deficiencies in the existing research, which leave open the possibility of an effect.

There is stronger evidence that melatonin can inhibit the growth of cancer cells in laboratory culture and in animals. Data on the possible relation of melatonin levels to risk of subsequent breast cancer in humans are limited and inconclusive. Studies investigating the effect of light exposure (which affects melatonin) on breast cancer risk in humans have given some evidence for an association, but left it unclear whether, if there is an association, it is causal in nature.

There is no consistent evidence, from research using cells, animals and humans, that EMF exposure is a cause of breast cancer, nor has any mechanism for such an association been demonstrated.

The report concludes with recommendations for further research.

Overall, the evidence that melatonin, and the timing and extent of light exposure, may affect breast cancer risk is intriguing but not conclusive. In aggregate, the evidence to date does not support the hypothesis that exposure to power frequency EMFs affects melatonin levels or the risk of breast cancer.

1 Introduction

Exposure to power frequency electromagnetic fields (EMFs) is ubiquitous in modern life. Sources of exposure include the national grid system, the local electricity supply network, and mains wiring in homes, offices and other buildings. In addition, EMFs are produced by all machines, appliances and devices powered by electricity: electric fields are related to voltage differences, whereas magnetic fields are associated with the flow of electric current.

In recent years there has been concern about possible health effects of exposure to EMFs arising from the electricity supply system (50 Hz in the UK) and in particular about a possible increased risk of childhood leukaemia. This has been a public issue in many countries and been the subject of a number of national and international reviews (eg Ahlbom et al, 2000; IARC, 2001).

The suggestion has also been made that exposure to EMFs may increase the risk of breast cancer. This was first made almost 20 years ago, and has led to a great deal of research: melatonin, a hormone produced by the pineal gland in the brain, has been suggested as playing a pivotal role in this.

The purpose of this report is to consider the available scientific evidence from studies with humans, animals and cells relating to power frequency EMFs, melatonin and the risk of breast cancer. Specifically, it has addressed three questions, namely, whether:

- (a) EMFs can affect the production or action of melatonin,
- (b) melatonin can affect the risk of breast cancer,
- (c) EMFs can affect the risk of breast cancer.

1.1 Background to the report

Melatonin is a hormone produced by the pineal gland in a distinct daily rhythm governed by daylength. Levels of melatonin in the blood are very low during the day and are elevated at night in all animals, including humans (Arendt, 1995). Melatonin has been shown to influence the control of daily activities such as the sleep/wake cycle and seasonal rhythms such as those of reproduction in animals that respond to daylength. It has been found to diminish the effect of some chemical carcinogens on mammary tissue in animal experiments and to reduce the rate of growth of human breast cancer cells in culture.

Cohen and colleagues first proposed the hypothesis that diminished function of the pineal gland may promote the development of human breast cancer (Cohen et al, 1978). In addition, Stevens (1987) suggested that chronic exposure to electric fields (or to visible light at night) may reduce melatonin secretion by the pineal gland and so increase the risk of breast cancer.

Taken together, these ideas form the basis of the so-called melatonin hypothesis, which attempts to link electrical power use with increased risk of breast cancer via melatonin. This hypothesis has aroused wide interest and attention and has stimulated considerable experimental and epidemiological research.

Given the diversity of effects caused by melatonin and the complexity of its actions, long-term changes in this hormone might have wide ranging consequences for health. It has been proposed that these might include altered incidence of solid tumours or leukaemias, as well as effects on ageing and neurodegenerative diseases. While these endpoints are of potential interest, most melatonin-related EMF research has been focused on breast cancer.

As part of a previous comprehensive evaluation of the potential of power frequency EMFs to cause cancer, the Advisory Group on Non-ionising Radiation (AGNIR) concluded that melatonin rhythms were not substantially affected by exposure to magnetic fields (AGNIR, 2001), although the preliminary data from one investigation suggested some effects may occur, possibility in a sensitive subgroup of the study population. Since the publication of that report, further studies have been published relevant to an understanding of possible effects of EMFs on melatonin. As a consequence, the Board of the NRPB asked the Advisory Group to examine the evidence related to effects of EMFs on the production of melatonin and to advise on whether this could be implicated in the development of breast cancer.

1.2 Structure of the report

The report has been written in eight chapters. Following this introduction which provides an overall background and perspective to the report, the next two chapters give further background information important to the interpretation of the epidemiological and experimental evidence.

Chapter 2 describes the main physical characteristics of EMFs and common sources of exposure. The likely size of the fields in various residential and occupational settings is considered, as are fluctuations in exposure that occur during the day and night. Exposure to either an electric or magnetic field will induce electric fields and currents in biological tissues that depend on the magnitude of the external field.

Chapter 3 reviews the basic physiology of the pineal gland and its role in the production and secretion of melatonin into the blood. The roles of melatonin in reproductive and other seasonal functions that depend on the response of organisms to the length of the day are discussed as well as those such as sleep activity that have a near 24 hour period (circadian rhythm). The characteristics of melatonin production in humans and the effects of melatonin treatment are also described.

In addition, there is compelling evidence that hormones substantially influence the development of breast cancer. This evidence is reviewed in Chapter 3 together with information on the possibility that melatonin may mediate or interact with the mechanism of sex hormone action. A range of other possible mechanisms have been postulated whereby melatonin could influence cell growth and differentiation, through changes in the communication between cells, and in the immune system. These possibilities are also reviewed, as are melatonin receptors and their pharmacology.

The next three chapters consider the experimental and epidemiological evidence for each of the three main themes of the report. The experimental studies are further divided into investigations using cultures of cells (*in vitro* studies), studies using animals (*in vivo* studies), and studies using human volunteers.

Chapter 4 considers the effect of EMFs on melatonin production and action. If exposure to EMFs is to influence the development of cancer through an effect on melatonin secretion, then an understanding of how such fields may influence melatonin is essential.

The *in vitro* evidence divides into two types of study: those investigating effects on the production of melatonin by cultures of pineal cells, and those investigating effects on the action of melatonin on cells. Most *in vivo* studies have used rats and mice, although some studies have used Djungarian hamsters. Very few data have been obtained using non-human primates. Laboratory-based studies in humans investigating the effects of exposure to magnetic fields are described. Few of these studies, however, have used appropriate control procedures or monitored conditions sufficiently to enable the potential of magnetic fields to be properly assessed. Only a few epidemiological studies on EMF exposures and melatonin have been published. These are mainly concerned with melatonin levels in relation to residential exposures in women, and occupational exposures in both women and men.

Chapter 5 explores the relationship between melatonin and risk of breast cancer. As indicated above, the potential relationship between melatonin and the development of cancer is a subject that has aroused much interest after early work suggesting that pineal secretions decreased the rates of cell proliferation and transformation. *In vitro* studies of melatonin and breast cancer that have focused on this aspect are reviewed: more than 60 *in vitro* studies have been published, with the vast majority using the same breast cancer cell line (MCF-7) that was first described in 1973.

The potential of melatonin to modulate the incidence and growth of mammary tumours in various animal models of breast cancer is also reviewed. Most of these studies have assessed the effects of melatonin treatment on the growth of chemically induced mammary tumours; however, a few studies investigated effects on transplantable tumours, while others used normal and transgenic mouse strains that express a high spontaneous incidence of mammary tumours. Further studies are described that investigated the effects caused by inducing changes in pineal function either by removing the pineal gland (pinealectomy) or by altering the apparent day length.

Studies assessing melatonin secretion in breast cancer patients and in individuals without cancer are also reviewed in Chapter 5. These studies are difficult to interpret because the presence of breast cancer may have affected melatonin levels, and also because many of the studies lack control of potentially confounding factors that may affect melatonin secretion. Two cohort studies have given data on risk of subsequent breast cancer in women for whom information was available on urinary melatonin metabolite levels; these are reviewed. Epidemiological studies of several groups thought to have an unusual extent or timing of light exposure, as for instance shift workers or blind women, are also reviewed for evidence that might be relevant to the melatonin hypothesis. None of these latter studies, however, has furnished any data directly on melatonin levels.

Chapter 6 considers whether EMFs can affect the risk of breast cancer. Previously, the Advisory Group (AGNIR, 2001) concluded that there was no convincing evidence that exposure to EMFs at levels likely to be encountered directly damaged DNA (ie was genotoxic) or that EMFs could bring about the transformation of cells in culture. EMFs were therefore considered unlikely to initiate cancer. Since the completion of that report, some *in vitro* studies specifically relevant to breast cancer have been published; these newer studies are reviewed.

Although limitations exist in using carcinogen-induced rodent models of breast cancer, they are a widely used assay, and a number of animal studies are described which have used such models to study the effects of magnetic fields on mammary tumour development.

The final part of Chapter 6 considers the relation of EMF exposure to risk of female breast cancer. This includes the studies of breast cancer risks in women in relation to residential proximity to electricity transmission power lines, and to use of electric blankets.

The principal conclusions of the Advisory Group are given in Chapter 7, and the report finishes with recommendations for further research in Chapter 8. A glossary of less familiar scientific terms is also included.

1.3 References

- AGNIR (2001). ELF electromagnetic fields and the risk of cancer. Report of an Advisory Group on Non-ionising Radiation. *Doc NRPB*, **12**(1), 1–179.
- Ahlbom A, Day N, Feychting M, Roman E, Skinner J, Dockerty J, Linet M, McBride M, Michaelis J, Olsen JH, Tynes T and Verkasalo PK (2000). A pooled analysis of magnetic fields and childhood leukaemia. *Br J Cancer*, **83**(5), 692–8.
- Arendt J (1995). Melatonin and the Mammalian Pineal Gland. London, Chapman Hall.
- Cohen M, Lippman M and Chabner B (1978). Role of pineal gland in aetiology and treatment of breast cancer. *Lancet*, **2**(8094), 814–16.
- IARC (2001). Non-ionizing radiation, Part I: Static and extremely low frequency electric and magnetic fields. IARC Monographs on the Evaluation of Carcinogen Risks to Humans, Volume 80. Lyon, World Health Organization/International Agency for Research on Cancer.

Stevens RG (1987). Electric power use and breast cancer: a hypothesis. Am J Epidemiol, 125, 556-61.

2 Sources, Measurements and Dosimetry

The aim of this chapter is to describe the main physical characteristics of power frequency electromagnetic fields (EMFs), the common sources of exposure, and to consider aspects of exposure that might be relevant to the melatonin hypothesis. Exposures at work and in the home are considered at the end of the chapter.

2.1 Characterisation of power frequency EMFs

Power frequency EMFs consist of electric and magnetic fields. The electric field describes the force created by electric charges, and the magnetic field describes the force caused by moving charges in the form of electric current. In the UK the fields alternate at a frequency of 50 Hz, the frequency of the electricity supply. The unit of electric field strength is normally volts per metre (V m⁻¹) or kilovolts per metre (kV m⁻¹). Magnetic field strength is usually measured in units of ampere per metre (A m⁻¹). A related quantity is magnetic flux density, measured in tesla (T), millitesla (mT), microtesla (μ T) or nanotesla (nT). The other unit frequently seen in the literature is milligauss (10 mG = 1 μ T).

EMFs are vector quantities, with a direction and magnitude. The fields can be linearly or elliptically polarised, depending on the arrangement of source current. Figure 2.1 shows the polarisation ellipse, which describes the rotation of the magnetic flux density vector (**B**) generated by sources of electric current. A single source current produces a linearly polarised field vector that oscillates in a discrete direction in space. Several current sources with different phases produce an elliptically polarised vector, which rotates in space. The fields in homes tend to have limited elliptical polarisation (Swanson, 1999). Those from power lines, however, have varying degrees of ellipticity depending on the source current arrangement.

In general, the electricity supply produces undistorted sinusoidal waveforms. However, in some situations the EMFs may contain additional frequency components called harmonics, which are multiples of the fundamental frequency.

2.2 Sources of power frequency EMFs

The sources of power frequency EMFs are divided broadly into those produced by natural processes and those generated by human activity. Naturally occurring EMFs arise from electrical processes associated with the Earth and the atmosphere. In most environments the dominant source of exposure is that associated with the generation, transmission and use of electricity. People are exposed directly through the use of electrical appliances or equipment, or incidentally through working close to heating systems and power supplies.





2.2.1 Natural sources

2.2.1.1 Electric fields

Processes in the atmosphere and magnetosphere produce a wide range of signals with frequencies reaching up to several megahertz (1 MHz = 10^{6} Hz). Extremely low frequency variations arise mainly from the effects of solar activity in the ionosphere and atmospheric effects such as lightning discharges that cause the resonant oscillations in the Earth–ionosphere cavity known as Schumann resonances. The natural electric field strength at the power frequency of 50 or 60 Hz is about 10^{-4} V m⁻¹ (EC, 1996).

2.2.1.2 Magnetic fields

The Earth's magnetic field changes continually at periods ranging from a few milliseconds up to 10^{12} years. The Schumann resonances produce magnetic fields around 10^{-5} µT at frequencies of 6–60 Hz. The measurement of signals with frequencies below 100 Hz is extremely difficult because of the interference from man-made signals. At 50/60 Hz the natural magnetic field is typically around 10^{-6} µT (Polk, 1974).

2.2.2 Man-made sources

The intensities of man-made EMFs at power frequencies usually far exceed those produced naturally. The principal sources are high voltage transmission lines, and electrical appliances and equipment used in industry and the home. Although the demand for electrical devices in modern society has led to increasing use of electricity, exposures have not necessarily increased in the same way, and in some situations new technologies have resulted in lower exposures.

2.2.2.1 Electric fields

High voltage (HV) power lines produce some of the largest electric fields encountered in the environment. In the UK, HV transmission lines operate principally at voltages of 400 kV and 275 kV and the distribution lines work at voltages from 132 kV down to 0.4 kV.

The strength of the electric field produced by a power line depends primarily on the operating voltage, the conductor size and geometry, the number of circuits, the phase arrangement, and the distance from the line. Maximum field strengths up to several kilovolts per metre are encountered at ground level below the highest voltage lines. For example, a 400 kV dual circuit transmission line can produce 11 kV m⁻¹ near to ground level at the lowest point of the conductors; however, in practice, due to conservative design margins, the levels are usually no more than a few kilovolts per metre. The fields decay with increasing distance, depending on how the phase conductors are arranged. Electric field strengths of a few hundred volts per metre are encountered a few tens of metres from an HV transmission line, decreasing to a few tens of volts per metre at 100 m. In general, the fields become weaker as the operating voltage decreases; field strengths near the ground beneath distribution lines range typically from a few tens to a few hundreds of volts per metre.

The strongest electric fields inside homes are usually encountered close to the surfaces of domestic appliances, light fixtures and supply wiring. Electric field strengths up to several kilovolts per metre have been measured at the surface of certain appliances, although maximum field strengths of no more than several hundred volts per metre are more typical. The high fields near to appliances are usually very localised and field strengths weaken typically to a few tens of volts per metre at distances of a few tens of centimetres. Appliances with electric motors produce a broad spectrum of signals, with typical emission maxima near to 10-20 kHz and field strengths around 20 V m^{-1} (EC, 1996).

Electric fields are readily perturbed by conducting objects, including building materials and people, and the levels encountered in homes are very variable. The electric field strength inside homes away from appliances and electrical wiring is typically in the range $0-20 \text{ V m}^{-1}$ (Swanson, 1999). In a pilot study carried out as part of the UK Childhood Cancer Study (UKCCS), arithmetic mean values of 13 V m⁻¹ (standard deviation, SD 14.2) and 10.5 V m⁻¹ (SD 13.1) were recorded for the bed and centre of the family living room, respectively (UKCCS Investigators, 2002). Inside buildings near to transmission lines, the electric fields are heavily attenuated, usually by a factor of between 10 and 1000 depending on the structure of the building. In these situations, the highest fields associated with the line, usually encountered close to open windows, are typically of the order of a few hundred volts per metre.

Electricity substations are another source of electric fields, although those encountered at the boundaries of substations are usually very weak due to effective screening. These are less than a few volts per metre

and certainly no more than a few hundred volts per metre near the largest installations. Further away from substations, connecting overhead power lines become the main source of electric field exposure. Electric fields from buried cables are negligible due to the shielding provided by the cable sheath and ground material (Maslanyj, 1996).

2.2.2.2 Magnetic fields

Common sources of exposure

There are many different man-made sources of magnetic fields in the environment. Figure 2.2 shows the pattern of magnetic field in a typical home in the UK (Swanson, 1999). The background levels generally arise from power lines and net currents in distribution circuits and conducting services (Merchant et al, 1994; Swanson, 1999). In some circumstances, currents flowing in the wiring inside the home can be important (Maslanyj et al, 2005).

Some of the largest magnetic fields encountered in homes are produced by appliances, and the fields they produce have been reviewed extensively elsewhere (Gauger, 1985; Preece et al, 1997; Maslanyj and Allen, 1998; AGNIR, 2001; ICNIRP, 2003). The strength of the field from an appliance is determined by the magnitude of the current used, the size and shape of conducting parts, the number of turns of wires in coils, and whether any shielding has been included in the design. Magnetic fields up to a few thousand microtesla can occur close to the body when using small appliances such as electric shavers. More



FIGURE 2.2 Power frequency magnetic field measured on a 1 m grid at 1 m above ground floor level in a typical home in the UK. After Swanson, 1999

typically, maximum fields are rarely more than a few hundred microtesla and the levels usually fall with distance, becoming negligible within 1–2 m or so. Large electrical appliances often produce smaller but more extensive fields because of the position and structure of the field source, usually the transformer, inside the equipment. Preece et al (1997) reported mean magnetic fields of 0.26 and 0.07 μ T at 50 cm and 1 m, respectively, in front of TV sets in the UK.

HV power lines are potentially important sources of magnetic fields in homes located near to them. The main determinants of the strength of the field are the load current(s), the relative phasing of circuits, the spacing of conductors, and the distance from the line. Maximum fields up to a few tens of microtesla can occur close to the ground beneath the largest HV lines operating at maximum rating in the UK. However, because in practice they normally operate below rated conditions, the levels actually encountered rarely exceed more than a few microtesla. The fields weaken with distance to give typical values of a few tenths of a microtesla at distances of several tens of metres from the lines. In general, the maximum field decreases as the operating voltage decreases, because currents and conductor separations become progressively smaller.

HV underground cables can result in larger accessible magnetic fields than overhead power lines because they can be approached more closely at ground level. A buried 400 kV cable operating at 2 kA can produce fields in excess of 100 μ T at 1 m above ground level directly over the cable. However, because the phase conductors are spaced more closely than for the equivalent overhead line, the field falls more rapidly, approaching background levels within a few tens of metres, and the region of elevated field associated with the cable is more restricted. The HV cables that operate at lower voltages tend to have phase conductors that are enclosed in a single metallic casing, which produce much weaker fields due to more efficient field cancellation – usually no more than a few tenths of a microtesla close to the ground.

Electricity substations are a common source of magnetic fields. Maximum fields of a few microtesla have been encountered at the boundary of the local area substations that serve residential areas in the UK, and the levels fall rapidly to become indistinguishable from normal domestic background, usually within a few metres. The maximum fields tend to occur opposite the feed pillar, transformer and switching units of the substation, and continue along the connecting power cables and lines (Maslanyj, 1996).

Sources of exposure in the workplace

Sources of exposure to power frequency EMFs in the workplace have been reviewed extensively elsewhere (Allen et al, 1994; AGNIR, 2001; Cooper, 2002; ICNIRP, 2003).

The main occupational environment where strong electric fields are encountered is the power industry, and particularly in electricity generation and transmission. Electric field strengths in the range of a few volts per metre to several kilovolts per metre have been reported close to electricity generating equipment at power stations in the UK (Cooper, 2002); higher levels may be encountered at specific locations in substations.

In general, data on electric fields in the workplace are sparse because many industrial processes rely on large currents rather than high voltages; thus most of the interest has been in magnetic fields. The industries most commonly associated with high magnetic fields are the electrical utilities, electrically powered transport systems, and those where the fields generated are integral to the processes used in the electrochemical, welding and induction heating industries (ICNIRP, 2003).

The main sources of magnetic fields in power stations are the HV power line conductors, where fields in excess of 1000 μ T have been reported. Bus bars and switchgear produce a few tens to a few hundred microtesla, and generators and alternators at power stations produce fields between a few tens and a few hundred microtesla (Cooper, 2002).

Mainline electric trains in the UK have an overhead 25 kV, 50 Hz supply, and on-board rectification can give rise to alternating components in the static or 'quasi-static' field in the traction parts of the train. Magnetic fields up to 50 μ T have been measured on a train, most likely related to the layout of the auxiliary power supplies (Allen et al, 1994).

The rectification processes used in the electrochemical industry can produce harmonics up to several hundred hertz at multiples of the power frequency. Fields up to several hundred microtesla have been measured close to rectifier banks and ambient levels of several microtesla have been measured in the areas around cells.

In the UK, AC welding equipment usually operates at 50 Hz. Magnetic fields of more than 1000 μ T have been measured at the surface of a welding cable and in excess of 100 μ T close to the power supply, and harmonics are often present (Allen et al, 1994).

Induction heaters operating at frequencies of a few tens of hertz are used for volume heating of metals. Power frequency fields of several hundred microtesla have been reported close to a 6 MW copper billet (Allen et al, 1994). Arc furnaces that are used for processing metal can operate at fundamental frequencies of a few tens of hertz and waveforms can have significant harmonic components. Fields of a few hundred microtesla have been measured within a few metres of this type of furnace (Cooper, 2002).

There is a range of other industrial sources that produce power frequency magnetic fields. Pulsed magnetic fields are used commercially to permanently magnetise small ferromagnetic components. The procedure often relies on charging and discharging through a coil, and the pulse produces an infinite range of frequencies. Peak fields of several millites can be produced at the coil, falling to several hundred microtes within a few centimetres (Cooper, 2002).

Crack detection systems use AC or DC power supplies, and the latter can produce a rectified current that may contain harmonics of the fundamental power frequency. The current is delivered continuously or in pulses. Some systems incorporate a demagnetising function whereby an alternating current is gradually reduced to zero amplitude over a period of time. Static and time-varying fields of a few tens of millitesla occur at positions occupied by personnel whilst operating the equipment (Cooper, 2002).

Tape erasers, or degaussers, typically use a magnetic field to erase data stored on tapes or other magnetic media. Maximum magnetic fields of a few millitesla can occur at a few centimetres from such devices, falling to a few hundred microtesla within a few tens of centimetres, and falling again to several tens of microtesla at the trunk of the operator (Allen et al, 1994).

The magnetic fields produced by electric motors can be quite large, occur over a wide range of frequencies, and exhibit great spatial variation. Magnetic fields up to a few millitesla have been measured at the surface of electric motors, falling to a few microtesla at distances of a few tens of centimetres (Maslanyj and Allen, 1998).

2.3 Exposure assessment

An exposure assessment might be carried out for a variety of reasons, including the testing of compliance with exposure guidelines, identifying a hazard in an occupational setting, or for research purposes. The main design aspects of an assessment are summarised in Figure 2.3.



FIGURE 2.3 Main design aspects of an EMF exposure assessment

2.3.1 Dosimetry

The first consideration in an exposure assessment is to determine what physical quantity inside the body is of interest. Ideally this is the dosimetric quantity that can be related directly to a biological effect. Under existing guidelines from the International Commission on Non-Ionizing Radiation Protection (ICNIRP), the dosimetric quantity for EMFs is the induced current density (ICNIRP, 1998), although there is a growing view that the appropriate quantity should be induced electric field strength (NRPB, 2004). Numerical modelling methods are used to link the internal dosimetric quantity and external magnetic and electric field strengths. These external field quantities are relatively easily measured and used in the first stage of assessing compliance with maximum guideline values.

An exposure metric is a numerical quantity that is used to summarise a particular characteristic of EMF exposure. The metric may combine frequency, time and spatial parameters of the field to which people are exposed, to produce a single summary measurement. In the case of the ICNIRP guidelines the important exposure quantities are the unperturbed root-mean-square (RMS) electric and magnetic field strengths.

It is not known what aspect of exposure (if any) is most directly related to health outcomes, and various other exposure metrics have been used in epidemiological studies, the most common being the peak, cumulative and time weighted average (TWA) exposure – the RMS field value averaged over time. In relation to appliances, the peak exposure will depend on the proximity of the particular source in use, whereas the TWA will tend to reflect the extent and frequency of use. Alternative metrics include the median or geometric mean used to determine the background level when exposures are log normally distributed, time above threshold representing prolonged periods of high field exposure, and others reflecting short-term variations. Some of the alternative metrics may be related closely to the TWA but this is not always the case.

Modern instruments have enabled more sophisticated metrics to be evaluated, reflecting field characteristics such as frequency, harmonic content, ellipticity, direction in relation to the Earth's static field, time variation, rate of change with time (d**B**/dt), and frequency and magnitude of transients.

Transients are rapidly changing signals that tend to appear at irregular intervals. They contain a broad range of frequencies that may extend into the megahertz range (NRC, 1997). Transients occur in all environments where switching of electrical appliances or equipment occurs and, because they are characterised by a high rate of change of field, they may induce large current densities within individuals (Kaune et al, 1997; Kavet, 1998).

Relatively few studies have assessed exposure to electric fields, partly because of the difficulty of making electric field measurements that properly take into account the perturbing effect of the bodies of subjects (Kaune and Gillis, 1981; Deno and Silva, 1984; Chartier et al, 1985; Kaune et al, 2002). In the UKCCS pilot study of electric fields, the measurements of field strengths were made in the absence of the perturbing effect of people (UKCCS Investigators, 2002).

2.3.2 Measurement procedures

2.3.2.1 Source characterisation

It is important to consider the source operating characteristics in the development of an exposure assessment. Knowledge of the source, and how it is used, facilitates the selection of appropriate instrumentation and the measurements that are necessary. Close to many sources the fields are inhomogeneous and assessing them may require complex measurement procedures and calculations. In some situations, particularly in occupational settings, it may be necessary to capture the waveform in order to identify the range of frequencies that is likely to be encountered. Information about the design and operation of appliances and equipment can usually be obtained from the manufacturer or operator.

2.3.2.2 Instrumentation

The development of sophisticated measurement instruments has played an important role in advancing exposure assessments in EMF research. Some of the meter parameters that need to be taken into account are frequency response, sampling rate, dynamic range, and accuracy. Meters are normally calibrated to a traceable standard and reliability checks may be required during the course of the assessment. Modern meters usually incorporate a digital computer and signal processing circuitry, which allow different metrics to be computed from time-series data, and information about waveforms may also be recorded (Bowman et al, 1998).

Electric field meters

Most electric field meters measure the induced current flowing between two electrically isolated metal plates or shells that are placed in the field. Modern sensors measure simultaneously the RMS orthogonal components of electric field strength, and these may be logged for a specific time interval.

Measuring electric fields presents unique difficulties and the measurements are also difficult to interpret (Kheifets et al, 1997). Common objects in the environment perturb the field. A personal exposure meter will reflect the perturbation of the field by the body. The field recorded is very dependent on where the device is worn, the posture of the subject, and the relative location of any sources. The meter itself perturbs the field and this has to be taken into account in the design, calibration and use of the meter.

Careful procedures are required to ensure that people carrying out the measurements do not affect the sensor.

Magnetic field meters

Magnetic fields are usually measured using coils that are shielded from the effect of electric fields. The field strength is normally determined by measuring the currents induced in three orthogonally mounted sensor coils. In contrast to electric fields, any field perturbation is minimal.

2.3.2.3 Sampling strategy

The main purpose of an exposure assessment in epidemiological studies is the classification of study subjects into groups with differing exposures. Exposure is most often characterised either by indirect surrogates of TWA field strength or by direct measures of the field averaged over periods of time typically from a few seconds to many hours (Kaune et al, 2001). In general, two broad measurement approaches are available: spot measurements, in which field meters are placed in locations to reflect those occupied by the subject, and personal monitoring, in which exposure meters are worn by the subject.

The use of spot measurements in epidemiological studies should be approached carefully. Whereas appropriate weighting of spot measurements for children has been found to yield good correspondence with personal exposure (Kaune et al, 1994), a similar approach in a study of adults has explained less than half the variability in the exposure data (Kavet et al, 1992).

Personal exposure meters integrate the general background field with the other fields that might be encountered by the subject whilst performing activities, such as using an electrical appliance. The time variation observed in the data is due both to the temporal variation of the field and to the spatial variation as the subject moves from one location to another. Careful account needs to be taken of the periods when the meter is not worn, the position of the meter on the body, and the variation of the field outside the record period.

A key challenge in most of the studies is how best to use contemporaneous measurements to assess the exposure incurred during some predefined period usually before the diagnosis of disease. An important question is whether a subject's behaviour or work pattern is likely to have changed as a result of the disease, and whether the measurements are likely to reflect previous exposures. Control subjects, if they are more active than cases, could have higher exposures recorded on personal meters, or alternatively cases could be assigned higher exposures if they are less active in a high field area. The measurements may be supplemented with information on lifestyle and working patterns, obtained from questionnaires, which can be used to refine the exposure measure.

One of the limitations of the measurements used in epidemiological studies is that they are acquired over a time interval that may not necessarily characterise the full range of field variation. Long assessment periods tend to increase the cost of the study, consequently a number of studies have utilised surrogates of exposure, relying on exposure conditions being predictable in certain environments. The wire-code surrogate, used widely in the USA (NRC, 1997), is based on power line conductor configuration and distance from the subject's home. The job title surrogate used in

occupational studies relies on the exposure being predictable for certain job types. A number of epidemiological studies have used the load-current data usually available for HV power lines to predict historical exposure.

2.3.2.4 Time variation of power frequency EMFs

An important aspect to consider in relation to the melatonin hypothesis, is the variation of magnetic field strength in time. The variation of the background magnetic field in homes generally follows a diurnal pattern related to electrical power consumption (Silva et al, 1989; Dovan et al, 1993; Kaune et al, 2001; Banks et al, 2002). In general, the highest field levels are observed in the evening and the lowest during the night (Figure 2.4). Banks et al (2002) carried out a detailed investigation of residential 60 Hz magnetic fields in two metropolitan areas of the USA. Measurements over 24 hours were repeated on a regular bi-monthly schedule over a period of one year. A systematic and appreciable diurnal effect was found, suggesting that evening spot measurements overestimate the long-term exposure by 20% or more. The study concluded that 24 hour measurements are likely to produce a fairly reliable estimate of exposure. However, because the study was based on spot rather than personal exposure measurements, the results apply strictly to people who are residentially stable. Ideally, assessments in the home should be made over at least 24 hours (Swanson, 1996). The detailed pattern is not always easy to predict because the background field arises mainly from net currents in the local supply circuits.

Magnetic field levels in homes are not necessarily low overnight, and there are a number of situations where high night-time exposure could occur:

- (a) in homes close to electricity cables that supply night-time industry, or in areas where low tariff energy is used overnight,
- (b) in homes where energy is consumed overnight through the use of night storage heaters or underfloor heating,
- (c) through the overnight use of electric bed-warming appliances such as electric blankets.

An example of night-time elevation of the magnetic field in a home that uses night storage heaters and off-peak demand electricity is shown in Figure 2.5. Many underfloor heating systems draw current during the night, relying on off-peak electricity and the heat capacity of the floor to provide warmth during the day.

Kaune et al (2001) found that residential magnetic fields measured in bedrooms showed no strong weekday–weekend dependence; however, personal exposure data showed that adults at work or out of the house on weekdays incurred higher personal exposures than on weekends.

Seasonal changes in residential magnetic fields have been observed in a number of studies. Banks et al (2002) found a small but significant seasonal effect (-3% to +4%) in homes in the USA. Kaune et al (2001) found that bedroom magnetic fields, and to a lesser extent personal exposure, were substantially larger in winter than summer, suggesting that epidemiological studies should control for the date when measurements are made. The seasonal maximum fields usually coincide with the time of maximum electricity usage, normally in the winter. Deadman et al (1999) made 48 hour personal exposure measurements on 382 Canadian children in five provinces, and found that arithmetic mean exposures



FIGURE 2.4 Typical diurnal variation of the power frequency magnetic field in a home. Values represent hourly arithmetic mean percentage change from the long-term exposure estimate and 95% confidence intervals. Estimates based on spot measurements of the magnetic field in 51 homes. After Banks et al, 2002



FIGURE 2.5 Variation of the magnetic field in a home that uses night storage heaters and low tariff electricity. The resultant field in the broadband frequency range (40–800 Hz) was measured in a bedroom away from operating appliances. Data from the UKCCS Investigators, by permission



FIGURE 2.6 Seasonal variation of the background magnetic field in a UK home. Values represent the 24 hour average field over a period of one year. After Swanson and Renew, 1994

were higher in winter (November–March; 0.137μ T; 95% confidence interval, Cl, $0.114-0.160 \mu$ T) than in summer (April–October; 0.109μ T; 95% Cl $0.096-0.123 \mu$ T). In the UK, background fields in homes vary seasonally (Figure 2.6), broadly following the annual variation of load on the relevant 415 V distribution circuit (Swanson and Renew, 1994).

The load currents on HV power lines have been shown to exhibit diurnal, weekday–weekend, seasonal and other variations, suggesting that contemporaneous short-term measurements in homes near power lines are unlikely to provide good estimates for historical exposure (Kaune et al, 1998).

With the development of logging meters, there has been interest in the short-term temporal variability of magnetic fields and related exposure metrics. In the study carried out by Kaune et al (2001), the short-term variability in personal exposure data was attributable predominantly to the physical movement of a subject through a spatially varying field. The lowest variability in personal exposure was observed when a subject was sleeping, even when night-day temporal variability was similar. The study concluded that spot measurements were unlikely to provide reasonable surrogates for the temporal variability of exposure.

The temporal changes of electric fields in the home are less well documented. In general, because of the voltage stability of the power supply, electric fields vary less than magnetic fields. The changes that do occur depend on the use of specific appliances and circuits inside the home, which are difficult to predict. The average electric field levels in the UKCCS homes apparently varied little over the day, based on measurements of the vertical field component (UKCCS Investigators, 2002).

2.4 Exposure environments

2.4.1 Residential exposure

The background magnetic field levels encountered in UK homes typically fall in the range from 0.01 to 0.1 μ T, and average residential exposures above 0.4 μ T occur for about 0.5% of the population (AGNIR, 2001). The background field encountered in homes in most countries is usually less than 0.2 μ T (Swanson and Kaune, 1999).

HV power lines can be a major contributor to long-term TWA magnetic field exposure (Kavet et al, 1992; Merchant et al, 1994; Forssén et al, 2002), raising exposure levels typically by an order of magnitude. However, averaged over the whole population, HV lines contribute only a small fraction of the collective exposure – for example, 5% of total population exposure in the UK (Swanson, 1999). In a recent study based on the EMF part of the UK Childhood Cancer Study (UKCCS Investigators, 1999; Maslanyj et al, 2005), less than 2.5% of homes had average exposures above 0.2 μ T, and HV power lines operating at 275 kV and above explained only 9% of these exposures. Low voltage (LV) sources including currents associated with the supply, and sources inside the home including suspected wiring faults, accounted for most of the exposures. House wiring may produce elevated fields inside the home under certain conditions, for instance, if field cancellation is imperfect, either because live and neutral wires are not close together or because of imbalance between live and return currents.

The contribution of household electrical appliances to exposure is an important factor to consider in the design and evaluation of studies (AGNIR, 2001). Most domestic appliances will not give rise to appreciable additional TWA exposure, because the fields they produce are very localised and the periods of use are relatively short (Delpizzo, 1990).

The proportional contribution of appliances to TWA exposure varies depending on the background field. In the UK, Swanson (1999) found that appliances contribute on average up to one-third of exposure, and between 3% and 50% for the majority of homes. In a study conducted by Mezei et al (2001), mean magnetic fields from various domestic appliances correlated only weakly with mean daily exposure, suggesting that studies focusing on a single appliance or a small number of appliances are unlikely to give a good indication of TWA exposure. The same appliances, however, could be significant determinants of peak and above-threshold exposures.

Appliances can be the dominant source of exposure to body extremities (Mader and Peralta, 1986). However, in the absence of an organ-specific effect, exposure tends to be estimated from fields averaged over the entire body. Fields of a few tens of microtesla reduce to a few tenths of a microtesla or even a few hundredths of a microtesla when averaged over the whole body. The contribution of an appliance field to exposure is often further reduced when averaged over time. Hairdryers can produce intense magnetic fields at 10 cm from the head, compared with ambient levels; however, average use of a few minutes per day means they contribute less than 5% to the average daily exposure (Kaune et al, 2002). Home sewing machines produce magnetic fields in the abdominal area of the body that are elevated by a factor of nearly three over the ambient level; however, based on average daily use, they contribute only about 2% to the daily exposure (Kaune et al, 2002). Kaune et al (2000a) found no statistically significant differences between magnetic fields calculated at each distance for TV sets used for viewing programmes and TV sets used for playing video games. However, children's exposure levels were larger, by a factor of three, when TV sets were used to play video games rather than watch TV programmes. The major reason for the difference was children sat on average about 75 cm closer to the sets while playing games.

In relation to the melatonin hypothesis, certain appliances, such as electric blankets, night storage heaters and underfloor heating, can contribute significantly to overnight exposure, depending on design and pattern of use (Delpizzo, 1990). Electric blankets operating at 240 V can produce magnetic fields averaged in the region of the body of 0.2 μ T for underblankets, and 0.25 μ T for overblankets, and the fields are likely to be 2 to 2.5 times higher for appliances designed to operate at 110 V. Underblankets are insignificant sources of cumulative exposure under most realistic conditions of use; however, overblankets designed for all-night use may result in substantial exposure. For water bed heaters, which are designed for all-year-round use, the exposure can vary greatly, depending on the size and position of the heating element. The heaters that are located directly under the user, common in the older designs, can produce body averaged fields of 0.4–0.5 μ T (Delpizzo, 1990). Underfloor heating systems operate by way of electric coils that are embedded in floors and connected during off-peak periods, usually overnight. The magnetic fields associated with underfloor heating depend on the configuration and depth of the cables, and the current flowing in them. Typical magnetic fields up to 1.5 μT can occur at floor level falling to a few tenths of a microtesla at 1 m above the floor (NRPB, 2004). In the residential study reported by Delpizzo (1990), concrete slab coil heaters operating at 240 V produced whole-body average fields on a bed of $1-6 \mu$ T; a standing adult would be exposed to a whole-body average field of up to 6.5 μ T, and a child playing on the floor would be exposed to an average field of 15 μ T or more. Systems operating at commercial premises can give up to a few hundred microtesla at floor level falling to a few tens of microtesla at 1 m above the floor (ICNIRP, 2003). As with electric blankets, the TWA exposure will depend on pattern of use.

2.4.1.1 Other frequencies

The frequencies of the ambient magnetic fields in homes are usually the dominant 50/60 Hz component and odd-harmonics whose strengths grow progressively weaker with increasing frequency (Kaune et al, 2000a). For most residential sources, the strengths of the even-harmonic components are markedly smaller than those of the odd-harmonic components (Zafanella, 1993). Some appliances produce different frequency spectra from the ambient fields in homes. TV sets, in particular, contain evenharmonic components (Kaune et al, 2000a), notably the 120 Hz related to the vertical deflection coils and frequency components in the very low frequency range related to the horizontal deflection coils.

There has also been interest in the transient magnetic field events caused by electrical switching in residential and occupational environments. Transients, because of their high frequency content, induce considerably stronger instantaneous currents and electric fields inside the body than similar magnitude power frequency magnetic fields (Kaune et al, 1997). In a study of magnetic transients in the home reported by Kaune et al (2000b), the diurnal pattern of transient activity was similar to that observed for power frequency magnetic fields.

2.4.2 Workplace exposure

Exposure in the workplace has been considered in detail by AGNIR (2001). Most of this information comes from industries where high levels of exposure have been anticipated due to the presence of high

voltages or currents. Studies have also been conducted where health concerns have been expressed about particular appliances such as sewing machines (Kelsh et al, 2003). Most of the information available concerns magnetic fields.

Occupational magnetic field exposure can be more difficult to characterise than residential exposure. The sources may produce a range of field strengths and frequencies, and the extent and pattern of exposure can be complicated. The fields produced by industrial processes can be very large, vary in space markedly and change during operation. Spot measurements may be of limited use in assessing exposure in these situations and personal exposure measurements may provide a better alternative.

Sources such as transformers and electric motors generate fields that tend to decrease with the cube of the distance from the source. The exposure of some organs may be substantially different from that measured by a meter worn at a convenient position. Delpizzo (1990) showed that hip-worn meters consistently underestimate both whole-body average exposure and head exposure. Chest-worn meters were found to provide a generally better measure of whole-body exposure. Exposure may also be difficult to define unless the subject's position is reasonably well described by the task undertaken. Whilst there may be the potential for the maximum exposure of a working subject to be high, the TWA exposure will be determined to a great extent by actual working practice (Kelsh et al, 2000).

The magnetic fields produced by industrial equipment may have non-sinusoidal waveforms and different spectral components to the main power supply. Harmonics can be generated by transformer and motor cores operating under high load conditions, thyristor control of power to equipment, ill-matched oscillator circuits, and AC supply rectification. Pulsed magnetic fields and transients are difficult to measure: the wide frequency range associated with the signals might exceed the instrument bandwidth, and if the instrument's response time is greater than the duration of the pulse or transient, the instantaneous maximum field strength may be much greater than the indicated reading. In such environments it is important to characterise the waveforms and spectral content fully so that appropriate measurement instrumentation can be used (Chadwick, 1997).

Various surrogate schemes based on job titles or occupational task have been used for exposure classification purposes. Recent studies have tended to rely on measurement data linked to job titles, whereby measurements are summarised by occupational categories which represent similar groups of job titles (Kelsh et al, 2000). The measurement studies suggest that job title might not necessarily reflect the true exposure, and may be appropriate only for a few working practices found in certain industries. Large variations in exposure can occur within the same job category, probably due to specific differences in the processes and work practices (van Tongeren et al, 2004). In some cases the work environment is at least as important as the nature of the work. For instance, consistently high exposures might be encountered in a power distribution utility, regardless of the job concerned (Kelsh et al, 2000).

A number of exposure assessment studies of power industry staff have employed integrating exposure meters to estimate mean exposure. In the UK electricity supply industry, Merchant et al (1994) found that transmission staff at substation sites or working on lines encountered the highest fields (geometric mean TWA of 1.16 μ T). Power station and distribution workers incurred exposures in the geometric mean TWA range 200–500 μ T. Magnetic field exposure has been investigated for a range of industries in a study carried out as part of a UK adult brain tumour study (van Tongeren et al, 2004). Occupational

magnetic field exposure was found to be the main determinant of the overall TWA exposure, which included periods at home and travelling. In many modern office-based environments the average background levels might be similar to residential levels. However, field levels up to several microtesla can be encountered close to copy machines, and prolonged use of such equipment and of visual display units and fax machines can increase long-term exposure.

In relation to the melatonin hypothesis, very few studies have included comprehensive exposure assessments that take into account the timing of exposure. Consideration of exposure during shift work has rarely been considered.

2.5 Summary

Power frequency EMFs are encountered everywhere electricity is used, in and outside the home and at work. The electric field describes the force created by electric charges, and the magnetic field describes the force caused by moving charges in the form of electric current.

The electric fields encountered in homes normally range in strength between zero and several hundred volts per metre. Electric field strengths up to several kilovolts per metre can be encountered close to the surfaces of household appliances, light fixtures and supply wiring. The domestic background level away from appliances and electrical wiring is typically in the range $0-20 \text{ V m}^{-1}$. Maximum field strengths up to several kilovolts per metre can be encountered at ground level under high voltage lines; however, because electric fields are easily screened, levels inside homes are very much lower.

The background magnetic field levels encountered in UK homes mostly fall in the range from 0.01 to 0.1 μ T and average domestic exposures at or above 0.4 μ T occur for about 0.5% of the population. These generally arise from power lines, and net currents in supply circuits and conducting services. Electrical appliances give rise to the largest magnetic fields encountered in homes. Nonetheless, most appliances do not give rise to appreciable exposures because the magnetic field they produce is localised and the periods of use are short. Maximum fields up to a few tens of microtesla could occur beneath the largest power lines at maximum rating; however, because the lines normally operate below maximum rated conditions, the levels encountered rarely exceed more than a few microtesla.

Information on electric fields in occupational environments is sparse because most industrial processes rely on large currents rather than high voltages; thus most of the interest has been in magnetic fields. The industries most commonly associated with high magnetic fields are the electrical utilities, electrically powered transport systems, and those where the fields are produced directly as part of an industrial process such as in the electrochemical, welding and induction heating industries.

Accurate exposure assessment is one of the main challenges of EMF research. The field levels vary in time and space and exposure can be represented by a number of different metrics. The important dosimetric quantity in the ICNIRP exposure guidelines, is induced current density, and the related exposure metrics are unperturbed RMS electric and magnetic field strengths. It is not known what aspect of exposure (if any) is most directly related to health outcomes, and a range of exposure metrics have been used. Modern instruments allow more sophisticated metrics to be evaluated, reflecting field characteristics such as frequency, harmonic content, ellipticity, field direction, and time variation. An important aspect of power frequency magnetic fields in relation to the melatonin hypothesis, is the variation in time, which

broadly follows the pattern of electricity use, and certain appliances, such as electric blankets, night storage heaters and underfloor heating, can contribute substantially to overnight exposure, depending on design and pattern of use.

2.6 References

- AGNIR (2001). ELF electromagnetic fields and the risk of cancer. Report of an Advisory Group on Non-ionising Radiation. *Doc NRPB*, **12**(1), 1–179.
- Allen SG, Blackwell RP, Chadwick PJ, Driscoll CMH, Pearson AJ, Unsworth C and Whillock MJ (1994). Review of occupational exposure to optical radiation and electric and magnetic fields with regard to the proposed CEC Physical Agents Directive. Chilton, NRPB-R265.
- Banks SR, Thomas W, Mandel JS, Kaune WT, Wacholder S, Tarone RE and Linet MS (2002). Temporal trends and misclassification on residential 60 Hz magnetic field measurements. *Bioelectromagnetics*, **23**, 196–205.
- Bowman JD, Kelsh MA, and Kaune WT (1998). Manual for measuring occupational electric and magnetic field exposures. US Department of Health and Human Resources. Cincinnati OH, National Institute for Occupational Safety and Health.
- Chartier VL, Bracken TD and Capon AS (1985). BPA study of occupational exposure to 60 Hz electric fields. *IEEE Trans Power Appar Syst*, **PAS-104**, 733–44.
- Cooper TJ (2002). Occupational exposure to electric and magnetic fields in the context of the ICNIRP guidelines. Chilton, NRPB-W24.
- Chadwick PJ (1997). Investigation of the suitability of EMDEX magnetic field dosemeters for assessment of the exposure to induction heating workers. Sudbury, HSE Report 128/1997.
- Deadman J-E, Armstrong BG, McBride ML, Gallagher R and Theriault G (1999). Exposures of children in Canada to 60 Hz magnetic and electric fields. *Scand J Work Environ Health*, **25**, 368–75.
- Delpizzo V (1990). A model to assess personal exposure to ELF magnetic fields from common household sources. *Bioelectromagnetics*, **11**, 139–47.
- Deno DW and Silva JM (1984). Method for evaluating human exposure to 60 Hz electric fields. *IEEE Trans Power Appar Syst*, **PAS-103**, 1699–706.
- Dovan T, Kaune WT and Savitz DA (1993). Repeatability of residential power frequency magnetic fields and wire codes. *Bioelectromagnetics*, **14**, 145–59.
- EC (1996). Non-ionizing radiation: sources, exposure and health effects. Luxembourg, European Commission, CE-92-95-449-EN-C.
- Forssén UM, Ahlbom A and Feychting M (2002). Relative contribution of residential and occupational magnetic field exposure over twenty-four hours among people living close to and far from a power line. *Bioelectromagnetics*, **23**, 239–44.
- Gauger JR (1985). Household appliance magnetic field survey. IEEE Trans Power Appar Syst, PAS-104, 2436.
- ICNIRP (1998). Guidelines for limiting exposure to time-varying electric, magnetic and electromagnetic fields (up to 300 GHz). *Health Phys*, **74**, 494–522.
- ICNIRP (2003). Exposure to static and low frequency electromagnetic fields, biological effect and health consequences (0–100 kHz) (R Matthes et al, eds). Oberscheleissheim, Germany, ICNIRP13/2003, www.icnirp.org.
- Kaune WT and Gillis MF (1981). General properties of the interaction between animals and ELF electric fields. *Bioelectromagnetics*, **2**, 1–11.
- Kaune WT, Darby SD, Gardner SN, Hrubec Z, Iriye RN and Linet MS (1994). Development of a protocol for assessing time-weighted-average exposures of young children to power frequency magnetic fields. *Bioelectromagnetics*, **15**, 33–51.
- Kaune WT, Gutman JL and Kavet R (1997). Comparison of coupling of humans to electric and magnetic fields with frequencies between 100 Hz and 100 kHz. *Bioelectromagnetics*, **18**, 67–76.

Kaune WT, Feychting M, Ahlbom A, Ulrich RM and Savitz DA (1998). Temporal characteristics of transmission-line loadings in the Swedish Childhood Cancer Study. *Bioelectromagnetics*, **19**, 354–65.

- Kaune WT, Miller MC, Linet MS, Hatch EE, Kleinerman RA, Wacholder S, Mohr AH, Tarone RH and Haines C (2000a). Children's exposure to magnetic fields produced by US television sets used for viewing programs and playing video games. *Bioelectromagnetics*, **21**, 214–27.
- Kaune WT, Bracken TD, Senior RS, Rankin RK, Niple JC and Kavet R (2000b). Rate of occurrence of transient magnetic field events in US residences. *Bioelectromagnetics*, **21**, 197–213.
- Kaune WT, Davis S, Stevens RG, Mirick DK and Kheifets L (2001). Measuring temporal variability in residential magnetic field exposures. *Bioelectromagnetics*, **22**, 232–45.
- Kaune WT, Miller MC, Linet MS, Hatch EE, Kleinerman RA, Wacholder S, Mohr AH, Tarone RE and Haines C (2002). Magnetic fields produced by hand-held hair dryers, stereo headsets, home sewing machines and electric clocks. *Bioelectromagnetics*, **23**, 14–25.
- Kavet R (1998). ELF magnetic fields, transients and TWA metrics. In *Exposure Metrics and Dosimetry for EMF Epidemiology* (AF McKinlay and MH Repacholi, eds). *Radiat Prot Dosim*, **83**, 29–40.
- Kavet R, Silva JM and Thornton D (1992). Magnetic field exposure assessment for adult residents of Main who live near and far away from overhead transmission lines. *Bioelectromagnetics*, **13**, 35–55.
- Kelsh MA, Kheifets L and Smith R (2000). The impact of work environment, utility and sampling design on occupational magnetic field exposure summaries. *Am Ind Hyg Assoc J*, **61**, 174–82.
- Kelsh MA, Bracken TD, Sahl JD, Shum M and Ebi KL (2003). Occupational magnetic field exposures of garment workers: results of personal and survey measurements. *Bioelectromagnetics*, **24**, 316–26.
- Kheifets LI, London SJ and Peters JM (1997). Leukaemia risk and occupational electric field exposure in Los Angeles County, California. *Am J Epidemiol*, **146**(1), 87–90.
- Mader DL and Peralta SB (1986). Residential exposure to 60 Hz magnetic fields from appliances. *Health Phys*, **51**(2), 215–25.
- Maslanyj MP (1996). Power-frequency electromagnetic fields associated with local area substations. Chilton, NRPB-M751.
- Maslanyj MP and Allen SG (1998). A review of electromagnetic fields associated with motorised appliances. London, HSE Contract Research Report 172.
- Maslanyj MP, Mee TJ and Allen SG (2005). Investigation and identification of sources of residential magnetic field exposures in the United Kingdom Childhood Cancer Study (UKCCS). Chilton, HPA-RPD-005.
- Merchant CJ, Renew DC and Swanson J (1994). Exposures to power frequency magnetic fields in the home. *J Radiol Prot*, **14**, 77–87.
- Mezei G, Kheifets LI, Nelson LM, Mills KM, Iriye R and Kelsey JL (2001). Household appliance use and residential exposure to 60 Hz magnetic fields. *J Exp Anal Environ Epidemiol*, **11**, 41–9.
- NRC (National Research Council) (1996). Possible health effects of exposure to residential electric and magnetic fields. Washington DC, National Academy Press.
- NRPB (2004). Review of the scientific evidence for limiting exposure to electromagnetic fields (0–300 GHz). *Doc NRPB*, **15**(3), 1–224.
- Polk C (1974). Propagation, amplitude and temporal variation of extremely low frequency (0–100 Hz) electromagnetic fields. In *Biologic and Clinical Effects of Low-frequency Magnetic and Electric Fields* (JG Llaurao et al, eds). Springfield, Thomas.
- Preece AW, Grainger P, Golding J and Kaune W (1997). Magnetic fields from domestic appliances in the UK. *Phys Med Biol*, **42**, 67–76.
- Silva M, Hummon N, Rutter D and Hooper C (1989). Power frequency magnetic fields in the home. *IEEE Trans Power Deliv*, **4**, 465–77.
- Swanson J (1996). Net currents in underground distribution circuits in the UK: implications for assessing magnetic field exposures. *J Radiol Prot*, **16**(4), 275–86.
- Swanson J (1999). Residential power-frequency electric and magnetic fields: sources and exposures. *Radiat Prot Dosim*, **83**(1–2), 9–14.
- Swanson J and Kaune WT (1999). Comparison of residential power-frequency magnetic fields away from appliances in different countries. *Bioelectromagnetics*, **20**, 244–54.
- Swanson J and Renew DC (1994). Power-frequency fields and people. Engin Sci Educ J, April, 71-9.
- UKCCS (UK Childhood Cancer Study) Investigators (1999). Exposure to power frequency magnetic fields and the risk of childhood cancer: a case-control study. *Lancet*, **354**(9194), 1925–31.
- UKCCS (UK Childhood Cancer Study) Investigators (2002). Exposure to power frequency electric fields and the risk of childhood cancer in the UK. *Br J Cancer*, **87**, 1257–66.
- van Tongeren M, Mee T, Whatmough PW, Broad L, Maslanyj M, Allen S, Muir K and McKinney P (2004). Assessing occupational and domestic ELF magnetic field exposure in the UK adult brain tumour study: results of a feasibility study. *Radiat Prot Dosim*, **108**, 227–36.
- Zafanella LE (1993). Survey of residential magnetic fields sources. Volume 1; Goals, results and conclusions. Report TR-102759-VI. Palo Alto CA, Electric Power Research Institute.

3 The Pineal Gland and Melatonin

A vast amount of data exists concerning the physiology of the pineal gland and its principal hormone, melatonin. Aspects of this literature are summarised and reviewed below. In particular, the characteristics of melatonin production and the factors that affect the synthesis and metabolism of melatonin are described, with emphasis placed on effects of the light–dark cycle. The interactions of melatonin with other hormones, and possible mechanisms whereby melatonin may influence cellular physiology, are also described.

3.1 Basic physiology

3.1.1 Structure of the pineal

The pineal gland (epiphysis cerebri) is a small, unpaired central structure, essentially an appendage of the brain. Great variation in size and position is seen even within species. In humans the pineal weighs around 100 to 150 mg. It assumes a shape resembling a pine cone (hence pineal) and, again owing to its shape, has been referred to as the 'penis of the brain'.

The mammalian pineal gland is a secretory organ, whereas in fish and amphibians it is directly photoreceptive, and in reptiles and birds it has a mixed photoreceptor and secretory function. The extracranial parietal (parapineal, frontal) organ found in some lower vertebrates has been referred to as the 'third eye'. The principal cellular component is the pinealocyte, and elements of its photoreceptive evolutionary history remain in both structure and function. In some species, including humans, calcified lumps are frequently present in pineal tissue after puberty, although this calcification does not appear to be associated with a decline in metabolic activity except in the sense that activity declines in general with ageing. The gland is richly vascularised. Its principal innervation is sympathetic and arises from the superior cervical ganglion. In addition, good evidence has been presented for parasympathetic, commissural, and peptidergic innervation. Its primary function in all species studied to date is to transduce information concerning light–dark cycles to body physiology, particularly for the organisation of body rhythms. This information is encoded in the secretion patterns of the major pineal hormone melatonin (V-acetyl-5-methoxytryptamine) (Arendt, 1995; Klein et al, 1997; Korf et al, 1998).

3.1.2 Synthesis and metabolism of melatonin

Melatonin is synthesised within pinealocytes – cell types derived from photoreceptors – from tryptophan via the pathway shown in Figures 3.1 and 3.2. Most synthetic activity occurs during the dark phase, with a major increase (7- to 150-fold) in the activity of serotonin- Λ -acetyltransferase (arylalkylamine Λ -acetyltransferase – AA-NAT – commonly abbreviated as NAT). The rhythm of production is endogenous in that it is generated in the suprachiasmatic nuclei (SCN), the major central rhythm-generating system or 'clock' in mammals (the pineal itself is a self-sustaining 'clock' in some, if not all, lower vertebrates). The



FIGURE 3.1 Diagram of the major sympathetic innervation of the pineal gland (top) and the synthesis of melatonin from tryptophan (bottom). From Tamarkin et al, 1985, by permission



FIGURE 3.2 Sympathetic control of melatonin synthesis in the rodent pineal gland. From Ganguly et al, 2002, by permission

mammalian melatonin rhythm is generated by a closed loop negative feedback of clock gene expression in the SCN, *Clock* and *Bmal* being positive stimulatory elements, *Per and Cry* negative elements, subsequently influencing, clock-controlled genes (Figure 3.3). *Per* and *NAT mRNA* oscillate in the pineal, although post-transcription control is evident in some species. The rhythm is synchronised to 24 hours primarily by the light–dark cycle acting via the retina and the retinohypothalamic projection to the SCN. The cDNAs encoding both NAT and the *O*-methylating enzyme hydroxyindole-*O*-methyltransferase have been cloned, and studies of molecular regulation of melatonin production show some species differences. It is likely that the human enzyme is regulated primarily at a post-transcriptional level, whereas in rodents the key event appears to be cyclic adenosine monophosphate (cAMP)-dependent phosphorylation of a transcription factor that binds to the NAT promoter. Rapid decline in activity with light treatment at night appears to depend on proteasomal proteolysis of NAT following dephosphorylation and removal of a protective 14-3-3 protein (Klein et al, 2003; Simonneaux and Ribelayga, 2003).

Phosphorylation of the transcription factor CREB (cAMP-responsive element-binding protein) appears to be an important step in the signal transduction cascade that activates melatonin biosynthesis in the mammalian pineal organ. According to distribution studies of NAT mRNA, this enzyme is expressed in the pineal gland, retina and, to a much lesser extent, some other brain areas, the pituitary, and the testis, but, apart from the pineal, these structures contribute little to circulating concentrations in mammals. There is some evidence for high melatonin levels (one to three orders of magnitude higher than in plasma) and synthesis in bone marrow, the Harderian gland and the gut (Djeridane et al, 1998; Maestroni, 2000; Bubenik, 2002); however, it is possible that part of the melatonin found in these structures is derived from the pineal gland itself.

Within the rodent retina a self-sustaining 'clock' maintains rhythmic production of melatonin *in vitro* as it does in many lower vertebrates (Tosini and Menaker, 1996). Whether this pattern is true in humans remains to be seen.

Melatonin is metabolised primarily within the liver by 6-hydroxylation, followed by sulphate and/or glucuronide conjugation (Kopin et al, 1961; Arendt, 1995; Skene et al, 2001) (Figure 3.4). The principal metabolite in humans is 6-sulphatoxymelatonin (aMT6s), accounting for 50–80% of melatonin produced. A number of minor metabolites are also formed through ring splitting, cyclisation of the side chain, or demethylation (see Arendt, 1995, for a bibliography). In humans and rodents, exogenous oral or intravenous melatonin has a short metabolic half-life (20–60 minutes, depending on the author and species), with a large hepatic first-pass effect and a biphasic elimination pattern. In ruminants, longer half-lives are seen after oral administration.

3.1.3 Neural control of melatonin synthesis

In mammals, pineal denervation, or ganglionectomy, abolishes the rhythmic synthesis of melatonin and the light–dark control of its production. Noradrenalin is clearly the major transmitter and acts via β_1 -adrenoceptors with potentiation by α_1 -stimulation, but the role of neural serotonin is probably not negligible. A day–night variation is seen in pineal noradrenalin, with the highest values at night, approximately 180 degrees out of phase with the pineal serotonin rhythm. cAMP acts as a second messenger and stimulates NAT activity. β -adrenergic receptor-binding sites in the rat pineal vary over a 24 hour period, the lowest number being found toward the end of the dark phase and increasing shortly after lights on. A recent comprehensive review addresses this field (Simonneaux and Ribelayga, 2003).



FIGURE 3.3 Generation of circadian rhythms: clock gene feedback loops. A simplified model of the molecular clockwork of the circadian clock in mammals. A general feature is feedback of clock gene proteins on clock gene promoters. Inhibition of transcription of RNA stops the circadian clock, and inhibition of translation of proteins stops the circadian clock. BMAL1 and CLOCK proteins together activate the expression of clock genes (and clock-controlled genes) in the promoter of these genes. Both mRNA and protein products of Period (Per1, 2, 3) and Cryptochrome (Cry1, 2) genes oscillate over the 24 hour cycle. Simultaneous activation of the Per, Cry and Rev-Erb a promoters by CLOCK and BMAL1 is followed by transcription of mRNA. mRNA transcripts are transported from the nucleus into the cytoplasm, bind to ribosomes and initiate translation of proteins which accumulate in the cytoplasm. PER proteins bind to CRY proteins in different combinations, casein kinase IE (CKE) binds to and phosphorylates the PER-CRY dimers. Coordinated nuclear entry of CKI+PER+CRY trimers and binding to CLOCK and BMAL1 abolishes their transcriptional promotion. Per and Cry mRNA and proteins are short lived and following phosphorylation (P) and degradation the cycle begins again with Per and Cry transcription. A recently described clock gene product REV-ERB a protein represses BMAL1 and CLOCK gene expression and also represses CRY1 gene expression. The products of clockcontrolled genes exert output functions. Some 2-10% of all mammalian genes are clock-controlled genes (Fu and Lee, 2003). The light-dark cycle dictates the timing of oscillations, whereby BMAL1 is high and PER and CRY are low at the beginning of a circadian day. CLOCK does not oscillate. Solid lines, direct regulation; dashed lines, indirect regulation. Redrawn and adapted from Fu and Lee, 2003. See Lakin-Thomas, 2000, and Fu and Lee, 2003, for reviews



FIGURE 3.4 Metabolism of melatonin. From Arendt, 1995, by permission

3.1.4 Other control mechanisms

The pineal contains very large numbers of other neuromodulators, neuroreceptors and hormone receptors: evaluation of their physiological importance *in vivo* is in its infancy. They include vasoactive intestinal peptide, pituitary adenylate cyclase activating peptide, peptide histidine isoleucine, neuropeptide Y, vasopressin, oxytocin, somatostatin, substance P, calcitonin gene-related peptide, secretoneurin, hypocretin, delta-sleep inducing peptide, natriuretic peptide, angiotensin, opiate peptides, LHRH, and nitric oxide synthase. Current findings have been summarised by Simmonneaux and Ribelayga (2003) and most of these peptides appear to have modulatory effects on the primary noradrenergic stimulation of melatonin synthesis.

3.1.5 Miscellaneous factors influencing melatonin production

Melatonin production declines with age. This was one of the most consistent observations in the field (see Arendt, 1995); however, some controversy has arisen since in a small number of healthy human subjects this was not observed in cross-sectional studies (Zeitzer et al, 1999). Production of aMT6s declines with age (Bojkowski and Arendt, 1990) and, as yet, this is not a controversial observation. There are no consistent observations regarding melatonin production with respect to sex or weight. Numerous drugs are reported to influence melatonin. Substances that influence serotonin and catecholamines by changing either their availablility or their receptor activity are prominent. In particular, beta-adrenergic receptor antagonists, a commonly used anti-hypertensive medication, suppress melatonin. A number of anti-depressant serotonin and catecholamine re-uptake inhibitors, some anti-psychotics, and possibly caffeine, increase melatonin and/or change the timing of secretion. Some benzodiazepines and

non-steroidal anti-inflammatory drugs, alcohol and nicotine are reported to decrease circulating melatonin. Substances that compete for metabolic pathways influence plasma levels of melatonin and urinary levels of its major metabolite, aMT6s. Gonadal steroids may influence both metabolism and production, and there is evidence that the use of contraceptives influences circulating melatonin (Kostoglou-Athanassiou et al, 1998; Wright and Badia, 1999; Simmoneaux and Ribelayga, 2003; Thapan et al, unpublished observations; Thapan, 2001). Table 3.1 provides an overview of substances that affect melatonin production, secretion and metabolism.

Factor	Model	Effect(s) on melatonin	Comment	References
Light	Humans	Suppression (>30 lux broad spectrum white in controlled studies) intensity required depends on previous light exposure	Maximum effective wavelength 460–480 nm	Zeitzer et al, 2000 Thapan et al, 2001 Brainard et al, 2001 Owen and Arendt, 1992 Hebert et al, 2002
Light	Animals	Suppression (intensity depends on species and environment)	Maximum effective wavelength 480 nm in mice	Foster and Hankins, 2002
Light	Humans	Entrainment: with scheduled sleep/darkness <200 lux Without other time cues >200 and <1000 lux	Short wavelengths more effective than white	Zeitzer et al, 2000 Wright et al, 2001 Middleton et al, 2002 Warman et al, 2003 Lockley et al 2003
Light	Animals	Entrainment (intensity depends on species and environment)		Elliot, 1981
Timing of sleep or rest-activity cycle	Humans Animals	Changes in timing (and sometimes amplitude after abrupt phase shift), eg shift work, jet lag in humans	Partly a secondary effect, sleep modifies light-dark exposure	Barnes et al, 1998 Rajaratnam and Arendt, 2001 Turek and Van Reeth, 1988 Fevre-Montange et al, 1981
Timing of light- dark cycle	Humans Animals	Changes in timing (and sometimes amplitude after abrupt phase shift), eg shift work, jet lag in humans		Deacon and Arendt, 1996 Illnerova and Sumova, 1997
Posture	Humans	Increase on standing up after a period of recumbency at night	20 min required to stabilise, may not affect timing	Deacon and Arendt, 1994 Voultsios et al, 1997
Exercise	Humans	Increase at night, induces phase shifts	Very hard exercise	Buxton et al, 2003
Adrenergic β-receptor antagonists	Humans Animals <i>In vitro</i>	Decreased synthesis, can be almost completely suppressed overnight	Anti-hypertensives	Arendt et al, 1985a,b Simmoneaux and Ribelayga, 2003 Klein et al, 1997

TABLE 3.1 Various factors influencing melatonin secretion; reviews are cited where appropriate. Most references prior to 1995 can be found in Arendt, 1995

TABLE 3.1 Continued

Factor	Model	Effect(s) on melatonin	Comment	References
Some serotonin (5HT) uptake inhibitors	Humans	Increase with fluvoxamine	Antidepressant (probably metabolic effect)	Hartter et al, 2001 Skene et al, 1994
Noradrenalin uptake inhibitors	Humans	Increase/change in timing	Antidepressants	Checkley et al, 1986
MAO α inhibitors	Humans Animals	Increase, may change phase	Antidepressants	Arendt, 1989
Alpha- adrenoceptor antagonists	Humans	Decrease with alpha-1, increase with alpha-2		Palazidou et al, 1989a,b
Benzodiazepines	Humans Animals <i>In vitro</i>	Varied, eg decrease with diazepam, alprazolam, not triazolam, temazepam, zopiclone <i>In vitro</i> and <i>in vivo</i> results not necessarily similar	Probably via GABA related mechanisms	McIntyre et al, 1993 Monteleone et al, 1989 Niles, 1991 Copinschi et al, 1990 Mann et al, 1996 Cardinali and Golombek, 1998
Vasopressin	In vitro	Increase	Potentiates noradrenergic stimulation	Simmoneaux and Ribelayga, 2003
Dopamine	In vitro	Possible effects both inhibitory and stimulatory		Simmoneaux and Ribelayga, 2003
Acetylcholine	Animals <i>In vivo</i> microdialysis and <i>in vitro</i>	Inhibition of synthesis		Simmoneaux and Ribelayga, 2003
Glutamate	In vitro	Inhibition of synthesis	Inhibits noradrenalin stimulation	Simmoneaux and Ribelayga, 2003
GABA	In vitro	Inhibition of synthesis	Inhibits noradrenalin stimulation	Simmoneaux and Ribelayga, 2003
Nitric oxide	In vitro	Inhibition of synthesis		Simmoneaux and Ribelayga, 2003
Testosterone	Animals Humans	Stimulation	Castration reduces melatonin synthesis Treatment of male hypogonadism decreases melatonin, hyperandrogenic women have increased melatonin	Simmoneaux and Ribelayga, 2003 Puig-Domingo et al, 1992 Luboshitzky et al, 1997
Oral contraceptives	Humans	Increase	Possible metabolic effect	Wright and Badia, 1999 Simmoneaux and Ribelayga, 2003 Kostoglou-Athanassiou et al, 1998
Progesterone		Controversial		

TABLE 3.1 Continued

Factor	Model	Effect(s) on melatonin	Comment	References
Oestradiol	In vitro	Inhibition of synthesis		Simmoneaux and Ribelayga, 2003
Oestradiol	Animals	Decrease (inconsistent)	Ovariectomy increases melatonin in rats and sheep (transient) but inconsistent between species Possible metabolic effect by competition for the CYP1A2 metabolic pathway?	Simmoneaux and Ribelayga, 2003 Cheng, 2001
Oestradiol	Humans	Decrease (inconsistent, insufficient data)	Oestradiol treatment for delayed puberty lowered aMT6s in a case report No effect of ovarian suppression in precocious puberty Menopause may be associated with increased melatonin	Arendt et al, 1989 Berga et al, 1989 Okatani et al, 2000
Oestrous cycle	Animals <i>In vivo</i> microdialysis	Inconsistent in whole animals No differences	Pro-oestrous decline in rodents	Simmoneaux and Ribelayga, 2003
Menstrual cycle	Humans	Inconsistent and conflicting reports	Never properly assessed in constant routine conditions in humans	Arendt, 1995
Oestradiol, Progesterone	DMBA- treated rats	Decrease		Simmoneaux and Ribelayga, 2003
Nicotine	Humans	Possible increase, insufficient data		Tarquini et al, 1994
Alcohol	Humans	Decrease, dose dependent		Ekman et al, 1993
Caffeine	Humans	May delay clearance, ie increase		Shilo et al, 2002 Hartter et al, 2003
Some non-steroidal anti-inflammatory drugs	Humans	Decrease		Murphy et al, 1996
Chlorpromazine	Humans	Increase	Antipsychotic Metabolic effect	Ozaki et al, 1976 Smith et al, 1977 Beedham et al, 1987
Luteinising hormone	In vitro	Increased production	Few data	Simmoneaux and Ribelayga, 2003
Benserazide	Animals Humans	Decrease No amplitude change in Parkinson patients, possible phase change	Aromatic amino-acid decarboxylase inhibitor	Arendt et al, 1981 Fertl et al, 1991 Ho and Smith, 1982
Calcium channel blockers	Humans	Possible effects		Escames et al, 2004

3.2 Light-dark control of melatonin synthesis in animals and man

3.2.1 Darkness hormone

In all mammalian and other vertebrate species studied to date, whether nocturnal or diurnal, melatonin is normally synthesised and secreted during darkness. Remarkably, even the unicellular alga *Conyaulax* appears to produce melatonin during the dark phase, and it appears to be present in higher plants (see Arendt, 1995). Melatonin production is clearly a highly evolutionarily conserved phenomenon. In most vertebrates the rhythm is endogenous, ie internally generated, and in higher vertebrates it is driven by the central pacemaker, the suprachiasmatic nuclei (SCN). In birds and lower vertebrates the pineal is a self-sustaining oscillator itself. The rhythm persists in the absence of specific time cues (zeitbegers) and becomes 'free-running', in general assuming a period deviating slightly from 24 hours, and is thus a true circadian rhythm. Lesions of the SCN lead to loss of the vast majority of circadian rhythms such as locomotor activity, sleep, behaviour, hormones including melatonin, and urinary constituents (Moore and Eichler, 1972; Stephan and Zucker, 1972; Klein, 1985; Moore, 1996). Circadian rhythms are entrained (synchronised) to the 24 hour day primarily by light-dark cycles (Figure 3.5) acting via the retina. It should be noted that in spite of a publication (Campbell and Murphy, 1998) reporting an impact of light applied behind the knees on rhythms, this has been refuted by several further studies (Lockley et al. 1998; Wright and Czeisler, 2002). Many blind people with no light perception at all, ie lacking the lightdark time cue, show free-running melatonin and other rhythms (eq sleep, cortisol, core temperature) in a normal environment (Lewy and Newsome, 1983; Arendt et al, 1997; Lockley et al, 1997a). In addition to entraining the rhythm, daylength (photoperiod) determines the duration of night-time secretion both by direct suppression of melatonin and by determining the length of the signal emitted by the SCN (Lincoln et al, 1985; Arendt, 1986; Illnerova and Sumova, 1997). Factors (zeitgebers) other than light-dark cycles that are involved in entrainment include behavioural imposition such as forced activity and rest, social and nutritional (rhythmic feeding) cues, temperature variations, knowledge of clock time, certain drugs, possibly EMFs, and melatonin itself.



Sleep-wake, temperature and other rhythmic functions

FIGURE 3.5 Mammalian circadian rhythms are generated in the suprachiasmatic nuclei, and entrained by light with contributions from non-photic time cues. Melatonin provides a closed loop to this system

3.2.2 Melatonin secretion in relation to daylength

In most species, melatonin secretion is related to the length of the night: the longer the night, the longer the duration of secretion. This phenomenon has been particularly well demonstrated in sheep, in which melatonin levels rise within a few minutes of lights off and, in photoperiods of more than around 14 hours of light, do not decline until lights on (Arendt, 1986) (Figure 3.6).



FIGURE 3.6 Winter and summer melatonin profiles in sheep and humans. Note the changed duration from winter to summer in sheep with primarily a change in timing – earlier in summer than in winter in humans

The most consistent observation in humans is that melatonin profiles show a phase change from winter to summer, with earlier secretion in summer than in winter (see Arendt, 1995). However, if humans are kept strictly in darkness for 14 hours per day for a period of 2 months, the melatonin secretion pattern expands to cover almost the entire dark period and, concomitantly, in extended periods of 16 hours of light, the rhythm contracts to less than 9 hours, with accompanying changes in body temperature and sleep (Wehr et al, 1993, 2001). Small changes in duration have been seen between winter and summer in high latitudes. Short-term imposition of artificial short days leads to a change in the timing of endogenous melatonin (Vondrasova-Jelinkova et al, 1999; Rajaratnam et al, 2003). Many clinical studies have not controlled sufficiently for exposure to natural or artificial light, making interpretation of data difficult.

3.2.3 Light suppression of melatonin secretion

Even brief exposure to light of suitable intensity, duration and spectral quality, suppresses melatonin production at night; short wavelengths (around 465 nm in humans) are most effective (Brainard et al, 2001; Thapan et al, 2001; Foster and Hankins, 2002) (Figures 3.7 and 3.8). Short wavelengths are also more effective than white light for entrainment of the rhythm to 24 hour (Lockley et al, 2003; Warman et al, 2003). Since the action spectrum derived from irradiance response curves does not correspond to either scotopic or photopic action spectra, a new photoreceptor(s) system has been



FIGURE 3.7 Melatonin suppression in humans by 2500 lux broad spectrum white light (double asterisks) and partial suppression by domestic intensity light (300 lux, single asterisks) from 00.30 to 01.00 hours, compared with control conditions in darkness. From Bojkowski et al, 1987a, by permission

invoked (Lucas et al, 1999). No sex differences have been noted with regard to entrainment in humans; however, the burdensome nature of these studies means that the number of subjects investigated is small. Moreover the tendency to use healthy, young men as volunteers reduces further the possibility of comparisons.

The amount of light required to suppress melatonin secretion during the night varies from species to species, with the time of night, and with previous light exposure. In humans, it was originally observed that 2500 lux broad spectrum white light (domestic light is around 300–500 lux) is required to completely suppress melatonin at night (Lewy et al, 1980), However, much lower 'domestic' intensity light will partially suppress and shift the rhythm in humans (Bojkowski et al, 1987a; Zeitzer et al, 2000). Thus artificial light at night is likely to modify melatonin production. These observations have been of very considerable importance for a general appreciation of the role of light in human physiology – in particular, its importance in the control of human rhythms and in the treatment of winter depression (seasonal affective disorder) (Rosenthal et al, 1984).



FIGURE 3.8 Action spectrum for melatonin suppression in humans. From Thapan et al, 2001, by permission

3.2.4 Light at night, melatonin suppression and risk of disease

The evidence that melatonin has some oncostatic activity has led to hypotheses regarding the increased incidence of breast cancer in female shift workers (see Chapter 5) and light suppression of melatonin in night shift workers. Little evidence exists for a decline in melatonin during night shift work, but it is likely that this can occur. In a light-induced forced nine hour phase shift, a clear decline in aMT6s production was noted (Deacon and Arendt, 1996). Unpublished work (Hall, English and Arendt) suggests that during a three day fast rotation shift schedule (three early shifts, three late shifts, three night shifts, rest days), the amplitude of aMT6s is significantly reduced (by approximately 30%) during the three night shifts (Figure 3.9). In an even faster (two day) rotation, significant changes were not observed (Costa et al, 1994). Actual light exposure at night needs to be evaluated carefully in field studies. Light at night has numerous other effects. The mere fact of frequent disruption of all circadian rhythms, not just melatonin, is effectively a physiological insult.

The behavioural effects of light at night include increased alertness and performance, and an increase in body temperature (Campbell and Dawson, 1990; Strassman et al, 1991; Myers and Badia, 1993; Phipps-Nelson et al, 2003). Noradrenalin and acetylcholine are decreased, and serotonin, GABA and dopamine are increased (Roberts, 2000). A number of neurotransmitters in the SCN are modified by light (Roberts, 2000). A recent review on light and immunomodulation suggests that light impacts on the immune system either by central neuroendocrine mechanisms or by skin-mediated responses (Carlberg, 2000). The treatment of neonatal jaundice by light is another example of a non-visual effect. There is good evidence that increased light exposure in the early morning (one hour, 05.00–06.00 hours) will increase



FIGURE 3.9 24 hour production of 6-sulphatoxymelatonin (aMT6s) (mean ± SD), during a three day shift rotation, onshore, N=15. Note lower production during night shift. From Hall, English and Arendt (unpublished data)

luteinising hormone (LH) secretion by 69% – it has even been proposed that this might improve fertility (Yoon et al, 2003). Cortisol secretion (in men) is increased by more than 50% by bright light exposure from 05.00–08.00 hours, but not by afternoon (13.00–16.00 hours) treatment (Leproult et al, 2001). Exposure to bright full spectrum white light is now a common treatment for seasonal affective disorder (albeit with a large placebo effect), but in spite of much effort it has not been possible to confirm a role for melatonin. Rather, serotonergic mechanisms are implicated (Partonen and Magnussen, 2001).

Impaired sleep as a result of shift work (Akerstedt, 1998) is evidently associated with light at night. Sleep deprivation and impaired metabolism at night are associated with a large number of problems including an increase in risk factors for major disease: higher circulating triacylglycerol, glucose intolerance, insulin resistance (Morgan et al, 2003). Sleep and sleep loss are also associated with immunomodulatory effects (Shephard and Shek, 1997; Redwine et al, 2000).

Most importantly, perhaps, light directly influences the expression of the clock gene feedback loops driving circadian rhythms (Reppert, 2000). Disruption of clock gene function is associated with increased risk of cancer in recent animal studies (Fu et al, 2002; Fu and Lee, 2003; Filipski et al, 2004). Moreover, a length polymorphism in the clock gene *per3* is associated with increased risk of breast

cancer in a very recent study (Zhu et al, 2005). Thus a more plausible hypothesis regarding the risk of breast cancer in subjects exposed to light at night would involve the whole circadian axis rather than just melatonin suppression.

3.2.5 Entrainment of the melatonin rhythm

A single daily light pulse of suitable intensity and duration in otherwise constant darkness is sufficient to phase shift and synchronise the melatonin rhythm to 24 hours in animals (Elliot, 1981). Phase shifting and entrainment have been demonstrated in humans with suitable intensity and duration of light treatment (Arendt and Broadway, 1986, 1987; Czeisler et al, 1986; Broadway et al, 1987; Shanahan and Czeisler, 1991; Middleton et al, 2002). However, the relative contribution of light to the entrainment of melatonin in a normal environment remains to be fully determined. Studies in Antarctica suggest that a structured social routine in a dim light environment (with no natural sunlight) suffices to synchronise melatonin to 24 hours (Broadway et al, 1987; Arendt, 1995). However, as many blind people with no conscious or unconscious light perception living in a normal social environment show desynchronised melatonin and other circadian rhythms, it is clear that at least some perception of light is of primary importance (Arendt et al, 1997; Lockley et al, 1997a,b; Hack et al, 2003).

3.3 Role of melatonin in photoperiodic seasonal functions in animals

3.3.1 Photoperiodism

Most species show seasonal variations in their physiology and behaviour, even humans. The reproductive cycle is timed so that environmental conditions are propitious for growth of the young, and variations in behaviour, pelage (coat growth and colour), appetite, body weight, and fat are such that survival in ambient temperature conditions is optimised and camouflage protects against predators (Hoffmann, 1979; Arendt, 1986; Goldman, 2001). When seasonal functions are primarily timed by daylength, species are referred to as being photoperiodic. Photoperiod is often critical for the timing of pubertal development (Foster et al, 1988). In general, puberty is reached only during the adult mating season. It is now clear that in photoperiodic mammals and marsupials, an intact innervated pineal gland is essential for the perception of photoperiodic change (Lincoln and Short, 1980; Arendt, 1986; Karsch et al, 1988; Lincoln and Richardson, 1998). Most information is derived from studies on reproductive function in hamsters and sheep.

3.3.2 Role of melatonin

Pinealectomy removes the vast majority of circulating melatonin in rodents, primates, and ungulates. It was therefore the first pineal hormone to be investigated as a pineal photoneuroendocrine transducer. It is possible to administer melatonin by daily infusion or feeding to generate at will circulating profiles with a duration characteristic of particular photoperiods in an intact or pinealectomised animal. In this way it has become clear that a particular melatonin duration is a necessary and sufficient condition for induction of a given seasonal response and is equipotent with a particular photoperiod. Long-duration melatonin is equivalent to long days.

Interpretation of the signal, as with daylength, depends on the physiology (for example, long- or shortday breeder) of the species in question. In sheep, the evidence is good that long days or short-duration melatonin can time the whole seasonal cycle, at least of reproduction, and act as a seasonal zeitgeber for a presumed endogenous annual rhythm. Animals become refractory to a specific duration of melatonin as they do to a particular photoperiod. For example, a period of long days (or a long-day melatonin signal) is required before a short-day melatonin signal advances the reproductive cycle in sheep (Arendt, 1986; Lincoln and Richardson, 1998; Goldman, 2001).

3.3.3 Puberty and development

The photoperiod via melatonin secretion determines the timing of puberty in some species, provided that a sufficient degree of physical maturity has been reached (Foster et al, 1988). Interestingly, photoperiod perception by the fetus is present before birth in rodents and ungulates and ensures a rate of development appropriate to environmental conditions (Davis and Mannion, 1988; Deveson et al, 1992). Melatonin crosses the placenta in a number of species, and melatonin injection in the mother can dictate the timing of postnatal reproductive development (Weaver and Reppert, 1986).

The laboratory rat is only marginally photoperiodic. Nevertheless, injections of melatonin during the late light phase, during a small window in the late dark phase, or even via continuous release implants, specifically during the period of pubertal development, delay reproductive maturity in both males and females (Sizonenko et al, 1985). Full sexual maturity is eventually achieved; thus the system is not permanently compromised. Moreover, *in vitro* melatonin inhibits gonadotropin-releasing hormone (GnRH)-induced LH release by cultured rat pituitary glands from prepubertal animals (Martin and Klein, 1976; Symons et al, 1985). These observations constitute the main evidence for a possible role of melatonin in the pubertal development of humans.

3.3.4 Mechanisms

The mechanism of action of melatonin with regard to seasonal variation in reproductive competence and the timing of puberty in animals is thought to involve an influence of melatonin on steroid feedback mechanisms in the brain, together with a direct influence on the pituitary gland via melatonin receptors (Goldman, 2001). For example, anoestrous sheep exposed to artificial short days in summer rapidly show increases in LH secretion and changes in the pulsatile pattern of secretion associated with onset of reproductive activity and oestrous (Bittman et al, 1985). There is clear evidence that the large seasonal changes in prolactin secretion in ungulates are mediated by a direct action of melatonin within the pars tuberalis (PT) of the pituitary via the MT1 receptor (Lincoln, 2002). Melatonin duration influences clock gene expression within the PT (Messager et al, 2000; Hazlerigg et al, 2001) but not within the SCN (according to Poirel et al, 2003). Figure 3.10 summarises melatonin actions with respect to seasonal and circadian rhythms.

Rhythmic clock gene expression in the anterior pituitary appears to depend on sensitisation of the adenosine A2b receptor occurring through activation of the melatonin (MT1) receptor (von Gall et al, 2002a,b). This observation leads to the possibility that melatonin widely influences peripheral gene expression.



Pars tuberalis

FIGURE 3.10 Control of production and the functions of melatonin with regard to seasonal and circadian timing mechanisms. The melatonin rhythm is generated by a closed loop negative feedback of clock gene expression in the SCN – *Clock* and *Bmal* being positive stimulatory elements, *Per* and *Cry* negative elements, and *CCG*, clock-controlled genes. *Per* and *NAT* mRNA oscillate in the pineal, although post-transcription control is evident in some species. Melatonin influences SCN activity via two or more receptors. MT2 appears to be the phase shifting receptor in rodents, whereas MT1 is associated with suppression of SCN electrical activity. The MT2 receptor was first characterised in the retina and influences dopamine release. Melatonin conveys photoperiodic information influencing the pattern of Per expression in the pars tuberalis for the control of seasonal prolactin variations via an MT1 receptor. Melatonin target sites in the hypothalamus influencing seasonal variations in reproductive hormones have yet to be fully defined. Based, with permission, on an original diagram by Dr Elisabeth Maywood, MRC Laboratory of Molecular Biology, Neurobiology Division, Hills Road, Cambridge, CB2 2QH, UK

3.3.5 Non-reproductive seasonal functions

The pineal gland via melatonin secretion probably plays a role in all photoperiod-dependent functions in mammals. Evidence exists to substantiate this statement with respect to behaviour, body weight, coat constitution and colour (for example, the white winter coat of some polar species), prolactin variations, antler growth, thyroid activity, appetite, thermoregulation, delayed implantation, embryonic diapause, and hibernation (see reviews by Arendt, 1986, and Goldman, 2001).

3.4 Role of the pineal in mammalian circadian rhythms

Until quite recently, opinion was that the pineal did not have an important role in the mammalian circadian system. Some strains of (healthy, prolific) laboratory mice have virtually no detectable melatonin. Pinealectomised rats show no obvious circadian abnormalities unless 'challenged'. Thus, in rats, pinealectomy increases the rate of re-entrainment to forced phase shifts of the light–dark cycle, and pinealectomy of hamsters in constant light leads to major disruption of the circadian system (Armstrong and Redman, 1985; Cassone, 1992). In hamsters, there is good evidence that maternal melatonin can influence the circadian timing of the fetus (Davis and Mannion, 1988). Melatonin is found in milk (of goats, cows, humans and rats) with increased levels at night. The hypothesis that melatonin in milk influences the circadian rhythms of the neonate has been refuted, at least in rats (Rowe and Kennaway, 2002). In humans it is likely that the phase shifting effects of light do not depend on melatonin suppression, but that the presence of endogenous or exogenous melatonin can, in some circumstances, modulate the effects of light on the circadian system (Arendt, 2003). It is quite possible to sleep out of phase with melatonin (although sleep is somewhat compromised). A substantial body of work implicates melatonin in circadian thermoregulation (see Badia et al, 1992, for a review). Many such effects may involve the thyroid gland. Most evidence for a circadian role of melatonin derives from timed administration.

3.5 Effects of timed administration of melatonin in animals

3.5.1 Behaviour, hormones, and temperature

In rats, daily melatonin injections synchronise free-running activity and temperature rhythms in constant darkness and are reported to partially or completely synchronise disrupted activity rhythms in constant light, although the latter observation is somewhat controversial (Redman et al, 1983; Chesworth et al, 1987). A phase–response curve to single injections of melatonin can be demonstrated with small phase advances of at most one hour during the late subjective day (Armstrong and Redman, 1985). Timed administration hastens adaptation of activity and melatonin production to forced phase shift and can change the direction of re-entrainment (Illnerova et al, 1989). Some, but not all, strains of adult hamster can be synchronised by melatonin administration, and fetal hamsters can be entrained by maternal injections of melatonin at 24 hour intervals in specific circadian phases (Davis and Mannion, 1988).

3.5.2 Gestation

In the rat, gestation length depends on the ambient light–dark cycle. Small advances or delays in parturition can be induced by daylengths shorter or longer than 24 hours, and the effect can partially be mimicked by timed melatonin administration (Bosc, 1987).

3.5.3 Oestrous cycle

The pineal is involved in circadian timing, so the presumption must be that it is concerned with timing of the LH surge and, indeed, with general oestrous timing. In rats, timed melatonin administration has been

demonstrated to mimic the effects of extending the light–dark cycle on timing of the LH surge. Observations of the melatonin rhythm itself show a decreased amplitude during proestrus in rodents, but with conflicting reports in other species (Rollag et al, 1979, and see Arendt, 1995, for a review).

3.5.4 Ageing and immune function

A fairly consistent observation in pineal research (but with some controversy) is the decline in amplitude of the melatonin rhythm in old age (see Arendt, 1995, for references). Pinealectomy is reported to accelerate the ageing process in animals; however; both acceleration and deceleration of ageing have been reported in shortened photoperiod (ie longer duration endogenous melatonin) (Holmes and Sugden, 1976; Armstrong and Redman, 1991; Aujard et al, 2001; Place et al, 2004). The possible antiageing effects of melatonin have generated considerable publicity (Trentini et al, 1992; Pierpaoli and Regelson, 1994; Reppert and Weaver, 1995; Arendt, 1996; Turek, 1996). Several hypotheses have been put forward to explain these sometimes flawed, insubstantial, but interesting observations. Melatonin, when appropriately administered, has (usually) stimulatory effects on aspects of the immune system, and longer survival with cancer has been claimed (Lissoni et al, 1993; Maestroni, 1999). One recent review considers that such effects are peripheral and primarily use nuclear melatonin receptors (RZR/ROR alpha and RZR beta) rather than the membrane receptors (MT1 and MT2) associated with rhythmic functions (Carlberg, 2000). Another explanation considers that appropriately timed daily melatonin administration optimises circadian relationships, especially of phase, and increases circadian amplitude (see Armstrong and Redman, 1991, and Arendt, 1995, for references). Such effects should lead to improved sleep, and thus enhanced immune function (Redwine et al, 2000).

3.5.5 In vitro phase shifts

The metabolic activity of the rodent SCN *in vivo* and the electrical activity of various *in vitro* SCN preparations can be modified by melatonin; it inhibits 2-deoxyglucose uptake into the nuclei in the late biological day with no effect at other times and inhibits electrical activity, also during the late biological day (Cassone et al, 1988; Shibata et al, 1989). There is very convincing evidence of the phase-advancing effect of melatonin on the circadian rhythm of electrical activity in cultured SCN neurons (McArthur et al, 1991). The effect was large, acute, and time dependent, with shifts of up to several hours being observed. Thus melatonin acts directly on a central biological clock to change its phase.

3.5.6 Retinal rhythms

Melatonin appears to function as a paracrine signal within the retina. It influences dopamine release (Dubocovich, 1983) and may enhance retinal function in low intensity light by inducing photomechanical changes and regulating turnover rates of the photoreceptive apparatuses of rods, cones, and the surrounding pigment epithelium (Reme, 1986; luvone and Gan, 1995; luvone et al, 2002). It is possible that retinal rhythmicity influences the central pacemaker. However, this question remains speculative.

3.5.7 Mechanisms

Using gene knockout technology and pharmacological manipulations, the results to date suggest that the phase-shifting melatonin receptor is MT2, whilst MT1 is associated with acute suppression of SCN electrical activity (Liu et al, 1997) in addition to its important actions within the pars tuberalis (Masana and Dubocovich, 2001; Ross and Morgan, 2002; Dubocovich et al, 2003). Several other physiological responses have been ascribed to MT1 and MT2 receptors, including (MT1) melatonin-mediated potentiation of adreneregic vasoconstriction and (MT2) modulation of dopamine release in the retina (Mahle et al, 1997; Doolen et al, 1998; Masana and Dubocovich, 2001). Genetic polymorphism has been identified within melatonin receptors and further investigation of these polymorphisms in relation to photoperioidism, human disease, sensitivity to melatonin, etc, is ongoing (Ebisawa et al, 2000; Migaud et al, 2002).

3.6 The pineal in human physiology and pathology

Clearly, the importance of the pineal in humans depends on the importance of light in human physiology. It is reasonable to assume that the pineal conveys information concerning light–dark cycles for the organisation of seasonal and circadian rhythms in humans as in animals. Pinealectomy in humans removes virtually all plasma melatonin (for review, see Arendt, 1995). Other consequences of the operation consist of diffuse neurological problems that do not add up to a consistent functional effect and may be more related to non-specific effects of the operation. Recent work suggests that melatonin is absent or very low in subjects with treated or untreated pineal germinomas (Murata et al, 1998), but the consequences remain to be defined. Interestingly, one study, as yet unpublished, suggests that seasonality is lost in pinealectomised humans (Macchi et al, 2002).

3.6.1 Measurement of melatonin and its primary metabolite 6-sulphatoxymelatonin

The most specific method of melatonin measurement is gas chromatography/mass spectrometry (GCMS) (Lewy and Markey, 1978). However, this technique is expensive, time consuming and has low throughput. Thus most investigators use either radioimmunoassay (RIA), which predates GCMS (Arendt et al, 1975), or enzyme-linked immunoassay (ELISA) (Middleton, in press). High performance liquid chromatography (HPLC) may also be used but does not have the throughput possibilities or, usually, the sensitivity of RIA and ELISA. The most reliable and specific RIAs and ELISAs are GCMS validated. Pre-extraction is required for most systems measuring plasma melatonin, although at least one GCMS validated assay operates successfully without extraction (Fraser et al, 1983a,b). By contrast, saliva measurements can usually be carried out without extraction (Vakkuri, 1985; McIntyre et al, 1987; English et al, 1993; Voultsios et al, 1997). The first urinary aMT6s assay to be developed was validated against GCMS measuring total urinary 6-hydroxyindoles rather than a specific GCMS assay (Arendt et al, 1985b; Aldhous and Arendt, 1988). To date, no specific GCMS assay exists for aMT6s.

Numerous commercial kits and reagents are available and a recent review addresses comparisons of different methods (Middleton, in press). In general, the sensitivity of assays has improved since the first publications, with analytical limits of detection for plasma and saliva melatonin now around 0.3 to

5.0 pg ml⁻¹, and for urinary aMT6s of 0.2 ng ml⁻¹ or less. Reproducibility is reasonable, coefficients of variation are usually less than 15% and can be far less. These assays require care and attention to possible contaminants, usually listed by suppliers. For example, melatonin in powder form, opened in a laboratory, can contaminate a whole room which then requires in-depth cleaning. Direct measurements in saliva can be affected by food and drink, notably caffeine: subjects should wash their mouths out prior to sampling. Urinary aMT6s is robust and does not degrade at room temperature for 24 hours, at 4°C for 2 days and at -20° C for at least 2 years (Bojkowski et al, 1987b) and possibly as much as 15 years (Griefahn et al, 2001) without preservative.

If a 'pure' representation of melatonin production is required it is essential to control light exposure, posture and drug ingestion (see Table 3.1). The 'gold standard' for assessment of melatonin rhythms is the so-called 'constant routine' where subjects remain recumbent or semi-recumbent for at least 24 hours, in very dim light, awake, and with identical hourly snacks (Mills et al, 1978; Duffy and Dijk, 2002). This is evidently impossible in field situations. Here the use of urinary aMT6s enables long-term non-invasive monitoring of melatonin production. Ideally, light exposure and sleep times should be measured simultaneously (eg using an Actiwatch-L, Cambridge Neurotechnology Ltd).

Sampling frequency for melatonin measurement in plasma or saliva is usually hourly or half-hourly, depending on the degree of resolution needed. Very frequent sampling may lead to apparent episodic secretion (English et al, 1987; Claustrat et al, 1997). This may be in part an artefact of assay noise. However, the presence of two peaks of secretion has frequently been noted (Arendt, 1979, 1985; Wehr et al, 1995). There is undoubtedly a correlation between the total overnight melatonin production and the value obtained from a single early morning or 02.00 hours sample in normally entrained individuals (Arendt, 1978; Graham et al, 1998). However, a whole dimension of information (timing of the rise and fall, and the duration of secretion) is lost by using only a single measurement.

Sampling frequency for urinary aMT6s varies, but for field studies it is usually collected over 3 or 4 hour periods with a longer oversleep collection. A 24 hour urine collection will provide an estimate of overall production but not timing or duration of secretion (Bojkowski et al, 1987b). 'Overnight' total excretion has also been used, but small phase shifts (advance or delay) can lead to underestimation of production and if a subject has very delayed (or less commonly, advanced) circadian rhythms much of the peak production may be lost. A single morning urine sample in a normally entrained individual should provide values which correlate with overall night-time production. However, once again timing and duration are lost by this approach. Moreover, in subjects who are out of phase the relationship will no longer hold. Evidently the resolution of 3-4 hourly sampling will be less than 30 or 60 minute plasma or saliva melatonin measures. However, using these sampling intervals there is a very robust correlation with both the timing and amplitude of plasma melatonin (Arendt et al, 1985b; Bojkowski et al, 1987b; Ross et al, 1995). Naidoo (Naidoo and Arendt, unpublished observations) measured aMT6s in urine sampled at hourly intervals for 24 hours in 14 healthy volunteers. The parameters of acrophase (calculated peak time using cosine curve fitting techniques) and amplitude were derived using the hourly data and subsequently using the same data at 2, 3 and 4 hour intervals, with an 'oversleep' 8 hour interval inserted. The difference in calculated acrophase was on average 12 minutes between hourly data and 3 hourly data, although the significance of the cosine fit decreased with decreasing numbers of data points used (Naidoo, 1999). With 4 hourly sampling (and oversleep) intervals, a 48 hour collection period will provide significant cosinor fits in general.

It is possible to measure melatonin or aMT6s concentration in urine samples and relate the data to creatinine concentrations, in place of collecting the entire urine production for each collection period and measuring the volume. However, creatinine itself has a minor but significant circadian rhythm (difference between highest and lowest average values is reported to be from 6–29% of the average lowest value – Haus and Touiton, 1992), peaking during the early evening (Kanabrocki et al, 1988; Haus and Touiton, 1992). Thus the characteristics of the aMT6s or melatonin rhythms may be distorted by this method.

The timing and duration of melatonin secretion are its critical features with regard to physiological functions. The relevance of small changes in melatonin amplitude remains obscure, particularly since there is an enormous individual variation between normal healthy subjects (see Table 3.2).

Figure 3.11 indicates the markers used to characterise melatonin and aMT6s rhythms. Area under the curve, total 24 hour excretion (aMT6s) or cosinor derived amplitude are used to assess total secretion. At present there is no standard definition of onset/offset (and hence duration). The so-called dim light melatonin onset (DLMO) initially proposed by Lewy and Sack (1989) uses a specific plasma concentration



FIGURE 3.11 Phase markers for melatonin in plasma or saliva, or 6-sulphatoxymelatonin (aMT6s) in urine

above which melatonin secretion is considered to have started. However, in this case the amplitude of the rhythm influences the onset (and offset) time. Other definitions of onset/offset include values exceeding twice or thrice the standard deviation of mean baseline levels. With a noisy baseline the 'mid-range crossing' or 25% rise methods avoid problems (Figure 3.11). More sophisticated curve fitting techniques have been employed and appear to give reliable data (Brown et al, 1997; Gamst et al, 2004). One in particular bases the algorithm on the supposed switch on and switch off of melatonin synthesis (preceding the final drop to baseline in the morning) (Brown et al, 1997); another estimates the 'Synoff' or switch off of melatonin secretion using iterative regression fits (Revell et al, 2005). It is advisable to use all the methods available. Much depends on the assay in question since with different limits of detection onset and offset will vary. In recent studies Hoppen (Hoppen, Middleton, Stone and Arendt, unpublished observations) evaluated the stability of melatonin onset in saliva. Using values exceeding twice baseline and continuing to rise, melatonin onset in saliva was at 22.30 hours and with the mid-range crossing method was at 23.18 hours (12 young male subjects). When onset time was repeatedly assessed (five times) in the same 12 subjects at intervals of at least 2 weeks, the variation in onset time was 28±10 minutes and 24±9 minutes, X±SD, respectively, illustrating the reliability of the timing of melatonin production in free-living subjects (Hoppen, 2002) (see also the small variations found by Voultsios et al, 1997). The duration of melatonin secretion following 8 days of 16 hours of near-darkness, <5 lux, and recumbency, assessed recently in constant routine conditions was 9 hours ± 36 minutes (plasma, 8 young men) (Rajaratnam et al, 2003).

3.6.2 Human melatonin production

Both the amplitude and the duration of melatonin secretion in combination define production per 24 hours. Total urinary aMT6s per 24 hours provides an integrated measure of melatonin production in normal, healthy, unmedicated volunteers.

In a 'normal' environment, melatonin is secreted during the night in healthy humans as in other species. The rhythm is endogenous, with a period usually greater than 24 hours (Arendt et al, 1985a; Wever, 1989; Middleton et al, 1996; Wright et al, 2001). As is evident from studies in blind subjects, it is entrained primarily by the light-dark cycle with minor contributions of non-photic zeitgebers (Arendt, 2000). The average maximum levels attained in plasma in adults are around $60-70 \text{ pg m}^{-1}$ when measured with high specificity assays (see Arendt, 1995). Salivary melatonin reflects well the plasma profile but concentrations are approximately 30% of those in plasma. Mean maximum concentrations of aMT6s in plasma attain 80–100 pg ml⁻¹. Minimum concentrations of both compounds are usually below 10 pg ml⁻¹. Peak concentrations of melatonin in plasma and saliva normally occur between 02.00 and 05.00 hours. The onset of secretion is usually around 21.00 to 23.00 hours and the offset around 07.00 to 09.00 hours in adults in temperate zones (depending on the definition of onset and offset, see Figure 3.12). There is a strong correlation of both amplitude and timing of the melatonin rhythm in plasma and saliva with aMT6s in normal, healthy individuals, as previously noted. Drug treatment and disease may modify this relationship. The appearance and peak levels of aMT6s in plasma are delayed by 1 to 2 hours and the morning decline by 3 to 4 hours with respect to melatonin. Between 50 and 80% of exogenous melatonin is converted to aMT6s (Middleton et al, 1997) and most appears in the overnight sample (24.00 to 08.00 hours) in a normal environment. It is low, but rarely undetectable, in the afternoon and early evening.



FIGURE 3.12 Mean concentrations (+ SEM) of melatonin in plasma (N=133, error bars within the symbol), saliva (N=28) and 6-sulphatoxymelatonin (aMT6s) in urine (N=88), healthy volunteers, all ages, M and F, all measurements by radioimmunoassay (Fraser et al, 1983a,b; Aldhous and Arendt, 1988; English et al, 1993)

Possibly the most striking characteristic of the normal human melatonin rhythm is its reproducibility from day to day and from week to week in normal individuals, rather like a hormonal fingerprint (Arendt, 1979, 1988; Bojkowski et al, 1987b; Voultsios et al, 1997; Klerman et al, 2002). This stability leads to the extensive use of melatonin in plasma or saliva, and aMT6s in urine, as marker rhythms for circadian phase – for example, in the investigation of sleep disorders, and evaluating adaptation to abrupt phase shifts as in shift work and jet lag. The very large inter-individual variations have been ascribed to the size of the pineal gland rather than to variations in enzymic activity, at least in sheep (Gomez Brunet et al, 2002) (Figure 3.13). A small number of apparently normal individuals have no detectable melatonin in plasma at all times of day.

In view of this inter-individual variation it is necessary to use subjects as their own controls in small studies. Table 3.2 shows some normal values for aMT6s, using the method of Aldhous and Arendt (1988), from young, healthy volunteers. There are no differences with sex in this series. A decline with age is apparent, as previously observed on numerous occasions (reviewed by Arendt, 1995). Importantly the timing of the rhythm is earlier with increasing age: a factor rarely taken into consideration. A power



FIGURE 3.13 Examples of high (28 year old male) and low (30 year old male) melatonin plasma levels in normal, healthy volunteers compared with average data

TABLE 3.2 6-sulphatoxymelatonin (aMT6s) excretion in healthy, predominantly young male
volunteers, N=85, examples of 'normal values' for total excretion per 24 hour and peak time. Note the
very large range (maximum-minimum). Younger subjects (16-34) excrete more aMT6s than the older
group and have later acrophases (ANOVA p=0.010). Previously reported normal values over the whole
age range can be found elsewhere (Bojkowski and Arendt, 1990; Skene et al, 1990)

Subjects	Age (years), X±SD	Total aMT6s excretion (ng per 24 h), <i>X</i> ± SD	Rhythm acrophase decimal (h), X± SD, N	Rhythm amplitude (ng per h), X± SD, N
All	26.4 ± 8.3	16072 ± 10898	4.4 ± 1.3	913 ± 776
Maximum	56	60536	8.1	4733
Minimum	16	540	0.9	70
All female (N=24)	28.21 ± 8.8	16827 ± 12741	4.55 ± 1.13	965 ± 943
All male (N=61)	25.66 ± 8.1	15774 ± 10185	4.38 ± 1.33	892 ± 707
16–24 years (N=49)	21.49 ± 2.1	16045 ± 10565	4.78 ± 1.2	904 ± 747
25–34 years (N=23)	28.5 ± 2.9	17856 ±14099	4.01 ± 1.3	1070 ± 1027
35–56 years (N=13)	43.4 ± 7.1	11387 ± 5908	3.95 ± 1.1	585 ± 385

calculation using the mean and standard deviation of aMT6s excreted per 24 hours indicates that for a 20% change to be found in a given population with 83% power and a significance level of p=0.05, 200 subjects must be studied.

Skene et al (1990) analysed 24 hour aMT6s production (6 hourly samples) in 160 healthy women, aged 40 to 69 years. These data are shown in Figures 5.2 and 5.3 in Chapter 5. They illustrate the age-related decline in 24 hour aMT6s production.

The amplitude of the melatonin rhythm appears to be an inherited characteristic, at least in sheep (Zarazaga et al, 1998), as may be the entrained phase (as a function of inherited periodicity). A polymorphism in the clock gene *per3* is strongly associated with extreme diurnal preference which in turn is related to long or short circadian period (Duffy et al, 1999; Gibertini et al, 1999).

Whilst many blind subjects with no perception of light (conscious or unconscious) show their intrinsic periodicity in a normal environment, ie they 'free-run', or have an abnormal phase of melatonin, the amount of melatonin or aMT6s produced is comparable to that of sighted individuals (Orth et al, 1979; Lewy and Newsome, 1983; Lockley et al, 1997a) and the duration of melatonin secretion does not differ from normal sighted subjects (Klerman et al, 2001). Only a small number of subjects have been investigated to date (Table 3.3). Light suppression of melatonin in the blind is of course only possible in subjects with (sometimes unconscious, unperceived) retained circadian light perception (Czeisler et al, 1995). These observations suggest that in a normal entrained environment, light exposure of sighted subjects, even in the evening, does not impact greatly on melatonin production. Healthy male subjects kept in very dim light (<5 lux) for periods up to four weeks show free-running melatonin rhythms, but the production of aMT6s is not different from that found in normal lighting (Middleton et al, 1996). It is worth noting that urban humans with artificial lighting are not normally exposed to substantial changes in photoperiod.

Possible decreased incidence of breast cancer in blind subjects has been addressed as potentially relevant to the role of melatonin in breast cancer aetiology (see Chapter 5). As discussed above, there is no

TABLE 3.3 6-sulphatoxymelatonin (aMT6s) production and intrinsic rhythm periodicity (tau) in sighted healthy subjects in a normal environment, in blind subjects, and in sighted subjects kept for 3–4 weeks in continuous very dim light (<8 lux). Published data (Middleton et al, 1996, 1997; Lockley et al, 2000; Hack et al, 2003) and normal laboratory values using reagents from Stockgrand Ltd, University of Surrey

	aMT6s (ng per 24 h),		Intrinsic period decimal (h),
Subjects	X± SD	Ν	X± SD
Normal healthy sighted adults			
All ages, M and F	12601 ± 9528	182	24.00 entrained
Registered blind, no perception of light			
All ages, M and F	9350 ± 6380	30	24.49 ± 0.17 free-running
Registered blind, some light perception			
All ages, M and F	12650 ± 7450	19	24.00 entrained
Normal sighted young men, ages 20-30 years			
Normal environment	16410 ± 10020	65	24.00 entrained
Light <8 lux, partial temporal isolation	18900 ± 9120	12	24.33 ± 0.15 free-running

evidence that totally blind people have more melatonin than sighted people. However, distinctions are not always made between 'blind' subjects with no conscious perception of light but who may retain circadian photoreception, and the totally blind with no photoreception at all.

3.6.3 Seasonal variations

In temperate and polar zones, a delay in the timing of melatonin secretion in winter is often observed (Broadway et al, 1987; Bojkowski and Arendt, 1988; Midwinter and Arendt, 1991; Yoneyama et al, 1999). It is likely that decreased strength of the light-dark time cue in winter and a lengthening of the night lead to weak entrainment of the rhythm, since increasing light exposure (one hour bright white light in the early morning and evening) is sufficient to restore summer phase position (Broadway et al, 1987). There are also some inconsistent reports of an increase in duration of secretion and hence quantity of melatonin secreted in winter (Beck-Friis et al, 1984; Makkison and Arendt, 1991). Levels may also be increased in winter in high Arctic latitudes (Kauppila et al, 1987; Kivela et al, 1988), and it has been suggested that a reduced risk of cancer found in Arctic residents is due to greater melatonin secretion during the Arctic winter (see Chapter 5). There is certainly evidence for increased melatonin levels in the Arctic winter (which has been associated with anovulatory cycles (Kauppila et al, 1987). However, in Antarctic studies no difference in peak or daytime values was found between summer and winter, rather a change of timing of the rhythm (later in winter) was noted. If daytime melatonin appears to be increased, this may well in some cases be due to a delay in the timing of the peak levels and 'spill over' into the morning hours (Broadway et al, 1987; Midwinter and Arendt, 1991). This is particularly important to bear in mind if only a single morning sample (of plasma, saliva or urine) is taken.

Any delay or advance in melatonin secretion means that simply taking an early morning urine or plasma sample can give data suggesting increased or decreased production.

3.6.4 Association with core temperature

Many associations of melatonin with temperature exist in humans. The most striking is the reciprocal relationship in circadian profiles, where the temperature nadir correlates closely with the peak of melatonin (Figure 3.14). The increase in core temperature induced by light at night, and associated with melatonin suppression (see Table 3.1), could be at least partially opposed by replacement melatonin. Possibly half of the night-time decline in core temperature might be ascribed to the hypothermic effects of endogenous melatonin. A causal relationship is also indicated as exogenous melatonin can acutely depress body temperature in humans (Strassman et al, 1991; Cagnacci et al, 1992). The ovulatory rise in temperature during the menstrual cycle has been associated with a reported decline in the amplitude of melatonin, but the decline in melatonin is not a consistent observation.

3.6.5 Association with sleep

Obvious correlations are noted between melatonin production at night and sleep and, again, specific causal relationships may exist. However, sleep deprivation does not abolish the melatonin rhythm and, in very dim light, does not apparently affect secretion (Akerstedt et al, 1979). During sleep deprivation,



FIGURE 3.14 Relationship of plasma melatonin to other major circadian rhythms. Note the close correspondence between the core temperature nadir, the alertness nadir, the peak accident rate, slowest reaction time and fat metabolism with the melatonin peak. Reproduced from Rajaratnam and Arendt, 2001, by permission

self-rated fatigue exhibits a circadian rhythm that is closely correlated with plasma melatonin levels (Akerstedt et al, 1979). The close relationship between the evening rise of circulating melatonin and the evening increase in sleep propensity suggests a causal relationship (Wehr et al, 2001). There is a correlation between the timing of sleep spindles and certain other electroencephalographic (EEG) characteristics and the circadian phase of melatonin (Dijk et al, 1997). Careful observations in constant routine conditions of the profiles of melatonin, core body temperature, sleep propensity, sleepiness, cortisol, REM, etc, have led to a definition of biological night in humans (Wehr et al, 2001). This corresponds to the period during which melatonin is secreted, with an apparent switch process at 'dusk' and 'dawn' that is common to melatonin onset and offset, core temperature rise and decline, and increasing–decreasing sleep propensity. The so-called 'forbidden zone for sleep' or 'wake maintenance zone' occurs just prior to the onset of melatonin secretion (Shochat et al, 1997) and can be overcome by melatonin administration to advance the evening rise (Arendt et al, 1984, 1985a; Deacon and Arendt, 1995; Rajaratnam et al, 2004). The peak of melatonin secretion is also associated with the nadirs of alertness and performance (Carrier and Monk, 2000), together with increased blood lipid levels at night (Morgan et al, 1998) (Figure 3.14).

Patients suffering from non-24-hour sleep–wake disorder (largely blind subjects with no perception of light) intermittently secrete melatonin during the daytime. This is strongly associated with the presence of daytime naps (Lockley et al, 1997b). Smith-Magenis syndrome (caused by a deletion in chromosome 17p11.2) is associated *inter alia* with daytime melatonin production, poor night-time sleep, but with normal cortisol rhythms (De Leersnyder et al, 2001, 2003). Treatment of these patients with atenolol (a beta-adrenoreceptor antagonist) to suppress daytime melatonin was apparently successful in reducing daytime sleepiness, suggesting a contribution of the endogenous hormone to this phenomenon (De Leersnyder et al, 2001, 2003).

3.6.6 Association with posture

Changes in posture may affect circulating melatonin concentrations (Table 3.1). Abrupt transitions from recumbent to standing position increase melatonin, presumably because binding proteins retain this small molecule in blood whilst haemodynamic effects lead to diffusion of other unbound small molecules and water out of blood into the extracellular spaces, thereby concentrating bound molecules (Deacon and Arendt, 1994; Nathan et al, 1998). However, another study found no effect of posture itself or changes in posture on salivary melatonin (Voultsios et al, 1997).

3.6.7 Other associations

Obviously, any variable with a marked circadian rhythm shows correlations with melatonin – if necessary, displaced in time. Examples include cortisol, prolactin, thyroid-stimulating hormone, aspects of the immune system, and many others. Nocturnal exercise increases melatonin and can induce phase shifts (Strassman et al, 1989; Buxton et al, 2003). The relationships of stress and some other non-pharmacological interventions in modification of melatonin production are somewhat unclear and do not appear to play a major role in humans (see Table 3.1).

3.6.8 Development, puberty and ageing

Shortly after birth, very little melatonin or aMT6s is detectable in body fluids. A robust melatonin rhythm appears around 6 to 8 weeks of life (Kennaway et al, 1996). Whether in specific individuals this rhythm corresponds to the organisation and synchronisation of other circadian variables such as sleep remains a question of considerable interest. The plasma concentration of melatonin increases rapidly thereafter and reaches a lifetime peak on average at 3–5 years of age (for a review, see Arendt, 1995). The increment is much greater at night. Subsequently, a steady decrease is seen, with mean adult concentrations attained in the mid to late teens and the major decline occurring before puberty. Values remain relatively unchanged until 35–40 years of age (25–30 according to Kennaway et al, 1999), and a final decline in amplitude then takes place until (on average) low levels are seen in old age with the exception of one study using healthy elderly subjects (Kennaway et al, 1999; Zeitzer et al, 1999) Reports of an association of differences in secretion in adults with sex, height, or body weight are not consistent.

Although a lower melatonin concentration has been reported in children with precocious puberty and higher concentrations in those with delayed puberty and hypothalamic amenorrhea than in age-matched controls (reviewed by Arendt, 1995), these associations remain correlative and not causal. Ovarian suppression with a GnRH analogue in precocious girls was not accompanied by changes in melatonin secretion (Berga et al, 1989). However, in some case reports, induction of sexual development was associated with a decline in melatonin production (Arendt et al, 1989; Puig-Domingo et al, 1992).

Maternal melatonin appears to be transferred to the fetus in humans as it is in rats and hamsters (Okatani et al, 1998). Thus any manipulations of maternal melatonin levels may impact on the fetus. In photoperiodic species maternal melatonin influences timing of circadian phase and pubertal development (see Section 3.1). Moreover, circulating levels of plasma melatonin are reflected in the profile of melatonin in human milk (Illnerova et al, 1993). These observations imply that physiological 'insults' to the maternal circadian system (for example, by shift work or time-zone travel) may have effects on the progeny. However, in rats, manipulation of maternal milk melatonin does not appear to influence development (Rowe and Kennaway, 2002).

3.6.9 Menstrual cycle

Some of the very earliest reports on human melatonin described low preovulatory concentrations the morning before ovulation and suggested that low melatonin was facilitatory to the preovulatory LH peak. This observation is inconsistent, however, and more recent work indicates that neither the amplitude nor the phase of melatonin is altered in the course of the normal menstrual cycle (Brzezinski et al, 1988; Leibenluft et al, 1994; Parry et al, 1997; Wright and Badia, 1999). However, there is some evidence that oral contraceptives may increase melatonin (Kostoglou-Athanassiou et al, 1998; Wright and Badia 1999; Thapan, Arendt and Skene, unpublished data). In spite of numerous attempts made to relate melatonin to endogenous gonadal steroids in humans, little consistency emerges.

The effects of melatonin on core body temperature are reported to vary in the course of the cycle, and herein may lie a physiological function (Cagnacci et al, 1996, 1997). LH pulses are amplified in the early follicular phase by oral melatonin at 08.00 hours (Cagnacci et al, 1995). In very large doses (75–300 mg)

melatonin can suppress the ovulatory peak of LH secretion (Voordouw et al, 1992). However, attempts to develop melatonin as a contraceptive pill in combination with a synthetic progestin minipill have not been successful.

Very large doses (100 mg daily) of melatonin potentiate testosterone-induced LH suppression (Anderson et al, 1993b). A series of studies in males with and without hypogonadism has reinforced the perception that melatonin is essentially inhibitory to human reproductive function (see, for example, Luboshitzky et al, 1996, 1997, 2003; Luboshitzky and Lavie, 1999). Because humans appear to conceive more readily in long photoperiods, an explanation may reside in residual human photoperiodism (Roenneberg and Aschoff, 1990; Bronson, 2004). The results of these studies partially support the contention that melatonin, suitably administered, can inhibit human reproductive activity. Other data, however, indicate that over periods of eight days to several weeks, daily melatonin administration (low pharmacological doses) has no effect on a number of pituitary/gonadal hormones (Arendt et al, 1985a; Wright et al, 1986; Luboshitzky et al, 2000; Rajaratnam et al, 2003).

3.7 Effects of exogenous melatonin in humans

3.7.1 Early work

Enormous doses of up to 6.6 g (the daily production of about 200,000 people) in the daytime had no beneficial effects on Parkinsonism, Huntington's chorea, depression (which was worsened), and schizophrenia. Skin pigmentation was not affected: human pigment cells do not resemble amphibian melanophores in pigment migration phenomena. Small decreases in plasma LH and FSH were observed (see Arendt, 1995, for references). Large amounts of melatonin such as these may produce headache, abdominal cramps, and somnolence.

3.7.2 Acute effects

Much lower (2 mg intranasally, 0.3–240 mg orally or intravenously) doses of melatonin during the 'biological day', ie when endogenous melatonin levels are low, can induce transient sleepiness or sleep, and lower core body temperature, in suitably controlled circumstances. Posture is important; the greatest effects are seen with recumbent subjects in very dim light (see Arendt, 2003, 2005, for references). The initial evidence dates from 40 years ago when Aaron Lerner, who first isolated the substance, took 100 mg and described sleepiness after the dose. Early investigations used EEG characteristics to delimit an acute mild sedative and 'hypnotic' effect in both animals (cats, rats and chickens) and humans. Subsequently, a substantial body of literature, generally using much lower doses, has described advance shifts in the timing of sleep after early evening administration, transient sleepiness at several different times of day within two to four hours of the dose, time-dependent increases in sleep propensity, effects on the waking EEG comparable to but not identical with benzodiazepines, a lengthening of the first rapid eye movement episode after early evening administration, increases in the fast EEG frequencies after evening naps or night-time sleep, and 'beneficial effects' when taken at bedtime. The subject has been extensively reviewed recently (Zhdanova et al, 1997; and see *Sleep Medicine Reviews*, volume 9, 2004). A brief decline in performance but with no effect on memory, has

been reported after daytime melatonin administration (Lieberman et al, 1984; Graw et al, 2001). Acute oral doses of melatonin stimulate prolactin secretion (Wright et al, 1986; Waldhauser et al, 1987). This enhanced prolactin secretion may relate to the ability of melatonin in pharmacological amounts to inhibit some dopaminergic functions. Acute effects on other pituitary hormones are somewhat inconsistent, although recently a relationship between melatonin and vasopressin secretion has been established (Forsling et al, 2000). The acute pharmacological properties of melatonin in animals include sedation, hypothermia, anxiolysis, muscle hypotonia, decrease in locomotor activity with a rebound increase on increasing the dose, slight analgesia, slight protection against electroconvulsive shock, constriction of cerebral arteries, potentiation of noradrenalin-induced vasoconstriction, and very low toxicity (Guardiola-Lemaitre, 1997).

3.7.3 Phase shifting

In early work, daily feeding of low dose (2-5 mg) melatonin in the late afternoon advanced the timing of evening self-rated fatique, the endogenous melatonin rhythm, and the morning decline in prolactin when compared with placebo. No significant effects were seen on self-rated mood or on LH, FSH, testosterone, cortisol, growth hormone, or thyroxine. No deleterious effects were reported by the subjects (Arendt et al, 1985a). Thus in low doses, melatonin has some chronobiotic effects in humans. Melatonin has rapid, transient, mild sleep-inducing effects and lowers alertness and body temperature during the 3 to 4 hours after low doses (0.5-5 mg) during the daytime, these effects being opposite to the acute effects of bright light given at night. In the same dose range it is able to shift timing of the internal clock to both later and earlier times when administration is appropriately timed (Figure 3.15). As for light, the appropriate timing can be predicted from a phase-response curve in subjects whose body clock phase is known (Lewy et al, 1998) (Figure 3.15). The phase-response curve to melatonin is essentially the reverse of that to light. Melatonin given approximately 8-13 hours before core temperature minimum will produce phase advances and, when given around 1-4 hours after core temperature minimum, will produce phase delays, although there is some controversy regarding the ability of melatonin to phase delay the circadian system. A recent, carefully controlled study (Rajaratnam et al, 2003, 2004) showed convincingly that daily administration of a 'surge sustained' release preparation (1.5 mg) of melatonin at 16.00 hours followed by recumbency and very dim light for 16 hours led to substantial phase advances of a number of circadian 'marker' rhythms and an advanced timing and redistribution of sleep during the dark phase. No increase in total sleep time was seen, reinforcing the view that melatonin acts on the timing mechanisms of sleep rather than being a 'hypnotic'. There were no deleterious effects on pituitary/gonadal hormones or daytime alertness. In this dose, reinforced by recumbency and dim light, melatonin is clearly very effective at shifting the circadian clock and sleep timing (Rajaratnam et al, 2003, 2004).

3.7.4 Entrainment

Melatonin can maintain entrainment to 24 hours in sighted subjects kept in a dim light environment conducive to free-running (Middleton et al, 1997). Most importantly perhaps, recent data have shown that timed melatonin administration (0.5–5 mg at 24 hour intervals, usually at desired bedtime) can fully entrain (or synchronise) the free-running circadian rhythms of most blind subjects exhibiting this

phenomenon, with a consequent improvement in sleep and daytime alertness. These observations, from two independent laboratories, indicate that melatonin can be as effective as light for circadian rhythm management, and is the treatment of choice for non-24-hour sleep–wake disorder of the blind (Arendt, 2000; Lockley et al, 2000; Sack et al, 2000; Lewy et al, 2001; Hack et al, 2003).



FIGURE 3.15 'Phase response curve' (PRC) to melatonin administration (from data in Zaidan et al, 1994; Middleton et al, 1997; Lewy et al, 1998). Melatonin (0.05–10 mg) given in the 'biological evening' advances the internal clock as evidenced by changes in timing of the endogenous melatonin rhythm, together with those of sleep, core body temperature, cortisol and other circadian rhythms assessed to date. In the 'biological early morning' delays are reported but remain controversial. Timed light treatment has essentially opposite effects on circadian rhythm timing and the PRC to light is approximately a mirror image of that to melatonin. Circadian time (CT) is a convention whereby the circadian period is divided into 'circadian hours' and referenced either to the melatonin onset (CT14) or to the core body temperature minimum (CT0). In the present illustration, circadian period in an entrained individual is 24 hours; however, the technique permits comparison of data from free-running individuals with periods differing from 24 hours

3.7.5 Jet lag and shift work

Melatonin treatment timed to induce phase advances and delays has been used in the alleviation of jet lag in at least 14 studies of real-life and simulation conditions, 11 of which reported beneficial effects. Field studies show that self-rated jet lag both westward and eastward can be reduced on average by 50% with appropriately timed treatment (Arendt et al, 2005). A recent Cochrane review concluded that melatonin was effective if used correctly to adapt to time-zone change (Herxheimer and Petrie, 2002). Exposure to bright light sufficient to suppress melatonin secretion during the night is clearly beneficial to night-shift workers in terms of alertness and performance (Czeisler et al, 1990, Bjorvatn et al, 1999). Preliminary work suggested improved sleep and increased daytime alertness in night-shift workers receiving melatonin at the desired bedtime (during the day) during a night-shift week as compared with placebo and baseline conditions (Folkard et al, 1993). A number of recent studies have successfully used melatonin to adapt to simulated or real shift work (Burgess et al, 2002), although it has to be said that several reports in the literature have shown no beneficial effects. Questions of posture, light environment and timing need to be resolved in field studies.

3.7.6 Blindness

Only a small number of subjects have been studied to date, but undoubtedly most blind/visually impaired subjects reporting sleep problems derive benefit from melatonin ingestion at the desired bedtime or specifically timed according to the circadian phase (Arendt et al, 1997; Arendt, 2000). Melatonin was strikingly effective at improving sleep and behaviour in multiply handicapped children with or without visual impairment (Jan et al, 1994).

3.7.7 Sleep problems of old age

Increasing numbers of positive reports now indicate that some elderly subjects will derive benefit from melatonin administration, including Alzheimer patients. It is possible that very low doses are more useful than 2–5 mg and that improvement is seen when an underlying rhythm disorder is present rather than non-specific insomnia. Dose timing and formulation remain to be optimised.

3.7.8 Delayed-sleep phase insomnia

Patients with delayed-sleep phase insomnia cannot sleep at the socially acceptable time of night; they delay sleep onset until the early hours of the morning and sleep through much of the day. This condition has been successfully treated with bright light in the early morning to induce phase advances of the clock. In others, evening melatonin (5 mg at 22.00 hours or 5 hours ahead of endogenous melatonin onset) also advances sleep time significantly. Judicious, timed application of both melatonin and bright light as time cues may well be the treatment of choice for such rhythm disturbances (Arendt and Skene, 2005).
3.7.9 Cancer

The preceding sections on the effects of melatonin have largely dealt with it as a chronobiotic, in line with its accepted physiolological properties. The approach to its possible use in cancer treatment has been more concerned with free radical scavenging and immunostimulant properties. However, chosen treatment times (usually early or later evening) are likely to induce phase advances and conceivably more optimal circadian phase, with consequent benefits for sleep and the functions related to sleep.

A few reports of positive effects of combination therapy – for example, melatonin and tamoxifen, melatonin and interleukin – require confirmation (Lissoni et al, 1996, 2000, 2003). Recently, the emphasis on the potential use of melatonin in cancer chemotherapy has moved to propose that it is effective at reducing the toxicity and increasing the effectiveness of existing treatments (Vijayalaxmi et al, 2002). Another consideration is the use of melatonin to 'optimise' circadian phase with regard to timed cancer treatment (chronotherapy). With the current striking data concerning the influence of clock gene function on cancer development (Fu et al, 2002; Fu and Lee, 2003) the potential influence of melatonin on clock gene function urgently needs further investigation.

3.8 Mechanisms of melatonin action

Melatonin, as described above, has been implicated in a diverse range of actions, and it implements these actions through a variety of mechanisms. The specific effect induced depends on the type of cell and the particular molecules that the cell possesses to interact with the hormone. Some cells possess receptors that bind melatonin but the hormone also has other routes of action.

3.8.1 Melatonin receptors

The development of 2-¹²⁵I-iodomelatonin as a high specific activity ligand has permitted the identification of high affinity (equilibrium dissociation constant, K_d , of 25 to 175 pM), saturable, specific, and reversible melatonin binding to cell membranes in the central nervous system, initially in the SCN (Vanecek et al, 1987) and the pars tuberalis of the pituitary (Morgan and Williams, 1989) and subsequently in many brain and other areas, including cells of the immune system, a number of cancer cell lines, the gonads, the kidney and, importantly, the cardiovascular system. The SCN shows clear binding in human postmortem tissue (Reppert et al, 1996). Species variation of melatonin-binding sites in the brain is of course apparent. The most consistent (but not universal) binding site between mammalian species is the pars tuberalis, primarily implicated in transduction of the effects of photoperiod, via melatonin, on seasonal variations in prolactin secretion in ruminants (Lincoln and Clarke, 1994).

Krause and Dubocovich (1990) have demonstrated a functional melatonin receptor initially in rabbit and chicken retina (inhibition of calcium-dependent dopamine release) that is localised in dopamine-containing amacrine cells in the inner plexiform, in the outer and inner segments in mice, and possibly in the pigmented layer in some mammals.

The interaction of melatonin with nuclear receptors (RZR/ROR alpha and RZR beta) and intracellular proteins, such as calmodulin or tubulin-associated proteins has also been reported. The transcription

factor RZR/ROR alpha may mediate a direct gene regulatory action of the hormone. It has been hypothesised that while the effects of melatonin on circadian and seasonal rhythms appear to use the membrane receptors, peripheral effects of melatonin (such as immunomodulatory effects) may largely be mediated by RZR/ROR alpha (Carlberg, 2000).

3.8.2 Melatonin receptor pharmacology

White et al (1987) initially demonstrated that melatonin-induced pigment aggregation in amphibian melanophores is a pertussis toxin-sensitive system and that melatonin inhibits forskolin-activated cAMP formation. Inhibition of cAMP production may be a general feature of melatonin receptors. Intensive investigation of the properties of the pars tuberalis-binding site has revealed that physiological doses of melatonin inhibit forskolin-activated cAMP production *in vitro* in a time- and dose-related manner (Morgan and Williams, 1989). Other studies have provided good evidence that most binding sites are coupled to G proteins (Reppert, 1997). Guanosine triphosphate analogues, which interfere with the regeneration of G_1 -coupled receptors, decrease the affinity and sometimes the capacity of ¹²⁵I-melatonin binding in reptiles, birds and mammals.

Melatonin receptors have been cloned, and three subtypes were initially named Mel-1a, Mel-1b and Mel-1c (Reppert, 1996; Reppert et al, 1996). The Mel-1a receptor gene has been mapped to human chromosome 4q35.1. Its primary expression is in the pars tuberalis of the pituitary and the SCN. Mel-1b has been mapped to chromosome 11q21-22 and its main expression is in the retina and the brain. Mel-1c is not found in mammals. Two cloned mammalian receptors (Mel-1a and Mel-1b) have now been renamed MT1 and MT2 (Dubocovich et al, 1998, 2003; Masana and Dubocovich, 2002). They are a new family of G protein coupled receptors, have high affinity (K_d 20–160 pM) and inhibit forskolin-stimulated cAMP formation. Research is underway to investigate the possible relationships between polymorphisms in melatonin receptors and circadian rhythm disorders (Ebisawa et al, 2000).

3.8.3 Oestogen receptors

The effect of melatonin on oestrogen receptors is considered in Section 3.9.

3.8.4 Melatonin antagonists and agonists

Large numbers of putative and actual melatonin agonists together with some antagonists have now been described. It is expected that much new information in this area will be available shortly; however, none of these agents is available as yet as a registered medication in the UK.

3.8.5 Gap junction communication

Gap junction communication between adjacent cells plays a normal part in intercellular exchange of small molecules and has a vital role in regulating cell growth and differentiation. The loss or a decrease of this avenue of communication has been observed in many cancer cells or cells treated with carcinogenic agents. The role of gap junction intercellular communication has been reviewed by a number of authors (see, for example, Ruch, 1994; Sulkowski, 1999; Trosko, 2003).

There have been suggestions that melatonin may play a role in gap junction communication. Kojima et al (1997) found that melatonin markedly induced gap junction protein expression and gap junction intercellular communication, although this was achieved at a pharmacological concentration of melatonin (10^{-2} M) . At lower concentrations, Cos and Fernandez (2000) showed that melatonin $(10 \ \mu\text{M} \text{ or } 1 \ \text{nM})$ could increase the transfer of small molecules through gap junctions in the breast cancer cell line MCF-7. Similarly, Ubeda and colleagues showed that physiological levels of melatonin enhanced junctional transfer in mouse embryonic fibroblasts and hepatocytes (Ubeda et al, 1995; Blackman et al, 2001).

3.8.6 Free radicals

Free radicals are created by the transfer of an electron between molecules; the result is unstable, transitory and potentially damaging to biological systems, particularly if DNA is involved in the electron transfer. However, the production of free radicals is a normal occurrence – it is a process that occurs continuously in biological systems and for which there are very effective defence mechanisms.

Many studies have assessed the role of melatonin as a scavenger of free radicals. Being an easily oxidised molecule, melatonin does have antioxidant activity. It has been suggested that this activity was its primary evolutionary function in primitive species (Hardeland et al, 1995; Antolin et al, 1997). However, the quantities of exogenous melatonin required to generate significant antioxidant activity *in vivo* remain to be specified. It is evident that melatonin, at least at pharmacological levels, is an effective antioxidant and infers protection against the damage caused by production of free radicals (see reviews by Reiter, 1998, 2004, and Vijayalaxmi et al, 2002). Its role as a free radical scavenger at physiological concentrations and hence *in vivo* is less convincing, as there are many other cellular candidates such as glutathione, vitamin A, or vitamin E that are found in much higher natural concentrations than melatonin, and therefore potentially play a greater role in cellular defence. There is some evidence that the overall antioxidant activity of blood is correlated with melatonin content, at least in chicks (Albarran et al, 2001). Correlation does not of course establish causality. It is difficult to reconcile the fact that most melatonin produced is excreted as aMT6s, whereas if the molecule was oxidised other primary metabolites would be likely (eg *N*-acetyl *N*-formyl 5-methoxykynurenamine).

3.8.7 Immunology

Melatonin is thought to play an important enhancing role in the immune system. The status of the immune system is an important factor not only in combating infections but also with respect to the development of cancerous cells. Immunodepression can be induced experimentally in animals by the removal or suppression of melatonin synthesis and can be counteracted by melatonin supplementation. Natural fluctuations in the duration of melatonin production also occur and are related to the photoperiod. These seasonal variations have been linked to immune parameters (reviewed by Nelson et al, 1995) – for example, short daylength was shown to enhance immune function in laboratory studies. Therefore immunity appears to be linked to the season and mediated through melatonin production. One route of possible melatonin enhancement of the immune system may be mediated through T-helper cells. These cells express melatonin receptors on the cell membrane that once activated stimulate the release of cell cytokines to aid in the immune response. Melatonin may also stimulate the immune response by influencing specific progenitor cells such as natural killer cells, monocytes and pre-B cells. These studies have been reviewed elsewhere (Fraschini et al, 1998; Maestroni, 1999, 2001).

3.8.8 Other mechanisms

Several other mechanisms of melatonin action have been proposed. For example, it has been suggested that intracellular calcium is modulated by melatonin via its effect on the calmodulin-calcium ion signalling pathway (Dai et al, 2002). Mediavilla et al (1999) suggested that melatonin was involved in control of the cell cycle by inducing the expression of key proteins. Blask et al (1997) showed that glutathione was required for melatonin action; indeed the breast cancer line HS578T, which is normally insensitive to melatonin, could be made responsive by raising the intracellular concentration of glutathione.

3.9 Melatonin, other hormones and breast cancer

Substantial evidence exists that steroid hormones heavily influence the natural history of breast cancer. This is reviewed below together with the observation that melatonin may mediate or interact with the mechanism of hormone action.

3.9.1 Hormones and breast cancer

There is compelling evidence that hormones, in particular oestrogens, both increase risk of developing breast cancer and maintain the continued growth/progression of established tumours.

3.9.2 Role of hormones in the development of breast cancer

Many aetiological factors associated with breast cancer have a hormonal component (Miller, 1996). Thus, enhanced susceptibility to breast cancer appears linked to increased oestrogen exposure whether from endogenous (Henderson et al, 1982; Toniolo et al, 1995; Berrino et al, 1996; Dorgan et al, 1997; Thomas et al, 1997a,b,c; Hendersen and Feigelson, 2000) or exogenous sources (Steinberg et al, 1991; Agarwal and Judd, 1999; Ursin, 2002). Conversely, oestrogen deprivation protects against the disease (Feinleib, 1968; Trichopoulos et al, 1972). For example, ablation of ovarian function before the age of 35 years reduces subsequent appearance of breast cancer by two-thirds (Feinleib, 1968). The mechanisms by which this is achieved are still not completely understood, but there are several possibilities:

- (a) oestrogens can increase the proliferation of breast epithelial cells which leads to genetic mistakes and results in a transformed cellular phenotype (Fiegelson et al, 1996),
- (b) oestrogens can accelerate the growth of occult cancers leading to increased incidence of overt disease (Miller, 1996),
- (c) there is also accumulating evidence that oestrogen metabolites are genotoxic (Yager and Liehr, 1996); in particular, metabolism of oestrogens via catechols (2- or 4-hydroxyestradiol or hydroxyestrone) may produce reactive quinones, which can directly interact with and mutate DNA, initiating carcinogenesis (Roy and Liehr, 1999; Santen et al, 1999; Raftogianis et al, 2000).

The enzymes responsible for this metabolic pathway may be raised in women at risk of breast cancer (Fiegelson et al, 1996; Santen et al, 1999; Raftogianis et al, 2000; Yue et al, 2003).

3.9.3 Oestrogens and established breast cancer

Ablative procedures that either destroy ovarian function in premenopausal women or reduce circulating oestrogens in postmenopausal women may cause the regression of established breast cancer (Luft et al, 1952; Huggins and Dao, 1953; Dao and Huggins, 1955; Pearson and Ray, 1959; Beatson, 1896; Newsome et al, 1997). Consequently, drugs that inhibit either the synthesis or action of hormones occupy a central role in the therapy of women with breast cancer (Jensen et al, 1967; Horwitz et al, 1975; McGuire, 1978).

Given these associations between reproductive hormones and the natural history of breast cancer, it is important to examine relationships between melatonin and these hormones – in particular, whether melatonin might influence levels of hormones or their signalling pathways.

3.9.4 Melatonin and steroid hormone levels

One of the initial theories suggesting that a diminished function of the pineal gland might promote the development of breast cancer hypothesised that melatonin suppression may lead to increased levels of reproductive hormones, particularly oestradiol, thereby stimulating the growth and proliferation of hormone-sensitive cells in the breast (Cohen et al, 1978; Stevens, 1987).

Lifetime changes in patterns of steroid hormones and melatonin differ considerably. Thus whilst oestrogens and other steroid hormones fall dramatically at the menopause with cessation of ovarian function (Miller, 1996), this general pattern is not seen in melatonin secretion (see Section 3.6) and there are no dramatic changes in melatonin levels at the menopause.

In contrast, within women of the same menopausal status there is some evidence of an inverse relationship between steroid hormones and melatonin. For example, a significant negative correlation has been observed between peak serum melatonin concentrations and those of 17-beta-oestradiol in premenopausal women (Okatani et al, 2000). Additionally, administration of melatonin reduces oestrogens in women. For example, melatonin (2 mg daily for 6 months) significantly reduced oestradiol concentrations in postmenopausal women aged between 64 and 80 years (Pawlikowski et al, 2002). (These observations would be consistent with the proposed action of melatonin in down-regulating hormones of the neuroendocrine axis leading to a decrease in circulating levels of steroids.) Conversely, daily oral administration of oestrogen (conjugated oestrogen, 0.625 mg) suppressed nocturnal melatonin (Okatani et al, 2000). There has also been a report of a relationship between plasma, nocturnal melatonin concentrations and the oestrogen receptor status of breast cancers (Tamarkin et al, 1982). Thus, women with oestrogen receptor positive (ER+) breast cancers had significantly lower concentrations of melatonin than patients with oestrogen receptor negative (ER-) breast tumours and healthy women.

Definitive studies in which melatonin and other circulating hormones have been measured and related to risk of breast cancer are comparatively few in number and have not produced conclusive evidence that melatonin levels are associated with breast cancer. Most notably, the limited data currently available from prospective studies are inconclusive on whether women who subsequently developed breast cancer had different melatonin levels from those who did not. Interestingly, previous studies derived from the same or similar populations of women had shown increased levels of circulating oestrogens in women who subsequently developed breast cancer (Thomas et al, 1997a,b; Hankinson et al, 1998).

3.9.5 Interaction of melatonin with hormone signalling

Evidence of a direct anti-oestrogenic effect of melatonin on breast cancer cells has come from *in vitro* studies. Most investigations have used the MCF-7 breast cancer cell line, which possesses oestrogen receptors (Soule et al, 1973) and whose growth appears dependent upon oestrogen, both in culture (Lippman et al, 1977) and when grown as xenografts in immunosuppressed animals (Soule and McGrath, 1980). In these experimental models cell proliferation induced by oestrogen was inhibited by melatonin at concentrations close to those found in nocturnal human plasma. More recently, results have been presented which suggest that these oncostatic properties of melatonin are dependent upon a viable oestrogen response pathway. Such data include the observations that melatonin:

- (a) inhibits proliferation only in cells expressing ERα (Cos and Sanchez-Barcelo, 2000) (although only MCF-7 cells were inhibited at physiological concentrations); melatonin had no effect on oestrogen insensitive breast tumour cell lines,
- (b) both blocks the mitogenic effects of oestrogen and antagonises oestrogen-induced invasion (Cos et al, 1998),
- (c) potentiates the sensitivity of MCF-7 cells to anti-oestrogens (Wilson et al, 1992),
- (d) inhibits the expression of oestrogen-regulated genes such as pS2 or cathepsin (Molis et al, 1995).

Whilst melatonin does not bind to the oestrogen receptor or interfere with the binding of oestradiol to its receptor (Molis et al, 1994), it decreases the expression of ER α and inhibits the binding of oestradiol-ER complexes to oestrogen response elements on DNA, suppressing oestrogen receptor gene transcription and decreasing the levels of oestrogen-receptor mRNA expression (Molis et al, 1994). The molecular basis of such interactions has not been completely elucidated but cAMP and other protein kinase activators appear to enhance ER-mediated transcription by mechanisms involving phosphorylation of the oestrogen receptor itself and receptor co-activators (Kiefer et al, 2002). It is thus relevant that melatonin, after binding to its membrane receptor, has been reported to decrease cAMP, thus potentially reducing co-activation of the oestrogen receptor pathway (Kiefer et al, 2002).

Conversely, there is a report that melatonin interacts with growth factors such as insulin and epidermal growth factor (EGF) to modulate the transcription of oestrogen receptor in the absence of oestrogen, an effect abolished by the addition of an anti-oestrogen (Cos and Blask, 1994; Ram et al, 1998). This suggests that melatonin and EGF co-operate to transactivate the oestrogen receptor, an effect that is not immediately compatible with melatonin's growth suppressive properties.

Whilst these data are persuasive of an effect of melatonin on oestrogen receptor signal transduction, which may translate into growth inhibitory effects in hormone sensitive cell lines, it should be noted that this has only been observed in model systems comprising a limited number of breast cancer cell lines. Furthermore, the literature is not completely consistent and others failed to show a reduction of oestrogen-stimulated cell growth by melatonin (see Table 5.1). Thus Cos and Blask (1994) and Baldwin et al (1998) found no inhibitory effect of melatonin when oestrogen was present and Bizzarri et al (2003) reported that melatonin blocks oestrogen stimulation of cell growth but only in the absence of fetal calf serum. These observations raise questions about the *in vivo* importance of melatonin as an oncostatic agent, especially in the presence of other naturally occurring hormones or growth factors.

Another potential interaction between melatonin and oestrogen has been reported in respect of oxidative damage to DNA. Thus melatonin has been observed to prevent oestrogen-induced DNA

damage and to synergise with oestrogen in reducing lipid peroxidation in liver and kidneys (no measurements were made in breast) (Karbownik et al, 2001).

Further research into these basic molecular mechanisms and their relevance to the clinical material is required before definitive conclusions can be reached.

3.9.6 Other hormones, growth factors and breast cancer

Although the links with the natural history of breast cancer are greatest for sex steroids, other hormones and growth factors such as prolactin and the insulin-like growth factor system have been associated with increased risk and progression of breast cancer.

Prolactin is a mitogen in the mammary gland in rodents (Reiter, 1980; Brzezinski, 1997) and also has a primary involvement in the development of mammary cancer in such animals (Vonderhaar, 1984, 1987). Whether prolactin has similar influences in humans is less clear. There are data that show:

- (a) the presence of specific receptors for prolactin on breast cancer cells, (Welsch and Nagasawa, 1977; Codegone et al, 1981; Peyrat et al, 1981; Biswas and Vonderhaar, 1987),
- (b) about 90% of primary breast cancers also express mRNA for prolactin receptors (Bonneterre et al, 1982),
- (c) the growth of breast cancer cells may be stimulated by prolactin (L'Hermite-Baleriaux et al, 1984; Vonderhaar, 1987, 1998),
- (d) melatonin may block the prolactin-induced growth of breast cancer cell lines (Malarkey et al, 1983),
- (e) melatonin acutely increases prolactin according to several reports (Vonderhaar and Biswas, 1987; Lemus-Wilson et al, 1995).

However, whilst the majority of breast cancer biopsies contain prolactin-like material and express specific prolactin receptors, no consistent correlation is evident between prolactin/prolactin receptors and the aetiology/prognosis of breast cancer (Love and Rose, 1985; Wang et al, 1986; Ingram et al, 1990; De Placido et al, 1990; Love et al, 1991; Holdaway et al, 1997). Furthermore, it has been difficult to demonstrate tumour regression in breast cancer patients treated with prolactin-inhibiting drugs (Henson et al, 1972; Pearson and Manni, 1978; Manni et al, 1989; Anderson et al, 1993a).

Growth hormone is involved in the development of normal breast and has been hypothesised to be involved in the development and progression of breast cancer (Waters and Conway-Campbell, 2004). However, these influences may be mediated through insulin-like growth factor (IGF-1) and a direct role in breast cancer development is still unproven. Furthermore the effects of melatonin on growth hormone levels in women are variable and usually insignificant (Waldhauser et al, 1987; Forsling et al, 1999; Rajaratnam et al, 2003).

There are both epidemiological and biological data to link the IGF system with risk of breast cancer. Thus the majority of studies in premenopausal women have shown that high circulating levels of IGF-1 and IGFBP-3 increase risk of developing breast cancer; however, there was no consistent effect in postmenopausal women (Fletcher et al, 2005). Laboratory studies also demonstrate that the insulin-like growth factor system may stimulate neoplastic growth through mitogenic and anti-apoptotic effects on breast cells (Schairer et al, 2004; Voskuil et al, 2004). It has also been suggested that the oncostatic

effects of melatonin on breast cancer cells may be mediated through IGF-1 (Kajdaniuk et al, 2002). This is based on

- (a) a negative correlation between circulating levels of melatonin and IGF-1 in women with breast cancer (Kajdaniuk et al, 2002),
- (b) differences in seasonal variation of IGF-1 and melatonin between women with breast cancer and controls (Holdaway et al, 1997),
- (c) melatonin therapy having clinical benefits in patients with advanced breast cancer, an effect associated with decreases in IGF-1 levels, these being significantly higher in responding tumours (Lissoni et al, 1995).

The immediate relevance of these studies, which involve small numbers of women with established breast cancer, to risk of the disease may be questioned.

3.10 Summary

The pineal gland is a photoneuroendocrine transducer organ, converting information about daylength (and possibly light intensity) into a hormonal signal: melatonin.

Melatonin is normally secreted during the night (the dark phase) in all species whether nocturnal or diurnal, and the duration of its night-time secretion indicates the length of the night. One definition of 'biological night' is the period of the 24 hour cycle for which melatonin is above baseline (usually daytime) concentrations. In seasonal mammals the profile of melatonin secretion provides an essential time cue for the organisation of seasonal activity (such as reproduction, winter or summer coat growth).

In so far as human physiology is dependent on daylength, melatonin is likely to serve as a photoperiodic signal. Its role within the circadian system of mammals including humans appears to be the reinforcement of 'night-time' physiology (for example, changes in core body temperature or sleep propensity), the modulation of the circadian phase shifting response to light, and in general to serve as an endogenous zeitgeber. Most is known of its actions on the central nervous system. Being highly liposoluble, it penetrates all tissues and body fluids and it therefore has the potential to influence peripheral oscillators. Evidence exists for an effect on clock gene expression in the pars tuberalis (the major site of melatonin receptor MT1). The presence of melatonin did not appear to be essential for maintained circadian function in a normal environment in early animal studies. However, pinealectomy is associated with increased incidence of cancer (chemically induced and spontaneous), hypertension, metabolic abnormalities and an abnormal response to abrupt phase shifts in rats. Endogenous melatonin probably serves to reinforce 'coupling' and to optimise phase within the circadian system, and thus influence the multitude of systems governed by circadian oscillators.

In mammals, melatonin is a 'hand' of the circadian clock rather than part of the clock mechanism. It is extensively used as the 'best' marker of the timing of the circadian clock either as the endogenous hormone or its major metabolite 6-sulphatoxymelatonin (aMT6s). There are very large individual variations in the concentrations recorded in plasma or saliva and in the urinary content of aMT6s. Thus for cross-sectional studies of changes in melatonin secretion, large study populations are required. A single sample, be this of morning urine, morning or night-time plasma or saliva, does not provide sufficient

information to define the characteristics of its secretion. There is an age-related decline in secretion and an earlier timing of the rhythm with age. Thus age-matched control populations or age-adjustment are required for comparative purposes. Light exposure must be controlled if the acute effects of light are to be avoided. Numerous drugs and other substances influence its secretion.

Light suppresses melatonin synthesis at night and timed exposure to light of suitable intensity and spectral composition shifts the melatonin rhythm and all other circadian rhythms investigated to date. Night-shift work and time-zone change may be associated with lower melatonin production, light at night being one of several possible causal factors. Few data address this point as yet.

Exogenous melatonin phase shifts and entrains circadian rhythms. It is successfully used in the treatment of circadian rhythm disorders. It possesses free radical scavenging and (in some reports) immunostimulatory properties, usually in 'pharmacological' doses. Whether these prove to be clinically useful remains to be seen.

Melatonin has a wide range of actions; the mechanism of melatonin action is via an effect on other biological molecules, although the particular mechanism depends on the type of cell. The actions can be mediated through binding to specific melatonin receptors or by modifying the influence of other molecules, such as inhibiting the binding of the oestrogen receptor to DNA. An area that has received much interest and speculation is the role of melatonin as a scavenger of free radicals. Melatonin is an effective antioxidant at least when used at pharmacological levels; however, it is one of many biological antioxidants and its role at physiological levels is not so clear. Several other mechanisms of melatonin action have also been demonstrated and they highlight the diversity of effects caused by this hormone and the complexity of its action.

While there is compelling evidence that oestrogens increase the risk of breast cancer, the hypothesis that suppression of melatonin may lead to increased levels of reproductive steroid hormones which in turn stimulate either the proliferation of breast cells or the growth of breast cancers is still hypothetical and without a substantial body of supportive evidence. Similarly, the case that melatonin may increase breast cancer risk through changing the levels of growth factors or hormones such as prolactin is not convincing. Data are limited and relate to small groups of women with established breast cancer.

Some *in vitro* data suggest that melatonin may have direct anti-oestrogenic effects on breast cancer cells. However, this has only been observed using a limited number of breast cancer cell lines, and other studies have failed to show any reduction of oestrogen-stimulated cell growth by melatonin.

3.11 References

Agarwal SK and Judd HL (1999). Estrogen replacement therapy and breast cancer. Fertil Steril, 71, 602–3.

- Akerstedt T (1998). Is there an optimal sleep-wake pattern in shift work? *Scand J Work Environ Health*, **24**(Suppl 3), 18–27.
- Akerstedt T, Froberg JE, Friberg Y and Wetterberg L (1979). Melatonin excretion, body temperature and subjective arousal during 64 hours of sleep deprivation. *Psychoneuroendocrinology*, **4**(3), 219–25.
- Albarran MT, Lopez-Burillo S, Pablos MI, Reiter RJ and Agapito MT (2001). Endogenous rhythms of melatonin, total antioxidant status and superoxide dismutase activity in several tissues of chick and their inhibition by light. *J Pineal Res*, **30**(4), 227–33.

- Aldhous ME and Arendt J (1988). Radioimmunoassay for 6-sulphatoxymelatonin in urine using an iodinated tracer. *Ann Clin Biochem*, **25**(Part 3), 298–303.
- Anderson E, Ferguson JE, Morten H, Shalet SM, Robinson EL and Howell A (1993a). Serum immunoreactive and bioactive lactogenic hormones in advanced breast cancer patients treated with bromocriptine and octreotide. *Eur J Cancer*, **29A**, 209–17.
- Anderson RA, Lincoln GA and Wu FC (1993b). Melatonin potentiates testosterone-induced suppression of luteinizing hormone secretion in normal men. *Hum Reprod*, **8**(11), 1819–22.
- Antolin I, Obst B, Burkhardt S and Hardeland R (1997). Antioxidative protection in a high-melatonin organism: the dinoflagellate *Conyaulax polyedra* is rescued from lethal oxidative stress by strongly elevated, but physiologically possible concentrations of melatonin. *J Pineal Res*, **23**(4), 182–90.
- Arendt J (1978). Melatonin assays in body fluids. J Neural Transm Suppl, 13, 265–78.
- Arendt J (1979). Radioimmunoassayable melatonin: circulating patterns in man and sheep. *Prog Brain Res*, **52**, 249–58.
- Arendt J (1985). Mammalian pineal rhythms. Pineal Res Rev, 3, 161-213.
- Arendt J (1986). Role of the pineal gland and melatonin in seasonal reproductive function in mammals. *Oxf Rev Reprod Biol*, **8**, 266–320.
- Arendt J (1988). Melatonin. Clin Endocrinol (Oxf), 29(2), 205-29.
- Arendt J (1989). Melatonin: a new probe in psychiatric investigation? Br J Psychiatry, 155, 585-90.
- Arendt J (1995). Melatonin and the Mammalian Pineal Gland. London, Chapman Hall.
- Arendt J (1996). Melatonin. BMJ, 312(7041), 1242-3.
- Arendt J (2000). Melatonin, circadian rhythms, and sleep. N Engl J Med, 343(15), 1114-16.
- Arendt J (2003). Importance and relevance of melatonin to human biological rhythms. *J Neuroendocrinol*, **15**(4), 427–31.
- Arendt J (2005). Melatonin: characteristics, concerns and prospects. J Biol Rhythms, 20(4), 291-303.
- Arendt J and Broadway J (1986). Phase response of human melatonin rhythms to bright light in Antarctica. *J Physiol*, **377**, 68P.
- Arendt J and Broadway J (1987). Light and melatonin as zeitgebers in man. Chronobiol Int, 4(2), 273-82.
- Arendt J and Skene DJ (2005). Melatonin as a chronobiotic. Sleep Med Rev, 9, 25-39.
- Arendt J, Paunier L and Sizonenko PC (1975). Melatonin radioimmunoassay. J Clin Endocrinol Metab, **40**(2), 347–50.
- Arendt J, Ho A K, Laud C, Marston A, Nohria V, Smith JA and Symons AM (1981). Differential effect of benserazide (Ro4-4602) on the concentration of indoleamines in rat pineal and hypothalamus. *Br J Pharmacol*, **72**(2), 257–62.
- Arendt J, Borbely AA, Franey C and Wright J (1984). The effects of chronic, small doses of melatonin given in the late afternoon on fatigue in man: a preliminary study. *Neurosci Lett*, **45**(3), 317–21.
- Arendt J, Bojkowski C, Folkard S, Franey C, Marks V, Minors D, Waterhouse J, Wever RA, Wildgruber C and Wright J (1985a). Some effects of melatonin and the control of its secretion in humans. *Ciba Found Symp*, 117, 266–83.
- Arendt J, Bojkowski C, Franey C, Wright J and Marks V (1985b). Immunoassay of 6-hydroxymelatonin sulfate in human plasma and urine: abolition of the urinary 24-hour rhythm with atenolol. *J Clin Endocrinol Metab*, **60**(6), 1166-73.
- Arendt J, Labib M H, Bojkowski C, Hanson S and Marks V (1989). Rapid decrease in melatonin production during successful treatment of delayed puberty. *Lancet*, **1**(8650), 1326.
- Arendt J, Skene D J, Middleton B, Lockley SW and Deacon S (1997). Efficacy of melatonin treatment in jet lag, shift work, and blindness. *J Biol Rhythms*, **12**(6), 604–17.
- Arendt J, Stone B and Skene DJ (2005). Jet lag and sleep disruption. In *Principles and Practice of Sleep Medicine* (T Roth et al, eds). Philadelphia, WB Saunders and Co.
- Armstrong SM and Redman J (1985). Melatonin administration: effects on rodent circadian rhythms. *Ciba Found Symp*, **117**, 188–207.

- Armstrong SM and Redman JR (1991). Melatonin: a chronobiotic with anti-aging properties? *Med Hypotheses*, **34**(4), 300–309.
- Aujard F, Dkhissi-Benyahya O, Fournier I, Claustrat B, Schilling A, Cooper HM and Perret M (2001). Artificially accelerated aging by shortened photoperiod alters early gene expression (Fos) in the suprachiasmatic nucleus and sulfatoxymelatonin excretion in a small primate, *Microcebus murinus*. *Neuroscience*, **105**(2), 403–12.
- Badia P, Myers B and Murphy P (1992). Melatonin and thermoregulation. In *Melatonin: Biosynthesis, Physiological Effects and Clinical Applications* (RJ Reiter and HS Yu, eds). Boca Raton FL, CRC Press.
- Baldwin WS, Travlos GS, Risinger JI and Barrett JC (1998). Melatonin does not inhibit estradiol-stimulated proliferation in MCF-7 and BG-1 cells. *Carcinogenesis*, **19**, 1895–900.
- Barnes RG, Deacon SJ, Forbes MJ and Arendt J (1998). Adaptation of the 6-sulphatoxymelatonin rhythm in shiftworkers on offshore oil installations during a 2-week 12-h night shift. *Neurosci Lett*, **241**(1), 9–12.
- Beatson GT (1896). On the treatment of inoperable cases of carcinoma of the mamma: suggestions for a new method of treatment with illustrative cases. *Lancet*, **2**, 104–7.
- Beck-Friis J, von Rosen D, Kjellman BF, Ljunggren JG and Wetterberg L (1984). Melatonin in relation to body measures, sex, age, season and the use of drugs in patients with major affective disorders and healthy subjects. *Psychoneuroendocrinology*, **9**(3), 261–77.
- Beedham C, Smith JA, Steele DL and Wright PA (1987). Chlorpromazine inhibition of melatonin metabolism by normal and induced rat liver microsomes. *Eur J Drug Metab Pharmacokinet*, **12**(4), 299–302.
- Berga SL, Jones KL, Kaufmann S and Yen SS (1989). Nocturnal melatonin levels are unaltered by ovarian suppression in girls with central precocious puberty. *Fertil Steril*, **52**(6), 936–41.
- Berrino F, Muti P, Micheli A, Bolelli G, Krogh V, Sciajno R, Pisani P, Panico S and Secreto G (1996). Serum sex hormone levels after menopause and subsequent breast cancer. J Natl Cancer Inst, 88, 291–6.
- Biswas R and Vonderhaar BK (1987). Role of serum in prolactin responsiveness of MCF-7 human breast cancer cells in long term tissue culture. *Cancer Res*, **47**, 3509–14.
- Bittman EL, Kaynard AH, Olster DH, Robinson JE, Yellon SM and Karsch FJ (1985). Pineal melatonin mediates photoperiodic control of pulsatile luteinizing hormone secretion in the ewe. *Neuroendocrinology*, **40**(5), 409–18.
- Bizzarri M, Cucina A, Valente MG, Tagliaferri F, Borrelli V, Stipa F and Cavallaro A (2003). Melatonin and vitamin D3 increase TGF-beta1 release and induce growth inhibition in breast cancer cell cultures. *J Surg Res*, **110**, 332–7.
- Bjorvatn B, Kecklund G and Akerstedt T (1999). Bright light treatment used for adaptation to night work and re-adaptation back to day life. A field study at an oil platform in the North Sea. J Sleep Res, 8(2), 105–12.
- Blackman CF, Andrews PW, Ubeda A, Wang X, House DE, Trillo MA and Pimentel ME (2001). Physiological levels of melatonin enhance gap junction communication in primary cultures of mouse hepatocytes. *Cell Biol Toxicol*, **17**, 1–9.
- Blask DE, Wilson ST and Zalatan F (1997). Physiological melatonin inhibition of human breast cancer cell growth *in vitro:* evidence for a glutathione-mediated pathway. *Cancer Res*, **57**, 1909–14.
- Bojkowski CJ and Arendt J (1988). Annual changes in 6-sulphatoxymelatonin excretion in man. *Acta Endocrinol (Copenh)*, **117**(4), 470–76.
- Bojkowski CJ and Arendt J (1990). Factors influencing urinary 6-sulphatoxymelatonin, a major melatonin metabolite, in normal human subjects. *Clin Endocrinol (Oxf)*, **33**(4), 435–44.
- Bojkowski CJ, Aldhous ME, English J, Franey C, Poulton AL, Skene DJ and Arendt J (1987a). Suppression of nocturnal plasma melatonin and 6-sulphatoxymelatonin by bright and dim light in man. *Horm Metab Res*, **19**(9), 437–40.
- Bojkowski CJ, Arendt J, Shih MC and Markey SP (1987b). Melatonin secretion in humans assessed by measuring its metabolite, 6-sulfatoxymelatonin. *Clin Chem*, **33**(8), 1343–8.
- Bonneterre J, Peyrat JP, Vandewalle B, Beuscart R, Vie MC and Cappelaere P (1982). Prolactin receptors in human breast cancer. *Eur J Cancer Clin Oncol*, **18**, 1157–62.

- Bosc MJ (1987). Time of parturition in rats after melatonin administration or change of photoperiod. *J Reprod Fertil*, **80**(2), 563–8.
- Brainard GC, Hanifin JP, Greeson JM, Byrne B, Glickman G, Gerner E and Rollag MD (2001). Action spectrum for melatonin regulation in humans: evidence for a novel circadian photoreceptor. *J Neurosci*, **21**(16), 6405–12.
- Broadway J, Arendt J and Folkard S (1987). Bright light phase shifts the human melatonin rhythm during the Antarctic winter. *Neurosci Lett*, **79**, 185–9.
- Bronson FH (2004). Are humans seasonally photoperiodic? J Biol Rhythms, 19(3), 180-92.
- Brown EN, Choe Y, Shanahan TL and Czeisler CA (1997). A mathematical model of diurnal variations in human plasma melatonin levels. *Am J Physiol*, **272**(3 Part 1), E506–16.
- Brzezinski A (1997). Melatonin in humans. New Engl J Med, 336, 186-95.
- Brzezinski A, Lynch HJ, Seibel MM, Deng MH, Nadar TM and Wurtman RJ (1988). The circadian rhythm of plasma melatonin during the normal menstrual cycle and in amenorrheic women. *J Clin Endocrinol Metab*, **66**(5), 891–5.
- Bubenik GA (2002). Gastrointestinal melatonin: localization, function, and clinical relevance. *Dig Dis Sci*, **47**(10), 2336–48.
- Burgess HJ, Sharkey KM and Eastman CI (2002). Bright light, dark and melatonin can promote circadian adaptation in night shift workers. *Sleep Med Rev*, **6**(5), 407–20.
- Buxton OM, Lee CW, L'Hermite-Baleriaux M, Turek FW and Van Cauter E (2003). Exercise elicits phase shifts and acute alterations of melatonin that vary with circadian phase. *Am J Physiol Regul Integr Comp Physiol*, **284**(3), R714–24.
- Cagnacci A, Elliott JA and Yen SS (1992). Melatonin: a major regulator of the circadian rhythm of core temperature in humans. *J Clin Endocrinol Metab*, **75**(2), 447–52.
- Cagnacci A, Soldani R and Yen SS (1995). Exogenous melatonin enhances luteinizing hormone levels of women in the follicular but not in the luteal menstrual phase. *Fertil Steril*, **63**(5), 996–9.
- Cagnacci A, Soldani R, Laughlin GA and Yen SS (1996). Modification of circadian body temperature rhythm during the luteal menstrual phase: role of melatonin. *J Appl Physiol*, **80**(1), 25–9.
- Cagnacci A, Krauchi K, Wirz-Justice A and Volpe A (1997). Homeostatic versus circadian effects of melatonin on core body temperature in humans. *J Biol Rhythms*, **12**(6), 509–17.
- Campbell SS and Dawson D (1990). Enhancement of nighttime alertness and performance with bright ambient light. *Physiol Behav*, **48**(2), 317–20.
- Campbell SS and Murphy PJ (1998). Extraocular circadian phototransduction in humans. *Science*, **279**(5349), 396–9.
- Cardinali DP and Golombek DA (1998). The rhythmic GABAergic system. Neurochem Res, 23(5), 607-14.

Carlberg C (2000). Gene regulation by melatonin. Ann NY Acad Sci, 917, 387-96.

- Carrier J and Monk TH (2000). Circadian rhythms of performance: new trends. Chronobiol Int, 17(6), 719–32.
- Cassone VM (1992). The pineal gland influences rat circadian activity rhythms in constant light. *J Biol Rhythms*, **7**(1), 27–40.
- Cassone VM, Roberts MH and Moore RY (1988). Effects of melatonin on 2-deoxy-[1-¹⁴C]glucose uptake within rat suprachiasmatic nucleus. *Am J Physiol*, **255**(2 Part 2), R332–7.
- Checkley SA, Corn TH, Glass IB, Thompson C, Franey C and Arendt J (1986). Neuroendocrine and other studies of the mechanism of antidepressant action of desipramine. *Ciba Found Symp*, **123**, 126–47.
- Cheng ZN, Shu Y, Liu ZQ, Wang LS, Ou-Yang DS and Zhou HH (2001). Role of cytochrome p450 in estradiol metabolism *in vitro*. *Acta Pharmacol Sin*, **22**(2), 148–54.
- Chesworth MJ, Cassone VM and Armstrong SM (1987). Effects of daily melatonin injections on activity rhythms of rats in constant light. *Am J Physiol*, **253**(1 Part 2), R101–7.
- Claustrat B, Brun J, Geoffriau M, Zaidan R, Mallo C and Chazot G (1997). Nocturnal plasma melatonin profile and melatonin kinetics during infusion in status migrainosus. *Cephalalgia*, **17**(4), 511–17; discussion 487.
- Codegone ML, DiCarlo R, Muccioli G and Bussolati G (1981). Histology and cytometrics in human breast cancers assayed for the presence of prolactin receptors. *Tumori*, **67**, 549–52.

- Cohen M, Lipman M and Chabner B (1978). Role of pineal gland in aetiology and treatment of breast cancer. *Lancet*, **2**, 814–16.
- Copinschi G, van Onderbergen A, L'Hermite-Baleriaux M, Szyper M, Caufriez A, Bosson D, L'Hermite M, Robyn C, Turek FW and Van Cauter E (1990). Effects of the short-acting benzodiazepine triazolam, taken at bedtime, on circadian and sleep-related hormonal profiles in normal men. *Sleep*, **13**(3), 232–44.
- Cos S and Blask DE (1994). Melatonin modulates growth factor activity in MCF-7 human breast cancer cells. *J Pineal Res*, **17**, 25–32.
- Cos S and Fernandez R (2000). Melatonin effects on intercellular junctional communication in MCF-7 human breast cancer cells. *J Pineal Res*, **29**, 166–71.
- Cos S and Sanchez-Barcelo EJ (2000). Melatonin and mammary pathological growth. *Front Neuroendocrinol*, **21**, 133–70.
- Cos S, Fernandez R, Guezmes A and Sanchez-Barcelo EJ (1998). Influence of melatonin on invasive and metastatic properties of MCF-7 human breast cancer cells. *Cancer Res*, **58**, 4383–90.
- Costa G, Ghirlanda G, Tarondi, G, Minors D and Waterhouse J (1994). Evaluation of a rapidly rotating shift system for tolerance of nurses to nightwork. *Int Arch Occup Environ Health*, **65**(5), 305–11.
- Czeisler CA, Allan JS, Strogatz SJ, Ronda JM, Sanchez R, Rios CD, Freitag WO, Richardson GS and Kronauer RE (1986). Bright light resets the human circadian pacemaker independent of the timing of the sleep-wake cycle. *Science*, **233**(4764), 667–71.
- Czeisler CA, Johnson MP, Duffy JF, Brown EN, Ronda JM and Kronauer RE (1990). Exposure to bright light and darkness to treat physiologic maladaptation to night work. *N Engl J Med*, **322**(18), 1253–9.
- Czeisler CA, Shanahan TL, Kierman EB, Martens H, Brotman DJ, Emens JS, Klein T and Rizzo J F 3rd (1995). Suppression of melatonin secretion in some blind patients by exposure to bright light. *N Engl J Med*, **332**(1), 6–11.
- Dai J, Inscho EW, Yuan L and Hill SM (2002). Modulation of intracellular calcium and calmodulin by melatonin in MCF-7 human breast cancer cells. *J Pineal Res*, **32**, 112–19.
- Dao TL and Huggins C (1955). Bilateral adrenalectomy in the treatment of cancer of the breast. *Arch Surg*, **71**, 645–57.
- Davis FC and Mannion J (1988). Entrainment of hamster pup circadian rhythms by prenatal melatonin injections to the mother. *Am J Physiol*, **255**(3 Part 2), R439–48.
- De Leersnyder H, De Blois MC, Claustrat B, Romana S, Albrecht U, Von Kleist-Retzow JC, Delobel B, Viot G, Lynnet S, Bekemans M and Munnich A (2001). Inversion of the circadian rhythm of melatonin in the Smith-Magenis syndrome. *J Pediatr*, **139**(1), 111–16.
- De Leersnyder H, Bresson JL, de Blois MC, Souberbielle JC, Mogenet A, Delhotal-Landes B, Salefranque F and Munnich A (2003). Beta 1-adrenergic antagonists and melatonin reset the clock and restore sleep in a circadian disorder, Smith-Magenis syndrome. *J Med Genet*, **40**(1), 74–8.
- De Placido S, Gallo C, Perrone F, Marinelli A, Pagliarulo C, Carlomagno C, Petrella G, D'Istria M, Delrio G and Bianco AR (1990). Prolactin receptor does not correlate with oestrogen and progesterone receptors in primary breast cancer and lacks prognostic significance. Ten year results of the Naples adjuvant (GUN) study. *Br J Cancer*, **62**, 643–6.
- Deacon S and Arendt J (1994). Posture influences melatonin concentrations in plasma and saliva in humans. *Neurosci Lett*, **167**(1–2), 191–4.
- Deacon S and Arendt J (1995). Melatonin-induced temperature suppression and its acute phase-shifting effects correlate in a dose-dependent manner in humans. *Brain Res*, **688**(1–2), 77–85.
- Deacon S and Arendt J (1996). Adapting to phase shifts, I. An experimental model for jet lag and shift work. *Physiol Behav*, **59**(4–5), 665–73.
- Delagrange P, Atkinson J, Boutin JA, Casteilla L, Lesieur D, Misslin R, Pellissier S, Penicaud L and Renard P (2003). Therapeutic perspectives for melatonin agonists and antagonists. *J Neuroendocrinol*, **15**, 442–8.
- Deveson S, Forsyth IA and Arendt J (1992). Retardation of pubertal development by prenatal long days in goat kids born in autumn. *J Reprod Fertil*, **95**(2), 629–37.

- Dijk DJ, Shanahan TL, Duffy JF, Ronda JM and Czeisler CA (1997). Variation of electroencephalographic activity during non-rapid eye movement and rapid eye movement sleep with phase of circadian melatonin rhythm in humans. *J Physiol*, **505** (Part 3), 851–8.
- Djeridane Y, Vivien-Roels B, Simmonneaux V, Miguez JM and Pevet P (1998). Evidence for melatonin synthesis in rodent Harderian gland: a dynamic *in vitro* study. *J Pineal Res*, **25**(1), 54–64.
- Doolen S, Krause DN, Dubocovich ML and Duckles SP (1998). Melatonin mediates two distinct responses in vascular smooth muscle. *Eur J Pharmacol*, **345**(1), 67–9.
- Dorgan JF, Longcope C, Stephenson HE Jr, Falk RT, Miller R, Franz C, Kahle L, Campbell WS, Tangrea JA and Schatzkin A (1997). Serum sex hormone levels are related to breast cancer risk in postmenopausal women. *Environ Health Perspect*, **105**(Suppl 3), 583–5.
- Dubocovich ML (1983). Melatonin is a potent modulator of dopamine release in the retina. *Nature*, **306**(5945), 782-4.
- Dubocovich ML, Cardinali DP, Guardiola-Lemaitre B, et al (1998). Melatonin receptors. In *The IUPHAR Compendium of Receptor Characterization and Classification*. London, IUPHAR Media, pp 187–93.
- Dubocovich ML, Rivera-Bermudez MA, Gerdin MJ and Masana MI (2003). Molecular pharmacology, regulation and function of mammalian melatonin receptors. *Front Biosci*, **8**, 1093–108.
- Duffy JF and Dijk DJ (2002). Getting through to circadian oscillators: why use constant routines? *J Biol Rhythms*, **17**(1), 4–13.
- Duffy JF, Dijk DJ, Hall EF and Czeisler CA (1999). Relationship of endogenous circadian melatonin and temperature rhythms to self-reported preference for morning or evening activity in young and older people. *J Investig Med*, **47**(3), 141–50.
- Ebisawa T, Uchiyama M, Kajimura N, Kamei Y, Shibui K, Kim K, Kudo Y, Iwase T, Sugishita M, Jodoi T, Ikeda M, Ozeki Y, Watanabe T, Sekimoto M, Katoh M, Yamada N, Toyoshima R, Okawa M, Takahashi K and Yamauchi T (2000). Genetic polymorphisms of human melatonin 1b receptor gene in circadian rhythm sleep disorders and controls. *Neurosci Lett*, **280**(1), 29–32.
- Ekman AC, Leppaluoto J, Huttunen P, Arando K and Vakkuri O (1993). Ethanol inhibits melatonin secretion in healthy volunteers in a dose-dependent randomized double blind cross-over study. *J Clin Endocrinol Metab*, **77**(3), 780–83.
- Elliot JA (1981). Circadian rhythms, entrainment and photoperiodism in the Syrian Hamster. Bristol, Scientechnica.
- English J, Arendt J, Poulton A and Symons AM (1987). Short-term variations of circulating melatonin in the ewe. *J Pineal Res*, **4**(4), 359–66.
- English J, Middleton BA, Arendt J and Wirz-Justice A (1993). Rapid direct measurement of melatonin in saliva using an iodinated tracer and solid phase second antibody. *Ann Clin Biochem*, **30**(Part 4), 415–16.
- Escames G, Khaldy H, Leon J, Gonzalez L and Acuna-Castroviejo D (2004). Changes in iNOS activity, oxidative stress and melatonin levels in hypertensive patients treated with lacidipine. *J Hypertens*, **22**(3), 629–35.
- Feinleib M (1968). Breast cancer and artificial menopause: a cohort study. J Natl Cancer Inst, 41, 315-29.
- Fertl E, Auff E, Doppelbauer A and Waldhauser F (1991). Circadian secretion pattern of melatonin in Parkinson's disease. J Neural Transm Park Dis Dement Sect, **3**(1), 41–7.
- Fevre-Montange M, van Cauter E, Refetoff S, Desir D, Tourniaire J and Copinschi G (1981). Effects of 'jet lag' on hormonal patterns. II. Adaptation of melatonin circadian periodicity. *J Clin Endocrinol Metab*, **52**(4), 642–9.
- Fiegelson HS, Ross RK, Yu MC, Coetzee GA, Reichardt JK and Henderson BE (1996). Genetic susceptibility to cancer from exogenous and endogenous exposures. *J Cell Biochem*, **255**, 15–22.
- Filipski E, Delaunay F, King VM, Wu MW, Claustrat B, Grechez-Cassiau A, Guettier C, Hastings MJ and Francis L (2004). Effects of chronic jet lag on tumor progression in mice. *Cancer Res*, **64**(21), 7879–85.
- Fletcher O, Gibson L, Johnson N, Altmann DR, Holly JM, Ashworth A, Peto J, and Silva Idos S (2005). Polymorphisms and circulating levels in the insulin-like growth factor system and risk of breast cancer: a systematic review. *Cancer Epidemiol Biomarkers Prev*, **14**(1), 2–19.
- Folkard S, Arendt J and Clark M (1993). Can melatonin improve shift workers' tolerance of the night shift? Some preliminary findings. *Chronobiol Int*, **10**(5), 315–20.

Forsling ML (2000). Diurnal rhythms in neurohypophysial function. Exp Physiol, 85(Spec No), 179S-186S.

- Forsling ML, Wheeler MJ and Williams AJ (1999). The effect of melatonin administration on pituitary hormone secretion in man. *Clin Endocrinol (Oxf)*, **51**(5), 637–42.
- Foster DL, Ebling FJ and Claypool LE (1988). Timing of puberty by photoperiod. Reprod Nutr Dev, 28(2B), 349–64.
- Foster RG and Hankins MW (2002). Non-rod, non-cone photoreception in the vertebrates. *Prog Retin Eye Res*, **21**(6), 507–27.
- Fraschini F, Demartini G, Esposti D and Scaglione F (1998). Melatonin involvement in immunity and cancer. *Biol Signals*, **7**, 61–72.
- Fraser S, Cowen P, Franklin M and Lewy AJ (1983a). Direct radioimmunoassay and gas chromatography-mass spectrometry compared for determination of melatonin in plasma. *Clin Chem*, **29**(9), 1703–4.
- Fraser S, Cowen P, Franklin M, Francy C and Arendt J (1983b). Direct radioimmunoassay for melatonin in plasma. *Clin Chem*, **29**(2), 396–7.
- Fu L and Lee CC (2003). The circadian clock: pacemaker and tumour suppressor. Nat Rev Cancer, 3, 350-61.
- Fu L, Pelicano H, Liu J, Huang P and Lee C (2002). The circadian gene Period2 plays an important role in tumor suppression and DNA damage response *in vivo*. *Nat Rev Cancer*, **111**, 41–50.
- Gamst A, Wolfson T and Parry B (2004). Local polynomial regression modeling of human plasma melatonin levels. *J Biol Rhythms*, **19**(2), 164–74.
- Ganguly S, Coon SL and Klein DC (2002). Control of melatonin synthesis in the mammalian pineal gland: the critical role of serotonin acetylation. *Cell Tissue Res*, **309**(1), 127–37.
- Gibertini M, Graham C and Cook MR (1999). Self-report of circadian type reflects the phase of the melatonin rhythm. *Biol Psychol*, **50**(1), 19–33.
- Goldman BD (2001). Mammalian photoperiodic system: formal properties and neuroendocrine mechanisms of photoperiodic time measurement. *J Biol Rhythms*, **16**(4), 283–301.
- Gomez Brunet A, Malpaux B, Daveau A, Taragnat C and Chemineau P (2002). Genetic variability in melatonin secretion originates in the number of pinealocytes in sheep. *J Endocrinol*, **172**(2), 397–404.
- Graham C, Cook MR, Kavet R, Sastre A and Smith DK (1998). Prediction of nocturnal plasma melatonin from morning urinary measures. *J Pineal Res*, **24**(4), 230–38.
- Graw P, Werth E, Krauchi K, Gutzwiller F, Cajochen C and Wirz-Justice A (2001). Early morning melatonin administration impairs psychomotor vigilance. *Behav Brain Res*, **121**(1–2), 167–72.
- Griefahn B, Remer T, Blaszkewicz M and Brode P (2001). Long-term stability of 6-hydroxymelatonin sulfate in 24-h urine samples stored at -20 degrees C. *Endocrine*, **15**(2), 199-202.
- Guardiola-Lemaitre B (1997). Toxicology of melatonin. J Biol Rhythms, 12(6), 697-706.
- Hack LM, Lockley SW, Arendt J and Skene DJ (2003). The effects of low-dose 0.5-mg melatonin on the freerunning circadian rhythms of blind subjects. *J Biol Rhythms*, **18**(5), 420–29.
- Hankinson SE, Willett WC, Manson JE, Colditz GA, Hunter DJ, Spiegelman D, Barbieri RL and Speizer FE (1998). Plasma sex steroid hormone levels and risk of breast cancer in postmenopausal women. *J Natl Cancer Inst*, **90**, 1292–9.
- Hardeland R, Balzer I, Poeggeler B, Fuhrberg B, Uria H, Behrmann G, Wolf R, Meyer TJ and Reiter RJ (1995). On the primary functions of melatonin in evolution: mediation of photoperiodic signals in a unicell, photooxidation, and scavenging of free radicals. *J Pineal Res*, **18**(2), 104–11.
- Hartter S, Nordmark A, Rose DM, Bertilsson L, Tybring G and Laine K (2003). Effects of caffeine intake on the pharmacokinetics of melatonin, a probe drug for CYP1A2 activity. *Br J Clin Pharmacol*, **56**(6), 679–82.
- Hartter S, Wang X, Weigmann H, Friedberg T, Arand M, Oesch F and Hiemke C (2001). Differential effects of fluvoxamine and other antidepressants on the biotransformation of melatonin. *J Clin Psychopharmacol*, **21**(2), 167–74.
- Haus E and Touitou Y (1992). Chronobiology in laboratory medicine. In *Biologic Rhythms in Clinical and Laboratory Medicine* (Y Touitou and E Haus, eds). Berlin, Springer-Verlag, pp 673–708.
- Hazlerigg DG, Morgan PJ and Messager S (2001). Decoding photoperiodic time and melatonin in mammals: what can we learn from the pars tuberalis? *J Biol Rhythms*, **16**(4), 326–35.

- Hebert M, Martin SK, Lee C and Eastman CI (2002). The effects of prior light history on the suppression of melatonin by light in humans. *J Pineal Res*, **33**(4), 198–203.
- Henderson BE and Feigelson HS (2000). Hormonal carcinogenesis. Carcinogenesis, 21, 427-33.
- Henderson BE, Ross RK, Pike MC and Casagrande JT (1982). Endogenous hormones as a major factor in human cancer. *Cancer Res*, **24**, 3232–9.
- Henson JC, Coune A and Staquet M (1972). Clinical trial of 2-Br-α-ergocryptine (CB154) in advanced breast cancer. *Eur J Cancer*, **8**, 155–6.
- Herxheimer A and Petrie KJ (2002). Melatonin for the prevention and treatment of jet lag. Cochrane Database Syst Rev (2), CD001520.
- Ho AK and Smith JA (1982). Effect of benserazide on the levels of pineal 5-hydroxytryptamine, melatonin synthesising enzymes and serum melatonin. *Biochem Pharmacol*, **31**(13), 2251–5.
- Hoffmann K (1979). Photoperiod, pineal, melatonin and reproduction in hamsters. Prog Brain Res, 52, 397-415.
- Holdaway IM, Mason BH, Gibbs EE, Rajasoorya C, Lethaby A, Hopkins KD, Evans MC, Lim T and Schooler B (1997). Seasonal variation in the secretion of mammotrophic hormones in normal women and women with previous breast cancer. *Breast Cancer Res Treat*, **42**, 15–22.
- Holmes SW and Sugden D (1976). The effect of melatonin on pinealectomy-induced hypertension in the rat. *Br J Pharmacol*, **56**(3), 360P–361P.
- Hoppen K (2002). The effects of light on alertness and performance in relation to melatonin secretion. PhD thesis, University of Surrey.
- Horwitz KB, McGuire WL, Pearson OH and Segaloff A (1975). Predicting response to endocrine therapy in human breast cancer: a hypothesis. *Science*, **189**, 726–7.
- Huggins C and Dao TL (1953). Adrenalectomy and oophorectomy in treatment of advanced carcinoma of the breast. *JAMA*, **151**, 1388–94.
- Illnerova H and Sumova A (1997). Photic entrainment of the mammalian rhythm in melatonin production. *J Biol Rhythms*, **12**(6), 547–55.
- Illnerova H, Trentini GP and Maslova L (1989). Melatonin accelerates reentrainment of the circadian rhythm of its own production after an 8-h advance of the light–dark cycle. J Comp Physiol [A], **166**(1), 97–102.
- Illnerova H, Buresova M and Presl J (1993). Melatonin rhythm in human milk. J Clin Endocrinol Metab, **77**(3), 838–41.
- Ingram DM, Nottage EM and Roberts AN (1990). Prolactin and breast cancer risk. Med J Aust, 153, 469-73.
- Iuvone PM and Gan J (1995). Functional interaction of melatonin receptors and D1 dopamine receptors in cultured chick retinal neurons. *J Neurosci*, **15**(3 Part 2), 2179–85.
- Iuvone PM, Brown AD, Haque R, Weller J Zawilska J B, Chaurasia SS, Ma M and Klein DC (2002). Retinal melatonin production: role of proteasomal proteolysis in circadian and photic control of arylalkylamine N-acetyltransferase. *Invest Ophthalmol Vis Sci*, **43**(2), 564–72.
- Jan JE, Espezel H and Appleton RE (1994). The treatment of sleep disorders with melatonin. *Dev Med Child Neurol*, **36**(2), 97–107.
- Jensen EV, DeSombre ER and Jungblut PP (1967). Endogenous factors influencing host-tumor balance. In *Estrogen Receptors in Hormone-responsive Tissues and Tumors* (RW Wissler et al, eds). University of Chicago Press, pp 15–30.
- Kajdaniuk D, Marek B, Kos-Kudla B, Zwirska-Korczala K, Ostrowska Z, Butner B and Szymszal J (2002). Does the negative correlation found in breast cancer patients between plasma melatonin and insulin-like growth factor-I concentrations imply the existence of an additional mechanism of oncostatic melatonin influence involved in defense? *Med Sci Monit*, 8(6), 457–61.
- Kanabrocki EL, Sothern RB, Scheving LE, Halberg F, Pauly JE, Greco J, Nemchausky BA, DeBartolo M, Kaplan E and McCormick JB (1988). Ten-year-replicated circadian profiles for 36 physiological, serological and urinary variables in healthy men. *Chronobiol Int*, **5**(3), 237–84.
- Karbownik M, Reiter RJ, Burkhardt S, Gitto E, Tan DX and Lewinski A (2001). Melatonin attenuates estradiolinduced oxidative damage to DNA: relevance for cancer prevention. *Exp Biol Med*, **226**(7), 707–12.

- Karsch FJ, Malpaux B, Wayne NL and Robinson JE (1988). Characteristics of the melatonin signal that provide the photoperiodic code for timing seasonal reproduction in the ewe. *Reprod Nutr Dev*, **28**(2B), 459–72.
- Kauppila A, Kivela A, Parkarinen A and Vakkuri O (1987). Inverse seasonal relationship between melatonin and ovarian activity in humans in a region with a strong seasonal contrast in luminosity. J Clin Endocrinol Metab, 65(5), 823–8.
- Kennaway DJ, Goble FC and Stamp GE (1996). Factors influencing the development of melatonin rhythmicity in humans. *J Clin Endocrinol Metab*, **81**(4), 1525–32.
- Kennaway DJ, Lushington K, Dawson D, Lack L, van den Heuvel C and Rogers N (1999). Urinary 6-sulfatoxymelatonin excretion and aging: new results and a critical review of the literature. J Pineal Res, 27(4), 210–20.
- Kiefer T, Ram PT, Yuan L and Hill SM (2002). Melatonin inhibits estrogen receptor transactivation and cAMP levels in breast cancer cells. *Breast Cancer Res Treat*, **71**, 37–45.
- Kivela A, Kauppila A, Ylostalo P, Vakkuri O and Leppaluoto J (1988). Seasonal, menstrual and circadian secretions of melatonin, gonadotropins and prolactin in women. *Acta Physiol Scand*, **132**(3), 321–7.
- Klein DC (1985). Photoneural regulation of the mammalian pineal gland. Ciba Found Symp, 117, 38-56.
- Klein DC, Coon SL, Roseboom PH, Weller JL, Bernard M, Gastel JA, Zatz M, Iuvone PM, Rodriguez IR, Begay V, Falcon J, Cahill GM, Cassone VM and Baler R (1997). The melatonin rhythm-generating enzyme: molecular regulation of serotonin N-acetyltransferase in the pineal gland. *Recent Prog Horm Res*, **52**, 307–57, discussion 357–8.
- Klein DC, Ganguly S, Coon SL, Shi Q, Gaildrat P, Morin F, Weller JL, Obsil T, Hickman A and Dyda F (2003). 14-3-3 proteins in pineal photoneuroendocrine transduction: how many roles? J Neuroendocrinol, 15(4), 370–77.
- Klerman EB, Gershengorn HB, Duffy JF and Kronauer RE (2002). Comparisons of the variability of three markers of the human circadian pacemaker. *J Biol Rhythms*, **17**(2), 181–93.
- Klerman EB, Zeitzer JM, Duffy JF, Khalsa SB and Czeisler CA (2001). Absence of an increase in the duration of the circadian melatonin secretory episode in totally blind human subjects. J Clin Endocrinol Metab, 86(7), 3166–70.
- Kojima T, Mochizuki C, Mitaka T and Mochizuki Y (1997). Effects of melatonin on proliferation, oxidative stress and Cx32 gap junction protein expression in primary cultures of adult rat hepatocytes. *Cell Struct Funct*, **22**, 347–56.
- Kopin IJ, Pare CM, Axelrod J and Weissbach H (1961). The fate of melatonin in animals. J Biol Chem, 236, 3072-5.
- Korf HW, Schomerus C and Stehle JH (1998). The pineal organ, its hormone melatonin, and the photoneuroendocrine system. *Adv Anat Embryol Cell Biol*, **146**, 1–100.
- Kostoglou-Athanassiou I, Athanassiou P, Treacher DF, Wheeler MJ and Forsling ML (1998). Neurohypophysial hormone and melatonin secretion over the natural and suppressed menstrual cycle in premenopausal women. *Clin Endocrinol (Oxf)*, **49**(2), 209–16.
- Krause DN and Dubocovich ML (1990). Regulatory sites in the melatonin system of mammals. *Trends Neurosci*, **13**, 464–70.
- L'Hermite-Baleriaux M, Casteels S, Vokaer A, Loriaus C, Noel G and L'Hermite M (1984). Prolactin and prolactin receptors in human breast disease. *Prog Cancer Res Ther*, **31**, 325–34.
- Lakin-Thomas PL (2000). Circadian rhythms: new functions for old clock genes. Trends Genet, 16(3), 135-42.
- Leibenluft E, Fiero PL and Rubinow DR (1994). Effects of the menstrual cycle on dependent variables in mood disorder research. *Arch Gen Psychiatry*, **51**(10), 761–81.
- Lemus-Wilson A, Kelly PA and Blask DE (1995). Melatonin blocks the stimulatory effects of prolactin on human breast cancer cell growth in culture. *Br J Cancer*, **72**, 1435–40.
- Leproult R, Colecchia EF, L'Hermite-Baleriaux M and Van Cauter E (2001). Transition from dim to bright light in the morning induces an immediate elevation of cortisol levels. *J Clin Endocrinol Metab*, **86**(1), 151–7.
- Lewy AJ and Markey SP (1978). Analysis of melatonin in human plasma by gas chromatography negative chemical ionization mass spectrometry. *Science*, **201**(4357), 741–3.
- Lewy AJ and Newsome DA (1983). Different types of melatonin circadian secretory rhythms in some blind subjects. *J Clin Endocrinol Metab*, **56**(6), 1103–7.

- Lewy AJ and Sack RL (1989). The dim light melatonin onset as a marker for circadian phase position. *Chronobiol Int*, **6**(1), 93–102.
- Lewy AJ, Wehr TA, Goodwin FK, Newsome DA and Markey SP (1980). Light suppresses melatonin secretion in humans. *Science*, **210**(4475), 1267–9.
- Lewy AJ, Bauer VK, Ahmed S, Thomas KH, Cutler NL, Singer CM, Moffit MT and Sack RL (1998). The human phase response curve (PRC) to melatonin is about 12 hours out of phase with the PRC to light. *Chronobiol Int*, **15**(1), 71–83.
- Lewy AJ, Bauer VK, Hasler BP, Kendall AR, Pires ML and Sack RL (2001). Capturing the circadian rhythms of freerunning blind people with 0.5 mg melatonin. *Brain Res*, **918**(1–2), 96–100.
- Lieberman HR, Waldhauser F, Garfield G, Lynch HJ and Wurtman RJ (1984). Effects of melatonin on human mood and performance. *Brain Res*, **323**(2), 201–7.
- Lincoln GA (2002). Neuroendocrine regulation of seasonal gonadotrophin and prolactin rhythms: lessons from the Soay ram model. *Reprod Suppl*, **59**, 131–47.
- Lincoln GA and Clarke IJ (1994). Photoperiodically induced cycles in the secretion of prolactin in hypothalamopituitary disconnected rams: evidence for translation of the melatonin signal in the pituitary gland. *J Neuroendocrinol*, **6**, 251–60.
- Lincoln GA and Richardson M (1998). Photo-neuroendocrine control of seasonal cycles in body weight, pelage growth and reproduction: lessons from the HPD sheep model. *Comp Biochem Physiol C Pharmacol Toxicol Endocrinol*, **119**(3), 283–94.
- Lincoln GA and Short RV (1980). Seasonal breeding: nature's contraceptive. Recent Prog Horm Res, 36, 1–52.
- Lincoln GA, Ebling FJ and Almeida OF (1985). Generation of melatonin rhythms. Ciba Found Symp, 117, 129-48.
- Lippman M, Monaco ME and Bolan G (1977). Effects of estrone, estradiol, and estriol on hormone-responsive human breast cancer in long-term tissue culture. *Cancer Res*, **37**, 1901–7.
- Lissoni P, Barni S, Tancini G, Rovelli F, Ardizzola A, Conti A and Maestroni GJ (1993). A study of the mechanisms involved in the immunostimulatory action of the pineal hormone in cancer patients. *Oncology*, **50**(6), 399–402.
- Lissoni P, Barni S, Meregalli S, Fossati V, Cazzaniga M, Esposti D and Tancini G (1995). Modulation of cancer endocrine therapy by melatonin: a phase II study of tamoxifen plus melatonin in metastatic breast cancer patients progressing under tamoxifen alone. *Br J Cancer*, **71**(4), 854–6.
- Lissoni P, Paolorossi F, Tancini G, Ardizzoia A, Barni S, Brivio F, Maestroni GJ and Chilelli M (1996). A phase II study of tamoxifen plus melatonin in metastatic solid tumour patients. *Br J Cancer*, **74**(9), 1466–8.
- Lissoni P, Bolis S, Brivio F and Fumagalli L (2000). A phase II study of neuroimmunotherapy with subcutaneous low-dose IL-2 plus the pineal hormone melatonin in untreatable advanced hematologic malignancies. *Anticancer Res*, **20**(3B), 2103–5.
- Lissoni P, Chilelli M, Villa S, Cerizza L and Tancini G (2003). Five years survival in metastatic non-small cell lung cancer patients treated with chemotherapy alone or chemotherapy and melatonin: a randomized trial. *J Pineal Res*, **35**(1), 12–15.
- Liu C, Weaver DR, Jin X, Shearman LP, Pieschl RL, Gribkoff VK and Reppert SM (1997). Molecular dissection of two distinct actions of melatonin on the suprachiasmatic circadian clock. *Neuron*, **19**(1), 91–102.
- Lockley SW, Brainard GC and Caeisler CA (2003). High sensitivity of the human circadian melatonin rhythm to resetting by short wavelength light. *J Clin Endocrinol Metab*, **88**(9), 4502–5.
- Lockley SW, Skene DJ, Arendt J, Tabandeh H, Bird AC and Defrance R (1997a). Relationship between melatonin rhythms and visual loss in the blind. *J Clin Endocrinol Metab*, **82**(11), 3763–70.
- Lockley SW, Skene DJ, Tabandeh H, Bird AC, Defrance R and Arendt J (1997b). Relationship between napping and melatonin in the blind. *J Biol Rhythms*, **12**(1), 16–25.
- Lockley SW, Skene DJ, Thapan K, English J, Ribeiro D, Haimov I, Hampton S, Middleton B, von Schantz M and Arendt J (1998). Extraocular light exposure does not suppress plasma melatonin in humans. J Clin Endocrinol Metab, 83(9), 3369–72.
- Lockley SW, Skene DJ, James K, Thapan K, Wright J and Arendt J (2000). Melatonin administration can entrain the free-running circadian system of blind subjects. *J Endocrinol*, **164**(1), R1–6.

- Love RR and Rose DP (1985). Elevated bioactive prolactin in women at risk for familial breast cancer. *Eur J Cancer Clin Oncol*, **21**, 1553–4.
- Love RR, Rose DR, Surawicz TS and Newcomb PA (1991). Prolactin and growth hormone levels in premenopausal women with breast cancer and healthy women with a strong family history of breast cancer. *Cancer*, **68**, 1401–5.
- Luboshitzky R and Lavie P (1999). Melatonin and sex hormone interrelationships a review. *J Pediatr Endocrinol Metab*, **12**(3), 355–62.
- Luboshitzky R, Wagner O, Lavi S, Herer P and Lavie P (1996). Abnormal melatonin secretion in male patients with hypogonadism. *J Mol Neurosci*, **7**(2), 91–8.
- Luboshitzky R, Wagner O, Lavi S, Herer P and Lavie P (1997). Abnormal melatonin secretion in hypogonadal men: the effect of testosterone treatment. *Clin Endocrinol (Oxf)*, **47**(4), 463–9.
- Luboshitzky R, Levi M, Shen-Orr Z, Blumenfeld Z, Herer P and Lavie P (2000). Long-term melatonin administration does not alter pituitary-gonadal hormone secretion in normal men. *Hum Reprod*, **15**(1), 60–65.
- Luboshitzky R, Shen-Orr Z, Herer P and Nave R (2003). Urinary 6-sulfatoxymelatonin excretion in hyperandrogenic women with polycystic ovary syndrome: the effect of ethinyl estradiol-cyproterone acetate treatment. *Gynecol Endocrinol*, **17**(6), 441–7.
- Lucas RJ, Freedman MS, Munoz M, Garcia-Fernandez JM and Foster RG (1999). Regulation of the mammalian pineal by non-rod, non-cone, ocular photoreceptors. *Science*, **284**(5413), 505–7.
- Luft R, Olivecrona H and Sjogren B (1952). Hypofysektomi pa manniska. J Nordic Med, 14, 351-4.
- Macchi MM, Bruce JA, Boulos Z, Cooper TB, Terman JS and Terman M (2002). Sleep, chronotype and seasonality after pineal resection in humans: initial findings. *Soc Res Biol Rhythms Abs*, **9**, 157.
- Maestroni GJ (1999). Therapeutic potential of melatonin in immunodeficiency states, viral diseases, and cancer. *Adv Exp Med Biol*, **467**, 217–26.
- Maestroni GJ (2000). Neurohormones and catecholamines as functional components of the bone marrow microenvironment. *Ann NY Acad Sci*, **917**, 29–37.
- Maestroni GJ (2001) The immunotherapeutic potential of melatonin. Expert Opinion Investig Drugs, 10, 467-76.
- Mahle CD, Goggins GD, Agarwal P, Ryan E and Watson AJ (1997). Melatonin modulates vascular smooth muscle tone. J Biol Rhythms, **12**(6), 690–96.
- Makkison I and Arendt J (1991). Melatonin secretion on two different Antarctic Bases (68oS and 75oS). J Interdis Cycle Res, 22, 149–50.
- Malarkey WB, Kennedy M, Allred LE and Milo G (1983). Physiological concentrations of prolactin can promote the growth of human breast tumor cells in culture. *J Clin Endocrinol Metab*, **56**, 673–7.
- Mann K, Bauer H, Hiemke C, Roschke J, Wetzel H and Benkert O (1996). Acute, subchronic and discontinuation effects of zopiclone on sleep EEG and nocturnal melatonin secretion. *Eur Neuropsychopharmacol*, **6**(3), 163–8.
- Manni A, Boucher AE, Demers LM, Harvey HA, Lipton A, Simmonds MA and Bartholomew M (1989). Endocrine effects of combined somatostatin analog and bromocriptine therapy in women with advanced breast cancer. *Breast Cancer Res Treat*, **14**, 289–98.
- Martin JE and Klein DC (1976). Melatonin inhibition of the neonatal pituitary response to luteinizing hormonereleasing factor. *Science*, **191**(4224), 301–2.
- Masana MI and Dubocovich ML (2001). Melatonin receptor signaling: finding the path through the dark. *Sci STKE*, **2001**(107), PE39.
- Masana MI and Dubocovich ML (2002). Melatonin receptor signaling: finding the path through the dark. *Science*, **107**, 1–5.
- McArthur AJ, Gillette MU and Prosser RA (1991). Melatonin directly resets the rat suprachiasmatic circadian clock *in vitro. Brain Res*, **565**(1), 158–61.
- McGuire WL (1978). Steroid receptors in human breast cancer. Cancer Res, 38, 4289-91.
- McIntyre IM, Norman TR, Burrows GF and Armstrong SM (1987). Melatonin rhythm in human plasma and saliva. *J Pineal Res*, **4**(2), 177–83.

- McIntyre IM, Norman TR, Burrows GD and Armstrong SM (1993). Alterations to plasma melatonin and cortisol after evening alprazolam administration in humans. *Chronobiol Int*, **10**(3), 205–13.
- Mediavilla MD, Cos S and Sanchez-Barcelo EJ (1999). Melatonin increases p53 and p21WAF1 expression in MCF-7 human breast cancer cells *in vitro*. *Life Sci*, **65**, 415–20.
- Messager S, Hazlerigg DG, Mercer JG and Morgan PJ (2000). Photoperiod differentially regulates the expression of Per1 and ICER in the pars tuberalis and the suprachiasmatic nucleus of the Siberian hamster. *Eur J Neurosci*, **12**(8), 2865–70.
- Middleton B (in press). Measurement of melatonin and 6-sulphatoxymelatonin. In *Hormone Assays in Biological* Fluids (M Hutchinson and M Wheeler, eds). Totowa NJ, Humana Press.
- Middleton B, Arendt J and Stone BM (1996). Human circadian rhythms in constant dim light (8 lux) with knowledge of clock time. *J Sleep Res*, **5**(2), 69–76.
- Middleton B, Arendt J and Stone BM (1997). Complex effects of melatonin on human circadian rhythms in constant dim light. *J Biol Rhythms*, **12**(5), 467–77.
- Middleton B, Stone BM and Arendt J (2002). Human circadian phase in 12:12 h, 200: <8 lux and 1000: <8 lux light-dark cycles, without scheduled sleep or activity. *Neurosci Lett*, **329**(1), 41-4.
- Midwinter MJ and Arendt J (1991). Adaptation of the melatonin rhythm in human subjects following night-shift work in Antarctica. *Neurosci Lett*, **122**(2), 195–8.
- Migaud M, Gavet S and Pelletier J (2002). Partial cloning and polymorphism of the melatonin1a (Mel1a) receptor gene in two breeds of goat with different reproductive seasonality. *Reproduction*, **124**(1), 59–64.
- Miller WR (1996). Estrogens and endocrine therapy for breast cancer. In *Estrogen and Breast Cancer*. Austin TX, RG Landes Company, pp 125–50.
- Mills JN, Minors DS and Waterhouse JM (1978). Adaptation to abrupt time shifts of the oscillator(s) controlling human circadian rhythms. J Physiol, 285, 455–70.
- Molis TM, Spriggs LL and Hill SM (1994). Modulation of estrogen receptor mRNA expression by melatonin in MCF-7 human breast cancer cells. *Mol Endocrinol*, **8**, 1681–90.
- Molis TM, Spriggs LL, Jupiter Y and Hill SM (1995). Melatonin modulation of estrogen-regulated proteins, growth factors, and proto-oncogenes in human breast cancer. *J Pineal Res*, **18**, 93–103.
- Monteleone P, Forziati D, Orazzo C and Maj M (1989). Preliminary observations on the suppression of nocturnal plasma melatonin levels by short-term administration of diazepam in humans. *J Pineal Res*, **6**(3), 253–8.
- Moore RY (1996). Neural control of the pineal gland. Behav Brain Res, 73(1-2), 125-30.
- Moore RY and Eichler VB (1972). Loss of circadian adrenal corticosterone rhythm following suprachiasmatic lesion in rat. *Brain Res*, **42**, 201–6.
- Morgan L, Arendt J, Owens D, Folkard S, Hampton S, Deacon S, English J, Ribeiro D and Taylor K (1998). Effects of the endogenous clock and sleep time on melatonin, insulin, glucose and lipid metabolism. *J Endocrinol*, **157**(3), 443–51.
- Morgan L, Hampton S, Gibbs M and Arendt J (2003). Circadian aspects of postprandial metabolism. *Chronobiol Int*, **20**, 795–808.
- Morgan PJ and Williams LM (1989). Central melatonin receptors; implications for a mode of action. *Experientia*, **45**, 955–65.
- Murata J, Sawamura Y, Ikeda J, Hashimoto S and Honma K (1998). Twenty-four hour rhythm of melatonin in patients with a history of pineal and/or hypothalamo-neurohypophyseal germinoma. *J Pineal Res*, **25**(3), 159–66.
- Murphy PJ, Myers BL and Badia P (1996). Nonsteroidal anti-inflammatory drugs alter body temperature and suppress melatonin in humans. *Physiol Behav*, **59**(1), 133–9.
- Myers BL and Badia P (1993). Immediate effects of different light intensities on body temperature and alertness. *Physiol Behav*, **54**(1), 199–202.
- Naidoo R (1999). Investigation of rhythmic endocrine function in intensive care with emphasis on melatonin. PhD Thesis, University of Surrey.
- Nathan PJ, Jeyaseelan AS, Burrows GD and Norman TR (1998). Modulation of plasma melatonin concentrations by changes in posture. *J Pineal Res*, **24**, 219–23.

- Nelson RJ, Demas GE, Klein SL and Kriegsfeld LJ (1995). The influence of season, photoperiod, and pineal melatonin on immune function. *J Pineal Res*, **19**, 149–65.
- Newsome HH, Brown PW, Terz JJ and Lawrence WJ (1977). Medical and surgical adrenalectomy in patients with advanced breast carcinoma. *Cancer*, **39**, 542–6.
- Niles L (1991). Melatonin interaction with the benzodiazepine-GABA receptor complex in the CNS. *Adv Exp Med Biol*, **294**, 267–77.
- Okatani Y, Okamoto K, Hayashi K, Wakatsuki A, Tamura S and Sagara Y (1998). Maternal-fetal transfer of melatonin in pregnant women near term. *J Pineal Res*, **25**(3), 129–34.
- Okatani Y, Morioka N and Wakatsuki A (2000). Changes in nocturnal melatonin secretion in perimenopausal women: correlation with endogenous estrogen concentrations. *J Pineal Res*, **28**(2), 111–18.
- Orth DN, Besser GM, King PH and Nicholson WE (1979). Free-running circadian plasma cortisol rhythm in a blind human subject. *Clin Endocrinol (Oxf)*, **10**(6), 603–17.
- Owen J and Arendt J (1992). Melatonin suppression in human subjects by bright and dim light in antarctica: time and season-dependent effects. *Neurosci Lett*, **137**(2), 181–4.
- Ozaki Y, Lynch HJ and Wurtman RJ (1976). Melatonin in rat pineal, plasma, and urine: 24-hour rhythmicity and effect of chlorpromazine. *Endocrinology*, **98**(6), 1418–24.
- Palazidou E, Franey C, Arendt J, Stahl S and Checkley S (1989a). Evidence for a functional role of alpha-1 adrenoceptors in the regulation of melatonin secretion in man. *Psychoneuroendocrinology*, **14**(1–2), 131–5.
- Palazidou E, Papadopoulos A, Sitsen A, Stahl S and Checkley S (1989b). An alpha 2 adrenoceptor antagonist, Org 3770, enhances nocturnal melatonin secretion in man. *Psychopharmacology (Berl*, **97**(1), 115–17.
- Parry BL, Berga SL, Mostofi N, Klauber MR and Resnick A (1997). Plasma melatonin circadian rhythms during the menstrual cycle and after light therapy in premenstrual dysphoric disorder and normal control subjects. *J Biol Rhythms*, **12**(1), 47–64.
- Partonen T and Magnussen A (2001). Seasonal Affective Disorder: Practice and Research. Oxford University Press.
- Pawlikowski M, Kolomecka M, Wojtczak A and Karasek M (2002). Effects of six months melatonin treatment on sleep quality and serum concentrations of estradiol, cortisol, dehydroepiandrosterone sulfate, and somatomedin C in elderly women. *Neuro Endocrinol Lett*, **23**(Suppl 1), 17–19.
- Pearson OH and Manni A (1978). Hormonal control of breast cancer growth in women and rats. In *Current Topics in Experimental Endocrinology* (L Martin and VHT James, eds). New York, Academic Press, pp 75–92.
- Pearson OH and Ray BS (1959). Results of hypophysectomy in the treatment of metastic mammary carcinoma. *Cancer*, **12**, 85–93.
- Peyrat JP, DeWailly D, Djiane J, Kelly PA, Vandewalle B, Bonneterre J and Le Febvre J (1981). Total prolactin binding sites in human breast cancer biopsies. *Breast Cancer Res Treat*, **1**, 369–73.
- Phipps-Nelson J, Redman JR, Dijk DJ and Rajaratnam SM (2003). Daytime exposure to bright light, as compared to dim light, decreases sleepiness and improves psychomotor vigilance performance. *Sleep*, **26**(6), 695–700.
- Pierpaoli W and Regelson W (1994). Pineal control of aging: effect of melatonin and pineal grafting on aging mice. *Proc Natl Acad Sci USA*, **91**(2), 787–91.
- Place NJ, Tuthill CR, Schoomer EE, Tramontin AD and Zucker I (2004). Short day lengths delay reproductive aging. *Biol Reprod*, **71**(3), 987–92.
- Poirel VJ, Boggio V, Dardente H, Pevet P, Masson-Pevet M and Gauer F (2003). Contrary to other non-photic cues, acute melatonin injection does not induce immediate changes of clock gene mRNA expression in the rat suprachiasmatic nuclei. *Neuroscience*, **120**(3), 745–55.
- Puig-Domingo M, Webb SM, Serrano J, Peinado MA, Corcoy R, Ruscalleda J, Reiter RJ and de Leiva A (1992). Brief report: melatonin-related hypogonadotropic hypogonadism. *N Engl J Med*, **327**(19), 1356–9.
- Raftogianis R, Creveling C, Weinshilboum R and Weisz J (2000). Estrogen metabolism by conjugation. *J Natl Cancer Inst Monogr*, **27**, 113–24.
- Rajaratnam SM and Arendt J (2001). Health in a 24-h society. Lancet, 358(9286), 999-1005.
- Rajaratnam SM, Dijk DJ, Middleton B, Stone BM and Arendt J (2003). Melatonin phase-shifts human circadian rhythms with no evidence of changes in the duration of endogenous melatonin secretion or the 24-hour production of reproductive hormones. *J Clin Endocrinol Metab*, **88**(9), 4303–9.

- Rajaratnam SM, Middleton B, Stone BM, Arendt J and Dijk DJ (2004). Melatonin advances the circadian timing of EEG sleep and directly facilitates sleep without altering its duration in extended sleep opportunities. *J Physiol*, **551**(Part 1), 339–51.
- Ram PT, Kiefer T, Silverman M, Song Y, Brown GM and Hill SM (1998). Estrogen receptor transactivation in MCF-7 breast cancer cells by melatonin and growth factors. *Mol Cel Endocrinol*, **141**, 53–64.
- Redman J, Armstrong S and Ng KT (1983). Free-running activity rhythms in the rat: entrainment by melatonin. *Science*, **219**(4588), 1089–91.
- Redwine L, Hauger RL, Gillin JC and Irwin M (2000). Effects of sleep and sleep deprivation on interleukin-6, growth hormone, cortisol, and melatonin levels in humans. *J Clin Endocrinol Metab*, **85**(10), 3597–603.
- Reiter RJ (1980). The pineal gland and its hormones in the control of reproduction in mammals. *Endocrine Rev*, **1**, 109–31.
- Reiter RJ (1998). Oxidative damage in the central nervous system: protection by melatonin. *Prog Neurobiol*, **56**, 359–84.
- Reiter RJ (2004). Mechanisms of cancer inhibition by melatonin. J Pineal Res, 37(3), 213-14.
- Reme C (1986). Visual cells of the vertebrate retina. Renewal processes, rhythms, and light. *Naturwissenschaften*, **73**(3), 117–24.
- Reppert SM (1996). Nature's knockout: the Mel1b receptor is not necessary for reproductive and circadian responses to melatonin in Siberian hamsters. *Mol Endocrinol*, **10**, 1478–87.
- Reppert SM (1997). Melatonin receptors: molecular biology of a new family of G-protein-coupled receptors. *J Biol Rhythms*, **12**, 528-31.
- Reppert SM (2000). Cellular and molecular basis of circadian timing in mammals. Semin Perinatol, 24(4), 243-6.
- Reppert SM and Weaver DR (1995). Melatonin madness. Cell, 83(7), 1059-62.
- Reppert SM, Weaver DR and Godson C (1996). Melatonin receptors step into the light: cloning and classification of subtypes. *Trends Pharmacol Sci*, **17**, 100–102.
- Revell VL, Arendt J, Terman M and Skene D J (2005). Short wavelength sensitivity of the human circadian pacemaker to phase advancing light. *J Biol Rhythms*, **20**(3), 270–72.
- Roberts JE (2000). Light and immunomodulation. Ann NY Acad Sci, 917, 435-45.
- Roenneberg T and Aschoff J (1990). Annual rhythm of human reproduction: I. Biology, sociology, or both? *J Biol Rhythms*, **5**(3), 195–216.
- Rollag MD, Chen HJ, Ferguson BN and Reiter RJ (1979). Pineal melatonin content throughout the hamster estrous cycle. *Proc Soc Experi Biol Med*, **162**(1), 211–13.
- Rosenthal NE, Sack DA, Gillin JC, Lewy AJ, Goodwin FK, Davenport Y, Mueller PS, Newsome DA and Wehr TA (1984). Seasonal affective disorder. A description of the syndrome and preliminary findings with light therapy. *Arch Gen Psychiatry*, **41**, 72–9.
- Ross AW and Morgan PJ (2002). The pars tuberalis as a target of the central clock. Cell Tissue Res, 309(1), 163-71.
- Ross JK, Arendt J, Horne J and Hatson W (1995). Night-shift work in Antarctica: sleep characteristics and bright light treatment. *Physiol Behav*, **57**(6), 1169–74.
- Rowe SA and Kennaway DJ (2002). Melatonin in rat milk and the likelihood of its role in postnatal maternal entrainment of rhythms. *Am J Physiol Regul Integr Comp Physiol*, **282**(3), R797–804.
- Roy D and Liehr JG (1999). Estrogen, DNA damage and mutations. Mutat Res, 424, 107–15.
- Ruch RJ (1994). The role of gap junctional intercellular communication in neoplasia. *Ann Clin Lab Sci*, **24**, 216–31.
- Sack RL, Brandes RW, Kendall AR and Lewy AJ (2000). Entrainment of free-running circadian rhythms by melatonin in blind people. *N Engl J Med*, **343**(15), 1070–77.
- Santen RJ, Yue, W, Naftolin F, Mor G and Berstein L (1999). The potential of aromatase inhibitors in breast cancer prevention. *Endocrine Related Cancer*, **6**, 235–43.
- Schairer C, Hill D, Sturgeon SR, Fears T, Pollak M, Mies C, Ziegler RG, Hoover RN and Sherman ME (2004). Serum concentrations of IGF-1, IGFBP-3 and c-peptide and risk of hyperplasia and cancer of the breast in postmenopausal women. *Int J Cancer*, **20**(5), 773–9.

- Shanahan TL and Czeisler CA (1991). Light exposure induces equivalent phase shifts of the endogenous circadian rhythms of circulating plasma melatonin and core body temperature in men. *J Clin Endocrinol Metab*, **73**(2), 227–35.
- Shephard RJ and Shek PN (1997). Interactions between sleep, other body rhythms, immune responses, and exercise. *Can J Appl Physiol*, **22**(2), 95–116.
- Shibata S, Cassone VM and Moore RY (1989). Effects of melatonin on neuronal activity in the rat suprachiasmatic nucleus *in vitro*. *Neurosci Lett*, **97**(1–2), 140–44.
- Shilo L, Sabbah H, Hadri R, Kovatz S, Weinberg, U, Dolev S, Dagan Y and Shenkman L (2002). The effects of coffee consumption on sleep and melatonin secretion. *Sleep Med*, **3**(3), 271–3.
- Shochat T, Luboshitzky R and Lavie P (1997). Nocturnal melatonin onset is phase locked to the primary sleep gate. *Am J Physiol*, **273**(1 Part 2), R364–70.
- Simonneaux V and Ribelayga C (2003). Generation of the melatonin endocrine message in mammals: a review of the complex regulation of melatonin synthesis by norepinephrine, peptides, and other pineal transmitters. *Pharmacol Rev*, **55**(2), 325–95.
- Sizonenko PC, Lang U, Rivest RW and Aubert ML (1985). The pineal and pubertal development. *Ciba Found Symp*, **117**, 208–30.
- Skene DJ, Bojkowski CJ, Currie JE, Wright J, Boulter PS and Arendt J (1990). 6-sulphatoxymelatonin production in breast cancer patients. *J Pineal Res*, **8**(3), 269–76.
- Skene DJ, Bojkowski CJ and Arendt J (1994). Comparison of the effects of acute fluvoxamine and desipramine administration on melatonin and cortisol production in humans. *Br J Clin Pharmacol*, **37**(2), 181–6.
- Skene DJ, Papagiannidou E, Hashemi E, Snelling J, Lewis DF, Fernandez M and Ioannides C (2001). Contribution of CYP1A2 in the hepatic metabolism of melatonin: studies with isolated microsomal preparations and liver slices. *J Pineal Res*, **31**(4), 333–42.
- Smith JA, Mee TJ and Barnes JL (1977). Elevated melatonin serum concentrations in psychiatric patients treated with chlorpromazine. *J Pharm Pharmacol*, **29**(Suppl), 30P.
- Soule HD and McGrath CM (1980). Estrogen responsive proliferation of clonal human breast carcinoma cells in athymic mice. *Cancer Lett*, **10**, 177.
- Soule HD, Vazguez J, Long A, Albert S and Brennan M (1973). A human cell line from a pleural effusion derived from a breast carcinoma. *J Natl Cancer Inst*, **51**, 1409–16.
- Steinberg KK, Thacker SB, Smith S J, Stroup DF, Zack MM, Flanders WD and Berkelman RL (1991). A meta-analysis of the effect of estrogen replacement therapy on the risk of breast cancer (published erratum in *JAMA*, 1991, **266**, 1362). *JAMA*, **265**, 1985–90.
- Stephan FK and Zucker I (1972). Circadian rhythm in drinking behaviour and locomotor activity of rats are eliminated by hypothalamic lesion. *Proc Natl Acad Sci USA*, **69**, 1583–6.
- Stevens RG (1987). Electric power use and breast cancer: a hypothesis. Am J Epidemiol, 125, 556-61.
- Strassman RJ, Appenzeller O, Lewy AJ, Qualis CR and Peake GT (1989). Increase in plasma melatonin, betaendorphin, and cortisol after a 28.5-mile mountain race: relationship to performance and lack of effect of naltrexone. J Clin Endocrinol Metab, **69**(3), 540–45.
- Strassman RJ, Qualls CR, Lisansky EJ and Peake GT (1991). Elevated rectal temperature produced by all-night bright light is reversed by melatonin infusion in men. *J Appl Physiol*, **71**(6), 2178–82.
- Sulkowski S, Sulkowska M and Skrzydlewska E (1999). Gap junctional intercellular communication and carcinogenesis. *Polish J Pathol*, **50**, 227–33.
- Symons AM, Arendt J and Pryde S J (1985). Differential effects of melatonin on the stimulated release of LH from dispersed pituitary cells of the prepubertal female rat. *J Endocrinol*, **107**(1), 107–12.
- Tamarkin L, Danforth D, Lichter A, DeMoss E, Cohen M, Chabner B and Lipman M (1982). Decreased nocturnal plasma melatonin peak in patients with estrogen receptor positive breast cancer. *Science*, **216**, 1003–5.
- Tamarkin L, Baird CJ and Aimeida OF (1985). Melatonin: a coordinating signal for mammalian reproduction? *Science*, **227**(4688), 714–20.
- Tarquini B, Perfetto F, Poli R and Tarquini R (1994). Daytime circulating melatonin levels in smokers. *Tumori*, **80**(3), 229–32.

- Thapan K (2001). The spectral sensitivity of light-induced melatonin suppression in humans. Thesis, University of Surrey.
- Thapan K, Arendt J and Skene DJ (2001). An action spectrum for melatonin suppression: evidence for a novel non-rod, non-cone photoreceptor system in humans. *J Physiol*, **535**(Part 1), 261–7.
- Thomas HV, Key TJ, Allen DS, Moore JW, Dowsett M, Fentiman IS and Wing DY (1997a). A prospective study of endogenous serum hormone concentrations and breast cancer risk in premenopausal women on the island of Guernsey. *Br J Cancer*, **76**, 401–5.
- Thomas HV, Key TJ, Allen DS, Moore JW, Dowsett M, Fentiman IS and Wing DY (1997b). A prospective study of endogenous serum hormone concentrations and breast cancer risk in postmenopausal women on the island of Guernsey. *Br J Cancer*, **75**, 1075–9.
- Thomas HV, Reeves GK and Key TJA (1997c). Endogenous estrogen and postmenopausal breast cancer: a quantitative review. *Cancer Causes Control*, **8**, 922–8.
- Toniolo PG, Levitz M, Zeleniuch-Jacquotte A, Banerjee S, Koenig KL, Shore RE, Strax P and Pasternack BS (1995). A prospective study of endogenous estrogens and breast cancer in postmenopausal women. *J Natl Cancer Inst*, **87**, 190–97.
- Tosini G and Menaker M (1996). Circadian rhythms in cultured mammalian retina. Science, 272(5260), 419–21.
- Trentini GP, Genazzani AR, Criscuolo M, Petraglia F, De Gaetani C, Ficarra G, Bidzinska B, Migaldi M and Genazzani AD (1992). Melatonin treatment delays reproductive aging of female rat via the opiatergic system. *Neuroendocrinology*, **56**(3), 364–70.
- Trichopoulos D, MacMahon B and Cole P (1972). Menopause and breast cancer risk. J Natl Cancer Inst, 48, 605–13.
- Trosko JE (2003). The role of stem cells and gap junctional intercellular communication in carcinogenesis. *J Biochem Mol Biol*, **36**, 43–8.
- Turek FW (1996). Melatonin hype hard to swallow. Nature, 379(6563), 295-6.
- Turek FW and van Reeth O (1988). Manipulation of the circadian clock with benzodiazepines: implications for altering the sleep-wake cycle. *Pharmacopsychiatry*, **21**(1), 38–42.
- Ubeda A, Trillo MA, House DE and Blackman CF (1995). Melatonin enhances junctional transfer in normal C3H/10T1/2 cells. *Cancer Lett*, **91**, 241–5.
- Ursin G, Tseng CC, Paganini-Hill A, Enger S, Wan PC, Formenti S, Pike MC and Ross RK (2002). Does menopausal hormone replacement therapy interact with known factors to increase risk of breast cancer? *J Clin Oncol*, **20**, 699–706.
- Vakkuri O (1985). Diurnal rhythm of melatonin in human saliva. Acta Physiol Scand, 124(3), 409-12.
- Vanecek J, Pavlik A and Illnerova H (1987). Hypothalamic melatonin receptor sites revealed by autoradiography. Brain Res, **453**, 359–62.
- Vijayalaxmi, Thomas CR Jr, Reiter RJ and Herman TS (2002). Melatonin: from basic research to cancer treatment clinics. *J Clin Oncol*, **20**, 2575–601.
- von Gall C, Garabette ML, Kell CA, Frenzel S, Dehghani F, Schumm-Draeger PM, Weaver DR, Korf HW, Hastings MH and Stehle JH (2002a). Rhythmic gene expression in pituitary depends on heterologous sensitization by the neurohormone melatonin. *Nat Neurosci*, **5**(3), 234–8.
- von Gall C, Stehle JH and Weaver DR (2002b). Mammalian melatonin receptors: molecular biology and signal transduction. *Cell Tissue Res*, **309**, 151–62.
- Vonderhaar BK (1984). Hormones and growth factors in mammary gland development. In *Control of Cell Growth and Proliferation* (C M Veneziale, ed). New York, Van Norstrand, Reinhold and Co, pp 11–13.
- Vonderhaar BK (1987). Prolactin: transport, function and receptors in mammary gland development and differentiation. In *The Mammary Gland* (MC Neville and CW Daniel, eds). New York, Plenum, pp 383–438.
- Vonderhaar BK (1998). Prolactin: the forgotten hormone of human breast cancer. *Pharmacol Ther*, **79**(2), 169–78.
- Vonderhaar BK and Biswas R (1987). Prolactin effects and regulation of its receptors in human mammary tumor cells. In *Cellular and Molecular Biology of Mammary Cancer* (D Medina et al, eds). New York, Plenum, pp 205–19.

- Vondrasova-Jelinkova D, Hajek I and Illnerova H (1999). Adjustment of the human melatonin and cortisol rhythms to shortening of the natural summer photoperiod. *Brain Res*, **816**(1), 249–53.
- Voordouw BC, Euser R, Verdonk RE, Alberda BT, de Jong FH, Drogendijk AC, Fauser BC and Cohen M (1992). Melatonin and melatonin-progestin combinations alter pituitary-ovarian function in women and can inhibit ovulation. J Clin Endocrinol Metab, 74(1), 108–17.
- Voskuil DW, Bosma A, Vrieling A, Rookus MA and van't Veer LJ (2004). Insulin-like growth factor (IGF)-system mRNA quantities in normal and tumor breast tissue of women with sporadic and familial breast cancer risk. *Breast Cancer Res Treat*, **84**(3), 225–33.
- Voultsios A, Kennaway DJ and Dawson D (1997). Salivary melatonin as a circadian phase marker: validation and comparison to plasma melatonin. *J Biol Rhythms*, **12**, 457–66.
- Waldhauser F, Lieberman HR, Lynch HJ, Waldhauser M, Herkner K, Frisch H, Vierhapper H, Waldhausl W, Schemper M and Wurtman RJ (1987). A pharmacological dose of melatonin increases PRL levels in males without altering those of GH, LH, FSH, TSH, testosterone or cortisol. *Neuroendocrinology*, **46**(2), 125–30.
- Wang DY, Hampson S, Kwa HG, Moore JW, Bulbrook RD, Fentiman IS, Hayward JL, King RJ, Millis RR, Rubens RD and Allen DS (1986). Serum prolactin levels in women with breast cancer and their relationship to survival. *Eur J Cancer Clin Oncol*, **22**, 487–92.
- Warman VL, Dijk DJ, Warman GR, Arendt J and Skene DJ (2003). Phase advancing human circadian rhythms with short wavelength light. *Neurosci Lett*, **342**(1–2), 37–40.
- Waters MJ and Conway-Campbell BL (2004). The oncogenic potential of autocrine human growth hormone in breast cancer. *Proc Natl Acad Sci USA*, **101**(42), 14992–3.
- Weaver DR and Reppert SM (1986). Maternal melatonin communicates daylength to the fetus in Djungarian hamsters. *Endocrinology*, **119**(6), 2861–3.
- Wehr TA (2001). Photoperiodism in humans and other primates: evidence and implications. *J Biol Rhythms*, **16**(4), 348–64.
- Wehr TA, Moul DE, Barbato G, Giesen HA, Seidel JA, Barker C and Bender C (1993). Conservation of photoperiodresponsive mechanisms in humans. *Am J Physiol*, **265**(4 Part 2), R846–57.
- Wehr TA, Schwartz PJ, Turner EJ, Feldman-Naim S, Drake CL and Rosenthal NE (1995). Bimodal patterns of human melatonin secretion consistent with a two-oscillator model of regulation. *Neurosci Lett*, **194**(1-2), 105-8.
- Wehr TA, Aeschbach D and Duncan WC Jr (2001). Evidence for a biological dawn and dusk in the human circadian timing system. *J Physiol*, **535**(Part 3), 937–51.
- Welsch CW and Nagasawa H (1977). Prolactin and murine mammary tumorigenesis: a review. *Cancer Res*, **37**, 951–63.
- Wever RA (1989). Light effects on human circadian rhythms: a review of recent Andechs experiments. *J Biol Rhythms*, **4**(2), 161–85.
- White BH, Sekura RD and Rollag MD (1987). Pertussis toxin blocks melatonin-induced aggregation in Xenopus dermal melanophores. J Comp Physiol [B], **157**, 153–9.
- Wilson S T, Blask D E and Lemus-Wilson A M (1992). Melatonin augments the sensitivity of MCF-7 human breast cancer cells to tamoxifen *in vitro*. J Clin Endoc Metabolism, **75**, 669–70.
- Wright J, Aldhous M, Franey C, English J and Arendt J (1986). The effects of exogenous melatonin on endocrine function in man. *Clin Endocrinol (Oxf)*, **24**(4), 375–82.
- Wright KP Jr and Badia P (1999). Effects of menstrual cycle phase and oral contraceptives on alertness, cognitive performance, and circadian rhythms during sleep deprivation. *Behav Brain Res*, **103**(2), 185–94.
- Wright KP Jr and Czeisler CA (2002). Absence of circadian phase resetting in response to bright light behind the knees. *Science*, **297**(5581), 571.
- Wright KP Jr, Hughes RJ, Kronauer RE, Dijk DJ and Czeisler CA (2001). Intrinsic near-24-h pacemaker period determines limits of circadian entrainment to a weak synchronizer in humans. *Proc Natl Acad Sci USA*, 98(24), 14027–32.
- Yager JD and Liehr JG (1996). Molecular mechanisms of estrogen carcinogenesis. *Ann Rev Pharmacol Toxicol*, **36**, 203–32.

- Yoneyama S, Hashimoto S and Honma K (1999). Seasonal changes of human circadian rhythms in Antarctica. *Am J Physiol*, **277**(4 Part 2), R1091–7.
- Yoon IY, Kripke DF, Elliott JA and Youngstedt DS (2003). Luteinizing hormone following light exposure in healthy young men. *Neurosci Lett*, **341**(1), 25–8.
- Yue W, Santen RJ, Wang JP, Li Y, Verderame MF, Bocchinfuso WP, Korach KS, Devansan P, Todorovic R, Rogan RG and Cavalieri EL (2003). Genotoxic metabolites of estradiol in breast: potential mechanism of estradiol induced carcinogenesis. J Steroid Biochem Mol Biol, 86(3–5), 477–86.
- Zaidan R, Geoffriau M, Brun J, Taillard J, Bureau C, Chazot G and Claustrat B (1994). Melatonin is able to influence its secretion in humans: description of a phase-response curve. *Neuroendocrinology*, **60**(1), 105–12.
- Zarazaga L A, Malpaux B, Bodin L and Chemineau P (1998). The large variability in melatonin blood levels in ewes is under strong genetic influence. *Am J Physiol*, **274**(4 Part 1), E607–10.
- Zeitzer JM, Daniels JE, Duffy JF, Klerman EB, Shanahan TL, Dijk DJ and Czeisler CA (1999). Do plasma melatonin concentrations decline with age? *Am J Med*, **107**(5), 432–6.
- Zeitzer JM, Dijk DJ, Kronauer R, Brown E and Czeisler C (2000). Sensitivity of the human circadian pacemaker to nocturnal light: melatonin phase resetting and suppression. *J Physiol*, **526** (Part 3), 695–702.
- Zhdanova IV, Lynch HJ and Wurtman RJ (1997). Melatonin: a sleep-promoting hormone. Sleep, 20(10), 899–907.
- Zhu Y, Brown HN, Zhang Y, Stevens RG and Zheng T (2005). Period3 structural variation: a circadian biomarker associated with breast cancer in young women. *Cancer Epidemiol Biomarkers Prev*, **14**, 268–70.

4 Effects of EMFs on Melatonin

4.1 In vitro studies

In vitro studies of exposure to EMFs divide into two types of investigation: effects on the production of melatonin by cells from the pineal gland; and effects on the action of melatonin on cells.

4.1.1 Effects on melatonin production in vitro

There are only a few studies that have investigated the effect of magnetic fields on melatonin production *in vitro*. All used rodents as the source of pineal gland cells but there are marked differences in their methodology. Most used power frequencies (50 or 60 Hz), but the field strength (50 μ T – 1 mT) and duration (1 hour – 12 hours) differ between the studies.

Both direct and indirect measures have been used to show cellular responses to melatonin. Direct measures include melatonin content or melatonin release from cells. Indirect measures can be made from the activity of *N*-acetyltransferase (NAT), an enzyme involved in the synthesis of melatonin, or of hydroxyindole-*O*-methyltransferase (HIOMT), an enzyme responsible for methylation and hence release of melatonin from the cells. Most of the studies have stimulated pharmacologically the production of melatonin in the isolated glands by the addition of noradrenaline (NA) or isoproterenol. NA is naturally released from the nerve endings in the pineal gland in the intact animal at the onset of the darkness and initiates the synthesis of melatonin. Isoproterenol (isoprenaline) is a beta-adrenergic receptor agonist and will mimic NA through its stimulation of cyclic adenosine monophosphate (cAMP) production.

Lerchl et al (1991) exposed pineal glands from young rats, removed during the day-light period, to a combination of a static field (44 μ T) and a low frequency magnetic field (44 μ T at 33.7 Hz). These are the theoretical conditions for cyclotron resonance of the calcium ion. One experiment was performed. The glands were stimulated by the addition of NA (10⁻⁶ M): seven were exposed to the field for 2.5 hours, and a further seven were used as controls. Exposure caused a reduction in NAT activity, melatonin production and melatonin release into the culture medium. As the authors commented, this result was surprising as calcium normally promotes NAT activity.

Rosen et al (1998) also used pineal glands from the rat, but this study was different to the other studies in that the pineal gland was separated into individual cells. The isolated cells were exposed for 12 hours to a 50 μ T, 60 Hz field and melatonin production was stimulated by the addition of NA at 10⁻⁶ M; the total amount of melatonin released from the cells was measured. The overall experimental protocol was well designed with two matched exposure systems both capable of being active or sham; sham–sham experiments were also run. One potentially interesting aspect of the protocol was to run some experiments blinded and compare these results with the non-blinded results. In the blinded experiments, three of the five did not show a reduction due to field exposure, and one even had a 31% increase. However, the five blinded experiments combined showed a 26% reduction, although the claim by the authors that this was a significant reduction is statistically dubious and certainly could not be regarded as robust. The authors pointed out that in these three particular experiments there was a poor response to NA stimulation and hence no increase in the level of melatonin production by the cells. Thus there was a failure to stimulate production rather than a failure of the magnetic field to suppress melatonin synthesis. The overall result of the ten experiments was that magnetic field exposure caused a statistically significant 46% reduction in stimulated melatonin release (although the standard deviations quoted in the publication for the individual experiments are erroneous).

Chacon (2000) used rat pineal glands to study NAT activity. The glands were isolated during the rats' dark cycle; the procedure was carried out in dim red light to avoid exposing the animals and glands to white light. The NAT activity was predicted to be at its natural peak at this time; no pharmacological stimulation was given. Four experiments were performed which investigated three magnetic field strengths (10, 100 and 1000 μ T) at 50 Hz and sham exposure; six test and six control glands were used in each experiment. The enzyme activity decreased by approximately 20% after a one hour exposure to the highest field strength tested (1000 μ T) but was not significantly altered by field strengths of 10 or 100 μ T. The interpretation of the result may be complicated by the removal of the pineal gland during the rats' dark period, which may have had an effect on melatonin synthesis and a confounding effect on the result.

A study by Brendel et al (2000) used pineal glands from the Djungarian hamster. It also differed from the previous studies in that the glands were maintained in a flow system, so that changes of melatonin released from the glands could be monitored throughout the duration of the experiment. The experimental protocol appears to have been well designed with random allocation of exposure or sham to identical exposure systems and the experiments run blind. The glands were removed during the normal light period, and hence avoided the possible confounding factors faced by the study by Chacon. The pineal glands were exposed for eight hours to a field strength of 86 μ T at either 50 or 16.67 Hz (a frequency generated in some railway systems), using a rectangular waveform. The glands were stimulated for 30 minutes with isoproterenol (10^{-7} M) to increase melatonin production. Melatonin release was assessed at hourly intervals. Although an apparently well-conducted study, the interpretation of the results was not convincing. The authors concluded that EMF exposure inhibited melatonin production in both the 50 and 16.67 Hz experiments. However, there was only one time point in one of four experiments using 50 Hz where the release of melatonin was statistically different from the shamexposed values. A similar single point difference was obtained for the 16.67 Hz experiments. From the graphs of the data, it is equally tempting to conclude that the sham-exposed values of melatonin release have varied at these time points in comparison with the pattern of the other experiments and the pattern of melatonin release is unchanged between the experiments. There is no convincing evidence in this study that EMF exposure causes inhibition of melatonin production.

Lewy et al (2003) used rat pineal glands isolated in the morning and hence during the 12 hour light period. The glands were exposed for four hours to a 50 Hz magnetic field at 1 mT. The activity of enzymes NAT and HIOMT was measured, as well as the release of melatonin into the incubation liquid. Two exposure systems were used that could be set to either sham or exposure. The glands were stimulated with NA (10^{-6} M), which caused a modest increase in both enzymes that was not suppressed by the exposure to the magnetic field. However, exposing the glands to the field for 30 minutes prior to the NA enhanced the stimulation (10-25%) compared with simultaneous exposure to NA and the

magnetic field. Addition of NA markedly increased the release of melatonin into the incubation liquid. Field exposure given simultaneously with NA or 30 minutes prior to NA administration caused a significant increase (approximately 50%) in melatonin release. This result is in contrast to the other studies where if an effect was found, it was a decrease. There was no change in melatonin release due to field exposure in glands that had not been stimulated by NA.

A study by Tripp et al (2003) was similar to the Brendel et al investigation in that it used a flow system to detect changes of melatonin release during the course of the exposure. The glands were obtained during the animals' light period and were exposed or sham exposed in identical systems. The exposure was for 4 hours to a circularly polarised 50 Hz magnetic field at 500 μ T. Samples were taken every 30 minutes; the process used remote collection to avoid potential artefacts involved in manual collection. The glands were not stimulated pharmacologically and no field-dependent changes in melatonin release were detected.

To reconcile the various results above and find a common pattern of response to magnetic fields is difficult (Table 4.1). The 'cyclotron resonance' study of Lerchl and colleagues is not directly comparable to the studies using power frequency fields as the frequency was slightly lower at 33.7 Hz and the static field was set at 44 μ T: it is the only study to have applied such fields to isolated pineal glands.

Exposure	Effect	Reference
'Ion-cyclotron resonance' for calcium	NA stimulation of melatonin production and release reduced	Lerchl et al, 1991
50 μT, 60 Hz for 12 h	NA stimulation of melatonin release reduced	Rosen et al, 1998
1 mT, 50 Hz for 1 h	NAT activity decreased	Chacon, 2000
86 μT,50 Hz or 16.67 Hz for 8 h	Isoproterenol stimulation of melatonin production reduced	Brendel et al, 2000
1 mT, 50 Hz for 4 h	NA stimulation of melatonin release increased	Lewy et al, 2003
0.5 mT, 50 Hz for 4 h	No effect on melatonin release	Tripp et al, 2003

TABLE 4.1 Studies of the effects of magnetic fields on melatonin production

Five of the studies reviewed used power frequency fields. The study by Tripp and colleagues used relatively high magnetic field strengths (500 μ T) for four hours, the glands were removed during the light period, so the NAT activity would be low and the cells were not pharmacologically stimulated; no effect of EMF exposure was found. This is in agreement with the Lewy et al study where exposure for four hours at 1 mT produced no effect in cells that had not been stimulated by NA. The study by Chacon also used non-stimulated cells; however, exposure to 1 mT for one hour caused a decrease in NAT activity. These glands were isolated during the dark cycle and NAT activity was expected to be at its highest, which may be equivalent to pharmacological stimulation in other authors' studies. Based on these three studies, it could be suggested that NAT activity needs to be high, either achieved naturally during the dark cycle or artificially by chemical stimulation, to see an effect of EMF exposure of field strengths between 500 and 1000 μ T. However, there is no agreement in which direction the effect is seen; Lewy et al found an increase in NAT activity and melatonin release, whereas Chacon showed a reduction in NAT activity. These contradictory findings cast doubt on whether the reported effects are real.

In the studies using lower field strengths (86 and 50 μ T) both Brendel et al and Rosen et al used relatively long exposures – 8 or 12 hours. Both stimulated the cells with chemicals and achieved reduced levels of melatonin after field exposure. Chacon also used similar field strengths (10–100 μ T) but field exposures of one hour did not cause a change in NAT activity. A possible explanation could be that, to see an effect, the NAT activity needs to be high and combined with a relatively long duration if the field strength is low. However, this postulation of an effect of low field strengths would be essentially based on two studies, neither of which in isolation could be regarded as robust.

4.1.2 Effects on the action of melatonin in vitro

The main interest in this area was caused by the claim that exposure to magnetic fields can block the inhibitory effect of melatonin on growth of breast cancer cells. The original work was reported by Liburdy et al (1993) in a study using a human oestrogen-responsive breast cancer cell line (MCF-7). They found that the proliferation of MCF-7 cells can be slowed by the addition of physiological concentrations of melatonin (1 nM). However, if the cells are simultaneously exposed to a 60 Hz, 1.2 μ T magnetic field, then the effect of melatonin on the rate of proliferation is reduced. The effects are fairly small and can only been seen after seven days in culture. They suggested that the magnetic field disrupted either the ligand/receptor interaction or the subsequent signalling pathway. The authors found no effect at a magnetic field strength of 0.2 μ T and suggested a threshold between 0.2 and 1.2 μ T. No effect was seen using field exposure alone.

A similar effect of a 60 Hz field was reported by Harland and Liburdy (1997) but using tamoxifen (100 nM) rather than melatonin to bring about the initial inhibition. The effect has been reported in other cell lines, namely a second breast cancer cell line, T47D (Harland et al, 1998), and a human glioma cell line 5F757 (Afzal and Liburdy, 1998). All of these studies come from Liburdy's group.

There has been some criticism of the initial study. A report from the National Institute of Environmental Health Sciences (NIEHS, 1999) made the following comments: 'There was some concern about the experimental design of these studies' and 'because the effect was small, the importance of these findings for human health is not clear'. Previously, AGNIR (2001) endorsed the NIEHS concern about the robustness of the effect and felt that the changes were small (10–20% growth over seven days).

Until the paper by Blackman et al (2001) there have been two cited independent replications of the Liburdy et al findings. These appeared as abstracts at the Bioelectromagnetics Society Annual Meeting in Florida 1998. Neither has appeared in peer-reviewed journals as a full paper. Blackman et al (2001) is the only peer-reviewed, full paper that sets out to replicate the Liburdy et al findings, although a similar effect has been shown using slightly different conditions (Ishido et al, 2001).

The Blackman et al study used MCF-7 cells and the experimental criteria supplied by Liburdy. However, the protocol appears to have been modified and improved on the original. Two incubators were used, each containing an exposure system. Both melatonin and tamoxifen effects were investigated. The melatonin part of the study was fairly weak, with poor controls, small sample size, exposures not random or blinded, and no test with magnetic field but without melatonin. The melatonin caused a 17% inhibition in cell numbers compared with controls; this was not statistically significant. The design of the tamoxifen study was more rigorous. The protocol was not an exact replication, however, since the

IN VITRO STUDIES

authors added only 25% of the tamoxifen concentration used by Liburdy. There were nine separate experiments, and in each experiment there was a total of 48 petri-dishes. Each incubator had 24 dishes, 12 with and 12 without tamoxifen. The assays were blinded, and incubator allocation was random. Tamoxifen caused a 25% inhibition in cell numbers, which was reduced to a 13% inhibition by exposure to a 60 Hz magnetic field at 1.2 μ T. This result confirmed the Liburdy et al findings, in which a 40% inhibition was reduced to 25% by EMF exposure. However, it should be noted that a 25% inhibition of cell numbers by tamoxifen implies that the experimental protocol was not optimal as a greater inhibitory effect of tamoxifen would be expected.

The effects of stronger magnetic fields were studied by Leman et al (2001) in three breast cancer cell lines that had, according to published reports, different metastatic capabilities: MDA-MB-435 cells were considered to have 'high' metastatic capabilities, MDA-MB-231 cells were considered 'low', and MCF-7 cells were considered 'non-metastatic'. Only their 'low' and 'non-metatastic' cells responded to melatonin and optimum inhibition was achieved at 1 mM concentration of melatonin (a million-fold higher than used in the Liburdy et al study). Cell growth was not affected in their culture conditions by exposure for 1 hour to a pulsed field at 300 μ T (pulse duration 20 ms with a repetition rate of 2 Hz) repeated for 3 days. The cells' invasive properties were evaluated in an *in-vitro* assay; MCF-7 cells had a low invasion rate which was not affected by addition of melatonin or EMF exposure. (The 'non-metastatic' definition of MCF-7 cells was based on the study by Shafie and Liotta (1980) which found the incidence of liver, lung and spleen metastases in athymic, ovariectomised nude mice inoculated with MCF-7 cells and supplemented with oestrogen to be 40–60% after 5–7 weeks.)

Ishido et al (2001) also mainly used higher magnetic fields strengths. MCF-7 cells (supplied by Liburdy) were exposed to 0, 1.2 or 100 μ T at 50 Hz for 7 days. Melatonin at concentrations of 10⁻⁹ M or higher induced inhibition of intracellular cAMP which was blocked by exposure to a 50 Hz field at 100 μ T. These authors also found that DNA synthesis was reduced 20% by melatonin at 10⁻¹¹ M (a concentration 100-fold less than used by Liburdy et al and not effective at causing cAMP accumulation in their system) and this reduction was blocked by exposure to a magnetic field at 1.2 μ T. This result was presumably only obtained from a single experiment, as the number of repeat experiments is not mentioned. The result is consistent, however, with the Liburdy et al findings.

All the experimental studies mentioned above were performed with the breast cancer cell line MCF-7; however, studies using these cells can be problematic. This cell line has been in existence for many years and its characteristics continue to change. This was recognised by Liburdy et al (1993) who commented that MCF-7 cells display heterogeneity to their response to melatonin and that sub-clones may change sensitivity to melatonin as passage number increases. This finding was endorsed by two of the groups who attempted to replicate the work of Liburdy et al. Morris et al (1998) had difficulty obtaining a melatonin responsive cell line, despite being supplied with cells from Liburdy; the cells also had different growth characteristics. Luben and Morgan (1998), who also had cells supplied by Liburdy, found the effects of magnetic fields variable and dependent on the source of cells, and the number of times the cells had been cultured, and suggested that the effects were specific to individual clone phenotypes. Although the MCF-7 cell line has undoubtedly provided a useful model to investigate effects on isolated breast cancer cells, it is only one possible model in cells that have been separated from their natural environment and therefore its implication for breast cancer in general is limited.

Other research groups, working on subjects unrelated to EMFs, also have had problems finding a consistent MCF-7 cell response to melatonin. Baldwin et al (1998) concluded that melatonin did not inhibit oestrogen stimulated MCF-7 cell growth, whereas Molis et al (1994) claimed inhibition. Ram et al (2000) found significant differences in the responsiveness of various stocks of MCF-7 cells to the growth inhibitory effects of melatonin (see Chapter 5).

Notwithstanding these limitations, the effect of EMFs on the inhibition of MCF-7 cell growth by melatonin has been independently replicated, and thus merits some attention. Although it must be emphasised that the effect is in isolated cells and thus does not imply an effect in a whole organism. Therefore, the inference that EMF exposure may have a detrimental effect on cancer growth in humans via a suppression of the action of melatonin or tamoxifen is not justified without more appropriate studies. Also it should be noted that a biological change is not the same as a health effect.

The importance of independent replication of the effect of EMF exposure on melatonin inhibition of MCF-7 cell growth lies not just with the biology, but also with our understanding of the physics. At present there is no generally accepted physical mechanism that can predict or explain how such weak magnetic fields can interact with biological systems.

Several studies have looked at the effect of EMF exposure on gap junction communication. In general, most of the studies show an inhibitory effect of EMF exposure on cell–cell communication (Schimmelpfeng et al, 1996; Chiang et al, 1999; Hu et al, 2002; Lohmann et al, 2002; Yamaguchi et al, 2002; Marino et al, 2003). However, the exposure conditions were not consistent between studies and varied in magnetic field strength, frequency and duration. These studies have not been replicated or, where replication has been tried, the inhibitory phenomenon was not shown (Griffin et al, 2000). A few studies have investigated the combined effect of melatonin and EMF exposure on gap junction communication. Physiological levels of melatonin enhanced transfer of a fluorescent dye between cells and this enhancement was blocked by 50 Hz EMF exposure at 160 mT (Ubeda et al, 1995).

The effect of a magnetic field on the rate of radical pair recombination is a well-established phenomenon in chemistry. The effect depends on the influence of magnetic fields on the recombination probability of the radicals in the first nanoseconds of their creation. It is a static magnetic field phenomenon but, because the recombination times are so short (nanoseconds), the power frequencies of 50 or 60 Hz would be essentially static for that fraction of a second. Although the effect of magnetic fields greater than 1 mT has been established, the effect of much weaker fields on biological systems is less certain and is still the subject of research (Brocklehurst and McLauchlan, 1996; Brocklehurst, 2002). A further complication is the geomagnetic field, which is typically about 50 μ T, but will vary markedly in the presence of some metal objects. The variation in the geomagnetic field due to walking near a steel object such as a filing cabinet or travelling in a lift, for example, will probably be much larger than that due to the presence of 50 or 60 Hz power frequencies.

4.1.3 Summary

Overall, the evidence that EMF exposure causes changes in melatonin production or release in isolated pineal glands is not convincing. This is despite the EMF field strengths used in the experiments being considerably higher than those usually encountered in the environment. Therefore a direct effect of EMF

exposure on isolated pineal glands or pineal cells is doubtful. However, this does not exclude an effect on animals as it may require an intact circadian system or input via a sensory organ, ie the eyes. Relatively few *in vitro* studies have been undertaken and some have serious limitations. All the studies have used cells from rodents and hence the results may not be directly applicable to human cells. There is scope for well-designed studies to investigate further the possibility of direct EMF effects on isolated pineal cells or glands.

The evidence that EMF exposure interferes with the action of melatonin on breast cancer cells *in vitro* is intriguing (Table 4.2) and there appears to be some supporting evidence in terms of independent replication using MCF-7 cells. However, this particular effect seems to be elusive: it requires a specific sub-clone of the MCF-7 cell line, and even this can transform and so lose its responsiveness. Overall, these results cannot not be regarded as a robust field-dependent effect. In view of these limitations, the significance of the biological findings is of doubtful relevance to other breast cancer cell lines, and of dubious significance to human health.

Exposure	Effect	Reference
1.2 $\mu\text{T},$ 60 Hz, for 7 days	EMF exposure partially blocked melatonin (10^{-9} M) inhibition in MCF-7 cells. Similar result with tamoxifen (10^{-7} M)	Liburdy et al, 1993 Harland and Liburdy, 1997
1.2 μT, 60 Hz for 7 days	EMF exposure partially blocked tamoxifen (2.5 10 ⁻⁸ M) inhibition of MCF-7 cells	Blackman et al, 2001
1.2 or 100 μT, 50 Hz for 7 days	Inhibition of DNA synthesis by melatonin (10^{-11} M) partially blocked by 1.2 μ T EMF exposure. 100 μ T blocked cAMP inhibition by melatonin (10^{-9} M) in MCF-7 cells	lshido et al, 2001
0.3 mT pulsed for 20 ms at 2 Hz for 1 h repeated over 3 days	Growth of MCF-7 cells unaffected	Leman et al, 2001

TABLE 4.2 Studies of the effects	of magnetic fields	on responses of	f cells to melatonin	or tamoxifen
----------------------------------	--------------------	-----------------	----------------------	--------------

4.2 In vivo studies

Various laboratory studies have investigated the effects of EMFs on melatonin rhythms in animals. Most of these have used rats, although some studies have used mice. Other studies have used seasonal breeders, such as Djungarian hamsters and sheep, and a few studies have used cattle or non-human primates.

4.2.1 Rodent studies

Attention was first focused on the potential effects of electric fields, before interest turned to magnetic fields. Early studies (Wilson et al, 1981, 1983, 1986; Reiter et al, 1988) reported that the exposure of rats to electric fields significantly suppressed pineal melatonin and the activity in the pineal gland of an enzyme (NAT) important in the synthesis of melatonin, and that this effect was transient, appearing within three weeks of exposure but recovering within three days following the cessation of exposure.

A similar suppression of pineal melatonin was reported following the prenatal and neonatal exposure of rats to power frequency electric fields; no simple dose–response relationship was apparent. However, a later study from the same laboratory (Sasser et al, 1991) briefly reported that they were unable to reproduce the reduction in pineal melatonin. Another laboratory (Grota et al, 1994) also reported that exposure to power frequency electric fields had no effect on pineal melatonin levels or NAT activity in rats, although serum melatonin levels were reduced. These studies are summarised in Table 4.3.

TABLE 4.3 Studies of effects of exposure to 60 Hz electric fields on melatonin levels in Sprague-Dawley rats

Assay	Exposure	Result	Reference
Pineal NAT and melatonin (male)	1.7–1.9 kV m ⁻¹ , 20 h per day, 30 days	No change in night pineal NAT, reduction in night melatonin (p<0.05)	Wilson et al, 1981, 1983
Pineal NAT and melatonin (male)	39 kV m ⁻¹ , 20 h per day, for 1, 2, 3 or 4 weeks	Significantly reduced melatonin and NAT activity after both 3 and 4 weeks exposure, withdrawal of the fields returned night melatonin and NAT to normal	Wilson et al, 1986
Pineal melatonin (male and female)	10, 65 or 130 kV m ⁻¹ , <i>in utero</i> and 23 days after birth	Reduced pineal melatonin levels (p<0.001), melatonin rhythm phase delayed by 1.4 h	Reiter et al, 1988
Pineal melatonin and NAT, serum melatonin activity (male)	35 kV m^{-1} , 20 h per day, for 30 days	No change in night pineal or melatonin levels, reduced night-time serum melatonin levels	Grota et al, 1994
Pineal melatonin (male and female)	20 h per day, for 30 days	No significant effect	Sasser et al, 1991

The difficulties in repeating the results of some of the earlier studies and the possible use of unreliable techniques have been noted by Brady and Reiter (1992). In particular, Reiter (1993) questioned whether the reported effects were an artefact of the method of melatonin measurement or some other methodological procedure.

More recent work has concentrated on studies of the effect of exposure to power frequency magnetic fields. These studies are summarised in Table 4.4. An extensive series of tests has been carried out by Kato and colleagues of the effects of exposure to circularly or linearly polarised power frequency magnetic fields on pineal and serum melatonin levels in male rats (Kato et al, 1993, 1994a–d, summarised in Kato and Shigemitsu, 1997). However, there is a major difficulty with the interpretation of many of these studies: the sham-exposed group was sometimes treated as a 'low dose' exposed group because the animals were exposed to stray magnetic fields (of less than 2%) generated by the exposure system. Thus statistical comparison was sometimes made with historical controls. Such procedures fail to allow for the inter-experimental variability that was reported in replicate studies by Kato and Shigemitsu (1997). Kato and colleagues seem to have taken these decisions *post hoc*, since the concurrent sham-exposed groups were treated in two different ways. This will also have increased the number of statistical comparisons made (based on the multiple use of Student's t-test), increasing the possibility of false positives.

Assay	Exposure	Result	Reference
Pineal and serum melatonin levels in Wistar-King rats	50 Hz, circularly polarised, 1, 5, 50 or 250 μT, for 6 weeks	Night-time and some daytime reductions in serum and pineal melatonin	Kato et al, 1993
Serum melatonin levels in Wistar- King rats	50 Hz, circularly polarised, 1 μT , for 6 weeks	Night-time melatonin levels reduced (p<0.05); returning to normal within 1 week	Kato et al, 1994a
Pineal and serum melatonin levels in (pigmented) Long-Evans rats	50 Hz, circularly polarised, 1 μ T, for 6 weeks	Night-time pineal and serum melatonin reduced (p<0.05)	Kato et al, 1994b
Serum melatonin levels in Wistar- King rats	50 Hz, horizontally or vertically polarised, 1 μ T, for 6 weeks	No significant effect	Kato et al, 1994c
'Antigonadotrophic' effect of melatonin on serum testosterone levels in Wistar-King rats	50 Hz, circularly polarised, 1, 5 or 50 μT , for 6 weeks	No significant effect	Kato et al, 1994d
Night-time serum melatonin levels and pineal NAT activity in Wistar rats	50 Hz, 1, 10 or 100 μT for 12 h (once) or 18 h per day for 30 days	Reduced melatonin and NAT activity after 100 µT (acute) (p<0.05) and 10 and 100 µT (chronic) (p<0.05)	Selmaoui and Touitou, 1995
Night-time excretion of melatonin urinary metabolite in Wistar rats	50 Hz, 1, 5, 100 or 500 μT for 24 h	No significant effects compared to baseline pre-exposure controls	Bakos et al, 1995, 1997, 1999
Night-time pineal melatonin levels in Sprague-Dawley rats (not DMBA treated)	50 Hz, 10 μT for 13 weeks	No significant effect	Mevissen et al, 1996a,b
Night-time serum melatonin, prolactin and oestradiol levels in Sprague-Dawley rats	50 Hz, 100 μT for 1 day, 1, 2, 4, 8 or 13 weeks	No consistent effects on melatonin; no effects on serum prolactin or oestradiol	Löscher et al, 1998
Night-time excretion of melatonin urinary metabolite in Sprague- Dawley rats	60 Hz, 1 mT continuous for 10 days or 6 weeks, or 1 mT intermittent for 2 days	No significant effect	John et al, 1998

TABLE 4.4 Studies of effects of exposure to magnetic fields on melatonin levels in rats

The first study (Kato et al, 1993) found that night-time pineal and serum melatonin levels were significantly reduced following six weeks exposure to circularly polarised power frequency magnetic fields of up to 250 μ T compared with levels in historical controls. In contrast, there was no difference between values in the exposed and concurrent sham-exposed groups. These results can be regarded as rather equivocal for the reasons outlined above. However, in a subsequent study (Kato et al, 1994a) a reduction in serum melatonin levels was observed in animals exposed to a circularly polarised magnetic field compared with sham-exposed animals. The next study (Kato et al, 1994b) reported the night-time suppression of serum and pineal melatonin in a different (pigmented) strain of rat in exposed animals compared with both sham-exposed animals and historical controls. In contrast to these results, a fourth study (Kato et al, 1994c) found that six weeks exposure to horizontally or vertically polarised power

frequency magnetic fields had no effect on pineal or serum melatonin compared with sham-exposed animals and historical controls. The reason for this difference between the effects of circularly polarised and horizontally or vertically polarised fields was not clear. In a final paper (Kato et al, 1994d) the idea was tested that a reduction in serum melatonin might be correlated with an increase in serum testosterone, given that melatonin is thought to have an 'antigonadotrophic' effect in some seasonally breeding animals. However, animals exposed to circularly polarised 50 Hz magnetic fields were found to have similar serum testosterone levels to their sham-exposed counterparts.

Other studies investigating the effects of magnetic field exposure on serum and pineal melatonin levels in rats have produced mostly negative results. Nevertheless, Selmaoui and Touitou (1995) reported that the acute exposure of rats to horizontally polarised power frequency magnetic fields significantly depressed night-time serum melatonin levels and NAT activity in the pineal gland; chronic exposure could produce a similar effect using a lower flux density (Touitou et al, 2002), suggesting a relationship between the effects of field intensity and duration of exposure. Sensitivity to magnetic fields may also depend on the age of the animals (Selmaoui and Touitou, 1999) as serum melatonin concentration and pineal enzyme activities in older animals were unaffected by magnetic field exposure. Bakos et al (1995, 1997, 1999, 2002) reported that exposure to a vertical or horizontal power frequency magnetic field had no consistent effects on the circadian excretion of the major urinary metabolite of melatonin. As part of a larger study of EMF effects on DMBA-induced mammary tumours and pineal function, Mevissen et al (1996a,b) found no effect of magnetic field exposure on pineal melatonin levels in rats not treated with DMBA. In addition, Löscher et al (1998) were unable to identify any consistent effects of power frequency magnetic field exposure for up to 13 weeks on night-time serum melatonin levels. Further, John et al (1998) reported that the exposure of rats for up to six weeks to power frequency magnetic fields under a variety of conditions had no effect on the circadian excretion of the major urinary metabolite of melatonin. Fedrowitz et al (2002) reported that exposure of rats to a magnetic field for two weeks had no significant effect on melatonin levels measured directly in the pineal or mammary tissues.

In addition, a few studies have been carried out using mice exposed to power frequency fields, and most have not reported any consistent field-dependent effects. A large-scale study (McCormick et al, 1995) briefly reported that exposure to continuous or intermittent magnetic fields had no effect on serum or pineal melatonin. Similarly, no effects on night-time plasma melatonin levels were observed in a small study using continuous, long-term exposure to a variable magnetic field (de Bruyn et al, 2001). As a part of a tumour promotion study, Heikkinen et al (1999) found no effect of chronic exposure to magnetic fields of varying intensity on the night-time excretion of a urinary metabolite of melatonin in mice exposed to ionising radiation at 4 Gy. However, Picazo et al (1998) described a significant reduction in the night-time serum melatonin levels of mice exposed up to sexual maturity for four generations.

4.2.2 Seasonal breeders

Several laboratories have investigated the effects of power frequency magnetic exposure on pineal activity, serum melatonin levels and reproductive development in seasonal breeding animals. These studies are summarised in Table 4.5.

Three laboratories examined these effects in Djungarian hamsters, in which the duration of melatonin secretion during the shortening days of autumn and winter inhibits reproductive activity. The most
TABLE 4.5	Studies of effects of exposure to magnetic fields on melatonin levels in seasonal bi	reeding
animals		

Assay	Exposure	Result	Reference
Night-time pineal and serum melatonin levels	60 Hz, 100 μT for 15 min, 2 h before dark period	Reduced and delayed night-time peak (p<0.05); effects not replicated	Yellon, 1994
Night-time pineal and serum melatonin levels	60 Hz, 100 μT for 15 min, 2 h before dark period	Reduced and delayed night-time peak (p<0.05); effects diminished	Yellon, 1996
Night-time pineal and serum melatonin levels; adult male reproductive status	60 Hz, 100 μT for 15 min, 2 h before dark period for 3 weeks	No effects on pineal and serum melatonin; no effects on melatonin-induced sexual atrophy	Yellon, 1996
Night-time pineal and serum melatonin levels; male puberty, assessed by testes weight	60 Hz, 100 μT for 15 min, 2 h before dark period from 16–25 days of age	Reduced and delayed night-time peak (p<0.05); not replicated; no effect on puberty	Truong et al, 1996
Night-time pineal and serum melatonin levels	60 Hz, 10 or 100 μT before or after dark onset or intermittent 100 μT, 15 or 60 min	No significant effects	Truong and Yellon, 1997
Night-time rise in pineal and serum melatonin levels; testicular weight	60 Hz, 100 μ T in complete darkness; 15 min per day for up to 21 days	No significant effects	Yellon and Truong, 1998
Night-time pineal and serum melatonin levels; testis cell numbers	50 Hz, 450 μΤ (peak) sinusoidal or 360 μΤ (peak) rectangular field, 56 days	Increased cell number and night-time serum melatonin after rectangular field exposure (p<0.05)	Niehaus et al, 1997
Night-time pineal melatonin levels, serum prolactin levels and testis and seminal vesicle weights in short day (regressed) animals	60 Hz, 100 or 500 $\mu T;$ CW and/or intermittent, starting 30 min or 2 h before onset of darkness; for up to 3 h for up to 42 days	Reduced pineal melatonin after acute (15 min) exposure (p<0.01); reduced gonad weight (p<0.05) but not melatonin after 42 day exposure	Wilson et al, 1999
Night-time serum melatonin levels and female puberty, detected by rise in serum progesterone	60 Hz, 6 kV m ⁻¹ and 4 μ T fields, for 10 months	No field-dependent effects, strong seasonal effects	Lee et al, 1993, 1995

complete data come from a series of studies by Yellon and colleagues. In the first study, Yellon (1994) found that acute exposure to a magnetic field two hours before the onset of darkness reduced and delayed the night-time rise in serum and pineal melatonin, but that this effect was diminished in a subsequent replicate study and absent in a third replicate study. Similarly, variable results on pineal and serum melatonin were reported by Yellon (1996) and Truong et al (1996). In addition, both studies found that magnetic field exposure had no effect on reproductive development, even in reproductively repressed hamsters on 'short day' (winter) schedules, which might be thought to be sensitive to reduced and delayed night-time melatonin levels of different magnetic field exposure parameters to those used in the previous experiments. Finally, Yellon and Truong (1998) reported that a brief exposure to a

magnetic field prior to the night-time rise in pineal and serum melatonin levels had no effect, even in complete darkness.

In contrast to the work of Yellon and his colleagues, Niehaus et al (1997) reported that the chronic exposure of Djungarian hamsters on 'long day' (summer) schedules to 'rectangular' power frequency magnetic fields resulted in increased testis cell numbers and night-time levels of serum melatonin, whereas exposure to sinusoidal power frequency magnetic fields had little effect. The authors concluded that the *in vivo* effects of magnetic fields may be dependent on their waveform, and that the rapidly changing waveform of the rectangular fields was a more effective biological stimulus. However, the results are not easy to interpret; increased duration of melatonin levels in the Djungarian hamster are usually accompanied by decreased testicular activity.

Wilson et al (1999) investigated the effects of exposure to magnetic fields on pineal melatonin levels, serum prolactin levels and testicular and seminal vesicle weights in Djungarian hamsters moved to a 'short day' light regimen in order to induce sexual regression. Night-time pineal melatonin levels were reduced following acute exposure but this effect diminished with prolonged exposure. In contrast, induced sexual regression, as indicated by the testicular and seminal vesicle weights, seemed to be enhanced rather than diminished by prolonged magnetic field exposure, suggesting a possible stress response.

Another set of studies investigating the effects of power frequency magnetic field on seasonal breeders concerned Suffolk sheep: these animals have a long gestational period and become reproductively active in the autumn, as daylength shortens. In two replicate studies (Lee et al, 1993, 1995), Suffolk ewe lambs were exposed outdoors to the magnetic fields generated by overhead transmission lines for about ten months. No field-dependent effects were reported on serum melatonin levels or on the onset of puberty.

4.2.3 Cattle

The effects of 60 Hz fields on melatonin levels in pregnant dairy cows have received some attention. No field-dependent effects on nocturnal melatonin levels were reported by Burchard et al (1998) following continuous exposure to a combined electric and magnetic field (10 kV m⁻¹ and 30 μ T) for several weeks. Melatonin levels were analysed from blood samples obtained every 30 minutes for 14 hours starting at 17.00 hours on day 25 of exposure. In a subsequent study, Burchard et al (2004) reported that continuous exposure to an electric field for four weeks was not associated with any changes in circulating levels of progesterone, melatonin, prolactin and insulin-like growth factor. Inconsistent changes in melatonin levels were noted, however. Blood samples were collected twice a week from catheters inserted into the jugular vein.

4.2.4 Non-human primates

Only two studies have investigated the effects of exposure to magnetic fields on melatonin rhythms in non-human primates. Rogers et al (1995a) reported that the chronic exposure of male baboons to power frequency EMFs had no effect on night-time serum melatonin levels. However, a preliminary study

(Rogers et al, 1995b), based on data from only two baboons, reported that a three week exposure to an irregular, intermittent sequence of a combination of electric and magnetic fields in which switching transients were generated, resulted in a marked suppression of the night-time rise in melatonin.

4.2.5 Summary

The possibility that melatonin rhythms are affected by exposure to power frequency EMFs has been investigated in a variety of mammals. Some, but not all, studies with rats report that exposure to power frequency EMFs results in a suppression of pineal and serum melatonin levels. The results of the early studies using electric fields could not be replicated and may have suffered from technical difficulties. The evidence from a series of studies using circularly polarised magnetic fields suggests that exposure suppresses night-time melatonin levels, but this result was sometimes weakened by inappropriate comparisons between exposed animals and historical controls. The data from other experiments were equivocal but mostly negative.

The evidence for an effect of exposure to power frequency EMFs on melatonin levels and melatonindependent reproductive status in seasonally breeding animals is mostly negative. A series of studies by one group reported reduced and delayed night-time peaks in pineal and serum melatonin in Djungarian hamsters. These effects could not be successfully replicated. No effects were seen on changes in reproductive status. Another study reported that 'rectangular' power frequency magnetic fields increased night-time serum melatonin levels. Testis cell numbers were increased, which is contrary to the expected inhibitory effect of melatonin in this species. A third group reported that sexual regression in male hamsters, induced by a short daylength, was enhanced rather than inhibited by magnetic field exposure. Finally, another group found no effect on serum melatonin and the onset of puberty in sheep.

Too few data exist to make any firm conclusions regarding the effects of magnetic fields on melatonin levels in cattle or non-human primates, although a preliminary study with baboons reported melatonin suppression in response to an irregular and intermittent exposure.

4.3 Human experimental studies

It is well established that exposure to light at night acutely suppresses melatonin in humans with conscious light perception (Lewy et al, 1980; Bojkowski et al, 1987; Zeitzer et al, 2000; Thapan et al, 2001), and subsequently causes shifts in the phase of circadian rhythms (Arendt and Broadway, 1986; Czeisler et al, 1986; Boivin and Czeisler, 1998; Warman et al, 2003b). These effects are comprehensively reviewed in Chapter 3.

Stevens (1987) proposed that magnetic fields might act in a similar way to light at night, suppressing the production of melatonin from the pineal gland, and that this reduction in night-time melatonin could result in an increased incidence of breast cancer. As a direct result of the melatonin hypothesis, the majority of studies that have been conducted to investigate the potential effects of magnetic fields in relation to light in humans have used melatonin suppression as a marker. Wever (1979) provided the first indication that exposure to low frequency EMFs may also adjust the human biological clock. He used a temporal isolation unit, shielded from external EMFs, to which controlled fields could be applied (or not)

during lengthy experiments (weeks). He reported that a 10 Hz square wave electric field at 2.5 V m⁻¹ could possibly act as a zeitgeber with respect to free-running rhythms. The field was either continuous or switched on for half of each 'circadian day' and periods of exposure were compared to periods with no exposure. Lighting was either continuous illumination (intensity not given but probably around 100–200 lux), or self-selected illumination. The lighting (and other electrical equipment operating within the unit) may have produced ambient 50 Hz electric fields of about 10–100 V m⁻¹. The experimental conditions varied to the extent that grouping of subjects was difficult to justify. Nevertheless there was consistent evidence that the field exposure (continuous or intermittent) could shorten free-running period, reduce the variability between individuals in free-running period, and perhaps prevent 'internal desynchronisation' of the circadian system. On the strength of these data, Wever proposed that the circadian system provided a sensitive model with which to test the effects of low frequency fields on biology. This would manifest itself as a change in the timing of the melatonin rhythm if treatment were to occur at an appropriate phase. However, the potential neuroanatomical pathways by which magnetic fields could exert an action on melatonin (suppression and/or phase-shifting) in humans (and other animals) remain obscure.

4.3.1 Experimental studies

Studies in humans investigating the effects of exposure to magnetic fields on melatonin rhythms, both in and out of the laboratory, have generally failed to provide consistent support for a field-dependent effect. While the results of a few laboratory-based volunteer studies suggest acute exposure may have an effect on the production or timing of the nightly melatonin rise, the majority of laboratory studies have not found any significant effects (Table 4.6). A similar pattern of results has been reported from epidemiology studies that have measured melatonin levels following chronic exposure at environmental levels: these studies are discussed in detail in Section 4.4.

It is particularly noteworthy that the most recent controlled data (Griefahn et al, 2001, 2002; Warman et al, 2003a) showed no acute effect of exposure to magnetic fields on amplitude or timing of the melatonin rhythm. Griefahn et al (2001, 2002) used partial constant routine conditions (see Chapter 3), generally considered to be the 'gold standard' for rhythm amplitude and phase assessment. In the Warman et al (2003a) study, correct sham controls were developed, baseline measures of melatonin levels were taken during the night preceding treatment, posture and light were controlled, and magnetic field exposure was carefully characterised. No significant effects of acute exposure (two hours) at 200–300 μ T were found on the timing or absolute levels of plasma melatonin (Figure 4.1). In addition, the timing of the core temperature rhythm was not affected. Statistical analysis of variance in melatonin onset data indicated that the protocol possessed 87% power to detect a change in the timing of melatonin onset as small as 20 minutes. In spite of this, no significant alteration in individual melatonin onset times was detected, and there was no significant effect of circadian time of exposure on the observation of a response relative to the sham data.

Correct sham controls require bifilar coil windings. Magnetic fields and sham fields may thus be created by unidirectional and bidirectional flow of current respectively to eliminate the confounding effects of vibration, electric fields, noise and heat. Exposure conditions must be monitored. Although Wood et al (1998) used correct sham controls, the apparent lack of control of light exposure seriously compromised their conclusions.



FIGURE 4.1 Mean melatonin profiles (N=19, mean \pm SEM) of subjects from 18.00 to 10.00 hours with sham exposure (A) or magnetic field exposure (200–300 μ T) during the melatonin rise (B). Closed circles denote the mean profile on baseline nights, open circles denote the mean profile on the sham (A) and exposure (B) legs. From Warman et al, 2003a

Two groups (Graham et al, 1996; Karasek et al, 1998) initially reported some positive effects of exposure to magnetic fields on melatonin. Graham and colleagues did not replicate their initial finding of an effect in a low melatonin subgroup in a subsequent experiment (Graham et al, 1997). Karasek and colleagues used a different exposure condition for their second attempt (Karasek et al, 2000) in which no field-related effects were found. The impressive series of studies by Graham and colleagues consistently used circularly polarised 60 Hz magnetic fields and convincingly demonstrated a lack of detectable effect in men or women, young or more elderly, with either continuous or intermittent exposure. It is possible that initial positive findings were due to chance with relatively small numbers of subjects.

Few early studies were conducted in a manner similar enough to allow direct comparisons to be drawn between them. Moreover in using melatonin as a marker, some researchers have neglected to bear in mind that it is an output from the circadian clock and as such its normal regulation is complex, time varying and extremely sensitive to light. As detailed in Chapter 3, the phase of the circadian clock can be altered by relatively modest amounts of light exposure (Boivin and Czeisler, 1998; Warman et al, 2003b) or by changes in behaviour (Buxton et al, 2003), and the time relative to the phase of the clock that a stimulus is administered can influence whether that stimulus has an effect or not (Khalsa et al, 2003). Standardisation of circadian phase such that the timing of the melatonin profile of all subjects is known prior to and during experimentation is thus essential to avoid confounding. Common methodological problems in studies with magnetic fields include single sampling, lack of appropriate baseline study nights, and lack of control for, or monitoring of, circadian phase (eg sleep–wake timing) and light exposure of subjects.

Reference	Exposure type	Design	Variable measured	Effect	Field strength
Wilson et al, 1990	Electric blankets 8 wk	Subjects' own controls N=28, F	aMT6s in overnight urine	7 showed decline and rebound post exposure with CPW	0.66 μT (max) conventional vs CPW blankets
Schiffman et al, 1994	MRI 01.00-02.00 h pulse sequence for imaging darkness	Subjects' own controls N=8, M	Plasma melatonin and cortisol	No effect	1.5 T control – darkness or light
Graham et al, 1996	CPMF, 60 Hz 23.00-07.00 h 1 h on, 1 h off <10 lux	Subjects' own controls N=33, M	Plasma melatonin hourly	No effect overall, lower in low melatonin group	1, 20 μT sham – field off
		Subjects' own controls N=40, M, low melatonin	Plasma melatonin hourly	No effect	20 μT sham – field off
Selmaoui et al, 1996	LPMF, 50 Hz continuous and 1 h on, 1 h off <30 lux	16 control, 16 test N=32, M, 20–30 y	Serum melatonin hourly aMT6s 3 hourly, 08.00–23.00 h, 9 h overnight	No effect	10 μT sham – field off
Graham et al, 1997	CPMF, 60 Hz, 23.00-07.00 h continuous <10 lux	Subjects' own controls N=40, M, 18–55 y	Plasma melatonin hourly	No effect in low or high melatonin group	20 μT sham – field off
Karasek et al, 1998	40 Hz, 20 min, 5 d per wk, 3 wk, 10.00 h or 18.00 h	Subjects' own controls N=12, M, middle-aged low back pain	Serum melatonin profiles	Lowers serum melatonin rhythm amplitude	2.9 mT pre-exposure (1 day) control
Wood et al, 1998	CPMF, 50 Hz during melatonin rise time 1.5–4 h light?	N=44, M, 18-49 y	Plasma melatonin 20–30 min hourly overnight	22 subjects had delayed melatonin rise time 12% lower peak	20 μT sham – bidirectional flow of current
Akerstedt et al, 1999	LPMF, 50 Hz 23.00–07.00 h continuous	Subjects' own controls N=18, M, 24–49 y	Plasma melatonin hourly	No effect on melatonin PRL, GH, cortisol, testosterone	1 μΤ
Graham et al, 2000a	CPMF, 60 Hz, 23.00-07.00 h 4 nights, 1 h on, 1 h off, <1 lux	Subjects' own controls N=30, M, 18–35 y	Melatonin and aMT6s AM urine aMT6s vs creatinine	No effect, less consistent on 4th night	28.3 μT vs <0.2 μT
Crasson et al, 2001	LPMF, 50 Hz continuous 30 min, intermittent on/off 15 s, 30 min, 50 lux 13.30 and 16.30 h	Subjects' own controls N=21, M, 20–35 y	Plasma melatonin hourly 20.00–07.00 h, urine volume recorded 19.00, 23.00, 07.00 h	No effect	100 μT sham – field off

TABLE 4.6 Human experimental studies of magnetic field exposure

TABLE 4.6 Continued

Reference	Exposure type	Design	Variable measured	Effect	Field strength
Karasek et al, 2000	200 Hz, Quatronic MRS 2000 8 min at 08.00 and 13.00 h, 3 wk, 5 d per wk	Subjects' own controls N=7, M, 32–42 y	Serum melatonin profile	No effect	25-80 μT
Hong et al, 2001	50 Hz electric blankets, 3 wk pre-exposure 11 wk exposure 2 wk post-exposure	Subjects' own controls N=9, M, 23–37 y	Melatonin in 5 urine samples daily 2 d, mid- week each condition	No effect	0.7 μT head 8.3 μT waist 3.5 μT feet
Graham et al, 2001a	CPMF, 60 Hz 23.00–07.00 h intermittent 1 h on, 1 h off <10 lux	Subjects' own controls N=53, F, 19–35 y 1 month apart 18–24 y match cycle days	Plasma melatonin and oestradiol hourly	No effect, no relation of melatonin to oestradiol	28.3 μT sham – field off <0.2 μT
Graham et al, 2001b	CPMF, 60 Hz 23.00–07.00 h continuous, intermittent 1h on, 1 h off <1 lux	Subjects' own controls N=24, M, 19–34 y	Melatonin and aMT6s AM urine aMT6s vs creatinine	No effect	127.3 μT sham – field off <0.2 μT
Graham et al, 2001c	CPMF, 60 Hz 23.00-07.00 h <10 lux	Subjects' own controls N=22 M and 24 F, 40–60 y	Melatonin and aMT6s AM urine aMT6s vs creatinine	No effect on melatonin, immune system, haematology	28.3 μT sham – field off <0.2 μT
Griefahn et al, 2001	16.7 Hz 18.00-02.00 h continuous <30 lux	Subjects' own controls N=7, M, 16–22 y	Constant, routine, saliva melatonin hourly, core body temperature, heart rate	No effect	200 μT sham – field off
Griefahn et al, 2002	16.7 Hz 18.00–02.00 h intermittent 15 s, <30 lux partial constant routine	Subjects' own controls N=12, M, 18–25 y	Saliva melatonin hourly, core body temperature, heart rate	No effect on melatonin or core body temperature, heart rate delayed	200 μT sham – field off
Kurokawa et al, 2003	LPMF, 50 Hz 23.00–07.00 h	Subjects' own controls N=10, M	Plasma melatonin hourly	No effect on melatonin, GH, PRL or cortisol	20 μT sham – field off
Warman et al, 2003a	CPMF, 50 Hz 2 h during melatonin rise time <10 lux partial constant routine	Subjects' own controls N=19, M, 18–35 y	Plasma melatonin hourly/half hourly 17.00–10.00 h, core body temperature	No effect	N=8, 200 μT N=11, 300 μT sham – bidirectional flow of current

Abbreviations: CPMF, circularly polarised magnetic field; CPW, continuous polymer wire; F, female; GH, growth hormone; LPMF, linearly polarised magnetic field; M, male; MRI, magnetic resonance imaging; PRL, prolactin

Although light is the factor that most significantly affects the clock, there are several non-photic stimuli that are known to influence circadian phase or apparent circulating levels of melatonin (as discussed in Chapter 3). Among these are posture (Deacon and Arendt, 1994), activity (Buxton et al, 2003), temperature (Dewasmes et al, 1994) and possibly even meal timing and content (Krauchi et al, 2002). A lack of standardisation of exposure to non-photic factors such as these may also produce changes in the melatonin profile, which can be falsely attributed to field exposure. However, the most important factor is lighting. Because relatively modest amounts of light commonly encountered in the home and laboratory setting can affect the circadian system greatly (Boivin and Czeisler, 1998; Warman et al, 2003b), wherever feasible, constant low levels (<10 lux) of light should be maintained throughout studies to avoid inadvertent light-induced phase shifting. In their study, Wood et al (1998) did not control for postural and lighting effects, and while they conducted baseline nights, they were not carried out immediately prior to each treatment as would be appropriate. Furthermore, different subjects appear to have received exposures of substantially different lengths (1.5-4 hours) and there are no data on whether exposure duration was correlated with observation of an effect. Overall, many of the earlier experimental studies investigating the effects of exposure to magnetic fields on melatonin profile suffer from a paucity of correct sham controls, a lack of explicit control of non-field-related factors, and inadequate field characterisation. Such studies are inadequate to assess the effects of magnetic fields on the melatonin profile.

The various factors that influence melatonin production and metabolism are listed in Table 3.1. Briefly, the most important are light exposure, circadian phase, sleep timing, posture, recent shift work or travel across time zones, beta-adrenergic antagonist and anti-depressant drugs. Most good studies also control for caffeine consumption, any medication including minor analgesics (eg non-steroidal anti-inflammatory drugs), use standardised meal times and meal content, and use subjects without extreme diurnal preference.

Single sampling to assess melatonin levels has been proposed as a cost-effective means of monitoring melatonin suppression in field-based studies (Cook et al, 2000). However, single sampling is innately risky due to the fact that one sample does not provide any temporal resolution of what is a time-varying process. Thus, without accurate personal monitoring of sleep-wake timing and light exposure of subjects prior to a single sample, and control of this timing, it is almost impossible to determine whether a difference in melatonin concentration in a single sample on different occasions is due to suppression or to a phase shift, or whether it is due to field exposure or to inadvertent exposure to light or changing sleep schedules. If a phase shift has occurred, a single sample can indicate an apparent increase or decrease in melatonin production when the overall effect is negligible (see Figure 4.2).

Some laboratory-based studies involve exposure of subjects to a sham condition on a separate occasion from the field exposure, suggesting circadian phase of the subjects could change between subsequent legs of the study. Without strict adherence to routine sleep–wake cycles between the legs of the study and without accurate assessment of circadian phase of subjects on entry to each leg (ie a baseline night), any changes in the timing or level of melatonin measured during the study cannot be attributed to field exposure.

If magnetic fields were a true zeitgeber and were able to phase shift the melatonin rhythm, the direction and magnitude of any shift would be expected to depend on the circadian timing of administration. Limited support for this hypothesis can be found in the data of Wood et al (1998), which suggested an



A Single 09:00 h sample (plasma or saliva) gives high, normal or low values depending on phase

FIGURE 4.2 Differences in melatonin levels due to phase changes, not amplitude changes, using a single morning saliva or plasma sample (A) or a single overnight urine collection (B)

effect of magnetic field exposure on the timing of the melatonin rise (a 'phase shift') in a 'responsive subgroup' treated (20 μ T, 1.5–4 hours) prior to or during the nightly melatonin onset, but not after it. In addition, Warman et al (2003a) noted that the variability of the timing of melatonin production was greater after field exposure than after sham exposure.

Although recent careful studies show no acute effects of EMFs or magnetic field alone on the melatonin profile, there are hints from the work of Graham et al (2000a), that longer term controlled exposure may eventually show some effect. They reported that intra-individual urinary measurements of melatonin and 6-sulphatoxymelatonin (aMT6s) were highly consistent over four exposure nights in the control condition. However, repeated nightly EMF exposure was associated with significantly reduced

consistency, suggesting a possible cumulative effect of EMF exposure on the stability of individual melatonin measurements over time. Moreover Griefahn et al (2002) reported that the rhythm of heart rate was delayed after acute exposure. Since, in general, all endogenous circadian rhythms are coordinated through the central circadian clock in the SCN, it is quite surprising that no effect was seen on the melatonin rhythm in this case. It is possible that the heart rate rhythm is influenced differently from melatonin downstream of the SCN.

Importantly, perhaps, sleep structure is modified by magnetic field exposure in a number of controlled studies (Akerstedt et al, 1999; Graham et al, 2000b) without concomitant effects on melatonin. Abnormal sleep is associated with a number of health problems, including compromised immune, hormonal and metabolic functions (Irwin, 2002; Spiegel et al, 2003).

Although most attention has been focused on investigating effects on melatonin, there have been a limited number of studies looking at circulating levels of other endocrine systems including the pituitary, thyroid, adrenal cortex and reproductive organs (see Table 4.6). As is the case with the melatonin studies, no conclusive effect of magnetic field exposure has been shown. In contrast, many reports of the effects of light on the timing of the circadian component of hormone rhythms exist, and most recently acute exposure to bright full spectrum white light in the early morning has been reported rapidly to increase production of both luteinising hormone and cortisol in young men (Yoon et al, 2003; Thorn et al, 2004).

4.3.2 Summary

Laboratory-based studies in humans investigating the effects of exposure to magnetic fields on melatonin rhythms have not provided consistent support for a field-dependent effect. While a few studies suggest acute exposure may have an effect on the production or timing of the nightly melatonin rise, the majority of studies have not found any significant modification of the rhythm. The lack of a consistent effect on the human melatonin profile may be due to inappropriate study designs, confounding, or the lack of an effect. The possibility remains that magnetic fields may affect the circadian clock in the long term and/or at a more subtle level.

4.4 Epidemiology

Observational epidemiology has considerable deficiencies compared with controlled trials as a method to investigate short-term effects of EMF exposure, and the epidemiological literature on EMFs and melatonin must therefore be considered cautiously, especially where data from volunteer trials are available. Three studies have been published of melatonin levels in relation to residential EMF exposures in women (Davis et al, 2001; Levallois et al, 2001; Youngstedt et al, 2002), one of occupational exposures in women (Juutilainen et al, 2000), and three of occupational or total exposures in men (Pfluger and Minder, 1996; Burch et al, 1998, 1999; Touitou et al, 2003). The studies including men have the complexity of interpretation that it is not certain to what extent factors affecting male levels would have the same effect in women (although for known factors so far, both sexes manifest similar effects).

4.4.1 Residential studies of women

Levallois et al (2001) found no relation of morning creatinine-adjusted urinary 6-sulphatoxymelatonin (aMT6s) concentration to proximity of residence to power lines or to measured 36 hour magnetic or spot residential electric fields, after adjustment for potential confounders. There were, however, significantly stronger relations of aMT6s to age and obesity (out of five variables for which the authors investigated effect modification) in women who lived close to power lines than in those who lived more distantly.

Davis et al (2001) assessed night-time urinary aMT6s concentrations normalised to creatinine in relation to nine magnetic field exposure variables derived from bedroom measures plus personal 24 hour metering. For one of these variables, bedroom magnetic field level, there was a borderline significant (depending on the potential confounders for which adjustment was made) association with aMT6s, significant in the subgroup of women who had used certain medications and a significant association at the time of year with fewest hours of darkness.

Youngstedt et al (2002) examined 24 hour urinary aMT6s excretion in 242 adults (226 women, 16 men*) aged 50–81 years in relation to 60 Hz magnetic field exposures levels in their beds, monitored over one week (but with no measurements of exposures in the rest of their lives, at home or occupationally). Fifteen per cent of the subjects had affective disorders at the time of observation (depression can affect melatonin levels) and some had had breast cancer. Regression analyses took account of several illumination variables, gained from 24 hour measurements, as well as age and usage of certain medications. All but six of the subjects had mean exposures of <0.5 μ T. No significant associations were found between measures of magnetic field exposure (mean, maximum, intermittency and variability) and several measures of aMT6s excretion including 24 hour mean and amplitude, and timing.

4.4.2 Occupational studies of women

Juutilainen et al (2000) analysed total aMT6s excretion over Thursday night and over Sunday night, and calculated the ratio of aMT6s concentrations in the two samples, in garment workers (who were stated to have high EMF exposures) and in office workers employed elsewhere. The Thursday/Sunday ratios for each group were close to unity, suggesting that several days of work did not affect aMT6s levels. Average Thursday night aMT6s excretion was significantly lower in exposed workers than in office workers, after allowing for confounders, but there was no dose–response relation.

4.4.3 Studies of occupational or total exposures in men

A comparison between electric train drivers (average exposure 20 μ T) and less-exposed railway workers (average exposure 1 μ T) found significantly reduced evening aMT6s concentrations (probably creatinine standardised) on work days compared with leisure days for the heavily-exposed but not the less-exposed men (Pfluger and Minder, 1996). This was not seen for morning levels, nor was there any dose–response relation. The two groups differed in their occupational night exposure to light, and it is not clear that adequate adjustment for this was made in the analyses.

* Although in principle it seems undesirable to combine both sexes, there were so few men that the results presumably apply essentially to women.

A study of electrical utility workers found no association between post-work aMT6s concentrations and mean occupational magnetic field exposures, but a significant reduction, after adjustment for confounders, in creatinine-adjusted aMT6s concentrations in men in the highest quartile compared with the lowest, for a measure of stability of exposure on the 2nd and 3rd days of the working week (Burch et al, 1999). There was a significant interaction with the effect of occupational light exposure, such that the magnetic field effect was only present for men with low light exposure. A study of 24 hour magnetic field exposure measures in the same men in relation to overnight aMT6s concentrations and total aMT6s excretion (Burch et al, 1998), with adjustment for confounders, found no relation for magnetic field intensity, intermittence or cumulative exposure. There was, however, a relation of stability of fields with aMT6s concentration (significant for 4th *cf* 1st quartile for residential fields) but not with total excretion.

In another group of male electrical utility workers, Burch et al (2000) found trends of lower nocturnal creatinine-adjusted urinary aMT6s concentration and lower overnight total urinary aMT6s excretion with greater occupational 60 Hz magnetic field exposure, adjusted for workplace light exposure, but this was only present for one subgroup of workers – those working in substations or three-phase environments for more than two hours per day, and not for those working in these environments for shorter times than this, or working for short or long hours in one-phase environments. The restriction to three-phase and substation environments rather than one-phase was stated to be explicable if circular or elliptical polarisation affects melatonin more than linearly polarised fields.

A study of a further group of male electrical utility workers by the same authors (Burch et al, 2002), found reduced nocturnal creatinine-adjusted aMT6s concentration and overnight aMT6s total excretion in men with high compared with low or medium workplace 60 Hz exposure, adjusted for light exposure at work (but not a consistent gradient with degree of exposure), restricted to analyses within men with high occupational mobile phone use (ie not present in those with medium or no such phone use).

A French study (Touitou et al, 2003) compared plasma melatonin and creatinine-normalised urinary aMT6s between 15 men who worked at (and lived 'near') high voltage substations in Paris and 15 white collar workers, without occupational magnetic field exposures, from the same company. The former group had had chronic occupational exposure to 50 Hz magnetic fields for 1–20 years, with weekly geometric mean exposures for the individual men (measured continuously over 7 days and nights) ranging from 0.1 to 2.6 μ T. The white collar workers had individual exposures ranging from 0.004 to 0.092 μ T. Considerable care was taken to avoid confounding differences between the two groups: for instance, all were non-smokers of a restricted age range, who did not do night work, and who were asked not to use electric razors or hair dryers in the 24 hours before samples were taken, and lights were turned off from 22.00 to 08.00 hours on the night of sample collection.

Blood samples were taken hourly from 20.00 to 08.00 hours and overnight urine collected. The hourly profile of plasma melatonin was similar between exposed and unexposed men. The aMT6s concentration in the first morning urine was not significantly different between the two groups. There were also no significant differences in plasma melatonin between the most highly exposed workers and the unexposed group. It was stated that the exposed men had not been 'on call' in the 48 hours before the samples were taken, but it is unclear whether this meant they had had no occupational exposures during that period, or indeed how recently before the samples were taken they had been occupationally or substantially exposed, and thus to what lag period after exposure the results relate.

4.4.4 Summary

In aggregate, the epidemiological studies of the relation of melatonin levels to EMF exposure do not give convincing evidence that EMFs affect the secretion of melatonin in humans. Although each of the published studies, except those by Youngstedt et al (2002) and Touitou et al (2003), has found some significant findings, usually in a subset of the data, there has been no consistency in the subgroup for which significant results were found, and indeed, in general, the significant results have not been re-examined for the same subgroup in subsequent studies. The authors of each of the positive studies proffered (apparently *post hoc*) reasons why effects might be detectable only in the particular (but very different) subgroup(s) identified in their studies. The positive findings have been such a restriction of effect. Hypotheses about why effects were found only in individuals who, for instance, had low light exposure or had used certain medications or had high phone use, would only carry conviction if subsequent studies examined the same subgroup and found that risks were present and restricted to this subgroup. Otherwise there must be concern that the significant finding could simply have been due to chance.

A further difficulty of interpretation is that as well as subgrouping in the above sense, there have also been many different measures of EMFs and of melatonin secretion used in different studies, and often several within one study (for instance, stability of fields, average fields, measures at home and at work, measures of melatonin on different days of the week, or ratios between days). It is not obvious *a priori* why the particular measures for which significant results have been found should alone have given significant relations.

The studies have measured melatonin in two different body fluids (plasma and metabolites in urine) and the urinary measures have been of two different types (creatinine-adjusted concentrations, and total excretion over 24 hours or overnight).

In most studies, melatonin measures have been made for one point in time or period of the day, rather than for 24 hours. Such measures are potentially highly susceptible to confounding by factors that alter the phase of the circadian clock, notably light exposure and sleep–wake timing (eg waking at a different time at weekends than during the week). Adjustment for (relatively crude) measures of such confounders could still leave considerable residual confounding (see Table 3.1 for further detail on confounders).

Finally, the studies published to date have mainly used urinary measures of melatonin*, which might lead to low response rates and hence potential selection bias with respect to the individuals who took part. In general, the studies either have reported low response rates or have not reported the response rate.

For these reasons, unless and until the specific significant findings in the published studies can be backed up by biological evidence or by replication in independent epidemiological studies, they can be regarded as no more than hypothesis-generating.

* Touitou et al (2003) also used plasma measures, but in addition to urinary measures.

4.5 References

- Afzal SMJ and Liburdy RP (1998). Magnetic fields reduce the growth inhibitory effects of tamoxifen in a human brain tumour cell line. In *Electricity and Magnetism in Biology and Medicine* (F Bersani, ed). Bologna, Italy, Plenum Press, pp 473–6.
- AGNIR (2001). ELF electromagnetic fields and the risk of cancer. Report of an Advisory Group on Non-ionising Radiation. *Doc NRPB*, **12**(1) 1–179.
- Akerstedt T, Arnetz B, Ficca G, Paulsson LE and Kallner A (1999). A 50-Hz electromagnetic field impairs sleep. *J Sleep Res*, **8**, 77–81.
- Arendt J and Broadway J (1986). Phase response of human melatonin rhythms to bright light in Antarctica. *J Physiol*, **377**, 68P.
- Baldwin WS, Travlos GS, Risinger JI and Barrett JC (1998). Melatonin does not inhibit estradiol-stimulated proliferation in MCF-7 and BG-1 cells. *Carcinogenesis*, **19**, 1895–1900.
- Bakos J, Nagy N, Thuroczy G and Szabo LD (1995). Sinusoidal 50 Hz, 500 μT magnetic field has no acute effect on urinary 6-sulphatoxymelatonin in Wistar rats. *Bioelectromagnetics*, **16**, 377–80.
- Bakos J, Nagy N, Thuroczy G and Szabo LD (1997). Urinary 6-sulphatoxymelatonin excretion is increased in rats after 24 hours of exposure to vertical 50 Hz, 100 μT magnetic field. *Bioelectromagnetics*, **18**, 190–92.
- Bakos J, Nagy N, Thuroczy G and Szabo LD (1999). Urinary 6-sulphatoxymelatonin excretion of rats is not changed by 24 hours of exposure to a horizontal 50-Hz, 100-µT magnetic field. *Electro- Magnetobiol*, **18**, 23–31.
- Bakos J, Nagy N, Thuroczy G and Szabo LD (2002). One week of exposure to 50 Hz, vertical magnetic field does not reduce urinary 6-sulphatoxymelatonin excretion of male Wistar rats. *Bioelectromagnetics*, **23**, 245–8.
- Blackman CF, Benane SG and House DE (2001). The influence of 1.2 μT, 60 Hz magnetic fields on melatonin- and tamoxifen-induced inhibition of MCF-7 cell growth. *Bioelectromagnetics*, **22**, 122-8.
- Boivin DB and Czeisler CA (1998). Resetting of circadian melatonin and cortisol rhythms in humans by ordinary room light. *Neuroreport*, **9**, 779–82.
- Bojkowski CJ, Aldhous ME, English J, Franey C, Poulton AL, Skene DJ and Arendt J (1987). Suppression of nocturnal plasma melatonin and 6-sulphatoxymelatonin by bright and dim light in man. *Horm Metab Res*, **19**, 437–40.
- Brady JV and Reiter RJ (1992). Neurobehavioral effects. IN Health Effects of Low-frequency Electric and Magnetic Fields. Oak Ridge Associated Universities, ppVII-1-40.
- Brendel H, Nielhaus M and Lerchl AA (2000). Direct suppressive effects of weak magnetic fields (50 Hz and 16²/₃ Hz) on melatonin synthesis in the pineal gland of Djungarian hamsters *(Phodopus sungorus). J Pineal Res*, **29**, 228–33.
- Brocklehurst B (2002). Magnetic fields and radical reactions: recent developments and their role in nature. *Chem Soc Rev*, **31**, 301–11.
- Brocklehurst B and McLauchlan KA (1996). Free radical mechanism for the effects of environmental electromagnetic fields on biological systems. *Int J Radiat Biol*, **69**, 3–24.
- Burch JD, Reif JS, Yost MG, Keefe TJ and Pitrat CA (1998). Nocturnal excretion of a urinary melatonin metabolite among electric utility workers. *Scand J Work Environ Health*, **24**, 183–9.
- Burch JD, Reif JS, Yost MG, Keefe TJ and Pitrat CA (1999). Reduced excretion of a melatonin metabolite in workers exposed to 60 Hz magnetic fields. *Am J Epidemiol*, **150**, 27–36.
- Burch JD, Reif JS, Noonan CW, Ichinose T, Bachand AM, Koleber TL and Yost MG (2002). Melatonin metabolite excretion among cellular telephone users. *Int J Radiat Biol*, **78**, 1029–36.
- Burch JD, Reif JS, Noonan CW and Yost MG (2000). Melatonin metabolite levels in workers exposed to 60-Hz magnetic fields: work in substations and with 3-phase conductors. *J Occup Environ Med*, **42**, 136–42.
- Burchard JF, Nguyen DH and Block E (1998). Effects of electric and magnetic fields on nocturnal melatonin concentrations in dairy cows. J Dairy Sci, 81, 722–7.
- Burchard JF, Nguyen DH, Monardes HG and Petitclerc D (2004). Lack of effect of 10 kV m⁻¹ 60 Hz electric field exposure on pregnant dairy heifer hormones. *Bioelectromagnetics*, **25**, 308–12.

- Buxton OM, Lee CW, L'Hermite-Baleriaux M, Turek FW and Van Cauter E (2003). Exercise elicits phase shifts and acute alterations of melatonin that vary with circadian phase. *Am J Physiol Regul Integr Comp Physiol*, **284**, R714–24.
- Chacon L (2000). 50-Hz sinusoidal magnetic field effect on *in vitro* pineal N-acetyltransferase activity. *Electro-Magnetobiol*, **19**, 339–43.
- Chiang H, Li CM, Fu YD and Lu DJ (1999). The mechanism of suppression of gap junctional intercellular communication by 50-Hz magnetic fields. *Electro- Magnetobiol*, **18**, 243–7.
- Cook MR, Graham C, Kavet R, Stevens RG, Davis S and Kheifets L (2000). Morning urinary assessment of nocturnal melatonin secretion in older women. *J Pineal Res*, **28**, 41–7.
- Crasson M, Beckers V, Pequeux C, Claustrat B and Legros JJ (2001). Daytime 50 Hz magnetic field exposure and plasma melatonin and urinary 6-sulfatoxymelatonin concentration profiles in humans. *J Pineal Res*, **31**, 234–41.
- Czeisler CA, Allan JS, Strogatz SH, Ronda JM, Sanchez R, Rios CD, Freitag WO, Richardson GS and Kronauer RE (1986). Bright light resets the human circadian pacemaker independent of the timing of the sleep–wake cycle. *Science*, **233**, 667–71.
- Davis S, Mirick DK and Stevens RG (2001). Night shift work, light at night, and risk of breast cancer. J Natl Cancer Inst, 93, 1557–62.
- de Bruyn L, de Jager L and Kuyl JM (2001). The influence of long-term exposure of mice to randomly varied power frequency magnetic fields on their nocturnal melatonin secretion patterns. *Environ Res*, **85**, 115–21.
- Deacon S and Arendt J (1994). Posture influences melatonin concentrations in plasma and saliva in humans. *Neurosci Lett*, **167**, 191–4.
- Dewasmes G, Nicolas A, Rodriguez D, Salame P, Eschenlauer R, Joly D and Muzet A (1994). Human core temperature minimum can be modified by ambient thermal transients. *Neurosci Lett*, **173**, 151–4.
- Fedrowitz M, Westermann J and Löscher W (2002). Magnetic field exposure increases cell proliferation but does not affect melatonin levels in the mammary gland of female Sprague-Dawley rats. *Cancer Res*, **62**, 1356–63.
- Graham C, Cook MR, Riffle DW, Gerkovich MM and Cohen HD (1996). Nocturnal melatonin levels in human volunteers exposed to intermittent 60 Hz magnetic fields. *Bioelectromagnetics*, **17**, 263–73.
- Graham C, Cook MR and Riffle DW (1997). Human melatonin during continuous magnetic field exposure. *Bioelectromagnetics*, **18**, 166–71.
- Graham C, Cook MR, Sastre A, Riffle DW and Gerkovich MM (2000a). Multi-night exposure to 60 Hz magnetic fields: effects on melatonin and its enzymatic metabolite. *J Pineal Res*, **28**, 1–8.
- Graham C, Sastre A, Cook MR and Gerkovich MM (2000b). Nocturnal magnetic field exposure: gender specific effects on heart rate variability and sleep. *Clin Neurophysiol*, **111**, 1936–41.
- Graham, C, Cook MR, Gerkovich MM and Sastre A (2001a). Examination of the melatonin hypothesis in women exposed at night to EMF or bright light. *Environ Health Perspect*, **109**, 501–7.
- Graham C, Cook MR, Gerkovich MM and Sastre A (2001b). Melatonin and 6-OHMS in high-intensity magnetic fields. *J Pineal Res*, **31**, 85–8.
- Graham C, Sastre A, Cook MR and Gerkovich MM (2001c). All-night exposure to EMF does not alter urinary melatonin, 6-OHMS or immune measures in older men and women. *J Pineal Res*, **31**, 109–13.
- Griefahn B, Kunemund C, Blaszkewicz M, Golka K, Mehnert P and Degen G (2001). Experiments on the effects of a continuous 16.7 Hz magnetic field on melatonin secretion, core body temperature, and heart rates in humans. *Bioelectromagnetics*, **22**, 581–8.
- Griefahn B, Kunemund C, Blaszkewicz M, Golka K and Degen G (2002). Experiments on effects of an intermittent 16.7-Hz magnetic field on salivary melatonin concentrations, rectal temperature, and heart rate in humans. *Int Arch Occup Environ Health*, **75**, 171–8.
- Griffin GD, Williams MW and Gailey PC (2000). Cellular communication in clone 9 cells exposed to magnetic fields. *Radiat Res*, **153**, 690–98.
- Grota LJ, Reiter RJ, Keng P and Michaelson S (1994). Electric field exposure alters serum melatonin but not pineal melatonin synthesis in male rats. *Bioelectromagnetics*, **15**, 427–37.

Harland JD and Liburdy RP (1997). Environmental magnetic fields inhibit the antiproliferative action of tamoxifen and melatonin in a human breast cancer cell line. *Bioelectromagnetics*, **18**(8), 555–62.

- Harland JD, Levine GA and Liburdy RP (1998). Differential inhibition of tamoxifen's oncostatic functions in a human breast cancer cell line by a 12 mG (1.2 μT) magnetic field. In *Electricity and Magnetism in Biology and Medicine* (F Bersani, ed). Bologna, Italy, Plenum Press, pp 465–8.
- Heikkinen P, Kumlin T, Laitinen JT, Komulainene H and Juutilainen J (1999). Chronic exposure to 50 Hz magnetic fields or 900-MHz electromagnetic fields does not alter nocturnal 6-hydroxymelatonin sulfate secretion in CBA/S mice. *Electro- Magnetobiol*, **18**, 33–42.
- Hong SC, Kurokawa Y, Kabuto M and Ohtsuka R (2001). Chronic exposure to ELF magnetic fields during night sleep with electric sheet: effects on diurnal melatonin rhythms in men. *Bioelectromagnetics*, **22**, 138–43.
- Hu GL, Fu YD, Zeng QL, Xu ZP and Chiang H (2002). Study on gap junctional intercellular communication inhibition by ELF magnetic fields using FRAP method. *Electromagn Biol Med*, **21**, 155–60.
- Irwin M (2002). Effects of sleep and sleep loss on immunity and cytokines. Brain Behav Immunol, 16, 503-12.
- Ishido M, Nitta H and Kabuto M (2001). Magnetic fields (MF) of 50 Hz at 1.2 μT as well as 100 μT cause uncoupling of inhibitory pathways of adenylyl cyclase mediated by melatonin 1a receptor in MF-sensitive MCF-7 cells. *Carcinogenesis (Oxford)*, **22**, 1043–8.
- John TM, Liu GY and Brown GM (1998). 60 Hz magnetic field exposure and urinary 6-sulphatoxymelatonin levels in the rat. *Bioelectromagnetics*, **19**, 172–80.
- Juutilainen J, Stevens RG, Anderson LE, Hansen NH, Kilpelainen M, Kumlin T, Laitinen JT, Sobel E and Wilson BW (2000). Nocturnal 6-hydroxymelatonin sulfate excretion in female workers exposed to magnetic fields. *J Pineal Res*, 28, 97–104.
- Karasek M, Woldanska-Okonska M, Czernicki J, Zylinska K and Swietoslawski J (1998). Chronic exposure to 2.9 mT, 40 Hz magnetic field reduces melatonin concentrations in humans. *J Pineal Res*, **25**, 240–44.
- Karasek M, Czernicki J, Woldanska-Okonska M, Zylinska K and Swietoslawski J (2000). Chronic exposure to 25–80 μT, 200-Hz magnetic field does not influence serum melatonin concentrations in patients with low back pain. *J Pineal Res*, **29**, 81–5.
- Kato M and Shigemitsu T (1997). Effects of 50 Hz magnetic fields on pineal function in the rat. In *The Melatonin Hypothesis, Breast Cancer and Use of Electric Power* (RG Stevens et al, eds). Columbus OH, Battelle Press, pp 337–76.
- Kato M, Honma K, Shigemitsu T and Shiga Y (1993). Effects of exposure to a circularly polarized 50-Hz magnetic field on plasma and pineal melatonin levels in rats. *Bioelectromagnetics*, **14**, 97–106.
- Kato M, Honma K, Shigemitsu T and Shiga Y (1994a). Recovery of nocturnal melatonin concentration takes place within one week following cessation of 50 Hz circularly polarized magnetic field exposure for six weeks. *Bioelectromagnetics*, **15**, 489–92.
- Kato M, Honma K, Shigemitsu T and Shiga Y (1994b). Circularly polarized 50-Hz magnetic field exposure reduces pineal gland and blood melatonin concentrations of Long-Evans rats. *Neurosci Lett*, **166**, 59–62.
- Kato M, Honma K, Shigemitsu T and Shiga Y (1994c). Horizontal or vertical 50-Hz, 1-µT magnetic fields have no effect on pineal gland or plasma melatonin concentration of albino rats. *Neurosci Lett*, **168**, 205–8.
- Kato M, Honma K, Shigemitsu T and Shiga Y (1994d). Circularly polarized, sinusoidal, 50 Hz magnetic field exposure does not influence plasma testosterone levels of rats. *Bioelectromagnetics*, **15**, 513–18.
- Khalsa SB, Jewett ME, Cajochen C and Czeisler CA (2003). A phase response curve to single bright light pulses in human subjects. *J Physiol*, **549**, 945–52.
- Krauchi K, Cajochen C, Werth E and Wirz-Justice A (2002). Alteration of internal circadian phase relationships after morning versus evening carbohydrate-rich meals in humans. *J Biol Rhythms*, **17**, 364–76.
- Kurokawa Y, Nitta H, Imai H and Kabuto M (2003). Acute exposure to 50 Hz magnetic fields with harmonics and transient components: lack of effects on nighttime hormonal secretion in men. *Bioelectromagnetics*, **24**, 12–20.
- Lee JM Jr, Stormshak F, Thompson JM, Thinesen P, Painter LJ, Olenchek EG, Hess DL, Forbes R and Foster DL (1993). Melatonin secretion and puberty in female lambs exposed to environmental electric and magnetic fields. *Biol Reprod*, **49**, 857–64.

- Lee JM Jr, Stormshak F, Thompson JM, Hess DL and Foster DL (1995). Melatonin and puberty in female lambs exposed to EMF: a replicate study. *Bioelectromagnetics*, **16**, 119–23.
- Leman ES, Sisken BF, Zimmer S and Anderson KW (2001). Studies of the interactions between melatonin and 2 Hz, 0.3 mT PEMF on the proliferation and invasion of human breast cancer cells. *Bioelectromagnetics*, **22**, 178–84.
- Lerchl A, Reiter RJ, Howes KA, Nonaka KO and Stokkan KA (1991). Evidence that extremely low frequency Ca(2+)-cyclotron resonance depresses pineal melatonin synthesis *in vitro*. *Neurosci Lett*, **124**, 213–15.
- Levallois P, Dumont M, Touitou Y, Gingras S, Masse B, Gauvin D, Kroger E, Bourdages M and Douville P (2001). Effects of electric and magnetic fields from high-power lines on female urinary excretion of 6-sulfatoxymelatonin. *Am J Epidemiol*, **154**, 601–9.
- Lewy AJ, Wehr TA, Goodwin FK, Newsome DA and Markey SP (1980). Light suppresses melatonin secretion in humans. *Science*, **210**, 1267–9.
- Lewy H, Massot O and Touitou Y (2003). Magnetic field (50 Hz) increases *N*-acetyltransferase, hydroxy-indole-*O*methyltransferase activity and melatonin release through an indirect pathway. *Int J Radiat Biol*, **79**, 431–5.
- Liburdy RP, Sloma TR, Sokolic R and Yaswen P (1993). ELF magnetic fields breast cancer and melatonin 60 Hz fields block melatonin's oncostatic action on ER-positive breast cancer cell proliferation. *J Pineal Res*, **14**, 89–97.
- Lohmann CH, Schwartz Z, Liu Y, Li Z, Albert SA, Simon BJ, Sylvia VL, Dean DD, Bonewald LF, Donahue HJ and Boyan BD (2002). Pulsed electromagnetic fields affect phenotype and connexin 43 protein expression in MLO-Y4 osteocyte-like cells and ROS 17/2.8 osteoblast-like cells. *J Bone Mineral Res*, **17**, S352.
- Löscher W, Mevissen M and Lerchl A (1998). Exposure of female rats to a 100-µT 50 Hz magnetic field does not induce consistent changes in nocturnal levels of melatonin. *Radiat Res*, **150**, 557–67.
- Luben RA and Morgan AP (1998). Independent replication of 60 Hz, 1.2 μT EMF effects on melatonin and tamoxifen responses of MCF-7 breast cancer cells *in vitro*. In Abstracts, 20th Annual Meeting of the Bioelectromagnetics Society, June 1998, Florida, A-3-4.
- McCormick DL, Cahill MA, Ryan BM, Findlay JC and Boorman GA (1995). Pineal function in B6C3F1 mice exposed to 60 Hz magnetic fields: time course studies. In Abstracts, 17th Annual Meeting of the Bioelectromagnetics Society, June 1995, Boston, Massachusetts, p 81.
- Marino AA, Kolomytkin OV and Frilot C (2003). Extracellular currents alter gap junction intercellular communication in synovial fibroblasts. *Bioelectromagnetics*, **24**, 199–205.
- Mevissen M, Lerchl A and Loscher W (1996a). Study on pineal function and DMBA-induced breast cancer formation in rats during exposure to a 100-mG, 50 Hz magnetic field. *J Toxicol Environ Health*, **48**, 169–85.
- Mevissen M, Lerchl A, Szamel M and Loscher W (1996b). Exposure of DMBA-treated female rats in a 50-Hz, 50 μT magnetic field: effects on mammary tumor growth, melatonin levels, and T lymphocyte activation. *Carcinogenesis*, **17**, 903–10.
- Molis TM, Spriggs LL and Hill SM (1994). Modulation of estrogen receptor mRNA expression by melatonin in MCF-7 human breast cancer cells. *Mol Endocrinol*, **8**, 1681–90.
- Morris JE, Chrisler WB, Miller DL, Sasser LB and Anderson LE (1998). *In vitro* exposure of MCF-7 human mammary cells to 60 Hz magnetic fields. In Abstracts, 20th Annual Meeting of the Bioelectromagnetics Society, June 1998, Florida, P-125A.
- Niehaus M, Bruggemeyer H, Behre HM and Lerchl A (1997). Growth retardation, testicular stimulation, and increased melatonin synthesis by weak magnetic fields (50 Hz) in Djungarian hamsters, *Phodopus sungorus. Biochem Biophys Res Commun*, **234**, 707–11.
- NIEHS (National Institute of Environmental Health Sciences) (1999). Health effects from exposure to power-line frequency electric and magnetic fields. Bethesda MD, National Institutes of Health, NIH Publication 99-4493.
- Pfluger DH and Minder CE (1996). Effects of exposure to 16.7 Hz magnetic fields on urinary 6-hydroxymelatonin sulfate excretion of Swiss railway workers. *J Pineal Res*, **21**, 91–100.
- Picazo ML, Catala MD, Romo MA and Bardasano JL (1998). Inhibition of melatonin in the plasma of thirdgeneration male mice under the action of ELF magnetic fields. *Electro- Magnetobiol*, **17**, 75–85.
- Ram PT, Yuan L, Dai J, Kiefer T, Klotz DM, Spriggs LL and Hill SM (2000). Differential responsiveness of MCF-7 human breast cancer cell line stocks to the pineal hormone, melatonin. *J Pineal Res*, **28**, 210–18.

- Reiter RJ (1993). Static and extremely low frequency electromagnetic field exposure: reported effects on the circadian production of melatonin. *J Cell Biochem*, **51**, 394–403.
- Reiter RJ, Anderson LE, Buschbom RL and Wilson BW (1988). Reduction of the nocturnal rise in pineal melatonin levels in rats exposed to 60-Hz electric fields *in utero* and for 23 days after birth. *Life Sci*, **42**, 2203–6.
- Rogers WR, Reiter RJ, Barlow-Walden L, Smith HD and Orr JL (1995a). Regularly scheduled, day-time, slow-onset 60 Hz electric and magnetic field exposure does not depress serum melatonin concentration in nonhuman primates. *Bioelectromagnetics*, **Suppl 3**, 111–18.
- Rogers WR, Reiter RJ, Smith HD and Barlow-Walden L (1995b). Rapid-onset/offset, variably scheduled 60 Hz electric and magnetic field exposure reduces nocturnal serum melatonin concentration in nonhuman primates. *Bioelectromagnetics*, **Suppl 3**, 119–22.
- Rosen LA, Barber I and Lyle DB (1998). A 0.5 G, 60 Hz magnetic field suppresses melatonin production in pinealocytes. *Bioelectromagnetics*, **19**, 123–7.
- Sasser LB, Morris JE, Buschbom RL, Miller DL and Anderson LE (1991). Effect of 60 Hz electric fields on pineal melatonin during various times of the dark period. IN Project Resumes, DOE Annual Review of Research on Biological Effects of 50 and 60 Hz Electric and Magnetic Fields. November 1991, Milwaukee, Wisconsin, pA-24.
- Schiffman JS, Lasch HM, Rollag MD, Flanders AE, Brainard GC and Burk DL Jr (1994). Effect of MR imaging on the normal human pineal body: measurement of plasma melatonin levels. *J Magn Reson Imag*, **4**, 7–11.
- Schimmelpfeng J, Stein JC and Dertinger H (1996). Action of 50 Hz magnetic fields on cyclic AMP and intracellular communication in monolayers and spheroids of mammalian cells. *Bioelectromagnetics*, **16**, 381–6.
- Selmaoui B and Touitou Y (1995). Sinusoidal 50-Hz magnetic fields depress rat pineal NAT activity and serum melatonin. Role of duration and intensity of exposure. *Life Sci*, **57**, 1351–8.
- Selmaoui B and Touitou Y (1999). Age-related differences in serum melatonin and pineal NAT activity and in the response of rat pineal to a 50-Hz magnetic field. *Life Sci*, **64**, 2291–7.
- Selmaoui B, Lambrozo J and Touitou Y (1996). Magnetic fields and pineal function in humans: evaluation of nocturnal acute exposure to extremely low frequency magnetic fields on serum melatonin and urinary 6-sulfatoxymelatonin circadian rhythms. *Life Sci*, **58**, 1539–49.
- Shafie SM and Liotta LA (1980). Formation of metastasis by human breast carcinoma cells (MCF-7) in nude mice. *Cancer Lett*, **11**, 81–7.
- Spiegel K, Leproult R and Van Cauter E (2003). Impact of sleep debt on physiological rhythms. *Rev Neurol (Paris)*, **159**, 6511–20.
- Stevens RG (1987). Electric power use and breast cancer. Am J Epidemiol, 125, 556-61.
- Thapan K, Arendt J and Skene DJ (2001). An action spectrum for melatonin suppression: evidence for a novel non-rod, non-cone photoreceptor system in humans. *J Physiol*, **535**, 261–7.
- Thorn L, Hucklebridge F, Esgate A, Evans P and Clow A (2004). The effect of dawn simulation on the cortisol response to awakening in healthy participants. *Psychoneuroendocrinology*, **29**, 925–30.
- Touitou Y, Selmaoui B, Lambrozo J and Auzeby A (2002). Evaluation of the effect of magnetic fields on the secretion of melatonin in humans and rats. Circadian study. *Bull Acad Natl Med*, **186**, 1625–39.
- Touitou Y, Lambrozo J, Camus F and Charbuy H (2003). Magnetic fields and the melatonin hypothesis: a study of workers chronically exposed to 50-Hz magnetic fields. *Am J Physiol Regul Integr Comp Physiol*, **284**, R1529–35.
- Tripp HM, Warman GR and Arendt J (2003). Circularly polarised MF (500 μT 50 Hz) does not acutely suppress melatonin secretion from cultured Wistar rat pineal glands. *Bioelectromagnetics*, **24**, 118–24.
- Truong H and Yellon SM (1997). Effect of various acute 60 Hz magnetic field exposures on the nocturnal melatonin rise in the adult Djungarian hamster. *J Pineal Res*, **22**, 177–83.
- Truong H, Smith JC and Yellon SM (1996). Photoperiod control of the melatonin rhythm and reproductive maturation in the juvenile Djungarian hamster: 60 Hz magnetic field exposure effects. *Biol Reprod*, **55**, 455–60.

- Warman GR, Tripp H, Warman VL and Arendt J (2003a). Acute exposure to circularly polarized 50-Hz magnetic fields of 200–300 μT does not affect the pattern of melatonin secretion in young men. *J Clin Endocrinol Metab*, **88**, 5668–73.
- Warman VL, Dijk DJ, Warman GR, Arendt J and Skene DJ (2003b). Phase advancing human circadian rhythms with short wavelength light. *Neurosci Lett*, **342**, 37–40.
- Wever RA (1979). The Circadian System of Man. New York, Springer Verlag Inc.
- Wilson BW, Anderson LE, Hilton DI and Phillips RD (1981). Chronic exposure to 60-Hz electric fields: effects on pineal function in the rat. *Bioelectromagnetics*, **2**, 371–80.
- Wilson BW, Anderson LE, Hilton DI and Phillips RD (1983). Erratum. Chronic exposure to 60-Hz electric fields: effects on pineal function in the rat [1981, **2**, 371–80]. *Bioelectromagnetics*, **4**, 293.
- Wilson BW, Chess EK and Anderson LE (1986). 60-Hz electric-field effects on pineal melatonin rhythms: time course for onset and recovery. *Bioelectromagnetics*, **7**, 239–42.
- Wilson BW, Wright CW, Morris JE, Buschbom RL, Brown DP, Miller DL, Sommers-Flannigan R and Anderson LE (1990). Evidence for an effect of ELF electromagnetic fields on human pineal gland function. *J Pineal Res*, **9**, 259–69.
- Wilson BW, Matt KS, Morris JE, Sasser LB, Miller DL and Anderson LE (1999). Effects of 60 Hz magnetic field exposure on the pineal and hypothalamic-pituitary-gonadal axis in the Siberian hamster (*Phodopus sungorus*). *Bioelectromagnetics*, **20**, 224–32.
- Wood AW, Armstrong SM, Sait ML, Devine L and Martin MJ (1998). Changes in human plasma melatonin profiles in response to 50 Hz magnetic field exposure. *J Pineal Res*, **25**, 116–27.
- Ubeda A, Trillo MA, House DE and Blackman CF (1995). A 50 Hz magnetic field blocks melatonin-induced enhancement of junctional transfer in normal C3H/10T1/2 cells. *Carcinogenesis*, **16**, 2945–9.
- Yamaguchi DT, Huang J, Ma D and Wang PKC (2002). Inhibition of gap junction intercellular communication by extremely low-frequency electromagnetic fields in osteoblast-like models is dependent on cell differentiation. J Cell Physiol, 190, 180–88.
- Yellon SM (1994). Acute 60 Hz magnetic field exposure effects on the melatonin rhythm in the pineal gland and circulation of the adult Djungarian hamster. *J Pineal Res*, **16**, 136–44.
- Yellon SM (1996). 60-Hz magnetic field exposure effects on the melatonin rhythm and photoperiod control of reproduction. *Am J Physiol*, **270**(5 Part 1), E816–21.
- Yellon SM and Truong HN (1998). Melatonin rhythm onset in the adult Siberian hamster: influence of photoperiod but not 60-Hz magnetic field exposure on melatonin content in the pineal gland and in circulation. *J Biol Rhythms*, **13**, 52–9.
- Yoon IY, Kripke DF, Elliott JA and Youngstedt SD (2003). Luteinizing hormone following light exposure in healthy young men. *Neurosci Lett*, **341**, 25–8.
- Youngstedt SD, Kripke DF, Elliott JA and Assmus JD (2002). No association of 6-sulfatoxymelatonin with in-bed 60-Hz magnetic field exposure or illumination level among older adults. *Environ Res*, **A89**, 201–9.
- Zeitzer JM, Dijk DJ, Kronauer R, Brown E and Czeisler C (2000). Sensitivity of the human circadian pacemaker to nocturnal light: melatonin phase resetting and suppression. *J Physiol*, **526**(Part 3), 695–702.

5 Melatonin and Breast Cancer

5.1 In vitro studies

In vitro studies of melatonin and breast cancer have focused on the possible oncostatic properties of melatonin. More than 60 studies have been published, with the vast majority of the work being undertaken on the breast cancer cell line MCF-7. Surprisingly few studies have used other breast cancer cell lines.

The MCF-7 is an oestrogen receptor positive cell line derived from a pleural effusion from a breast carcinoma (Soule et al, 1973) and, as described in Chapter 4, it has been used extensively as a cellular model of breast cancer.

Several authors have commented on the heterogeneity of the MCF-7 cells and that cellular responses may vary depending on the sub-clone and the conditions under which the cells are grown (Osborne et al, 1987; Liburdy et al, 1993; Ram et al, 2000). One reported response, namely that physiological concentrations of melatonin inhibit the growth of MCF-7 cells, may be of particular relevance here. There is not complete agreement that the growth of MCF-7 cells is affected by treatment with melatonin – some researchers (Shellard et al, 1989; Bartsch et al, 1992; Panzer et al, 1998; Papazisis et al, 1998) find no inhibitory effect of melatonin at physiological concentrations. However, there are a sufficient number of reports from a variety of independent laboratories showing inhibitory effects of melatonin on MCF-7 cells that these effects must be accepted, in at least some of the sub-clones (Blask and Hill, 1986; Cos and Sanchez-Barcelo, 1995; Karasek and Pawlikowski, 1999; Rato et al, 1999; Scott et al, 2001; Czeczuga-Semeniuk et al, 2002). Some researchers express the view that it is only a sub-clone of MCF-7 that is responsive in this way and that this inhibitory effect is not a general phenomenon of all breast cancer cells (Panzer et al, 1998). Since the majority of these studies have used MCF-7 cells, it is difficult to judge how common this inhibitory effect might be in other mammary cancer cells, but at least one other cell line has been shown to be responsive. Bizzarri et al (2003), working on a rat mammary cancer cell line (RM4), found that melatonin inhibited proliferation if 17-beta-oestradiol was present, although 1,25-dihydroxyvitamin D₃ was more potent.

The mechanisms by which melatonin inhibits the growth of cells have been investigated. There are suggestions that it acts on the binding of the oestrogen receptor to DNA and therefore blocks the stimulatory effect of oestrogen (see Chapter 3 for further details).

At higher than physiological levels of melatonin, other mechanisms have been proposed. Scott et al (2001) suggested a receptor-modulated pathway of cytotoxicity and an uncoupling of oxidative phosphorylation in cells treated with 100 nM melatonin. At high concentrations ($>10^{-5}$ M) melatonin is an effective scavenger of free-radicals (Baldwin and Barrett, 1998) and will protect the cell from damage and exert an oncostatic effect. However, this protective effect is achieved at pharmacological levels of melatonin; at physiological levels its role as a free-radical scavenger may not be significant.

5.1.1 Summary

The bulk of the *in vitro* studies of breast cancer have been undertaken on one particular cell line: the MCF-7 (Table 5.1). Some sub-clones have been demonstrated to be sensitive to physiological levels of melatonin in that growth can be inhibited under certain conditions. The cell line has proved useful in providing some insights into the potential mechanism of action. However, little work has been undertaken with other breast cancer cell lines. It is therefore of concern that the oncostatic action of melatonin may be confined to a sub-clone of MCF-7 cells and not applicable to breast cancer cells or cancer cells in general.

TABLE 5.1 Summary	/ of <i>in</i>	vitro studies	of the eff	ects of me	latonin on	mammalian (cells
THE STI Summar		77770 Staaros	or the orre	0000 01 1110		mannanan	00110

Results	Reference
Melatonin (10 ⁻⁹ – 10 ⁻⁵ M) inhibited MCF-7 cell proliferation only in absence of oestrogen	Baldwin et al, 1998
Melatonin (10 ⁻⁹ M) inhibited proliferation in rat mammary cancer cell line (RM4) in presence of 17-beta-oestradiol, but had no effect when serum was included in culture medium	Bizzarri et al, 2003
Melatonin (10 ⁻¹¹ – 10 ⁻⁹ M) inhibited MCF-7 cell proliferation, and blocked oestradiol stimulation, but required serum in the culture medium	Blask and Hill, 1986
Melatonin inhibited proliferation to a varying extent in three breast cancer cell lines, but only MCF-7 cells responded to physiological concentrations (10 ⁻⁹ M). No effect on oestrogen-insensitive cells	Hill et al, 1992
Inhibitory effect of melatonin (10 ⁻¹¹ – 10 ⁻⁷ M) varied between stocks of MCF-7 cells	Ram et al, 2000
Melatonin (10 ⁻⁵ M) inhibited proliferation of MCF-7 cells, but had no effect on the stimulatory action of oestradiol	Czeczuga-Semeniuk et al, 2002
Melatonin (10 ⁻⁹ M) only inhibited proliferation in fast growing MCF-7 cells	Cos and Sanchez-Barcelo, 1995
Melatonin inhibited proliferation in MCF-7 cells	Karasek and Pawlikowski, 1999
Melatonin (10 ⁻⁹ M) inhibited proliferation in MCF-7 cells	Liburdy et al 1993
Melatonin $(10^{-13} - 10^{-3} \text{ M})$ showed hardly any effect of inhibiting the growth of the six human cancer cell lines tested, including MCF-7 cells	Bartsch et al, 1992
No inhibitory effect of melatonin $(10^{-13} - 10^{-7} \text{ M})$ in MCF-7 cells, nor in three other non-breast cancer cell lines	Panzer et al, 1998
No inhibitory effect of melatonin (10 ⁻¹¹ – 10 ⁻⁹ M) on cell proliferation in breast cancer cell lines MCF-7 and T47D	Papazisis et al, 1998
No inhibitory effect of melatonin $(10^{-10} - 10^{-8} \text{ M})$ on MCF-7 cells, nor in three other non-breast cancer cell lines	Shellard et al, 1989

5.2 In vivo studies

The potential of melatonin to modulate the incidence and growth of mammary tumours has been determined using various animal models of breast cancer. Most studies have assessed the effects of exogenous melatonin treatment on the growth of chemically induced mammary tumours. However, a few studies have investigated effects on transplantable tumours, while others have used normal and transgenic mouse strains that express a high spontaneous incidence of mammary tumours. Further

studies have investigated the effects caused by inducing changes in pineal function either through pinealectomy or by altering the photoperiod.

Early studies found that treatment with melatonin had inconsistent effects on the growth of various tumours in rodents, with both stimulatory, inhibitory and null effects being reported (see Pawlikowski et al, 2002). However, Bartsch and Bartsch (1981) suggested that the effect of melatonin on tumour growth was dependent on the photoperiod and the time of administration of the treatment: application of melatonin in the morning caused an increase in tumour growth, whereas exposure in the evening caused an inhibition of growth. A similar differential effect of time of treatment on tumour development was reported by Wrba et al (1986) and Chatterjee and Banerji (1989), prompting the suggestion that (mammary) tumours may exhibit a diurnal rhythm in sensitivity to melatonin (Blask, 1997). The majority of subsequent studies have found that melatonin administration in the late afternoon and early evening has an inhibitory effect on a wide variety of tumour types in rats, mice and hamsters: these data have been reviewed by Anisimov (2003), Brainard et al (1999), Cos and Sánchez-Barceló (2000), Pawlikowski et al (2002) and Sánchez-Barceló et al (2003). The discrepancies between studies have been largely attributed to differences in treatment times and also to complex differences in tumour types and to differences between the animal species used.

5.2.1 Transplantable tumours

Two studies have investigated the effects of melatonin on transplantable mammary tumours. Anisimov et al (1973; quoted from Anisimov, 2003) reported that daily subcutaneous (sc) injection of melatonin caused a large decrease in the size of the mammary tumours in female mice. Similarly, Karmali et al (1978a,b) reported that injection of melatonin significantly inhibited the growth of R3230AC mammary tumours in female Fisher rats. These studies are summarised in Table 5.2.

Model	Treatment	Result	Reference
RSM carcinoma in C3HA mouse	Daily sc injection, 50 µg per mouse per day	Decrease in tumour size (p<0.05)	Anisimov et al, 1973
R3230AC tumour in the rat	1 mg, daily sc injection, beginning 2 days before transplant and continuing for 4 weeks	Reductions in tumour weight (p<0.05) and size	Karmali et al, 1978a,b

TABLE 5.2 Effects of melatonin on transplantable mammary tumours

5.2.2 Chemically induced tumours

Many animal studies (see Pawlikowski et al, 2002) have reported that treatment with melatonin reduces the growth of a variety of chemically induced tumours, including colon cancers induced by 7,12-dimethylbenz[a]anthracene (DMBA) in rats (Anisimov et al, 1997a, 2000) and skin papillomas induced by benzo(a)pyrene in mice (Kumar and Das, 2000).

More than a dozen studies (summarised in Table 5.3) have investigated the effects of exogenous melatonin treatment on mammary cancer induced by DMBA or *N*-nitroso-*N*-methylurea (NMU) in female rats. Some of these studies also investigated the effectiveness of combined melatonin treatment with drugs such as tamoxifen or raloxifene, but these particular results are not considered here.

Model	Carcinogen treatment	Melatonin treatment	Photoperiod (light:dark) h	Result	Reference
Sprague- Dawley rat	30 mg DMBA by gavage at 50 days old	100 μg by sc injection daily in morning from 43 to 243 days old	12:12	<i>Increase</i> in tumour incidence (p<0.005)	Hamilton, 1969
Sprague- Dawley rat	25 mg DMBA at 60 days old (route not specified)	200 μg by sc injection twice a week at 17.00 h from 60 to ≈135 days old	12:12	Significant decrease in latency (p<0.001). Incidence of tumours reduced by 25–30%	Aubert et al, 1980
Sprague- Dawley rat	15 mg DMBA, by gavage, at 50 days old	2.5 mg kg ⁻¹ by ip injection daily at 16.00 h, from 50 to 140 days old	12:12	Highly reduced incidence of tumours at 190 days old (p<0.002)	Tamarkin et al, 1981
Holtzman rat	10 mg DMBA by gavage, at 55–60 days old	500 μg by ip injection daily in late afternoon, from 52 to 145 days old	10:14	Incidence of tumours reduced by 38%	Shah et al, 1984
Sprague- Dawley rat	5 mg DMBA by iv injection, at 55 days old	250 μg by injection daily in late afternoon (from 16.00 to 18.00 h), from 76 to 181 days old	12:12	Slight, non- significant reduction in tumour incidence. Significant inhibition observed in combination with underfeeding (p<0.05)	Blask et al, 1986
Holtzman rat	20 mg DMBA by gavage, at 55 days old	100 μg per rat per day, in drinking water, from 20 to 140 days old	10:14	Reduced incidence of tumours at 180 days old (p<0.05)	Kothari, 1987
Holtzman rat	10 mg DMBA by gavage, at 55 days old	200 μg per rat per day, in drinking water, from 48 to 62 days old (initiation); or from 62 days old for 26 weeks (promotion)	10:14	Reduced incidence of tumours at 244 days old (initiation p<0.05; promotion p<0.0025)	Subramanian and Kothari, 1991a
Sprague- Dawley rat	3 doses of 10 mg* DMBA by gavage, on day of age 45, 50 and 55	20 μg ml ⁻¹ (0.45 mg per rat) in drinking water, from 15.00 to 08.00 h, 4 alternate days per week, from 33–37 days old for 26 weeks after DMBA treatment	12:12	Reduced incidence and frequency of tumours per group (p<0.001)	Môciková- Kalická et al, 2001
Wistar: Han rat	1.44 Gy γ-rays (⁶⁰ Co) over 15 days from 44–46 days old. 3 doses of 10 mg DMBA, by gavage	$100 \ \mu g \ ml^{-1}$ in drinking water, available continuously, from irradiation for 26 weeks	12:12	Total tumour volume decreased (p<0.05) but no significant effects on tumour incidence, frequency	Môciková et al, 2000

TABLE 5.3 Effects of melatonin on chemically induced tumours in female rats

 Dosage of DMBA as given by Môciková-Kalická et al (2001) in their abstract: the value of 10 μg given in their main text seems unlikely.

or latency

from 52-58 days

old

TABLE 5.3 Continued

Model	Carcinogen treatment	Melatonin treatment	Photoperiod (light:dark) h	Result	Reference
Sprague- Dawley rat	2 doses of 50 mg kg ⁻¹ NMU by iv injection, at 50 and 60 days old (initiation only) or at 50 and 57 days old	500 μg by sc injection daily in late afternoon, from 37 to 60 days old (initiation); from 85 days old for 16 weeks (delayed promotion); or from 57 days old for 19 weeks (promotion)	12:12	Initiation: no significant effects. Delayed promotion: slight, non- significant effects on tumour incidence and latency, number of tumours reduced (p<0.05). Promotion: tumour incidence and number reduced (p<0.05), no effect on latency	Blask et al, 1991
Sprague- Dawley rat	50 mg kg ⁻¹ NMU by ip injection, at 50 days old	200 µg per rat per day in drinking water, available continuously, for 180 days following surgical removal of primary tumour at ≤6 months	Not known	No effect on tumour incidence but latency of second generation tumours increased (p<0.001)	Kothari et al, 1995
Sprague- Dawley rat	50 mg kg ⁻¹ NMU by ip injection, at 50 days old	200 µg per rat per day in drinking water, available continuously, for 300 days	12:12	No effect on tumour incidence but latency increased (p<0.001)	Kothari et al, 1997
Sprague- Dawley rat	2 doses of 50 mg kg ⁻¹ NMU by ip injection, 7 days apart from 46–57 days old. 20 mg DMBA by gavage, at 50–54 days old	500 μg per rat daily in drinking water from 15.00 to 08.00 h, from 34–45 days old for 24 weeks	12:12	Non-significant changes in incidence and latency, except for decrease in mean tumour volume with DMBA (p<0.05)	Kubatka et al, 2001a
Sprague- Dawley rat	2 doses of 50 mg kg ⁻¹ NMU by ip injection, 7 days apart from 43–54 days old	110 μg per rat daily in drinking water, from 15.00 to 08.00 h, from 31–42 days old for 26 weeks	12:12	Non-significant decrease in incidence of tumours by 19%. Decrease in tumour frequency per group (p<0.05)	Kubatka et al, 2001b
Sprague- Dawley rat	2 doses of 50 mg kg ⁻¹ NMU by ip injection, 7 days apart from 51–58 days old. 3 doses of 10 mg DMBA by gavage, at 45–59 days old	500 μg per rat daily in drinking water, from 15.00 to 08.00 h, from 7 days before NMU treatment for 28 weeks, or 4 days before DMBA treatment for 14 weeks	12:12	Increased tumour growth over last 10 weeks with NMU (p<0.01). Decreased incidence of tumours (p<0.05) with DMBA. Other inhibitory effects not significant	Kubatka et al, 2002
LIO	50 mg kg ⁻¹ NMU by iv injection, at 2 months old	20 mg I ⁻¹ per day in drinking water, from 18.00 to 09.00 h for 3 days starting 2 days before NMU injection	14:10	Reduced number of malignant tumour bearing rats and reduced numbers of tumours per rat	Musatov et al, 1999

In an early study, Hamilton (1969) found that morning administration of melatonin resulted in an increase of DMBA-induced mammary tumours. However, subsequent studies have reported that late afternoon or evening treatment with melatonin generally inhibited chemically induced mammary tumorigenesis, with a reduced incidence and multiplicity of tumours and an increased latency period for tumour appearance (Aubert et al, 1980; Tamarkin et al, 1981; Shah et al, 1984; Blask et al, 1986, 1991; Kothari, 1987; Subramanian and Kothari, 1991a; Kothari et al, 1995, 1997; Musatov et al, 1999; Môciková et al, 2000; Kubatka et al, 2001a,b, 2002; Môciková-Kalická et al, 2001). Melatonin treatment also brought about consistent changes in the morphology of the mammary gland compared with control animals, with much reduced ductal branching with few terminal end buds and alveolar buds. However, melatonin had inconsistent effects on food intake or weight gain, although some studies reported significant decreases in weight and in food and water intake.

Despite there being many methodological and procedural differences between these studies, the overall results are comparable, and these studies indicate that the repeated, long-term treatment of female rats with high (supraphysiological) doses of exogenous melatonin can inhibit the growth and development of mammary tumours induced by chemical carcinogens. There is some evidence to suggest that higher doses of melatonin may be less effective at inhibiting mammary tumorigenesis than more physiological levels: for comparison with the values given in Table 5.3, Kubatka et al (2001a) estimated that, in the Sprague-Dawley rat, a dose of about 0.23 μ g melatonin produces a concentration of 1 nM. Further, it would seem that melatonin may act through mechanisms associated more with inhibiting effects on tumour promotion rather than with inhibiting tumour initiation.

5.2.3 Spontaneous tumours

A handful of studies have investigated the effects of melatonin on the development of mammary tumours using mouse models expressing a high incidence of spontaneous mammary cancers: these are summarised in Table 5.4. Two studies have used inbred strains. Subramanian and Kothari (1991b) found that prolonged, oral melatonin treatment caused a significant suppression of the development of mammary tumours in female C3H/Jax mice. About 65% of these animals spontaneously develop mammary tumours by eight to nine months old. At one year of age, 23% of mice treated with melatonin had developed adenocarcinomas as opposed to 63% of control animals (p<0.02). Treatment with melatonin also resulted in significant reductions (in three month old mice) in serum 17-beta-oestradiol levels (p<0.05) and in DNA synthesis in the mammary gland (p<0.02). As part of a study investigating the effects of melatonin on long-term health and tumour development in adult, female CBA mice, Anisimov et al (2001) found that intermittent, oral treatment with melatonin at night did not affect the incidence of spontaneous mammary tumours.

Other studies have used transgenic mice over-expressing genes involved in mammary cancer. Mediavilla et al (1997) injected young female mice carrying the N-*ras* proto-oncogene with melatonin or vehicle late in the evening, five times a week. After five months of treatment, the mammary glands were dissected and examined for histological and immunochemical changes. Compared to vehicle-treated animals, the animals given melatonin had a lower density of hyperplastic alveolar nodules (HANs), an absence of epithelial dysplastic cells and weak immunostaining of the N-ras protein. In addition, none of the treated animals developed mammary carcinomas.

Model	Treatment	Photoperiod (light:dark) h	Result	Reference
CH3/Jax mouse	25 μg per mouse per day in drinking water from 21 to 44 days old, and 50 μg per mouse per day from day 45 until 12 months old	10:14	Significantly reduced incidence of tumours (adenocarcinomas) (p<0.02)	Subramanian and Kothari, 1991b
CBA mouse	20 mg I^{-1} at night in drinking water, 5 consecutive days per month, from 6 months old	12:12	No significant change in incidence of tumours (adenoma or adenocarcinoma)	Anisimov et al, 2001
MMTV-LTR/N- <i>ras</i> transgenic mouse	Subcutaneous injection, 200 µg per mouse per day for 5 days per week for 5 months from 28 days old, 2 hours before lights off	14:10	Reduced incidence of premalignant lesions and adenocarcinomas. Reduced expression of N-ras protein on focal hyperplastic lesions (p<0.001)	Mediavilla et al, 1997
TG.NK/c- <i>neu</i> transgenic mouse	50–200 mg kg ⁻¹ daily by gavage for 30 weeks from 28 days old	12:12	Significant delay in appearance of palpable tumours (p<0.05)	Rao et al, 2000
HER-2/ <i>neu</i> transgenic mouse	20 mg l ⁻¹ at night in drinking water from 2 months to 9–13 months old. Calculated nightly dose of 0.076 mg per mouse	12:12	Significantly decreased incidence (p<0.0003) and size (p<0.05) of tumours. Expression of HER-2/neu mRNA reduced by 2.5 fold. Reduced life span (p<0.05)	Baturin et al, 2001 Anisimov et al, 2003

TABLE 5.4 Effects of melatonin on spontaneous mammary tumours

Rao et al (2000) reported that melatonin delayed the appearance of palpable mammary tumours and reduced the growth of these tumours in TG.NK female transgenic mice (which express the c-*neu* breast cancer oncogene under the control of mouse mammary tumour virus (MMTV) promoter). The animals were gavaged using a corn oil vehicle containing melatonin at 50, 100 or 200 mg per kilogram of body weight. These doses were administered no more than two hours before the start of the dark cycle each day for 30 weeks, starting at four weeks of age. A significant dose-related decrease was found for the incidence of tumours, and for the multiplicity of tumours per mouse. The authors noted that mortality was increased in the group treated with the high dose of melatonin, and suggested an interaction with the stresses associated with repeated gavage.

Anisimov and colleagues (Baturin et al, 2001; Anisimov et al, 2003) gave melatonin in drinking water at night (from 18.00 to 09.00 hours) to female FVB/N mice transgenic for the oncogene HER-2/neu from two to nine months of age. This treatment decreased the incidence and size of mammary adenocarcinomas, although the mean number of tumours per mouse was not changed. Treatment resulted in a reduction in the expression of HER-2/neu mRNA in mammary tumours. A reduction in life span in treated animals was attributed to melatonin-induced kidney insufficiency. Keeping the animals under constant light conditions was found to modulate the observed responses (see below) while interrupted treatment with melatonin (administered on only five nights per month, so the total dose of melatonin was relatively much smaller) increased the number of tumours per tumour-bearing mouse.

Overall, these studies suggest that treatment with exogenous melatonin has a generally inhibitory effect on mammary tumour development in mice. These effects appear dependent on factors such as the total dose of melatonin, dosing schedule, and the age of the animals.

5.2.4 Induced changes in pineal function

Some studies have indirectly assessed the potential of melatonin to affect the development of mammary carcinogenesis by causing changes in pineal function, either by manipulating the photoperiod or following pinealectomy. These studies have been reviewed recently by Brainard et al (1999) and Anisimov (2002, 2003).

Overall, these studies (Table 5.5) suggest that inhibition of pineal function with pinealectomy or with exposure to constant light (400–800 lux) stimulates chemically induced mammary tumorigenesis in female rats (Khaetski, 1965 (cited by Anisimov, 2002); Hamilton, 1969; Aubert et al, 1980; Tamarkin et al, 1981; Kothari, 1987; Kothari et al, 1982, 1984; Shah et al, 1984; Subramanian and Kothari 1991a; Anisimov et al, 1994, 1997; Travlos et al, 2001) and the induction of spontaneous mammary tumours in normal (Jöchle, 1964) and transgenic mice (Baturin et al, 2001). Exposure to constant light has also been reported to result in increased incidence of other spontaneous tumours and leukaemias in CBA female mice (Anisimov et al, 2004).

In addition, Blask et al (2002, 2003) reported that in ovary-intact, non-oestrogenised female nude rats bearing a xenograft of human MCF-7 breast cancer cells exposure to constant light produced an immediate and sustained increase in the growth rate of the tumour (Figure 5.1). After 15 days, the tumours were estimated to be significantly heavier than those of animals housed under a diurnal lighting schedule (p<0.05). Results suggested that this was mediated via a suppression of melatonin production (exposure of nude rats to constant light for five weeks was observed to suppress completely the nocturnal peak in circulating levels of melatonin), leading to an increased tumour update of linoleic acid and its metabolism to the tumour mitogen, 13-hydroxyoctadecadienoic acid (13-HODE). However, group sizes used were very small. Nevertheless, comparable results were also obtained using a tissue-isolated NMU-induced rat mammary tumour model as well as a rat hepatoma 7288CTC model (Blask et al, 2002), suggesting that inhibition of fatty acid metabolism may represent a common mode of action of melatonin.

Conversely, there is some evidence (summarised in Table 5.6) that treatments that stimulate pineal function, either through light deprivation or following surgical blinding, inhibit chemically induced mammary carcinogenesis (Chang et al, 1985; Anisimov et al, 1994, 1997b; Jull, 1996; see also Blask et al, 1988, and Sanchez-Barcelo et al, 1988). Anisimov (2002) cites early work by Kuralasov in 1979 that first demonstrated the inhibitory effects of light deprivation on the growth and development of transplantable (RMK-1) and DMBA-induced mammary tumours in rats. Survival of rats bearing tumours induced by DMBA was also 55% longer than controls.

A few of these studies have reported inconsistent or paradoxical results, possibly reflecting that the susceptibility of mammary tissues to carcinogenic insult is a complex response modulated by various environmental and other factors, including the level of endogenous melatonin.

TABLE 5.5 Effects of exposure to constant light (LL), pinealectomy (Px) or combined treatments (PxLL) on chemically induced, tissue-isolated xenografts, or spontaneous mammary tumours in female rats and mice

Model	Carcinogen treatment	Treatment	Result	Reference
Outbred rat	1.5 mg DMBA by 5 weekly iv injections	LL (4 weeks after last injection)	Significant increase in tumour multiplicity and growth rate; tumour latency reduced	Khaetski, 1965
Sprague- Dawley rat	30 mg DMBA by gavage, at 50 days old	LL	Tumour incidence increased (p<0.002)	Hamilton, 1969
Sprague-Dawley rat	25 mg DMBA at 60 days old	LL, Px or PxLL (Px at 58 days old)	Tumour latency <i>increased</i> (for LL, p<0.02)	Aubert et al, 1980
Sprague-Dawley rat	7 mg DMBA by gavage, at 50 days old	Px (Px at 20 days old)	Significantly increased incidence of tumours at 240 days old (p<0.002)	Tamarkin et al, 1981
Holtzman rat	10 mg DMBA by gavage, at 55–60 days old	LL or PxLL (Px at 1 day old)	Significantly increased incidence of tumours and significantly reduced latency period	Shah et al, 1984
Holtzman rat	20 mg DMBA by gavage, at 55 days old	LL, Px or PxLL (Px at 1-2 days old)	Increased incidence of tumours, and reduced latency, at 180 days old with LL and PxLL (in all cases, p<0.05). No effects with Px	Kothari et al, 1982, 1984; Kothari, 1987
Holtzman rat	10 mg DMBA by gavage, at 55 days old	LL, Px or PxLL (Px neonatally)	Increased incidence of tumours at 244 days old (during initiation or promotion)	Subramanian and Kothari, 1991a
Outbred rat	$50 \text{ mg kg}^{-1} \text{ NMU iv}$ injection, 3 times weekly from 55 days old	LL	Increased incidence of tumours, and <i>reduced</i> latency (in both cases, p<0.05)	Anisimov et al, 1994, 1997b
Fischer 344 rat	50 mg kg ⁻¹ NMU ip injection, at 50 days old	Px	No significant effects	Travlos et al, 2001
Inbred nude rat	Tissue-isolated human MCF-7 breast cancer xenografts, grown in BALB/c nude mice	LL on day 40 after implant	Significant increase in tumour growth rate (p<0.05); tumour weight doubled on day 55 (p<0.05)	Blask et al, 2003
C3H-A mice	Spontaneous tumours	LL	Reduced latency of tumours.	Jöchle, 1964
HER-2/ <i>neu</i> transgenic mouse	Spontaneous tumours	LL	Tumours multiplicity increased (p<0.02); mean latency <i>increased</i> (p<0.05)	Baturin et al, 2001



FIGURE 5.1 Effects of diurnal light (12L:12D) or constant light (12L:12L) on tumour growth in ovary intact, non-oestrogenised female nude rats bearing tissue-isolated MCF-7 human breast cancer xenografts. All rats (N=7) were maintained on 12L:12D until day 40 after implant at which time three rats were transferred to 12L:12L while the other four rats remained on 12L:12D. Each time point represents the mean (± SD) estimated tumour weights for both light treatment groups derived from serial measurements of each tumour over the course of the growth period. Tumour growth rates (slopes of the regression lines) between L:L and L:D groups were significantly different (p < 0.05). Redrawn from Blask et al, 2003, with permission

Model	Carcinogen treatment	Treatment	Result	Reference
Sprague- Dawley rat	20 mg DMBA by gavage, at 55 days old	DD and LL	3% and 87% of DD and 44% and 82% of LL animals developed fibroadenomas and adenocarcinomas respectively	Jull, 1966
Sprague- Dawley rat	3 x 5 mg DMBA by gavage, from 55–69 days old	Superior cervical ganglionectomy (GNx), or blinding plus olfactory bulbectomy (BA)	Generally reduced tumorigenesis: incidence of tumours reduced (GNx at 12 weeks, BA at 15 weeks, p<0.05); no. of tumours per rat reduced (GNx, p<0.05). No sham-operated controls	Chang et al, 1985
Outbred white rat	50 mg kg ⁻¹ NMU iv injection, 3 times weekly (DD) or single iv injection (XX) from 55 days old	DD or XX (at 30 days old)	Tumour incidence decreased by both treatments (p<0.05)	Anisimov et al, 1994, 1997b

TABLE 5.6	Effects of light deprivation (DD) or surgical blinding (XX) on chemically induced mammary
tumours iı	n female rats

5.2.5 Summary

A consistent pattern of results has emerged from studies investigating the effects of melatonin on mammary tumours in animal models. Treatment with melatonin in the evening, or manipulations that should increase the production of melatonin by the pineal gland, not only lead to an increase in the latency of chemically induced or spontaneous mammary tumours but generally also cause reductions in the incidence of tumours. In addition, melatonin may possess the greatest oncosuppresive potential during the promotional phase of tumorigenesis. Conversely, pinealectomy or exposure to constant light usually engenders the opposite effects, and causes increases in the incidence of tumours and acceleration of the growth of tumours. These responses on mammary carcinogenesis have been observed using a number of different models of breast cancer and using differing doses of melatonin, dosing regimes, and routes of administration. While these differences make detailed comparisons between the studies difficult, it suggests that the responses in animals generalise to some extent across different experimental conditions and are not limited to any specific circumstances. Nevertheless the importance of the time of treatment relative to the endogenous melatonin rhythm (or photoperiod) must be acknowledged. Other factors, including seasonal effects, may also modulate these observed responses, and these may account for some of the discrepancies between studies. While it might be possible to criticise the relevance of those studies where melatonin had been administered to animals at pharmacological doses, this cannot apply to those studies that manipulated the photoperiod. Further, the little evidence that exists suggests that physiological doses of melatonin are more effective at inhibiting tumorigenesis than doses at pharmacological levels.

5.3 Epidemiology: breast cancer risk in relation to melatonin levels

Studies in which melatonin has been measured prospectively in women who subsequently either developed breast cancer or remained disease free are rare. However, in a prospective nested casecontrol study amongst British women melatonin levels were not strongly associated with risk of breast cancer (Travis et al, 2004). In this study concentrations of 6-sulphatoxymelatonin (aMT6s) were measured by radioimmunoassay in 24 hour urine samples collected from women enrolled into a prospective investigation in which evidently normal women without any evidence of breast cancer were followed up for cancer incidence. Levels in 127 patients diagnosed with breast cancer during follow-up were not significantly different from 353 control subjects (not developing cancer) matched for age, date of recruitment, menopausal status and day of the menstrual cycle for premenopausal women or number of years past the menopause in postmenopausal women (comparing patients who had aMT6s levels in the highest third of measured values with patients who had levels in the lowest third, the odds ratio, OR, for breast cancer = 0.99, with a 95% confidence interval, Cl, 0.5–1.70). This study is important in that it is prospective and well conducted. There has been some criticism of the investigation because the sample size did not allow exclusion of a modest association between urinary aMT6s and risk. The investigation also focused on total daily production of melatonin and did not measure either peak nocturnal levels or the timing of the nocturnal peak. The latter would only be pertinent if cancer risk were mediated through circadian patterns rather than absolute levels (Hrushesky and Blask, 2004).

Brief results from a case–control study nested within the US Nurses Health Study II cohort have very recently been presented at a conference (Schernhammer and Hankinson, 2005). As only a one paragraph

abstract is available, we cannot evaluate this study. The study compared urinary aMT6s levels in first morning voided urine between 190 women who subsequently developed breast cancer during four years of follow-up and 376 controls matched on several variables. After adjustment for other breast cancer risk factors, the risk in women with the highest quartile of urinary aMT6s compared with those in the lowest quartile was borderline significantly reduced (RR 0.59, 95% CI 0.34–1.00). We have noted in Chapter 3 the difficulties in interpretation of single urine specimens rather than 24 hour collections.

There are a large number of case–control studies that have compared melatonin levels in patients with breast cancer and in women without the disease. However, it is difficult to interpret findings in patients with breast cancer because the disease itself, treatment and stress/changes in behaviour occurring after diagnosis or treatment may influence melatonin concentrations in the blood. Additionally, the hormone/growth factor milieu may have changed radically from the time of initiation of malignancy to



FIGURE 5.2 Age-related decline of 24 hour urinary 6-sulphatoxymelatonin (aMT6s) production (6-hourly samples) in 160 healthy women (results of ANOVA shown, compared to the 40–44 age group, negative correlation between age and aMT6s excretion, -0.269, p<0.001). From Skene et al, 1990, by permission

diagnosis of its presence; hence prospective investigations of cohorts of individuals who are apparently 'normal' but subsequently go on to develop cancer (or not) are much more informative.

These retrospective case–control studies have yielded conflicting results. Several have indicated lower concentrations of melatonin or metabolites of melatonin in the blood and/or urine of patients with breast cancer compared with measurements in breast-cancer-free women (Bartsch et al, 1981, 1989, 1991, 1997). Other case–control studies have found: the daytime serum concentration level of melatonin was higher in patients with breast cancer than in healthy control subjects (Lissoni et al, 1987, 1990); no association between the risk of breast cancer and the mean daytime nadir and night-time peak concentrations (Danforth et al, 1985); or the amounts of aMT6s excreted in urine samples (6-hourly for 24 hours) from women with malignant breast tumours (N=10) were similar to those in women with benign breast disease (N=14) and in a control group of 160 healthy age-matched women (Skene et al, 1990). There was, however, a highly significant decline in aMT6s with age, underlining the importance of age-matched controls (see Figures 5.2 and 5.3).

Additionally, most studies have failed to control for factors such as posture and environmental light that affect melatonin secretion and these may confound associations.



FIGURE 5.3 Individual 24 hour urinary 6-sulphatoxymelatonin (aMT6s) production from 160 healthy women without breast tumours together with that from 10 women who developed malignant breast tumours. Note the very large individual differences. From Skene et al, 1990, by permission

5.4 Epidemiology: breast cancer risk in relation to light exposure

Epidemiological studies of several groups thought to have unusual timing or extent of light exposure – for instance, blind women and shift workers – have been considered as giving evidence that might be relevant to the melatonin hypothesis, although in several instances this was not the objective of the studies. None of the studies has furnished any data directly on melatonin levels, and hence they provide only indirect evidence concerning the melatonin hypothesis. Therefore, consideration of the results in the present context needs to deal with both the extent to which the studies show a relation between shift work, blindness etc and risk of breast cancer, and also whether, if there is such a relation, melatonin is the likely explanation.

5.4.1 Breast cancer risks in blind women

Risk of breast cancer in blind women has been assessed in three cohort studies from the Nordic countries (Feychting et al, 1998; Verkasalo et al, 1999; Kliukiene et al, 2001) and two case–control studies from the USA (Davis, 1991; Hahn, 1991).

In Norway (Kliukiene et al, 2001) the standardised incidence ratio (SIR) for breast cancer in visually impaired women overall was close to unity (1.02) and there was no trend of risk with degree of visual impairment, although for totally blind women there was a non-significantly reduced SIR (0.64, 95% CI 0.21–1.49). The SIR was lower for women who became totally blind over the age of 65 years than for those who became totally blind at younger ages, although neither alone was significant. Based on small numbers, the SIR was lower for ever-married (0.42) than for never-married women (0.97) who were totally blind.

In Sweden (Feychting et al, 1998), too, there was a non-significantly reduced risk of breast cancer in women who were totally blind (0.82, 95% CI 0.47–1.34), but not in those who were severely visually impaired but not blind (1.06, 95% CI 0.92–1.21). In Finland the SIR for visually impaired women overall was not appreciably reduced (0.96, 95% CI 0.80–1.15), but there was a gradient of decreasing cancer risk with increasing visual impairment across five categories (p=0.036) (Verkasalo et al, 1999). The SIR for totally blind women was again non-significantly reduced (0.47, 95% CI 0.01–2.63).

A US case–control study based on diagnoses recorded on hospital discharge records (Hahn, 1991) compared the frequency of a diagnosis of profound visual impairment in both eyes between women who had breast cancer and women who had coronary heart disease or stroke diagnoses. Individuals discharged with diabetes were excluded from the study. Unfortunately the analyses were based on discharge episodes, not individual people, because no record linkage was available. As a consequence, the same individual could count more than once in the analysis. The use of cardiovascular disease patients as controls was unsatisfactory and a potential source of bias – for instance, such patients would be atypical with respect to menopausal status and smoking, which are related to cardiovascular disease risks. The analyses were not stated to have been age adjusted, so confounding by age may have occurred. (Stratified analyses were presented by very broad age groups, but these do not satisfactorily overcome this.) Also there was no information on the duration of blindness, so that onset of blindness in the subjects could even have occurred *after* the breast cancer developed. The odds ratio for breast

cancer in blind women was significantly reduced (0.57, 95% CI 0.35–0.92) and the author stated that there was little evidence that this was confounded by marital status. The effect of blindness was only present in women aged under 65, not older women.

A second US case–control study (Davis, 1991) attempted to replicate the study by Hahn, but with the advantages that record linkage was used to avoid duplicate inclusion of the same individual, and that a second (but not entirely satisfactory) control group, consisting of several cancer diagnoses, was used. The number of subjects was smaller than in Hahn's study, however. Breast cancer risk was not appreciably diminished using the same control diagnoses as Hahn had used (0.95, 95% CI 0.4–2.2), nor using the cancer controls (OR not presented).

5.4.1.1 Discussion

The methodologically superior cohort studies from the Nordic countries each showed no decrease in breast cancer risk for visually impaired women overall and a non-significantly reduced SIR for totally blind women. The larger US case–control study was the only one to show a significantly reduced risk, but it had too many serious methodological flaws to carry weight as evidence. The studies did not (with the exception of that by Hahn, 1991) publish results by age or menopausal status, but the data appear largely to relate to postmenopausal ages. Although the Finnish study showed a gradient of decreasing breast cancer risk with increasing visual impairment, this was not supported by the Norwegian and Swedish data. Overall, therefore, the evidence for an association between complete blindness and decreased risk of breast cancer is appreciable but not conclusive. Furthermore, there are several reasons why even if there is an association, it is far from obvious that it is due to melatonin levels or indeed any endocrinological consequence of blindness, rather than due to confounding by factors that might affect breast cancer risks in these women, or bias.

Numerous confounders might affect comparison of blind with sighted women, and almost none of these has been taken into account in the analyses discussed above. Reproductive factors might well differ between blind and sighted women, and indeed Kliukiene et al (2001) noted that in Norway the proportion of blind women who were nulliparous (89% of those born from 1935 onwards) was far greater than the proportion of women in the general population (39% of women born in the same period). The only, very partial, approach to addressing this problem was sub-analyses by marital status (and hence, by implication, by likelihood of having had any children) in two studies (Hahn, 1991; Kliukiene et al, 2001). Greater prevalence of nulliparity in blind women, however, would be expected to lead to *raised*, not reduced, risks of breast cancer.

Exercise levels could reasonably be lower in blind than sighted women, and Pukkala et al (1999) cite expert opinion that this is so. If this is the case, however, it would again be expected to lead to *increased* rather than decreased risk of breast cancer.

Age at menarche and age at menopause might differ between blind and sighted women, and Hahn (1991) cites evidence of earlier menarche in the blind, which would again bias the study in favour of finding an increased risk of breast cancer in blind women. Hormone replacement therapy and oral contraceptive use might differ between blind and sighted women, as might diet and alcohol intake. Such lifestyle-related factors might differ both because of the effects of blindness on lifestyle and also because blindness might affect employment opportunities, with consequent socio-economic effects. Pukkala et al

(1999) cite expert opinion that blind women do not drink more alcohol than others, but have worse diets (in terms of lack of fruit and high fat intake).

The diagnoses that lead to blindness might be associated with altered breast cancer risk – in particular, diabetes is a frequent cause of blindness and might be related to raised risk of breast cancer (Weiderpass et al, 1997; Wideroff et al, 1997). Also, macular degeneration risk is raised in smokers, and it is notable that the two studies that analysed it (Feychting et al, 1998; Pukkala et al, 1999) each found non-significantly raised standardised mortality ratios (SMRs) for lung cancer in totally blind women. Smoking, it has been suggested, may alter (increase or decrease) the risk of breast cancer (MacMahon et al, 1982; Gammon et al, 1998; Band et al, 2002).

Methodological features of the studies might also have affected risks. For instance, in the Finnish dataset (Pukkala et al, 1999) the blindness register was more complete in low social class geographical areas than higher social class areas, whereas the expected numbers were taken from the whole of Finland; as breast cancer risks tend to be lower in low social class areas, this would have given a bias toward finding a low risk of breast cancer in blind women in the cohort, although the authors state that adjustment for geography did not materially affect the conclusions. In the Swedish study (Feychting et al, 1998), inclusion of self-and relative-reported blindness as well as physician reports may have led to misclassification, but unless it was differential for cases as compared with non-cases, this would be expected to dilute and not exaggerate the effects found.

Finally, as discussed in Chapter 3, the limited available evidence does not suggest that melatonin secretion is increased in blind subjects.

5.4.2 Breast cancer risks in airline cabin staff

Breast cancer risks in female airline cabin staff have been assessed by cohort studies of cancer incidence in the Nordic countries (Pukkala et al, 1995; Lynge, 1996; Haldorsen et al, 2001; Rafnsson et al, 2001) and California (Reynolds et al, 2002), a cohort study of cancer mortality in Germany (Blettner et al, 2002), and a collaborative cohort study of cancer mortality in eight countries (Zeeb et al, 2003)*. Each study found somewhat raised breast cancer risk for airline cabin workers overall, with relative risks in the range from 1.1 to 1.9, although only in the Finnish (Pukkala et al, 1995) and Californian (Reynolds et al, 2002) studies was this significant. Risk in relation to duration (years) of exposed work was assessed in the Finnish, Norwegian, Icelandic, German and international cohorts, with no substantial trend of increasing risks with longer exposure, and in the Californian cohort, where there was a non-significant indication of greater risk after longer service. A nested case–control analysis in the Icelandic cohort, after extension of the follow-up and adjusting for reproductive variables, found risk was borderline significantly related to years of employment for the cohort overall, and was significantly related for employment before 1971, when jet aircraft were introduced on Icelandic international routes (Rafnsson et al, 2003). In the international

^{*} In addition: (a) a small cohort study has been published from Greece with only two breast cancers observed (two expected) (Paridou et al, 2003) and (b) a US retrospective cohort study of female cabin staff, published as a letter, reported a raised risk of breast cancer (Wartenberg and Stapleton, 1998). There was almost no description of the methods in the US study, however, and there was apparently a serious bias from omission of cohort members who died, and perhaps omission of cancers, because of reliance on membership of a retirement association for inclusion of subjects in the study (Wartenberg and Stapleton, 1999).
mortality study (Zeeb et al, 2003), risks were non-significantly greater for women employed only before 1971 than for other female cabin crew.

The Finnish study considered risk by duration after recruitment: all breast cancers occurred at least 15 years after recruitment, but no trend was seen of increasing risks subsequently. In Iceland, all but 2 of 26 cases occurred at least 15 years after recruitment.

In Norway the extent of employment at a young age was considered, but was not related to risk. In California, there was a non-significant indication of greater risk for women who started this work younger, and also for those who flew international rather than domestic routes in 1997.

5.4.2.1 Discussion

The extent to which airline cabin staff are exposed to light at night and to time shifts would depend greatly on the type of airline work undertaken – for instance, short haul flights within Europe would entail little, if any, time shift. No direct information was available in any of the studies, however, to categorise risks in relation to the extent of these variables. (Limited information was available (at one point in time) on domestic versus international routes from the USA (Reynolds et al, 2002), but these overlap considerably in length and time shift.)

Numerous confounders might affect risks of breast cancer in airline cabin staff, and in general adjustment had not been made for these within the published studies. Airline cabin staff would be expected less often to be parous than women in the general population, which would lead to greater breast cancer risk. Pukkala et al (1995) cite data showing that in Finland 20% of female cabin staff aged 35 years and over are nulliparous compared with 12% of the general population, and that cabin staff have fewer children on average than the general population. They note, however, that these differences in reproductive factors would not entirely explain the level of raised breast cancer risk found in their cohort. In Iceland parity was lower and age at birth of first child was later for cabin staff than the general population (Rafnsson et al, 2001). In the Norwegian study (Haldorsen et al, 2001), adjustment was made for parity and age at first birth, and in these analyses there was no trend of risk with duration of cabin employment (the adjusted analyses did not compare cabin staff with the general population); in the Icelandic nested case–control study, however, there was evidence of an effect of duration of employment after such adjustment.

A second potential confounder is ionising radiation exposure due to cosmic radiation at high altitudes. Blettner et al (2002), however, calculate that the doses from such exposures in cabin crew generally would only lead to a relative risk of 1.0 to 1.1 for breast cancer, based on BEIR V calculations of the effect of ionising radiation (BEIR, 1990).

A further factor that has been shown to be greater in airline staff than others is alcohol consumption (Hansen, 2001), which would be expected to lead to raised risk of breast cancer. Other factors that have been noted to differ between airline cabin staff and the general population (Blettner et al, 2002) are passive smoking, pesticide exposure on board, and EMF exposure, although there is not good evidence that any of these factors would affect breast cancer risk. In a more general sense, airline work might be associated with a relatively high social class lifestyle; it has been noted in two studies (Pukkala et al, 1995; Lynge, 1996), however, that the breast cancer SIR for cabin attendants was greater than that for social

class 1 in the same country, giving some evidence that the raised risk was not simply due to socioeconomic lifestyle factors.

In summary, there is appreciable evidence that breast cancer risk is raised in female airline cabin attendants, but in part this may well reflect their reproductive status, and beyond this there may be other confounding factors. We know of no data reporting decreased melatonin levels in airline cabin staff, although one study suggests greater variability in melatonin production assessed by aMT6s excretion, and another reports shifts in the timing of the rhythm but does not provide data on amplitude (Roach et al, 2002; Grajewski et al, 2003). Thus the evidence is weak that there is a raised risk specifically due to melatonin-related aspects of airline cabin work.

5.4.3 Shift work and risk of breast cancer

The relation of shift work to risk of breast cancer has been investigated in four published investigations, two of cohort design and two case-control studies. Tynes et al (1996) published a cohort study of 2619 Norwegian women who had been certified to work as radio and telegraph operators during 1920-80*. Of these women, 98% had worked at sea on merchant ships. Follow-up was exceptionally complete, with only 41 women lost plus 103 who had emigrated. The degree to which the women had worked shifts was imputed from histories of the work patterns over time on the merchant ships, which had been collected from seamen's records. It thus appears that work patterns were not known for each individual directly. 'Shift work' was categorised as 0, 1, 2 or 3, but it is unclear what these levels mean or indeed what defined shift work in this study. In nested case-control analyses matched on year of birth, there was no association between shift work and breast cancer risk at ages under 50 years, but a significant (p=0.01) trend in risk for ages older than this. The relative risk of breast cancer for the top category in this analysis (compared with no shift work) was 6.1 (95% CI 1.5–24.2). There was a strong correlation between the extent of shift work and duration of employment, and duration of employment was itself significantly associated with risk of breast cancer. After adjustment for duration of employment, the effect of shift work was present but not significant. Analyses adjusting for fertility were very limited because they were based on few cases.

The other cohort-based analysis derived from the US Nurses Cohort Study, in which ten years of follow-up were available for a cohort of 78,562 US nurses in relation to exposure variables originally ascertained by questionnaire (Schernhammer et al, 2001). The study is large, detailed, and has been highly influential, although follow-up has been somewhat incomplete (90% in 1994 – Hankinson et al, 1998). The questionnaire had included an enquiry about the number of years in total that the nurses had worked on rotating night shifts with at least three nights worked per month, in addition to days or evenings in that month. Risk of breast cancer increased significantly (p=0.02) with number of years working rotating night shifts, defined in this way, with an SIR for the longest category of exposure (\geq 30 years) compared with never working such shifts of 1.36 (95% Cl 1.04–1.78). The trend and the elevation of risk in the highest exposure category were significant in postmenopausal women, and although there were results in the same direction in premenopausal women, these results were not significant. There was evidence that the relation to shift work was specific to oestrogen receptor positive cancers. The analyses were adjusted for

* The authors have recently published a second paper about this cohort (Kliukiene et al, 2003), but this does not contain analyses of shift work.

a large number of potential confounders including age at first birth, parity, oral contraceptive use, body mass index, alcohol consumption, postmenopausal hormone use and menopausal status.

In a case–control study in Seattle, USA, Davis et al (2001) compared risk factors between 813 populationbased cases of breast cancer and 793 controls identified by random digit dialling. Information on 'graveyard' shift work, defined as eight hours work between 19.00 and 09.00 hours, was obtained by interview of the cases and controls. The odds ratio for breast cancer in relation to ever-work on the graveyard shift in the ten years before diagnosis was 1.6 (1.0–2.5). There was a significant relationship of risk to number of hours worked per week on the graveyard shift in the ten years before diagnosis, with a greatest relative risk of 2.3 for the top quartile in this analysis compared with no such shift work. The risk in relation to number of years of shift work was borderline significant, depending on the analytical method. The analysis was adjusted for four breast cancer risk factors, but these did not include menopausal status, age at menarche or age at first pregnancy.

The other published case-control study was conducted by record linkage in Denmark (Hansen, 2001). Over 7,000 women diagnosed with breast cancer at ages 30–54 years, who were identified from the Danish Cancer Registry, were compared with matched controls taken at random from the Central Population Register. Employment histories were gained from a national pension scheme with compulsory membership, and occupations were counted as night work if, in a 1976 survey, 60% or more of respondents in this occupation had 'night-time schedules'. Occupations were counted as not night work if less than 40% of respondents to this survey had reported night-time schedules. The odds ratio for breast cancer in women who had worked for at least half a year in an occupation deemed to be night work from the 1976 survey, compared with work in occupations that were deemed not to be night work, was 1.5 (1.2–1.7)*. Adjustment was made for a small number of potentially confounding variables, not including age at menarche, menopausal status or alcohol consumption. Only four occupations met the criterion to be considered night workers in this study, and each of them separately showed a relative risk greater than unity for breast cancer. It was also noted, however, that each had an alcohol consumption in the 1976 survey of more than three times the median for female employees overall. There was said to be a positive trend of risk of breast cancer with number of years of work at night, but it was not stated whether this trend was significant.

5.4.3.1 Discussion

Each of the four published studies on shift workers has found significant results, often with dose or duration-response effects, for breast cancer risk in relation to shift work. It should be noted, however, that the studies differed in the aspects of shift work/night work for which such relations were found, and it is not clear that each was evaluating a comparable variable with respect to effects on melatonin. Furthermore, the Norwegian shipping cohort (Tynes et al, 1996) may have involved time-zone shift as well as shift work, and it is possible that the extent of time-zone shift differed between shift workers and non-shift workers.

The studies also varied in the menopausal status group for which an association was found. Tynes et al (1996) found a relation for postmenopausal but not premenopausal women. Schernhammer et al

* 1.5 (1.3–1.7) in a table.

(2001) found an effect of similar magnitude for both premenopausal and menopausal women (although based on larger numbers, only significant for the latter). The other two studies, which did not analyse their data by menopausal status or age, were in one instance probably largely composed of postmenopausal women (ages 20–74 years – Davis et al, 2001), and in the other probably mainly premenopausal (ages 30–54 years – Hansen, 2001). There is no obvious reason why an effect of shift work via melatonin should apply only to premenopausal, or only to postmenopausal, breast cancer. However, the general epidemiology of breast cancer gives several indications that aetiology may differ between premenopausal and postmenopausal tumours, so a difference for shift work is not inherently implausible.

While the study by Schernhammer et al (2001) adjusted for a large range of potential risk factors, and thus appears unlikely to be confounded by known risk factors for breast cancer, the other three studies had partial if any such adjustment and therefore may have been confounded by other known risks factors for breast cancer. (The Tynes et al (1996) study, however, might be a special circumstance where confounding was minimal because all cohort members were from similar backgrounds and lived in similar conditions.) A notable potential confounder, because it was found to be raised in shift workers in the study by Hansen (2001), is alcohol consumption. Ever-use of oral contraceptives has also been found more common in shift workers as, slightly, has nulliparity (Davis et al, 2001). There might be confounding by exercise levels, both occupationally and outside the workplace. There is also a possibility that genetic factors might influence which individuals find night-shift work manageable, and hence there might be genetic differences between night or shift workers and other workers. For instance, it has been found that 'morning types' have more difficulty in adapting to night work, and are more likely to give it up, than 'evening types', and have different melatonin profiles (Griefahn et al, 2002). Furthermore, given the lack of complete knowledge about breast cancer aetiology, it is possible that there are other, as yet unknown, risk factors for breast cancer that are confounding - because shift work is strongly associated with many aspects of behaviour and lifestyle there is considerable scope for this to occur. As well as differences between shift workers and non-shift-workers within an occupational group, there could also be confounding differences between those occupations in which shift work occurs frequently and those in which it does not.

Methodological issues can be noted in certain of the shift worker studies that lead to some uncertainty about their results, but in general these were well-conducted studies and the results are not obviously the consequence of bias or errors in methodology. The study by Tynes et al (1996) leaves it unclear what shift work means as a variable within the study, and has no individual data on whether subjects were shift workers, but has the strength of an apparently well-conducted and complete cohort study. The Nurses Health Study asked only about rotating shift work, so that non-rotating (permanent) night workers would count as 'unexposed'. The study by Davis et al (2001) has the potential selection and reporting biases of a case–control study – in particular, that random digit dialling might have given bias in the types of individual included in the controls. Finally, the Danish record linkage study (Hansen, 2001) has the deficiency that no information was available for the individuals in the study concerning their shift work status, and this was deduced simply from a separate survey at one point in time. Overall, however, the studies can be regarded as supportive of an association of shift work with risk of breast cancer, although not conclusively so, but they are more unclear on whether an association, if one exists, is causal or a consequence of confounding.

Furthermore, even if shift work is causally associated with breast cancer risk, it would remain to be demonstrated whether effects on melatonin were the pathway for this risk. Although there is evidence that recent shift work is associated with diminished melatonin levels (Schernhammer et al, 2004; and see Chapter 3), this could be because melatonin is affected by shift work, or because women with lower melatonin levels choose to do shift work. It is very likely that work on the night shift with sufficient illumination will lead to suppression of melatonin secretion (see Chapter 3), although this remains to be demonstrated. It may, however, be more appropriate to consider the disruption to the whole circadian system as a result of rapid change of time cues. Animal work suggests that such disruption is associated with vulnerability to cancer progression (Filipski et al, 2004). Furthermore, oestradiol levels have been found raised in long-term shift workers (Schernhammer et al, 2004), giving an alternative potential pathway for causality.

5.4.4 Breast cancer risks in inhabitants of the polar regions

It has been hypothesised that winter darkness in the Arctic might lead to raised cumulative melatonin levels during these months, and hence to altered breast cancer risk (Erren, 2002). Limited data address this point. The Antarctic winter leads to a change in *timing* (delay) but not the amplitude of melatonin secretion and other circadian rhythms (Broadway et al, 1987, Yoneyama et al, 1999). One report describes free-running melatonin and other rhythms without apparent change of amplitude or duration of melatonin in four subjects (Kennaway and Van Dorp, 1991) and the available evidence from Northern Europe (summarised in Chapter 3) indicates an increase in melatonin levels on specific days of the menstrual cycle and in daytime in winter compared with summer (Kauppila et al, 1987; Kivela et al, 1988).

Low incidence of breast cancer has been found in Alaskan Indians (SIR 47) (Lanier et al, 1980) and Alaskan, Canadian and Greenland Eskimos (SIRs about 20–50) (Lanier et al, 1980; Miller and Gaudette, 1996). These data are ecological, however, relating to populations not individuals, and as such are weak evidence. Furthermore, in addition to severe potential for confounding by behavioural factors and exposures, there may be genetic differences between Arctic inhabitants and persons elsewhere that may influence cancer risks.

5.4.5 Light at night and risk of breast cancer

As well as analysing risks in relation to shift work, the case–control study by Davis et al (2001) discussed above examined breast cancer risks in relation to reported degree of light exposure at night. Breast cancer risk increased significantly with reported greater number of nights per week on which the subject was not asleep between 01.00 and 02.00 hours (stated to be the period of peak melatonin levels – we would expect it to be two hours or so later), and number of years of such a sleep pattern, after adjustment for a limited number of potential confounders.

Cases and controls did not differ significantly, however, in the extent of getting up and turning the light on at night, and in the lightness of the bedroom at night. The study analysed several variables, and even for those that were significant the results were not entirely consistent.

5.4.6 Summary

There is appreciable, but not conclusive, evidence for a relation of blindness to decreased risk of breast cancer and of airline cabin work and shift work to raised risk. Confounding is a potential explanation in each instance, but even if there are true associations it is far from clear, and there is no direct evidence, that the associations involve melatonin. None of the studies included data on melatonin levels.

The uncertainty is compounded by a lack of knowledge on which melatonin parameters (eg 24 hour total secretion, peak levels, circadian timing) might be relevant to cancer aetiology. Most epidemiological studies have assumed, if only implicitly, that 24 hour melatonin secretion is associated with their study variables. For blind women and airline cabin staff there is no evidence that cumulative melatonin secretion differs from that in normal women. For shift workers, there is some inconsistent evidence of reduced cumulative melatonin secretion (see Chapter 3) during night work in the short term. It is unclear, however, whether any long-term changes are a cause of taking up and remaining in shift work, or a consequence of it.

A further difficulty is that the exposure variables examined in these studies have often been ascertained in ways likely to lead to considerable misclassification (eg recall of sleep patterns over many years), and they were ascertained for particular time periods (eg the last ten years) that may not be those most relevant to aetiology, if there are any. Thus if there are true effects, they are likely to be greater than those shown by the studies.

Overall, the results on breast cancer risks in relation to blindness, airline cabin work and shift work are intriguing and deserve further study, but do not, at present, provide strong evidence for the melatonin hypothesis.

5.5 References

Anisimov VN (2002). The light-dark regimen and cancer development. *Neuro Endocrinol Lett*, **23**(Suppl 2), 28–36.

- Anisimov VN (2003). The role of pineal gland in breast cancer development. *Crit Rev Oncol Hematol*, **46**(3), 221–34.
- Anisimov VN, Morozov VG, Khavinson VKh and Dil'man VM (1973). Comparison of the anti-tumor activity of extracts of the epiphysis and hypothalamus, melatonin and sigetin in mice with transplanted mammary gland cancer (in Russian). *Vopr Onkol*, **19**(10), 99–101.
- Anisimov VN, Zhukova OV, Beniashvili DSh, Bilanishvili VG and Menabde MZ (1994). Light deprivation, electromagnetic fields and mammary carcinogenesis. *Adv Pineal Res*, **7**, 229–34.
- Anisimov VN, Popovich IG and Zabezhinski MA (1997a). Melatonin and colon carcinogenesis: I. Inhibitory effect of melatonin on development of intestinal tumors induced by 1,2-dimethylhydrazine in rats. *Carcinogenesis*, **18**(8), 1549–53.
- Anisimov VN, Zhukova OV, Beniashvili DSh, Bilanishvili VG, Menabde MZ and Gupta D (1997b). Effect of the light regime and electromagnetic fields on the mammary gland in female rats. *Biophysics*, **41**, 817–23.
- Anisimov VN, Popovich IG, Shtylik AV, Zabezhinski MA, Ben-Huh H, Gurevich P, Berman V, Tendler Y and Zusman I (2000). Melatonin and colon carcinogenesis. III. Effect of melatonin on proliferative activity and apoptosis in colon mucosa and colon tumors induced by 1,2-dimethylhydrazine in rats. *Exp Toxicol Pathol*, **52**(1), 71–6.
- Anisimov VN, Zavarzina NY, Zabezhinski MA, Popovich IG, Zimina OA, Shtylick AV, Arutjunyan AV, Oparina TI, Prokopenko VM, Mikhalski AI and Yashin AI (2001). Melatonin increases both life span and tumor incidence in female CBA mice. *Gerontol A Biol Sci Med Sci*, **56**(7), B311–23.

- Anisimov VN, Alimova IN, Baturin DA, Popovich IG, Zabezhinski MA, Manton KG, Semenchenko AV and Yashin AI (2003). The effect of melatonin treatment regimen on mammary adenocarcinoma development in HER-2/neu transgenic mice. *Int J Cancer*, **103**(3), 300–305.
- Anisimov VN, Baturin DA, Popovich IG, Zabezhinski MA, Manton KG, Semenchenko AV, Yashin AI (2004). Effect of exposure to light-at-night on life span and spontaneous carcinogenesis in female CBA mice. *Int J Cancer*, **111**(4), 475–9.
- Aubert C, Janiaud P and Lecalvez J (1980). Effect of pinealectomy and melatonin on mammary tumor growth in Sprague-Dawley rats under different conditions of lighting. *J Neural Transm*, **47**(2), 121–30.
- Baldwin WS and Barrett JC (1998). Melatonin attenuates hydrogen peroxide toxicity in MCF-7 cells only at pharmacological concentrations. *Biochem Biophys Res Commun*, **250**, 602–5.
- Baldwin WS, Travlos GS, Risinger JI and Barrett JC (1998). Melatonin does not inhibit estradiol-stimulated proliferation in MCF-7 and BG-1 cells. *Carcinogenesis*, **19**(11), 1895–900.
- Band PR, Le ND, Fang R and Deschamps M (2002). Carcinogenic and endocrine disrupting effects of cigarette smoke and risk of breast cancer. *Lancet*, **360**, 1044–9.
- Bartsch H and Bartsch C (1981). Effect of melatonin on experimental tumors under different photoperiods and times of administration. *J Neural Transm*, **52**(4), 269–79.
- Bartsch C, Bartsch H, Jain AK, Laumas KR and Wetterberg L (1981). Urinary melatonin levels in human breast cancer patients. *J Neural Transm*, **52**, 281–94.
- Bartsch C, Bartsch H, Fuchs U, Lippert TH, Bellmann O and Gupta D (1989). Stage-dependent depression of melatonin in patients with primary breast cancer. Correlation with prolactin, thyroid stimulating hormone, and steroid receptors. *Cancer*, 64, 426–33.
- Bartsch C, Bartsch H, Bellmann O and Lippert TH (1991). Depression of serum melatonin in patients with primary breast cancer is not due to an increased peripheral metabolism. *Cancer*, **67**, 1681–4.
- Bartsch H, Bartsch C, Simon WE, Flehmig B, Ebels I and Lippert TH (1992). Antitumor activity of the pineal gland: effect of unidentified substances versus the effect of melatonin. *Oncology*, **49**, 27–30.
- Bartsch C, Bartsch H, Karenovics A, Franz H, Peiker G and Mecke D (1997). Nocturnal urinary 6-sulphatoxymelatonin excretion is decreased in primary breast cancer patients compared to age-matched controls and shows negative correlation with tumor size. *J Pineal Res*, **23**, 53–8.
- Baturin DA, Alimova IN, Anisimov VN, Popovich IG, Zabezhinski MA, Provinciali M, Mancini R and Franceschi C (2001). The effect of light regimen and melatonin on the development of spontaneous mammary tumors in HER-2/neu transgenic mice is related to a downregulation of HER-2/neu gene expression. *Neuro Endocrinol Lett*, 22(6), 441–7.
- BEIR (Committee on the Biological Effects of Ionizing Radiation) (1990). *Health Effects of Exposure to Low Levels of Ionizing Radiation: BEIR V.* Washington DC, National Academy Press.
- Bizzarri M, Cucina A, Valente MG, Tagliaferri F, Borrelli V, Stipa F and Cavallaro A (2003). Melatonin and vitamin D3 increase TGF-beta1 release and induce growth inhibition in breast cancer cell cultures. *J Surg Res*, **110**, 332–37.
- Blask DE (1997). Systemic, cellular, and molecular aspects of melatonin action on experimental breast carcinogenesis. In *The Melatonin Hypothesis: Breast Cancer and the Use of Electric Power* (RG Stevens et al, eds). Columbus OH, Battelle Press, pp 189–230.
- Blask DE and Hill SM (1986). Effects of melatonin on cancer: studies on MCF-7 human breast cancer cells in culture. *J Neural Transm Suppl*, **21**, 433–49.
- Blask DE, Hill SM, Orstead KM and Massa JS (1986). Inhibitory effects of the pineal hormone melatonin and underfeeding during the promotional phase of 7,12-dimethylbenzanthracene-(DMBA)-induced mammary tumorigenesis. J Neural Transm, 67(1–2), 125–38.
- Blask DE, Hill SM and Pelletier DB (1988). Oncostatic signalling by the pineal gland and melatonin in the control of breast cancer. In *The Pineal Gland and Cancer* (D Gupta et al, eds). London, Brain Research Promotion, pp 221–32.
- Blask DE, Pelletier DB, Hill SM, Lemus-Wilson A, Grosso DS, Wilson ST, and Wise ME (1991). Pineal melatonin inhibition of tumor promotion in the N-nitroso-N-methylurea model of mammary carcinogenesis: potential involvement of antiestrogenic mechanisms *in vivo. J Cancer Res Clin Oncol*, **117**(6), 526–32.

- Blask DE, Dauchy RT, Sauer LA, Krause JA and Brainard GC (2002). Light during darkness, melatonin suppression and cancer progression. *Neuroendocrinol Lett*, **23**(Suppl 2), 52–6.
- Blask DE, Dauchy RT, Sauer LA, Krause JA and Brainard GC (2003). Growth and fatty acid metabolism of human breast cancer (MCF-7) xenografts in nude rats: impact of constant light-induced nocturnal melatonin suppression. *Breast Cancer Res Treat*, **79**(3), 313–20.
- Brainard GC, Kavet R and Kheifets LI (1999). The relationship between electromagnetic field and light exposures to melatonin and breast cancer risk: a review of the relevant literature. *J Pineal Res*, **26**(2), 65–100.
- Blettner M, Zeeb H, Langner I, Hammer GP and Schafft T (2002). Mortality from cancer and other causes among airline cabin attendants in Germany, 1960–1997. *Am J Epidemiol*, **156**, 556–65.
- Broadway J, Arendt J and Folkard S (1987). Bright light phase shifts the human melatonin rhythm during the Antarctic winter. *Neurosci Lett*, **79**(1–2), 185–9.
- Chang N, Spaulding TS and Tseng MT (1985). Inhibitory effects of superior cervical ganglionectomy on dimethylbenz(a)anthracene-induced mammary tumors in the rat. *J Pineal Res*, **2**, 331–40.
- Chatterjee S and Banerji TK (1989). Effects of melatonin on the growth of MtT/F4 anterior pituitary tumor: evidence for inhibition of tumor growth dependent upon the time of administration. *J Pineal Res*, **7**(4), 381–91.
- Cos S and Sanchez-Barcelo EJ (1995). Melatonin inhibition of MCF-7 human breast-cancer cells growth: influence of cell proliferation rate. *Cancer Lett*, **93**, 207–12.
- Cos S and Sánchez-Barceló EJ (2000). Melatonin and mammary pathological growth. *Front Neuroendocrinol*, **21**(2), 133–70.
- Czeczuga-Semeniuk E, Wolczynski S, Anchim T, Dzieciol J, Dabrowska M and Pietruczuk M (2002). Effect of melatonin and all-trans retinoic acid on the proliferation and induction of the apoptotic pathway in the culture of human breast cancer cell line MCF-7. *Pol J Pathol*, **53**, 59–65.
- Danforth DN Jr, Tamarkin L, Mulvihill JJ, Bagley CS and Lippman ME (1985). Plasma melatonin and the hormonedependency of human breast cancer. *J Clin Oncol*, **3**, 941–8.
- Davis R (1991). Light exposure and breast cancer. Epidemiology, 2, 458.
- Davis S, Mirick DK and Stevens RG (2001). Night shift work, light at night, and risk of breast cancer. J Natl Cancer Inst, 93, 1557–62.
- Erren TC (2002). Does light cause internal cancers? The problem and challenge of an ubiquitous exposure. *Neuroendocrinol Lett*, **23** (Suppl 2), 61–70.
- Feychting M, Osterlund B and Ahlbom A (1998). Reduced cancer incidence among the blind. *Epidemiology*, **9**, 490–94.
- Filipski E, Delaunay F, King VM, Wu MW, Claustrat B, Grechez-Cassiau A, Guettier C, Hastings MH and Francis L (2004). Effects of chronic jet lag on tumor progression in mice. *Cancer Res*, **64**(21), 7879–85.
- Gammon MD, Schoenberg JB, Teitelbaum SL, Brinton LA, Potischman N, Swanson CA, Brogan DJ, Coates RJ, Malone KE and Stanford JL (1998). Cigarette smoking and breast cancer risk among young women (United States). *Cancer Causes Control*, **9**, 589–90.
- Grajewski B, Nguyen MM, Whelan EA, Cole RJ and Hein MJ (2003). Measuring and identifying large-study metrics for circadian rhythm disruption in female flight attendants. *Scand J Work Environ Health*, **29**, 337–46.
- Griefahn B, Kunemund C, Golka K, Thier R and Degen G (2002). Melatonin synthesis: a possible indicator of intolerance to shiftwork. *Am J Ind Med*, **42**, 427–36.
- Hahn RA (1991). Profound bilateral blindness and the incidence of breast cancer. *Epidemiology*, 2, 208–10.
- Haldorsen R, Reitan JB and Tveten U (2001). Cancer incidence among Norwegian airline cabin attendants. *Int J Epidemiol*, **30**, 825–30.
- Hankinson SE, Willett WC, Colditz GA, Hunter DJ, Michaud DS, Deroo B, Rosner B, Speizer FE and Pollak M (1998). Circulating concentrations of insulin-like growth factor-I and risk of breast cancer. *Lancet*, **351**, 1393–6.
- Hamilton T (1969). Influence of environmental light and melatonin upon mammary tumour induction. *Br J Surg*, **56**(10), 764–66.
- Hansen J (2001). Breast cancer among women who work at night. Epidemiology, 12, 588-9.

- Hill SM, Spriggs LL, Simon MA, Muraoka H and Blask DE (1992). The growth inhibitory action of melatonin on human breast cancer cells is linked to the estrogen response system. *Cancer Lett*, **64**(3), 249–56.
- Hrushesky WJM and Blask DE (2004). Re: Melatonin and breast cancer: a prospective study. J Natl Cancer Inst, **96**(6), 888–9.
- Jöchle W (1964). Trends in photophysiologic concepts. Ann NY Acad Sci 117, 88–104.
- Jull JW (1966). The effect of infection, hormonal environment, and genetic constitution on mammary tumor induction in rats by 7,12-dimethylbenz(a)anthracene. *Cancer Res*, **26**, 2368–73.
- Karasek M and Pawlikowski M (1999). Antiproliferative effects of melatonin and CGP 52608. *Biol Signals Recept*, **8**, 75–8.
- Karmali RA, Horrobin DF, Ghayur T, Manku MS, Cunnane SC, Morgan RO, Ally AI, Karmazyn M and Oka M (1978a). Influence of agents which modulate thromboxane A2 synthesis or action on R3230AC mammary carcinoma. *Cancer Lett*, 5(4), 205–8.
- Karmali RA, Horrobin DF and Ghayur T (1978b). Role of pineal gland in aetiology and treatment of breast cancer. Lancet, Nov 4, **2**(8097), 1001–2.
- Kauppila A, Kivela A, Pakarinen A and Vakkuri O (1987). Inverse seasonal relationship between melatonin and ovarian activity in humans in a region with a strong seasonal contrast in luminosity. J Clin Endocrinol Metab, 65(5), 823–8.
- Kennaway DJ and Van Dorp CF (1991). Free-running rhythms of melatonin, cortisol, electrolytes, and sleep in humans in Antarctica. *Am J Physiol*, **260**(6 Part 2), R1137–44.
- Khaetski IK (1965). Effect of hypothalamo-pituitary lesions induced by constant illumination on development of induced mammary tumours in rats. *Vopr Exp Oncol (Kiev)*, **1**, 87–93.
- Kivela A, Kauppila A, Ylostalo P, Vakkuri O and Leppaluoto J (1988). Seasonal, menstrual and circadian secretions of melatonin, gonadotropins and prolactin in women. *Acta Physiol Scand*, **132**(3), 321–7.
- Kliukiene J, Tynes T and Andersen A (2001). Risk of breast cancer among Norwegian women with visual impairment. *Br J Cancer*, **84**, 387–99.
- Kliukiene J, Tynes T and Andersen A (2003). Follow-up of radio and telegraph operators with exposure to electromagentic fields and risk of breast cancer. *Eur J Cancer Prev*, **12**, 301–7.
- Kothari LS (1987). Influence of chronic melatonin on 9,10-dimethyl-1,2-benzanthracene-induced mammary tumors in female Holtzman rats exposed to continuous light. *Oncology*, **44**(1), 64–6.
- Kothari LS, Shah PN and Mhatre MC (1982). Effect of continuous light on the incidence of 9,10-dimethyl-1,2benzanthracene induced mammary tumors in female Holtzman rats. *Cancer Lett*, **16**, 313–17.
- Kothari LS, Shah PN and Mhatre MC (1984). Pineal ablation in varying photoperiods and the incidence of 9,10-dimethyl-1,2-benzanthracene induced mammary cancer in rats. *Cancer Lett*, **22**, 99–102.
- Kothari A, Borges A and Kothari L (1995). Chemoprevention by melatonin and combined melatonin-tamoxifen therapy of second generation nitroso-methylurea-induced mammary tumours in rats. *Eur J Cancer Prev*, **4**(6), 497–500.
- Kothari A, Borges A, Ingle A and Kothari L (1997). Combination of melatonin and tamoxifen as a chemoprophylaxis against N-nitroso-N-methylurea-induced rat mammary tumors. *Cancer Lett*, **111**(1–2), 59–66.
- Kubatka P, Bojková B, Môciková-Kalická K, Mníchová-Chamilová M, Adámeková E, Ahlers I, Ahlersová E and Cermáková M (2001a). Effects of tamoxifen and melatonin on mammary gland cancer induced by N-methyl-N-nitrosourea and by 7,12-dimethylbenz(a)anthracene, respectively, in female Sprague-Dawley rats. *Folia Biol* (Praha), **47**(1), 5–10.
- Kubatka P, Bojková B, Kalická K, Chamilová M, Adámeková E, Ahlers I, Ahlersová E and Cermáková M (2001b). Preventive effects of raloxifene and melatonin in N-methyl-N-nitrosourea-induced mammary carcinogenesis in female rats. *Neoplasma*, **48**(4), 313–19.
- Kubatka P, Kalická K, Chamilová M, Ahlersová E, Ahlers I, Bojková B and Adámeková E (2002). Nimesulide and melatonin in mammary carcinogenesis prevention in female Sprague-Dawley rats. *Neoplasma*, **49**(4), 255–9.
- Kumar CA and Das UN (2000). Effect of melatonin on two stage skin carcinogenesis in Swiss mice. *Med Sci Monit*, **6**(3), 471–5.

- Lanier AP, Blot WJ, Bender TR and Fraumeni JF Jr (1980). Cancer in Alaskan Indians, Eskimos, and Aleuts. J Natl Cancer Inst, 65, 1157–9.
- Liburdy RP, Sloma TR, Sokolic R and Yaswen P (1993). ELF magnetic fields breast cancer and melatonin 60 Hz fields block melatonin's oncostatic action on ER-positive breast cancer cell proliferation. *J Pineal Res*, **14**, 89–97.
- Lissoni P, Bastone A, Sala R, Mauri R, Rovelli F, Viviani S, Bajetta E, Esposti D, Esposti G, di Bella L, et al (1987). The clinical significance of melatonin serum determination in oncological patients and its correlations with GH and PRL blood levels. *Eur J Cancer Clin Oncol*, **23**, 949–57.
- Lissoni P, Crispino S, Barni S, Sormani A, Brivio F, Pelizzoni F, Brenna A, Bratina G and Tancini G (1990). Pineal gland and tumor cell kinetics: serum levels of melatonin in relation to Ki-67 labelling rate in breast cancer. *Oncology*, **47**, 275–7.
- Lynge E (1996). Risk of breast cancer is also increased among Danish female airline cabin attendants. *BMJ*, **312**, 253.
- MacMahon B, Trichopoulos D, Cole P and Brown J (1982). Cigarette smoking and urinary estrogens. *N Engl J Med*, **307**, 1062–65.
- Mediavilla MD, Güezmez A, Ramos S, Kothari L, Garijo F and Sánchez-Barceló EJ. (1997). Effects of melatonin on mammary gland lesions in transgenic mice overexpressing N-*ras* proto-oncogene. J Pineal Res, **22**(2), 86–94.
- Miller AB and Gaudette LA (1996). Cancers of skin, bone, connective tissues, brain, eye, thyroid and other specified and unspecified sites in Inuit. *Acta Oncol*, **35**, 607–16.
- Môciková K, Mníchová M, Kubatka P, Bojková B, Ahlers I and Ahlersová E (2000). Mammary carcinogenesis induced in Wistar:han rats by the combination of ionizing radiation and dimethylbenz(a)anthracene: prevention with melatonin. *Neoplasma*, **47**(4), 227–9.
- Môciková-Kalická K, Bojková B, Adámeková E, Mníchová-Chamilová M, Kubatka P, Ahlersová E and Ahlers I (2001). Preventive effect of indomethacin and melatonin on 7,12-dimethybenz(a)anthracene-induced mammary carcinogenesis in female Sprague-Dawley rats. A preliminary report. *Folia Biol (Praha)*, **47**(2), 75–9.
- Musatov SA, Anisimov VN, Andre V, Vigreux C, Godard T and Sichel F (1999). Effects of melatonin on N-nitroso-Nmethylurea-induced carcinogenesis in rats and mutagenesis *in vitro* (Ames test and COMET assay). *Cancer Lett*, **138**(1-2), 37-44.
- Osborne CK, Hobbs K and Trent JM (1987). Biological differences among MCF-7 human breast cancer cell lines from different laboratories. *Breast Cancer Res Treat*, **9**, 111–21.
- Panzer A, Lottering ML, Bianchi P, Glencross DK, Stark JH and Seegers JC (1998). Melatonin has no effect on the growth, morphology or cell cycle of human breast cancer (MCF-7), cervical cancer (HeLa), osteosarcoma (MG-63) or lymphoblastoid (TK6) cells. *Cancer Lett*, **122**, 17–23.
- Papazisis KT, Kouretas D, Geromichalos GD, Sivridis E, Tsekreli OK, Dimitriadis KA and Kortsaris AH (1998). Effects of melatonin on proliferation of cancer cell lines. *J Pineal Res*, **25**, 211–18.
- Paridou A, Velonakis E, Langner I, Zeeb H, Blettner M and Tzonou A (2003). Mortality among pilots and cabin crew in Greece, 1960–1997. *Int J Epidemiol*, **32**, 244–7.
- Pawlikowski M, Winczyk K and Karasek M (2002). Oncostatic action of melatonin: facts and question marks. *Neuro Endocrinol Lett*, **23**(Suppl 1), 24–9.
- Pukkala E, Auvinen A and Wahlberg G (1995). Incidence of cancer among Finnish airline cabin attendants, 1967–92. *BMJ*, **311**, 649–52.
- Pukkala E, Verkasalo PK, Ojarno M and Rudanko S-L (1999). Visual impairment and cancer: a population-based cohort study in Finland. *Cancer Causes Control*, **10**, 13–20.
- Rafnsson V, Tulinius H, Jonasson JG and Hrafnkelsson J (2001). Risk of breast cancer in female flight attendants: a population-based study (Iceland). *Cancer Causes Control*, **12**, 95–101.
- Rafnsson V, Sulem P, Tulinius H and Hrafnkelsson J (2003). Breast cancer risk in airline cabin attendants: a nested case-control study in Iceland. *Occup Environ Med*, **60**, 807–9.
- Ram PT, Yuan L, Dai J, Kiefer T, Klotz DM, Spriggs LL and Hill SM (2000). Differential responsiveness of MCF-7 human breast cancer cell line stocks to the pineal hormone, melatonin. *J Pineal Res*, **28**, 210–18.

- Rao GN, Ney E and Herbert RA (2000). Effect of melatonin and linolenic acid on mammary cancer in transgenic mice with c-*neu* breast cancer oncogene. *Breast Cancer Res Treat*, **64**(3), 287–96.
- Rato AG, Pedrero JG, Martinez MA, del Rio B, Lazo PS and Ramos S (1999). Melatonin blocks the activation of estrogen receptor for DNA binding. *Faseb J*, **13**, 857–68.
- Reynolds P, Cone J, Layefsky M, Goldberg DE and Hurley S (2002). Cancer incidence in California flight attendants (United States). *Cancer Causes Control*, **13**, 317–24.
- Roach GD, Rodgers M and Dawson D (2002). Circadian adaptation of aircrew to transmeridian flight. *Aviat Space Environ Med*, **73**, 1153–60.
- Sánchez-Barceló EJ, Cos S and Mediavilla MD (1988). Influence of pineal function on the initiation and growth of hormone-dependent breast tumors. Possible mechanisms. In *The Pineal Gland and Cancer* (D Gupta et al, eds). London, Brain Research Promotion, pp 221–32.
- Sánchez-Barceló EJ, Cos S, Fernández R and Mediavilla MD (2003). Melatonin and mammary cancer: a short review. *Endocr Relat Cancer*, **10**(2), 153–9.
- Schernhammer ES and Hankinson SE (2005). Urinary melatonin levels and breast cancer risk. Paper presented at 96th Annual Meeting of the American Association for Cancer Research, Anaheim Convention Center, Anaheim CA, April 2005.
- Schernhammer ES, Laden F, Speizer FE, Willett WC, Hunter DJ, Kawachi I and Colditz GA (2001). Rotating night shifts and risk of breast cancer in women participating in the nurses' health study. *J Natl Cancer Inst*, **93**, 1563–8.
- Schernhammer ES, Rosner B, Willett WC, Laden F, Colditz GA and Hankinson SE (2004). Epidemiology of urinary melatonin in women and its relation to other hormones and night work. *Cancer Epidemiol Biomarkers Prev*, 13, 936–43.
- Scott AE, Cosma GN, Frank AA, Wells RL and Gardner HS Jr (2001). Disruption of mitochondrial respiration by melatonin in MCF-7 cells. *Toxicol Appl Pharmacol*, **171**, 149–56.
- Shah PN, Mhatre MC and Kothari LS (1984). Effect of melatonin on mammary carcinogenesis in intact and pinealectomized rats in varying photoperiods. *Cancer Res*, **44**(8), 3403–7.
- Shellard SA, Whelan RD and Hill BT (1989). Growth inhibitory and cytotoxic effects of melatonin and its metabolites on human tumour cell lines *in vitro*. *Br J Cancer*, **60**, 288–90.
- Skene CJ, Bojkowski CJ, Currie JE, Wright J, Boulter PS and Arendt J (1990). 6-sulphatoxymelatonin production in breast cancer patients. *J Pineal Res*, **8**, 269–76.
- Soule HD, Vazguez J, Long A, Albert S and Brennan M (1973). A human cell line from a pleural effusion derived from a breast carcinoma. *J Natl Cancer Inst*, **51**, 1409–16.
- Subramanian A and Kothari L (1991a). Suppressive effect by melatonin on different phases of 9,10-dimethyl-1,2benzanthracene (DMBA)-induced rat mammary gland carcinogenesis. *Anticancer Drugs*, **2**(3), 297–303.
- Subramanian A and Kothari L (1991b). Melatonin, a suppressor of spontaneous murine mammary tumors. *J Pineal Res*, **10**(3), 136–40.
- Tamarkin L, Cohen M, Roselle D, Reichert C, Lippman M and Chabner B (1981). Melatonin inhibition and pinealectomy enhancement of 7,12-dimethylbenz(a)anthracene-induced mammary tumors in the rat. *Cancer Res*, **41**(11 Part 1), 4432–6.
- Travis RC, Allen DS, Fentiman IS and Key TJ (2004). Melatonin and breast cancer: a prospective study. J Natl Cancer Inst, 96, 475–82.
- Travlos GS, Wilson RE, Murrell JA, Chignell CF and Boorman GA (2001). The effect of short intermittent light exposures on the melatonin circadian rhythm and NMU-induced breast cancer in female F344/N rats. *Toxicol Pathol*, **29**(1), 126–36.
- Tynes T, Hannevik M, Andersen A, Vistnes AI and Haldorsen T (1996). Incidence of breast cancer in Norwegian female radio and telegraph operators. *Cancer Causes Control*, **7**, 197–204.
- Verkasalo PK, Pukkala E, Stevens RG, Ojamo M and Rudanko S-L (1999). Inverse association between breast cancer incidence and degree of visual impairment in Finland. *Br J Cancer*, **80**, 1459–60.
- Wartenberg D and Stapleton CP (1998). Risk of breast cancer is also increased among retired US female airline cabin attendants. *BMJ*, **316**, 1902.

- Wartenberg D and Stapleton CP (1999). Re: Risk of breast cancer among female airline cabin attendants. *BMJ*, **318**, 126.
- Weiderpass E, Gridley G, Persson I, Nyrén O, Ekbom A and Adami H-O (1997). Risk of endometrial and breast cancer in patients with diabetes mellitus. *Int J Cancer*, **71**, 360–63.
- Wideroff L, Gridley G, Mellemkjaer L, Chow W-H, Linet M, Keehn S, Borch-Johnsen K and Olsen JH (1997). Cancer incidence in a population-based cohort of patients hospitalized with diabetes mellitus in Denmark. *J Natl Cancer Inst*, **89**, 1360–65.
- Wrba H, Halberg F and Dutter A (1986). Melatonin circadian-stage-dependently delays breast tumor development in mice injected daily for several months. *Chronobiologia*, **13**(2), 123–8.
- Yoneyama S, Hashimoto S and Honma K (1999). Seasonal changes of human circadian rhythms in Antarctica. *Am J Physiol*, **277**(4 Part 2), R1091–7.
- Zeeb H, Blettner M, Langner I, Hammer GP, Ballard TJ, Santaquilani M, Gundestrup M, Storm H, Haldorsen T, Tveten U, Hammar N, Linnersjo A, Velonakis E, Tzonou A, Auvinen A, Pukkala E, Rafnsson V and Hrafnkelsson J (2003). Mortality from cancer and other causes among airline cabin attendants in Europe: a collaborative cohort study in eight countries. *Am J Epidemiol*, **158**, 35–46.

6 Effects of EMFs on Breast Cancer

6.1 In vitro studies

AGNIR (2001) reviewed the evidence from cellular studies linking EMF exposure with carcinogenesis, which included breast cancer. It was concluded that there was no convincing evidence that EMFs at the levels likely to be encountered in everyday life could be directly genotoxic or bring about the transformation of cells in culture. Exposure to EMFs was therefore unlikely to initiate carcinogenesis and there was no clear evidence that EMFs could affect biological processes. Since completion of the AGNIR report, some *in vitro* studies specifically relevant to breast cancer have been published. These and other studies are reviewed here.

In their study of gene expression in MCF-7 cells, Dees et al (1996) found no field-dependent changes after exposure to 60 Hz fields at 1.2, 100 or 900 μ T for exposure periods of 2–30 hours, the time depending on the particular experiment.

Ding et al (2001) also investigated gene expression and found no change in MCF-7 cells exposed to a 60 Hz field at 5 mT for 4, 8 or 24 hours. However, if a 24 hour exposure was given after x-rays (at 12 Gy), there was a significant decrease in apoptosis, but this effect was not seen at 72 hours, suggesting, at most, a transient suppression of x-ray-induced apoptosis.

Loberg et al (2000) investigated several breast cancer lines, including MCF-7 and normal breast epithelial cells. They found no evidence that exposure to a 60 Hz field at 1 mT for 72 hours altered cell viability or growth. This work complemented their earlier study (Loberg et al, 1999) on human breast epithelial cells which reported no effects on early gene expression from exposure to a 60 Hz field at 10–1000 μ T for 24 hours.

Supino et al (2001) exposed MCF-7 cells and normal human fibroblasts to a 50 Hz field at 20 or 500 μ T for 1 or 4 days. They found no effect on growth or viability of the cells at either field strength or for either duration.

Tofani et al (2001) investigated MCF-7 cells along with two other cell types (human adenocarcinoma and embryonic lung fibroblasts). In both transformed cell lines, but not in the non-transformed lung fibroblasts, there was increased cell death after exposure to magnetic fields strengths greater than 1 mT. The effect was independent of frequency but increased when static and 50 Hz fields were superimposed.

Yoshizawa et al (2002) exposed five cell lines derived from human tumours (including MCF-7) to magnetic fields for 3 days. Both 50 and 60 Hz were studied, as were linearly polarised, circularly polarised and elliptically polarised fields. The field strengths used were 2, 20, 100 and 500 μ T rms for the linearly polarised and 500 μ T rms for the rotating fields. No effects of any exposure on cell growth were found.

The consensus of the power frequency studies is that exposure to magnetic fields in the range 1.2 μ T to 5 mT for durations up to 4 days has no effect on gene expression or cell growth in the breast cancer cell

line MCF-7 (Table 6.1). What has not been answered by these studies is whether there is a co-promotional effect of EMF exposure or whether there is an effect of long-term exposure. In general, these newer studies give no cause for concern; they provide some further reassurance regarding the lack of direct effects of EMF exposure on breast cancer cells.

EMF exposure	Effect	Reference
1.2, 100 or 900 μT, 60 Hz for 2 to 30 h	No change in gene expression in MCF-7 cells	Dees et al, 1996
5 mT, 60 Hz for 4, 8 or 24 h	No change in gene expression in MCF-7 cells	Ding et al, 2001
1 mT, 60 Hz for 72 h	No effect on viability or growth in several breast cancer cell lines including MCF-7 cells	Loberg et al, 2000
20 or 500 μT, 50 Hz for 1 or 4 days	No effect on viability or growth of MCF-7 cells	Supino et al, 2001
>1 mT, frequency independent	Increased cell death in cancer cell lines, including MCF-7 cells	Tofani et al, 2001
2, 20, 100 or 500 μT, 50 or 60 Hz, for 3 days	No effect on cell growth in five tumour cell lines including MCF-7 cells	Yoshizawa et al, 2002

TABLE 6.1	In vitro studies of	magnetic field	exposure and	breast cancer
	<i></i>	ina que cere nera	chposal c alla	biodoc callect

6.2 *In vivo* studies

The induction of mammary tumours in female rats has long been used as a standard assay in the investigation of potential carcinogenesis, often using a chemical carcinogen such as 7,12-dimethylbenz(a)anthracene (DMBA) as an initiator in the two-stage initiator/promoter model of carcinogenesis. A number of studies have used this model to explore the potential of magnetic fields to cause tumour promotion. However, there are limitations in using carcinogen-induced rodent models. Thus, the mammary glands of rodents differ from the human breast in terms of structure, developmental stages and sensitivity to mutagens and hormones (Short and Drife, 1977; Tsubura et al, 1991; Anderson 1998; Cardiff and Wellings, 1999). In addition, the DMBA model has limitations in terms of being morphologically dissimilar to most breast cancers, being highly vascular and encapsulated, having a particular sensitivity to prolactin, and rarely forming metastases.

6.2.1 EMF effects on mammary tumour growth

In the first study to explore the effects of magnetic fields on the growth of induced mammary tumours, Beniashvili et al (1991) found that while chronic exposure to magnetic fields for 0.5 hour per day caused no effects on rats, exposure for 3 hours per day significantly increased the incidence of chemically induced mammary tumours and decreased their latency. The exposed animals also had a higher total number of large malignant mammary tumours (adenocarcinomas) than the control group. However, the experimental details were only briefly summarised. In particular, a detailed description of the exposure system was not provided, the strain and background mammary tumour incidence was not given, and the methods of counting tumour incidence and assessing tumour latency were not clearly described.

Subsequently, a series of medium-term studies of magnetic field effects on mammary tumour incidence were carried out by Löscher, Mevissen and colleagues (Löscher et al, 1993, 1994, 1997; Baum et al, 1995; Löscher and Mevissen, 1995; Mevissen et al, 1993a,b, 1996a,b, 1998). These authors attempted to control for possible confounding factors. The evidence from a full histopathological analysis of mammary tissue showed that, under two different exposure conditions (0.3–1.0 μ T and 100 μ T), there was no statistically significant effect on tumour incidence. These studies are summarised in Table 6.2.

Exposure	Result	Reference
50 Hz, 0.3-1.0 μT for 13 weeks	No effect on visible or histologically identified tumour incidence	Mevissen et al, 1993a,b
50 Hz, 10 μT for 13 weeks	No effect on incidence of visible tumours at autopsy	Mevissen et al, 1996a
50 Hz, 50 μ T for 13 weeks	Increased incidence of visible tumours at autopsy (p<0.05)	Mevissen et al, 1996b
50 Hz, 100 μT for 13 weeks	Increased incidence of visible tumours (p<0.05), no effect on histologically identified incidence, increased malignancy (p<0.05)	Löscher et al, 1993 Baum et al, 1995
50 Hz, 100 μT for 13 weeks	Increased incidence of visible tumours at autopsy (p<0.05)	Mevissen et al, 1998
50 Hz, 100 μT for 27 weeks	Increased mammary tumour incidence in exposed group (p<0.05)	Thun-Battersby et al, 1999
50/60 Hz, 100 or 500 μT for 13 or 26 weeks	No significant effect. Sham tumour incidence very high in two of three experiments	Anderson et al, 1999 Boorman et al, 1999 NTP, 1999
50 Hz intermittent fields of 250 or 500 μ T for 21 weeks	No significant effect	Ekström et al, 1998

TABLE 6.2 Effects of long-term exposure to power frequency magnetic fields on DMBA-induced mammary tumours in female Sprague-Dawley rats

In contrast, the incidence of palpable tumours (detected during exposure) and, more particularly, macroscopically visible tumours (detected during post-mortem examination) was significantly increased following exposure to 50 μ T (Mevissen et al, 1996b) and to 100 μ T (Löscher et al, 1993). Indeed, the percentage increase in the incidence of macroscopically visible tumours compared with their concurrent sham control showed a highly linear dose–response relationship over the flux density range 0.3–1.0 μ T up to 100 μ T (Löscher and Mevissen, 1995) but not up to 30 mT (Mevissen et al, 1993a). Thus, Löscher and Mevissen (1995) argued that magnetic field exposure did not alter the incidence of (neoplastic) mammary lesions but accelerated tumour growth, so that higher numbers of tumours were macroscopically visible when the rats were sacrificed. In addition, Baum et al (1995) reported that in rats exposed to 100 μ T, compared with sham-exposed animals, there was a statistically significant increase in the number with mammary gland adenocarcinomas. However, the total number of malignant tumours in the exposed group was not statistically significantly increased (AGNIR, 2001).

A particular difficulty with the interpretation of these studies is that there is considerable variation between experiments in the tumour incidence in the sham-exposed groups (AGNIR, 2001). For example, in the first 100 μ T experiment (Löscher et al, 1993; Baum et al, 1995) there was an unusually low incidence of visible tumours in the sham-exposed group compared with the overall mean sham incidence in the four studies from which the linear dose–response was derived (Löscher and Mevissen, 1995). In addition, contradictory results were seen in replicate studies carried out at 30 mT. These observations suggest the possibility that the authors actually lacked control over a factor or factors that influenced the outcome of their experiments.

In an analysis of these data, Löscher et al (1997) noted the possibility of seasonal influences on mammary tumour incidence. In particular, the low sham incidence of 34% coincided with exposure during autumn and early winter, in contrast to the other experiments. A replicate 100 μ T study (Mevissen et al, 1998) carried out at a different time of year to the original study reported that the incidence of macroscopically visible tumours in the sham-exposed group was 62%, almost double the incidence in the earlier study. In addition, significantly more (83%) of the exposed animals had developed macroscopically visible tumours. A re-analysis of all of these data still revealed a statistically significant linear correlation between flux density and increase in tumour incidence (Mevissen et al, 1998). However, this analysis does not account for the variability seen in the sham-exposed data. Thun-Battersby et al (1999) reported a significantly increased incidence of mammary tumours following 10 μ T exposure for 27 weeks following initiation by only 10 mg DMBA. This effect was attributed to a field-induced increase in the proliferative activity of the epithelial cells in the mammary tissue (Fedrowitz et al, 2002).

Independent replication of these results is clearly of importance. An attempted repeat and extension of the original study using the same outbred Sprague-Dawley strain of rat has been reported by one laboratory: the NTP study (Anderson et al, 1999; Boorman et al, 1999; NTP, 1999). This found no evidence that magnetic field exposure was associated with an earlier onset or an increased multiplicity of mammary tumours, nor was the mammary tumour incidence significantly increased in the one experiment (of three) in which an increased incidence could have been detected.

There were, however, clear differences in the responsiveness of the rats used in the NTP study to DMBA compared with those used by Löscher and colleagues and other differences in experimental protocol (Anderson et al, 2000); use of 20 mg DMBA in the first 13 week NTP study (Anderson et al, 1999) or 10 mg in the 26 week study (Boorman et al, 1999) resulted in a high incidence of mammary tumours in the sham-exposed groups, diminishing the sensitivity of these experiments. However, the second 13 week study, using only 8 mg DMBA, resulted in an incidence of around 40–50% in the sham group (Anderson et al, 1999). Ekström et al (1998) found no effect on DMBA-induced mammary tumour incidence in the same rat strain following prolonged exposure to intermittent power frequency magnetic fields. There were no statistically significant differences in the number of tumour bearing animals and no differences in the total number of tumours between the different groups. In addition, the rate of tumour appearance was the same in all groups. However, the description of the experimental protocol, statistical analysis and results was somewhat brief.

The issue of differential responsiveness of different stocks of the same strain of animals to magnetic fields was examined by Fedrowitz et al (2004). They reported that one substrain of Sprague-Dawley rat exhibited a significantly increased sensitivity to magnetic fields in the DMBA model compared with

another substrain. Thus genetic background was considered to play a crucial role in effects of magnetic field exposure.

6.2.2 EMF effects on melatonin levels and tumour growth

Other studies by Löscher and colleagues have attempted to correlate field-induced changes in the incidence and growth of DMBA-induced mammary tumours in rats with changes in the nocturnal levels of serum melatonin. In two experiments (Löscher et al, 1994; Mevissen et al, 1996a) it was reported that exposure for three months to a power frequency magnetic field, which had no significant effect on the incidence of DMBA-induced mammary tumours in female rats, was associated with significantly decreased nocturnal melatonin levels. Subsequently, a significantly increased incidence of DMBA-induced mammary tumours in female rat, 1996b, 1998); however, nocturnal melatonin levels were not significantly affected. Thus, EMF-induced changes in DMBA-induced mammary tumour incidence were not correlated with changes in nocturnal melatonin levels.

6.2.3 Summary

The effects of magnetic fields on mammary tumours have been investigated in animals using the DMBA model of breast cancer, but the evidence concerning field-dependent effects is equivocal.

Two laboratories have reported an increased incidence of chemically-induced mammary tumours in female rats exposed to power frequency magnetic fields. However, potential problems exist with these studies. The work of one laboratory was inadequately described, making it difficult to judge how well the study had been carried out, and the experiments from the other laboratory suffered from inter-experimental variability, suggesting some sort of experimental confounding.

In contrast, another two laboratories have reported a lack of effect of power frequency magnetic field exposure on chemically induced mammary tumours. One of these laboratories could not replicate the positive effects reported above, and the other found no effects following exposure to intermittent power frequency magnetic fields. However, potential problems also exist with these studies.

It has been suggested that the genetic background of animals may play a pivotal role in sensitivity to magnetic fields in the DMBA model of breast cancer. This may contribute to the observed inconsistency in outcomes between studies.

6.3 Epidemiology

Studies on the relation of EMF exposure to risk of breast cancer have been reviewed previously by AGNIR (2001) and elsewhere (Ahlbom et al, 2001), and therefore will not be considered in detail here. In brief, there have been several studies of breast cancer risks in women in relation to residential proximity to electricity transmission power lines. There have also been studies of breast cancer risk in women in relation to the use of electric blankets, and studies of breast cancer risks in occupations with potential high exposure to magnetic fields; the latter occupations tend to be male dominated, and as a

consequence many of these studies have been of breast cancer in men. There have been some studies showing raised risks for men (based on small numbers) (Tynes and Andersen, 1990; Demers et al, 1991; Matanoski et al, 1991; Feychting et al, 1998; Pollan et al, 2001) or for women mainly at premenopausal ages (Coogan et al, 1996; Feychting et al, 1998; Forssén et al, 2000; Kliukiene et al, 1999, 2003, 2004). A few studies have included data on both residential and occupational EMF exposures in women (Forssén et al, 2000; Kliukiene et al, 2004). These found some evidence for an association with breast cancer in women below 50 years of age. Overall, however, the literature does not provide substantial support for an association (AGNIR 2001; Ahlbom et al, 2001; McElroy et al, 2001; van Wijngaarden et al, 2001; Davis et al, 2002; Kabat et al, 2003; London et al, 2003; Schoenfeld et al, 2003; Forssén et al, 2005).

Few of the studies have included data on dose response, and the results of these have been inconsistent – most have not shown evidence favouring a relation (Brainard et al, 1999). There has generally been a lack of data on confounders, and only indirect indicators of exposure. Thus the studies provide only weak evidence on the possibility of an association.

6.4 References

- AGNIR (2001). ELF electromagnetic fields and the risk of cancer. Report of an Advisory Group on Non-ionising Radiation. *Doc NRPB*, **12** (1) 1–179.
- Ahlbom IC, Cardis E, Green A, Linet M, Savitz D and Swerdlow A (2001). Review of the epidemiologic literature on EMF and health. *Environ Health Perspect*, **Suppl 6**, 911–33.
- Anderson LE, Boorman GA, Morris JE, Sasser LB, Mann PC, Grumbein SL, Hailey JR, McNally A, Sills RC and Haseman JK (1999). Effect of 13 week magnetic field exposures on DMBA-initiated mammary gland carcinomas in female Sprague-Dawley rats. *Carcinogenesis*, **20**(8), 1615–20.
- Anderson LE, Morris JE, Sasser LB and Löscher W (2000). Effects of 50- or 60-hertz, 100 μT magnetic field exposure in the DMBA mammary cancer model in Sprague-Dawley rats: possible explanations for different results from two laboratories. *Environ Health Perspect*, **108**(9), 797–802.
- Anderson TJ (1998). Normal breast: myths, realities and prospects. *Mod Pathol*, **11**(2), 115–19.
- Baum A, Mevissen M, Kamino K, Mohr U and Löscher W (1995). A histopathological study on alterations in DMBAinduced mammary carcinogenesis in rats with 50 Hz, 100 μT magnetic field exposure. *Carcinogenesis*, **16**(1), 119–25.
- Beniashvili DS, Bilanishvili VG and Menabde MZ (1991). Low-frequency electromagnetic radiation enhances the induction of rat mammary tumors by nitrosomethyl urea. *Cancer Lett*, **61**(1), 75–9.
- Boorman GA, Anderson LE, Morris JE, Sasser LB, Mann PC, Grumbein SL, Hailey JR, MaNally A, Sills RC and Haseman JK (1999). Effect of 26 week magnetic field exposures in a DMBA initiation–promotion mammary gland model in Sprague-Dawley rats. *Carcinogenesis*, **20**(5), 899–904.
- Brainard CG, Kavet R and Kheifets LI (1999). The relationship between electromagnetic field and light exposures to melatonin and breast cancer risk: a review of the relevant literature. *J Pineal Res*, **26**, 65–100.
- Cardiff RD and Wellings SR (1999). The comparative pathology of human and mouse mammary glands. *J Mammary Gland Biol Neoplasia*, **4**, 105–22.
- Coogan PF, Clapp RW, Newcomb PA, Wenzl TB, Bogdan G, Mittendorf R, Baron JA and Longnecker MP (1996). Occupational exposure to 60-hertz magnetic fields and risk of breast cancer in women. *Epidemiology*, **7**, 459–64.
- Davis S, Mirick DK and Stevens RG (2002). Residential magnetic fields and the risk of breast cancer. *Am J Epidemiol*, **155**, 446–54.
- Dees C, Garrett S, Henley D and Travis C (1996). Effects of 60-Hz fields, estradiol and xenoestrogens on human breast cancer cells. *Radiat Res*, **146**, 444–52.

- Demers PA, Thomas DB, Rosenblatt KA, Jimenez LM, McTiernan A, Stalsberg H, Stemhagen A, Thompson WD, Curnen MG and Satariano W (1991). Occupational exposure to electromagnetic fields and breast cancer in men. *Am J Epidemiol*, **134**, 340–47.
- Ding G-R, Nakahara T, Tian F-R, Guo Y and Miyakoshi J (2001). Transient suppression of X-ray-induced apoptosis by exposure to power frequency magnetic fields in MCF-7 cells. *Biochem Biophys Res Commun*, **286**, 953–7.
- Ekstrom T, Mild KH and Holmberg B (1998). Mammary tumours in Sprague-Dawley rats after initiation with DMBA followed by exposure to 50 Hz electromagnetic fields in a promotional scheme. *Cancer Lett*, **123**(1), 107–11.
- Fedrowitz M, Westermann J and Löscher W (2002). Magnetic field exposure increases cell proliferation but does not affect melatonin levels in the mammary gland of female Sprague Dawley rats. *Cancer Res*, **62**, 1356–63.
- Fedrowitz M, Kamino K and Löscher W (2004). Significant differences in the effects of magnetic field exposure on 7,12-dimethylbenz(a)anthracene-induced mammary carcinogenesis in two substrains of Sprague-Dawley rats. *Cancer Res*, **64**, 243–51.
- Feychting M, Osterlund B and Ahlbom A (1998). Reduced cancer incidence among the blind. *Epidemiology*, **9**, 490–94.
- Forssén UM, Feychting M, Rutqvist LE, Floderus B and Ahlbom A (2000). Occupational and residential magnetic field exposure and breast cancer in females. *Epidemiology*, **11**, 24–9.
- Forssén UM, Rutqvist LE, Ahlbom A and Feychting M (2005). Occupational magnetic fields and female breast cancer: a case-control study using Swedish population registers and new exposure data. *Am J Epidemiol*, **161**, 250–59.
- Kabat GC, O'Leary ES, Schoenfeld ER, Greene JM, Grimson R, Henderson K, Kaune WT, Gammon MD, Britton JA, Teitelbaum SL, Neugut AI and Leske MC (2003). Electric blanket use and breast cancer on Long Island. *Epidemiology*, **14**, 514–20.
- Kliukiene J, Tynes T, Martinsen JI and Blaasaas KG AA (1999). Incidence of breast cancer in a Norwegian cohort of women with potential workplace exposure to 50 Hz magnetic fields. *Am J Ind Med*, **36**, 147–54.
- Kliukiene J, Tynes T and Andersen A (2003). Follow-up of radio and telegraph operators with exposure to electromagnetic fields and risk of breast cancer. *Eur J Cancer Prev*, **12**, 301–7.
- Kliukiene J, Tynes T and Andersen A (2004). Residential and occupational exposures to 50-Hz magnetic fields and breast cancer in women: a population-based study. *Am J Epidemiol*, **159**, 852–61.
- Loberg LI, Gauger JR, Buthod JL, Engdahl WR and McCormick DL (1999). Gene expression in human breast epithelial cells exposed to 60 Hz magnetic fields. *Carcinogenesis*, **20**, 1633–6.
- Loberg LI, Engdahl WR, Gauger JR and McCormick DL (2000). Cell viability and growth in a battery of human breast cancer cell lines exposed to 60 Hz magnetic fields. *Radiat Res*, **153**, 725–8.
- London SJ, Pogoda JM, Hwang KL, Langholz B, Monroe KR, Kolonel LN, Kaune WT, Peters JM and Henderson BE (2003). Residential magnetic field exposure and breast cancer risk: a nested case–control study from a multiethnic cohort in Los Angeles County, California. *Am J Epidemiol*, **158**, 969–80.
- Löscher W and Mevissen M (1995). Linear relationship between flux density and tumor co-promoting effect of prolonged magnetic field exposure in a breast cancer model. *Cancer Lett*, **96**(2), 175–80.
- Löscher W, Mevissen M, Lehmacher W and Stamm A (1993). Tumor promotion in a breast cancer model by exposure to a weak alternating magnetic field. *Cancer Lett*, **71**(1–3), 75–81.
- Löscher W, Wahnschaffe U, Mevissen M, Lerchl A and Stamm A (1994). Effects of weak alternating magnetic fields on nocturnal melatonin production and mammary carcinogenesis in rats. *Oncology*, **51**(3), 288–95.
- Löscher W, Mevissen M and Haussler B (1997). Seasonal influence on 7,12-dimethylbenz[a]anthracene-induced mammary carcinogenesis in Sprague-Dawley rats under controlled laboratory conditions. *Pharmacol Toxicol*, **81**(6), 265–70.
- McElroy JA, Newcomb PA, Remington PL, Egan KM, Titus-Ernstoff L, Trentham-Dietz A, Hampton JM, Baron JA, Stampfer MJ and Willett WC (2001). Electric blanket or mattress cover use and breast cancer incidence in women 50–79 years of age. *Epidemiology*, **12**, 613–17.
- Matanoski GM, Breysse PN and Elliott EA (1991). Electromagnetic field exposure and male breast cancer. *Lancet*, **337**, 737.

- Mevissen M, Stamm A, Buntenkötter S, Zwingleberg R, Wahnschaffe U and Löscher W (1993a). Effects of magnetic fields on mammary tumour development induced by 7,12-dimethylbenz(a)anthracene in rats, *Bioelectromagnetics*, **14**, 131–43.
- Mevissen M, Wahnschaffe U, Löscher W, Stamm A and Lerchl A (1993b). Effects of AC magnetic field on DMBAinduced mammary carcinogenisis in Sprague-Dawley rats. In *Electricity and Magnetism in Biology and Medicine* (M Blank, ed). San Francisco Press, pp 413–15.
- Mevissen M, Lerchl A and Löscher W (1996a). Study on pineal function and DMBA-induced breast cancer formation in rats during exposure to a 100-mG, 50 Hz magnetic field. *J Toxicol Environ Health*, **48**(2), 169–85.
- Mevissen M, Lerchl A, Szamel M and Löscher W (1996b). Exposure of DMBA-treated female rats in a 50-Hz, 50-μT magnetic field: effects on mammary tumor growth, melatonin levels, and T lymphocyte activation. *Carcinogenesis*, **17**(5), 903–10.
- Mevissen M, Haussler M, Lerchl A and Löscher W (1998). Acceleration of mammary tumorigenesis by exposure of 7,12-dimethylbenz[a]anthracene-treated female rats in a 50-Hz, 100-µT magnetic field: replication study. *J Toxicol Environ Health A*, **53**(5), 401–18.
- NTP (1999). NTP technical report on the toxicology and carcinogenesis studies of 60-Hz magnetic fields in F344/N rats and B6C3F1 mice. Washington DC, National Toxicology Program. NTP TR 488, NIH Publication No. 99-3979.
- Pollan M, Gustavsson P and Floderus B (2001). Breast cancer, occupation, and exposure to electromagnetic fields among Swedish men. *Am J Ind Med*, **39**, 276–85.
- Schoenfeld ER, O'Leary ES, Henderson K, Grimson R, Kabat GC, Ahnn S, Kaune WT, Gammon MD and Leske MC (2003). Electromagnetic fields and breast cancer on Long Island: a case–control study. *Am J Epidemiol*, **158**, 47–58.
- Short RV and Drife JO (1977). The aetiology of mammary cancer in man and animals. *Symp Zoological Soc London*, **41**, 211–30.
- Supino R, Bottone MG, Pellicciari C, Caserini C, Bottiroli G, Belleri M and Veicsteinas A (2001). Sinusoidal 50 Hz magnetic fields do not affect structural morphology and proliferation of human cells *in vitro*. *Histol Histopathol*, **16**, 719–26.
- Thun-Battersby S, Mevissen M and Löscher W (1999). Exposure of Sprague-Dawley rats to a 50-hertz, 100-µT magnetic field for 27 weeks facilitates mammary tumorigenesis in the 7,12-dimethylbenz[a]-anthracene model of breast cancer. *Cancer Res*, **59**(15), 3627–33.
- Tofani S, Barone D, Cintorino M, de Santi MM, Ferrara A, Orlassino R, Ossola P, Peroglio F, Rolfo K and Ronchetto F (2001). Static and ELF magnetic fields induce tumor growth inhibition and apoptosis. *Bioelectromagnetics*, **22**, 419–28.
- Tsubura A, Hatano T, Hayama S and Morii S (1991). Immunophenotypic difference of keratin expression in normal mammary glandular cells from five different species. *Acta Anaton*, **140**, 287–93.
- Tynes T and Andersen A (1990). Electromagnetic fields and male breast cancer. Lancet, 336, 1596.
- Van Wijngaarden E, Nylander-French LA, Millikan RC, Savitz DA and Loomis D (2001). Population-based case-control study of occupational exposure to electromagnetic fields and breast cancer. Ann Epidemiol, 11, 297–303.
- Yoshizawa H, Tsuchiya T, Mizoe H, Ozeki H, Kanao S, Yomori H, Sakane C, Hasebe S, Motomura T, Yamakawa T, Mizuno F, Hirose H and Otaka Y (2002). No effect of extremely low-frequency magnetic field observed on cell growth or initial response of cell proliferation in human cancer cell lines. *Bioelectromagnetics*, **23**, 355–68.

7 Conclusions

The present report by the independent Advisory Group on Non-ionising Radiation considers scientific research investigating the possibility that chronic exposure to power frequency electromagnetic fields (EMFs) may increase the risk of breast cancer via a reduction in melatonin secretion from the pineal gland. The report covers cellular, animal and human volunteer studies as well as epidemiological investigations. It also summarises the sources and extent of exposure to EMFs in the home and at work, and reviews the basic physiology of the pineal gland and the secretion of melatonin. The risk of cancer from exposure to extremely low frequency EMFs was the subject of a comprehensive review by the Advisory Group in 2001; this report extends and updates that information relating to breast cancer.

7.1 Sources and exposure assessment

Power frequency EMFs are encountered everywhere electricity is used, in and outside the home and at work.

Maximum electric field strengths of up to several kilovolts per metre can be encountered at ground level under high voltage lines in the UK; however, because electric fields are easily screened, ambient levels inside homes are usually very much lower, typically in the range $0-20 \text{ V m}^{-1}$.

Magnetic fields of up to a few tens of microtesla can occur beneath the largest power lines; however, the levels normally encountered rarely exceed more than a few microtesla. Magnetic fields are not easily screened and the levels inside homes usually arise from the external electricity supply circuits, with contributions from the wiring and electrical appliances inside the home. The background fields in UK homes typically fall in the range from 0.01 to 0.1 μ T; average residential exposures above 0.4 μ T occur for about 0.5% of the population

The industries most commonly associated with high magnetic field exposure are the electrical utilities, electrically powered transport systems and those where magnetic fields are produced directly as part of an industrial process such as welding or induction heating.

Characterisation of exposure is complex because of the wide range of measures that are necessary to describe fully the fields and the diversity of possible sources. It is not known what aspect of exposure, beyond the field strength values that form the basis for exposure guidelines, might most directly relate to health outcomes in human populations. However, instrumentation has improved and modern meters allow more flexible approaches, which include logging over time and details of exposure characteristics such as frequency, polarisation, spatial orientation and wave shape. It is also important to consider that power frequency magnetic fields vary with time broadly following the variation in use of electrical power.

7.2 The pineal gland and melatonin

The pineal gland is a photoneuroendocrine transducer organ, converting information about daylength (and possibly light intensity) into a hormonal signal: melatonin.

Melatonin is normally secreted during the night (the dark phase) in all species whether nocturnal or diurnal, and the duration of its night-time secretion indicates the length of the night. One definition of 'biological night' is the period of the 24 hour cycle for which melatonin is above baseline (usually daytime) concentrations. In photoperiodic mammals (where seasonal changes in physiology depend upon daylength), the profile of melatonin secretion provides an essential time cue for the organisation of seasonal activity, such as reproduction or coat growth.

In so far as human physiology is dependent on daylength, melatonin is likely to serve as a photoperiodic signal. Its role within the circadian system of mammals including humans appears to be the reinforcement of 'night-time' physiology (for example, changes in core body temperature, or sleep propensity), the modulation of the circadian phase shifting response to light, and in general to serve as an endogenous zeitgeber (time cue). Being highly liposoluble it penetrates all tissues and body fluids and it therefore has the potential to influence peripheral oscillators, including gene expression. Endogenous melatonin probably serves to reinforce 'coupling' and to optimise phase within the circadian system and thus influence the multitude of systems governed by circadian oscillators.

In mammals, melatonin is a 'hand' of the circadian clock rather than part of the clock mechanism. It is extensively used as the 'best' marker of the timing of the circadian clock as either the endogenous hormone or its major metabolite 6-sulphatoxymelatonin (aMT6s). There are very large variations between individuals in the concentrations recorded in plasma or saliva and in the urinary content of aMT6s. Thus for cross-sectional studies of changes in melatonin secretion, large study populations are required. A single sample, be this of morning urine, or morning or night-time plasma or saliva, does not provide sufficient information to define the characteristics of its secretion. There is an age-related decline in secretion (in most studies) and an earlier timing of the rhythm with age. Thus age-matched control populations (or age-adjustment) are required for comparative purposes. Light exposure must be controlled if the acute effects of light are to be avoided. Numerous drugs and other substances influence its secretion.

Light suppresses melatonin synthesis at night and timed exposure to light of suitable intensity and spectral composition shifts the melatonin rhythm and all other circadian rhythms investigated to date. Night-shift work and time-zone change may be associated with lower melatonin production, light at night being one of several possible causal factors.

Exogenous melatonin phase shifts and entrains circadian rhythms. Melatonin has a diverse range of actions implemented through a variety of cellular mechanisms. Some actions are mediated via specific receptors for melatonin, whereas other effects are more general – for example, its ability to scavenge free radicals is related to the hormone's chemical structure.

7.3 Effects of EMFs on melatonin

In vitro studies

The evidence that EMF exposure changes melatonin production in isolated rodent cells or glands is not convincing, despite the exposures being higher than those encountered in the environment.

The idea that EMF exposure can interfere with the cellular action of melatonin is intriguing and there is some independent supporting evidence. However, the effect is fairly small, not robust and has doubtful significance to human health.

In vivo studies

Many, but not all, studies using rats, mice and hamsters indicate that exposure to magnetic fields does not result in a consistent suppression of nocturnal melatonin levels. The best evidence for a fielddependent effect comes from studies with rats exposed to circularly polarised magnetic fields, although these results are not definitive. Too few data exist to make any firm conclusions regarding the effects of magnetic fields on melatonin levels in non-human primates. The results of existing studies using electric fields are unreliable.

Human experimental studies

While a few laboratory-based studies suggest that acute exposure of volunteers may have an effect on the production or timing of the nightly melatonin rise, the majority of studies have not found any field-dependent effects. The lack of a consistent magnetic-field-dependent effect on the human melatonin profile may be due to inappropriate study designs, confounding, or the absence of a real effect. The possibility remains that magnetic fields may affect the circadian clock at a more subtle level, and long-term effects have not been properly evaluated as far as melatonin is concerned.

Epidemiology

In aggregate, the epidemiological studies of the relation of melatonin levels to EMF exposure do not give convincing evidence that EMFs affect the secretion of melatonin in humans. Although most of the published studies have found some significant results, usually in a subset of the data, there has been no consistency in the subgroup for which significant results were found, and indeed in general the significant results have not been re-examined for the same subgroup in subsequent studies. The studies have varied in the measures of EMF exposure and of melatonin secretion that have been used, and the melatonin measures have often been unsatisfactory as markers of 24 hour total excretion. At present, therefore, the studies overall can be seen only as hypothesis-generating rather than giving clear evidence for an association.

7.4 Melatonin and breast cancer

In vitro studies

The oncostatic action of melatonin has been demonstrated mainly in one breast cancer cell line (MCF-7) of which only some sub-clones were responsive to physiological levels of melatonin. In the light of these limited findings there is concern that the oncostatic action may not be applicable to other breast cancer cells or cancers in general.

In vivo studies

Under the appropriate experimental conditions, treatment with exogenous melatonin in the late afternoon or early evening can result in well-defined oncosuppresive effects in animals with mammary tumours. Melatonin appears to have the greatest effect on the promotional phase of tumorigenesis. Conversely, manipulations that decrease endogenous melatonin levels, such as light-at-night, may increase the incidence and growth of tumours (which is consistent with the hypothesis that light-induced suppression of melatonin rhythms may be a risk factor for human breast cancer). A few studies have reported inconsistent or paradoxical results, possibly reflecting that the susceptibility of mammary tissues to carcinogenic insult is a complex response modulated by various environmental and other factors, including the level of endogenous melatonin.

Epidemiology: breast cancer risk in relation to melatonin levels

Two cohort studies have assessed the risk of subsequent breast cancer in women from whom data were available on urinary melatonin metabolite levels. The numbers of breast cancers included in each were not large, and the currently available results are inconclusive. Several case–control studies have compared melatonin levels in patients with breast cancer to those in women without this disease, but because the levels might have been altered by the presence of breast cancer, its treatment, and changes in behaviour as a consequence of these, the results do not allow conclusions on melatonin as a potential causal factor for breast cancer.

Epidemiology: breast cancer risk in relation to light exposure

There is appreciable, but not conclusive, evidence for a relation of blindness to decreased risk of breast cancer, and of airline cabin work and shift work to raised risk. Confounding is a potential explanation in each instance, but even if there are true causal associations it is far from clear, and there is no direct evidence, that the associations involve melatonin. The results currently available are intriguing, but do not provide strong evidence for the melatonin hypothesis.

Melatonin, other hormones and breast cancer

The hypothesis that decreased exposure to melatonin leads to increased levels of reproductive steroid hormones which in turn stimulate either the proliferation of breast cells or the growth of breast cancer remains speculative and is without a substantial body of supportive evidence. Similarly, the case that melatonin may increase breast cancer risk through changing the levels of growth factors or other hormones such as prolactin is not convincing: data relate to small groups of women with established breast cancer. However, there is an interesting and increasing body of evidence to suggest that melatonin may have direct anti-oestrogenic effects on breast cancer cells.

7.5 Effects of EMFs on breast cancer

In vitro studies

An earlier Advisory Group report found no convincing evidence of genotoxicity or transformation in cells due to exposure to EMFs. The studies undertaken since that report provide further evidence of a lack of a direct effect of exposure on breast cancer cells.

In vivo studies

The effect of magnetic fields on mammary tumours has been investigated in rats using the DMBA model of breast cancer. However, potential problems exist with these studies and the evidence concerning field-dependent effects remains equivocal: some studies report effects while others do not. Significant differences in sensitivity to the chemical carcinogen used in these studies have been observed in the same strain of rat, and this may reflect genetic differences between different stocks of these animals. It has been suggested that the genetic background of animals may play a pivotal role in sensitivity to magnetic fields in the DMBA model of breast cancer, and this may have contributed towards the inconsistency in experimental results so far obtained.

Epidemiology

Although some epidemiological studies of breast cancer risk in relation to EMF exposure have had positive findings, the literature overall does not support an association. The studies have generally only collected indirect data on EMF exposures, and have lacked data on confounders. They therefore provide only weak evidence on the possibility of an association.

7.6 Overall summary and conclusions

EMF exposure is ubiquitous in modern life. The hypothesis that chronic exposure to power frequency EMFs may increase the risk of breast cancer via a reduction in melatonin secretion was first made almost 20 years ago, and has led to a great deal of research. To review this hypothesis, the Advisory Group has addressed in this report evidence on three questions: whether EMFs affect melatonin production or action; whether melatonin affects risk of breast cancer; and whether EMFs affect risk of breast cancer.

Investigations using cells, animals and humans have not given consistent or convincing evidence that EMF exposure affects melatonin production or action. However, there are deficiencies in the existing research, which leave open the possibility of an effect.

There is stronger evidence that melatonin can inhibit the growth of cancer cells both *in vitro* and in animals. Data on the possible relation of melatonin levels to risk of subsequent breast cancer in humans are limited and inconclusive. Studies investigating the effect of light exposure (which affects melatonin) on breast cancer risk in humans have given some evidence for an association, but left it unclear whether, if there is an association, it is causal in nature.

There is no consistent evidence, from research using cells, animals and humans, that EMF exposure is a cause of breast cancer, nor has any mechanism for such an association been demonstrated.

Overall, the evidence that melatonin, and the timing and extent of light exposure, may affect breast cancer risk is intriguing but not conclusive. In aggregate, the evidence to date does not support the hypothesis that exposure to power frequency EMFs affects melatonin levels or risk of breast cancer.

8 Research Recommendations

The following recommendations are made for further research into possible relations between exposure to power frequency electromagnetic fields (EMFs), melatonin and the risk of breast cancer. Separate proposals are made regarding the effects of exposure to EMFs on melatonin, the effects of melatonin on breast cancer, and the effects of EMFs on breast cancer. Specific recommendations are made for laboratory investigations using cells, animals and humans, and for epidemiological research.

8.1 Effects of EMFs on melatonin

In vitro studies

There is scope for well-designed experiments on the effects of EMF exposure on isolated pineal glands, especially during a period of elevated melatonin synthesis.

In vivo studies

Generally, additional animal studies exploring the effects of EMFs on melatonin rhythms are not recommended, due to the low probability of finding consistent field-dependent effects. However, in order to confirm and characterise the effects of circularly polarised fields more fully, additional studies with these fields should be undertaken. These studies should aim to replicate the reported effects, but using concurrent sham-exposed animals, and not using historical control groups.

Human experimental studies

The lack of a consistent short-term effect on the major features of human melatonin profile does not eliminate the possibility that magnetic fields may affect the circadian clock at a more subtle level. Clarification of this issue will rely on the analysis of more sensitive indicators of changes in the rhythm-generating system, such as changes in clock gene expression in the suprachiasmatic nucleus or in peripheral oscillatory systems. In addition, although difficult to accomplish, longer-term controlled studies are required. A compromise is possible, which would avoid the difficulties of long-term controlled experiments. This would be the evaluation, in constant routine conditions, and controlling for all relevant factors, of melatonin phase and amplitude, and detailed profiles, in subjects whose magnetic field exposure has been carefully monitored over a designated time period. More importantly, however, other circadian rhythms should be taken into account. Any effect on melatonin amplitude or timing is likely to involve the whole circadian system.

Epidemiology

The existing literature does not suggest that further epidemiological study of the possible relation between EMFs and melatonin should be a priority, especially as human volunteer studies are in general methodologically preferable for investigating this issue. If further epidemiological studies are to be conducted, they need to include better information on circadian factors, especially light exposure, and need to re-examine apparent positives from the existing literature rather than solely to examine new subgroups or analytical categories.

In addition, careful consideration should be given to the characterisation of the EMF environment and the way in which the exposure assessment is conducted. The ability to define more carefully the relevant exposure of interest seems to be a crucial challenge for future studies. Exposure assessments should include comprehensive breakdowns by periods at home, work and elsewhere, and separate exposures during sleep and work, including shift time. The potential for diurnal and seasonal variability in the exposure data also needs to be taken into account.

8.2 Melatonin and breast cancer

In vitro studies

Melatonin may have direct anti-oestrogenic effects on breast cancer cells. Further research is required to determine the physiological relevance of these results (which have been derived from relatively artificial model systems, such as MCF-7 cells) and whether the observations will also be apparent in primary breast cancers.

In vivo studies

Studies with rodents have indicated that melatonin may possess oncostatic properties. Therefore further animal studies are suggested to establish and characterise these possibilities further. In this respect, the use of transgenic models which over-express genes associated with mammary cancer offer promise since there is no need for additional treatment with a chemical carcinogen, and hence there is the elimination of any variables associated with that process. In addition, application of melatonin in animals' drinking water is considered the least stressful route for administration for the animals, especially compared to repeated, long-term gavage or injection (although the latter may offer the potential for a greater control and consistency of dose). The effects of exposure to constant light are also considered important to explore further, and studies using the tissue-isolated, MCF-7 breast cancer xenograft preparation in nude rats are recommended in particular to confirm and extend the original observation.

Epidemiology: breast cancer risk in relation to melatonin levels

Case-control studies comparing melatonin levels in patients with breast cancer and in women without the disease are difficult to interpret because cancer itself, treatment and stress or behavioural changes associated with diagnosis and treatment may influence hormone concentrations. Furthermore, during the time interval between initiation and appearance of cancer, the hormone/growth factor milieu may have changed markedly. Prospective investigations of cohorts of individuals who are apparently 'normal' but subsequently go on to develop cancer or remain disease-free are much more informative. However, these need both to be well conducted and to take account of potential confounding factors. Large numbers of subjects will be required, as can be seen from previous investigations into the role of oestrogens in risk of breast cancer. The cost and labour involved in establishing such research simply to compare melatonin levels do not seem to be justified. However, this should not preclude the inclusion of melatonin measurements in sub-protocols of studies investigating risk of breast cancer for other primary reasons.

Epidemiology: breast cancer risk in relation to light exposure

The results that suggest the relation of blindness, airline cabin work and shift work to risk of breast cancer, give a strong case for further research to clarify these potentially important associations. Future epidemiological studies need to gain better data on potential confounders and, for airline and shift work, need to gain more detailed information on the occupational factor: ideally these breast cancer studies would include data on melatonin and other hormone levels. It would also be of value to investigate further the relation of blindness, airline cabin work and shift work to hormone levels.

8.3 EMFs and breast cancer

In vitro studies

There is some uncertainty about EMFs having co-promotional effects on breast cancer cells, and of possible long-term effects of exposure; however, both of these questions would be better answered by *in vivo* studies.

In vivo studies

Studies using the DMBA model of breast cancer have suggested that different substrains of animals may exhibit different sensitivity to magnetic fields. Therefore further studies are suggested to establish whether differing sensitivities exist, and if so then genomic studies are recommended to identify the genetic factors involved. It should be relatively straightforward, although time consuming, to use DNA microarrays or similar modern technologies to identify and characterise the genetic basis for these differences between substrains.

Epidemiology

The existing epidemiology leaves uncertainty on whether there is a relation of EMF exposure to risk of breast cancer. If further investigation of this is undertaken, it needs to include better exposure assessment, data on potential confounders, and analysis of dose and duration response relations.

Glossary

Acrophase The time of the peak of a rhythm, usually the peak time of the best-fitting mathematical function approximating the data.

Amplitude The amount of variability due to a given rhythm, usually defined as equal to one-half the peak-to-trough difference.

Biological/subjective day/night Biological/subjective night is often considered as the period of time during which melatonin is secreted in humans. In general, with no light and dark time cues biological/subjective night refers to the time during which night-time phenomena are manifest, eg running activity and peak melatonin secretion in nocturnal rodents, sleep and peak melatonin secretion in humans.

Change in timing A regularly occurring phenomenon which occurs earlier or later than usual

Circadian Occurring or recurring about (Latin – *circa*) once per day (Latin – *dies*). Biological circadian rhythms are internally generated and, in humans, have an intrinsic period (in the absence of time cues) which is usually slightly longer than 24 hours, other terms include circannual: about 1 year, ultradian or pulsatile: with a period shorter than 20 hours.

Entrainment An 'entrained' rhythm is synchronised with the correct phase with respect to a specific time cue, eg melatonin production at night.

Free-run/free-running A rhythm displaying its intrinsic period. Human free-running rhythms are usually slightly longer than 24 hours.

Gavage Direct delivery of solution (eg drug, nutrient) into the stomach using a small tube via the mouth; also known as intragastric (ig) intubation.

Intrinsic period (tau, τ) The period of a biological rhythm manifested in the absence of time cues.

Period The duration of one complete cycle of a rhythmic variation.

Phase A distinct stage in a process of rhythmic change, eg a circadian rhythm, the dark phase or the light phase of the day.

Phase shift An abrupt change in the timing of time cues maintaining circadian synchrony, eg after time zone change or starting night shift work

Photoperiod Strictly the length of the light phase of a particular light–dark cycle, but may be used to describe the whole light–dark cycle.

Photoperiodism The response of an organism to changes in the lengths of the daily periods of light.

Synchronisation The process by which a free-running rhythm is timed to coincide with a specific time cue or cues. A synchronised rhythm having the period of a specific time cue (eg 24 hours) does not necessarily have the 'correct' phase, eg melatonin peak production during the daytime.

Vehicle An inert carrier in which a chemical or other active agent is administered.

Zeitgeber, time cue or synchroniser A periodic stimulus capable of determining the timing, with respect to clock hour or calendar date, of a given endogenous rhythm.

Appendix

Publications of the independent Advisory Group on Non-ionising Radiation

- Electromagnetic fields and the risk of cancer. Report of an Advisory Group on Non-ionising Radiation. *Doc NRPB*, 3(1), 1–138 (1992).
- Electromagnetic fields and the risk of cancer. Summary of the views of the Advisory Group on Non-ionising Radiation on epidemiological studies published since its 1992 report. *Doc NRPB*, 4(5), 65–9 (1993).
- **3** Health effects related to the use of visual display units. Report of an Advisory Group on Non-ionising Radiation. *Doc NRPB*, **5**(2), 1–75 (1994).
- 4 Electromagnetic fields and the risk of cancer. Supplementary report by the Advisory Group on Non-ionising Radiation (12 April 1994). *Doc NRPB*, **5**(2), 77–81 (1994).
- 5 Health effects from ultraviolet radiation. Report of an Advisory Group on Non-ionising Radiation. Doc NRPB, 6(2), 7–190 (1995).
- **6** Use of sunbeds and cosmetic tanning. Statement by the Advisory Group on Non-ionising Radiation. *Radiol Prot Bull,* No. 218, 11–15 (1999).
- 7 The solar eclipse. Statement by the Advisory Group on Non-ionising Radiation. Chilton, NRPB Information Services Leaflet P8/99 (1999).
- 8 ELF electromagnetic fields and the risk of cancer. Report of an Advisory Group on Non-ionising Radiation. *Doc NRPB*, **12**(1), 1–179 (2001).
- **9** Possible health effects from terrestrial trunked radio (TETRA). Report of an Advisory Group on Non-ionising Radiation. *Doc NRPB*, **12**(2), 1–86 (2001).
- **10** ELF electromagnetic fields and neurodegenerative disease. Report of an Advisory Group on Non-ionising Radiation. *Doc NRPB*, **12**(4), 1–24 (2001).
- Health effects from ultraviolet radiation. Report of an Advisory Group on Non-ionising Radiation.
 Doc NRPB, 13(1), 1–276 (2002).
- **12** Health effects from radiofrequency electromagnetic fields. Report of an independent Advisory Group on Non-ionising Radiation. *Doc NRPB*, **14**(2), 1–177.
- 13 Particle deposition in the vicinity of power lines and possible effects on health. Report of an independent Advisory Group on Non-ionising Radiation and its Ad Hoc Group on Corona lons. Doc NRPB, 15(1), 1–55 (2004).

Health Protection Agency Centre for Radiation, Chemical and Environmental Hazards Radiation Protection Division Chilton Didcot Oxfordshire OX11 0RQ United Kingdom

Tel: +44 (0)1235 831600 Fax: +44 (0)1235 833891 email: information@hpa-rp.org.uk www.hpa.org.uk

Documents of the Health Protection Agency

Series B: Radiation, Chemical and Environmental Hazards RCE-1 February 2006 ISBN 0 85951 573 7 £30.00 © Health Protection Agency This report is printed on chlorine-free paper